THE FEDERAL DEMOCRATIC REPUBLIC OF ETHIOPIA CENTRAL STATISTICAL AGENCY

STATISTICAL REPORT ON
THE 2012 URBAN EMPLOYMENT UNEMPLOYMENT
SURVEY

Addis Ababa September 2012

STATISTICAL BULLETIN

Created by PDF Generator (http://www.alientools.com/), to remove this mark, please buy the softwar

TABLE OF CONTENT

Table of Cont	ents	i
List of Statisti	cal Tables	iii
Chapter I	Background of the Survey	1
1.1	Introduction	1
1.2	Objectives of the Survey	3
1.3	Content and Organization of the Survey Questionnaire	4
1.4	Training of Field Staff	5
1.5	Organization of the Field Work	6
1.6	Data Processing	7
1.7	Definitions of Terms and Concepts	
1.8	Organization of the Report	9
1.9	Abbreviations	
1.0	Abbiovations	.0
Chapter II	Survey Methodology	11
2.1	Coverage	
2.2	Sampling Frame	
2.2	Sample Design	
2.3 2.4	Sample Size and Selection Scheme	10
2.4	Sample Size and Selection Scheme	12
Chapter III	Cools Demographic Characteristics of	
Chapter III	Socio - Demographic Characteristics of Urban Population	15
2.4		
3.1	Introduction	15
3.2	Age- Sex Structure of Urban Population and	
	Other Socio - Demographic Data	15
Chapter IV	Economic Activity Status	39
4.1	Introduction	39
4.2	Data Collection Approaches of the Economic Activity Status of	
	Urban Population	41
	4.2.1 Current Activity Status Approach	41
	4.2.2 Usual Activity Status Approach	
4.3	Current Activity Status	
4.4	Economically Active and Non-Active Population and Activity Rate	43
Chapter V	Characteristics of the Currently urban Employed Population	63
5.1	Introduction	. 63
5.2	Employment to Population Ratio	. 63
5.3	Occupation and Industry	. 64
5.4	Employment Status	. 65
5.5	Number of Hours Worked	. 65
5.6	Earning from Paid Employment	
5.7	Absence from Work	
5.8	Availability to Work Additional Hours	
5.9	Formal and Informal Employment Sector	
	1 2	-

Chapter V I	Unemployment in Urban Areas	187
6.1 6.2 6.3	Introduction Measurement of Unemployment Size and Rate of Urban Unemployment	188
Annex I Annex II Annex III	Survey Questionnaire Estimation Procedures of Total, Ratio and Sampling Errors Estimates and CV's for Selected Tables	235
Deferences		242

LIST OF STATISTICAL TABLES

		Page
Table 3.1	Estimated Total Population of Urban Areas by Age Group, Sex and	
	Region: 2012	17
Table 3.2	Distribution of Urban Population, Households and Average Household Size By Region: 2012	
Table 3.3	Distribution of Urban Population by Region, Sex and Relationship to the	
Table 0.4	Head of Household: 2012	20
Table 3.4	Distribution of Urban Population Aged Ten Years and Above by	
	Age Group, Sex, Literacy Status and Educational Attainment - COUNTRY TOTAL: 2012	00
Table 3.5		22
Table 3.5	Distribution of Urban Population Aged Ten Years and	
	Above by Region, Sex, Training Status and Types of Training - COUNTRY TOTAL: 2012	24
Table 3.6	Distribution of Urban Population by Five Years Age Group,	24
Table 3.0	Sex and Sex Ratio - COUNTRY TOTAL: 2012	22
	Sex and Sex Ratio - COUNTRY TOTAL: 2012	33
Table 4.1.	Urban population Aged Ten Years and Above by Age Group,	
1 able 4.1.	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) - COUNTRY TOTAL: 2012	/11
Table 4.1.1	Urban population Aged Ten Years and Above by Age Group,	4 1
1 4.1.1	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) -TIGRAY REGION: 2012	42
Table 4.1.2	Urban population Aged Ten Years and Above by Age Group,	42
14016 4.1.2	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) - AFFAR REGION: 2012	43
Table 4.1.3	Urban population Aged Ten Years and Above by Age Group,	40
14016 4.1.5	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) - AMHARA REGION: 2012	11
Table 4.1.4	Urban population Aged Ten Years and Above by Age Group,	44
14516 7.1.7	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) – OROMIA REGION: 2012	45
Table 4.1.5	Urban population Aged Ten Years and Above by Age Group,	
14010 4.1.0	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) - SOMALI REGION: 2012	46
Table 4.1.6	Urban population Aged Ten Years and Above by Age Group,	·····
14510 41110	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) - BENISHANGUL GUMUZ REGION: 2012	47
Table 4.1.7	Urban population Aged Ten Years and Above by Age Group,	
14510 41111	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) - SNNP REGION: 2012	48
Table 4.1.8	Urban population Aged Ten Years and Above r by Age Group,	
14510 4.110	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) - GAMBELLA REGION: 2012	40
Table 4.1.9	Urban population Aged Ten Years and Above by Age Group,	
14510 4.115	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) - HARARI REGION: 2012	50
Table 4.1.10	Urban population Aged Ten Years and O Above by Age Group, Sex,	
. 40.0 7.11.10	Activity Status and Rate during the Last Seven Days	
	(Current Status Approach)-	
	ADDIS ABABA CITY ADMINISTRATION: 2012	51
Table 4.1.11	Urban population Aged Ten Years and Above by Age Group,	
. 40.0 7.11.11	Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach) DIDE DAWA ADMINISTRATION: 2012 52	

Table 4.2	Population of Major Towns Aged Ten Years and Above by Sex, Activity Status and Rate during the Last Seven Days	
	(Current Status Approach): 2012	53
Table 4.3	Economically Non - Active Population of Urban Areas	
	Aged Ten Years and Above by Age Group, Sex and Reasons	
	For Not Being Active During the Last Seven Days	
	(Current Status Approach) - COUNTRY TOTAL: 2012	55
Table 4.4	Economically Non-active Population of Urban Areas	
	Aged Ten Years and Above by Region, Sex and Reasons	
	For Not Being Active During the Last Seven days	
	(Current Status Approach):2012	57
Table 5.1	Urban Population Aged Ten Years and Above by Region, Age Group,	
	Sex and Employment to Population Ratio - COUNTRY TOTAL: 2012	67
Table 5.2	Urban Population of Major Towns Aged Ten Years and	
	Above by Age Group, Sex and Employment to	
	Population Ratio: 2012	71
Table 5.3	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Age Group, Sex and	
	Major Occupational Groups - COUNTRY TOTAL: 2012	77
Table 5.4	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Region, Sex and	
	Major Occupational Groups: 2012	79
Table 5.5	Currently Employed Population of Major Towns Aged Ten Years	
	and Above by Sex and	
	Major Occupational Groups: 2012	81
Table 5.6	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Region, Sex and	
	Minor Occupational Groups: 2012	83
Table 5.7	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Age Group, Sex and Major Industrial Divisions -	
T-11- 50	COUNTRY TOTAL: 2012	102
Table 5.8	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Region, Sex and Major Industrial Divisions –	101
Table F.O.	COUNTRY TOTAL: 2012	104
Table 5.9	Currently Employed Population of Major Towns Aged Ten Years	107
Table 5.10	and Above by Sex and Major Industrial Divisions: 2012	107
Table 5.10	Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Minor Industrial Divisions-	
	COUNTRY TOTAL: 2012	100
Table 5.11	Currently Employed Population of Urban Areas Aged Ten Years	108
Table 5.11	and Above by Age Group, Sex and Status in Employment -	
	COUNTRY TOTAL: 2012	136
Table 5.12	Currently Employed Population of Urban Areas Aged Ten Years	
14510 0.12	and Above by Region, Sex and Status in Employment: 2012	138
Table 5.13	Currently Employed Population of Major Towns Aged Ten Years	
14510 0110	and Above by Sex and Status in Employment: 2012	140
Table 5.14	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Sex, Major Occupation Groups and	
	Major Industrial Divisions - COUNTRY TOTAL: 2012	142
Table 5.15	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Sex, Major Industrial Divisions and	
	Employment Status – COUNTRY TOTAL: 2012	145
Table 5.16	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Major Occupational Groups, Sex and	
	Status in Employment - COUNTRY TOTAL: 2012	147

Table 5.17	Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Age Group and Number of Hours Worked Per Week- COUNTRY TOTAL: 2012	. 149
Table 5.18	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Region, Sex and Number of Hours	4 - 4
Table 5.40	Worked Per Week: 2012	. 151
Table 5.19	Currently Employed Population of Urban Areas Aged Ten Years and Above by Major Industrial Divisions, Sex and Number of Hours Worked Per Week - COUNTRY TOTAL: 2012	153
Table 5.20	Currently Employed Population of Urban Areas Aged Ten Years	. 100
1 4510 0.20	and Above by Sex, Major Occupational Groups and Number	
	of Hours Worked Per Week - COUNTRY TOTAL: 2012	155
Table 5.21	Currently Employed Population Urban Areas Aged Ten Years	
. 4.5.0 0.2.	and Above by Sex, Status in Employment and Number	
	of Hours Worked Per Week - COUNTRY TOTAL: 2012	. 157
Table 5.22	Paid Employees of Urban Population Aged Ten Years and	
	Above by Region, Sex and Type of Payment: 2012	. 159
Table 5.23	Paid Employees of Urban Population Aged Ten Years and	
	Above by Sex and Period of Payment –	
	COUNTRY TOTAL: 2012	. 161
Table 5.24	Paid Employees of Urban Population Aged Ten Years and	
	Above by Sex, Period of Payment and Amount of Total Payment –	
	COUNTRY TOTAL: 2012	. 162
Table 5, 25	Paid Employees of Urban Population Aged Ten Years and	
	Above by Major Industrial Divisions, Sex and Amount of	
	Payment per month - COUNTRY TOTAL: 2012	. 163
Table 5.26	Total Employed Population Not at Work by Region, Sex, and	
	Reason for Not Working: 2012	. 165
Table 5.27	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above by Age Group, Sex, Whether Available and Ready to	
	Work Additional Hours and Percent Available and Ready to Work	
	Additional Hours – COUNTRY TOTAL: 2012	. 166
Table 5.28	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above Who Sought or Available to Work Additional Hours and	
	Number of Hours Worked Per Week by Age Group and Sex –	
	COUNTRY TOTAL: 2012	. 167
Table 5.29	Currently Employed Population of Urban Areas Aged Ten Years	
	and Above Who Sought or Available to Work Additional Hours and	
	Type of Work Sought for by Region and Sex: 2012	. 169
Table 5.30	Currently Employed Population of Urban Areas Aged Ten Years and	
	Above by Age Group, Sex, Employment Sector and Percentage Share of Inform	
	COUNTRY TOTAL I: 2012	. 170
Table 5.31	Currently Employed Population of Urban Areas Aged Ten Years and Above	
	by Region, Sex, Employment Sector and Percentage Share of Informal: 2012	. 171
Table 5.32	Currently Employed Populations of Major Towns Aged Ten Years and	
	Above by Sex, Employment Sector and Percentage Share of Informal: 2012	. 172
Table 5.33	Distribution of Urban Population aged ten years and Above Working	
	in the Informal Sector by Major Occupational Group	
	and Region : 2012	. 174
Table 5.34	Distribution of Urban Population aged ten years and Above Working	
	in the Informal Sector by Region, Sex and	
	Major Branches of Industry: 2012	. 176
Table 5.35	Distribution of Urban Population aged ten years and Above Working	
	in the Informal Sector by Region, Sex, Major Branches of Industry and	
	Mean Number of Hours Worked Per Week: 2012	. 177
Table 5.36	Distribution of Urban Population aged ten years and Above Working	
	in the Informal Sector by Region, Sex Literacy Status and Educational Level	

	and: 2012	178
Table 5.37	Distribution of Urban Population Aged ten years and Above Working	
	in the Informal Sector by Region and Sex,	
	Size of Workers: 2012	180
Table 5.38	Distribution of Urban Population Aged ten years and Above Working	
	in the Informal Sector by Region, Sex, Type of	
	Working Place: 2012	181
Table 5.39	Distribution of Urban Population aged ten years and Above Working	
	in the Informal Sector by Regions, Order and	
	Type of Difficulties Faced: 2012	183
Table 6.1	Currently Unemployed Populations of Urban Areas	
	Aged Ten Years and Above by Region, Age Group, Sex and	
	Unemployment Rate: 2012	190
Table 6.2	Currently Unemployed Populations of Major Towns	
	Aged Ten Years and Above by Sex, Age Group and	
	Unemployment Rate: 2012	194
Table 6.3	Currently Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Age group, Sex, Literacy Status and	
	Educational Attainment, COUNTRY TOTAL: 2012	200
Table 6.4	Currently Unemployed Populations of Urban Areas Aged Ten Years	
	and Above by Region, Sex, Literacy Status and	
	Educational Attainment: 2012	202
Table 6.5	Currently Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Region, Sex, Training Status and	
	Types of Training: 2012	204
Table 6.6	Currently Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Age Group, Sex and Marital Status,	
	COUNTRY TOTAL: 2012	213
Table 6.7	Currently Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Region, Sex and Marital Status: 2012	214
Table 6.8	Currently Unemployed Population of Urban Areas Aged Ten Years	
	and Above Who Were Seeking for Work by Region, Sex and Steps	
	Taken to Seek Work: 2012	215
Table 6.9	Currently Unemployed Population of Urban Areas Aged ten Years	
1 410 610	and Above by Region, Sex and Type of job looking for: 2012	217
Table 6.10	Currently Unemployed Population of Urban Areas Aged Ten Years	
14510 0.10	and Above by Age Group, Sex, Previous Work Experience	
	and Duration of Unemployment, COUNTRY TOTAL: 2012	218
Table 6.11	Currently Unemployed Population of Urban Areas Aged Ten Years	
Tuble 0.11	and Above by Region, Sex, Previous Work Experience and	
	Duration of Unemployment: 2012	220
Table 6.12	Currently Unemployed Population of Urban Areas Aged Ten Years	
. anio VIII	and Above by Region, Sex, Types of Problem Faced to	
	Establish Own Business: 2012	222
	LUMPHUH OWIT DUUHIGUU EVIL	

CAHAPTER I

BACKGROUND OF THE SURVEY

1.1 Introduction

Statistical information on all aspects of the population is vital for the design, implementation, monitoring and evaluation of economic and social development plan and policy issues. Labour force survey is one of the most important sources of data for assessing the role of the population of the country in the economic and social development process. It is useful to indicate the extent of available and unutilized human resources that must be absorbed by the national economy to ensure full employment and economic well being of the population. Statistics on the labour force further present the economic activity status and its relationship to other social and economic characteristics of the population. Seasonal and other variations as well as changes over time in the size, distribution and characteristics of employed and unemployed population can be monitored using up-to-date information from labour force surveys. It serves as an input for assessing the achevements of the Millennium Development Goals (MDGs). Furthermore, labour force data is also useful as a springboard for monitoring and evaluation of the five years growth and transformation plan of the country.

Despite the significance of the labour force data, the availability of reliable and timely labour force data were inadequate. The lack of reliable and timely data on different aspects of the population hinders the monitoring and evaluation of changes of developmental activities.

The Central Statistical Agency (CSA) has been providing labour force and related data at different levels with various contents and details in order to fill the gap in data requirement for the purpose of planning, monitoring and evaluation. These include the 1976 Addis Ababa Manpower and Housing Sample Survey, the 1978 Survey on Population and Housing Characteristics of Seventeen Major Towns, the 1980/81 and 1987/88 Rural Labour Force Surveys (RLFS). The 1984, 1994 and 2007 Population

and Housing Censuses and the 1999 and 2005 National Labour Force Surveys provided a comprehensive national labour force data representing both urban and rural areas.

Considering the development process of the country and dynamic nature of the sectors and fulfilling the demands of different data users, the CSA has been launched Urban Employment Unemployment Survey program (UEUS) since 2003. The Agency conducted UEUS in seven rounds. The statistical and analytical report of the October 2003, April 2004, and April 2006 surveys were published together in bulletin number 301, 319, 373, respectively. Whereas the May 2009, May 2010 and March 2011 surveys, the statistical and analytical results were published separately. The statistical report found in bulletin number 469, 499 and 521, while the analytical report in bulletin number 477, 504 and 530, respectively.

The 2012 Urban Employment and Unemployment Survey conducted from March 12 - March 27, 2012 is the seven series. Unlike the previous surveys the May 2009, May 2010, March 2011, and the recent survey of March 2012 survey have collected information from selected major towns with population size 100,000 and above and including region capital cities.

In addition, data on economic activities of children were also collected to measure child labour in urban areas. For this purpose, the former minimum age limit 10 years was lower down to 5 years since May 2009 and the data in each subsequent survey was also collected from those persons aged five years and above. This is because children in rural and urban areas used to work at their early age such as collection of fire wood, looking after cattle, shoeshine, street vendor, petty trading...etc.

The survey primarily provides data on the size, distribution and other sociodemographic characteristics of employed and unemployed population. It also provide data on active population and activity status and employment in the informal sector, and its distribution in the various sectors of the economy during the survey reference period. For the purpose of measuring the economic activity status based on Ethiopian situation, the lower age limit was fixed in to ten years. This statistical report is therefore mainly aimed at provide information on the economic characteristics of the population aged ten years and above. Furthermore, the 2012 UEUS provide data on employment in the informal sector, their spatial distribution and at various sectors.

This statistical report presents a brief introduction of each chapter including basic concepts based on ILO standard definitions together with statistical tables.

1.2. Objectives of the Survey

The general objective of the 2012 Urban Employment and Unemployment Survey is to provide statistical data on the characteristics and size of the economic activity status i.e. employed, unemployed population of the country at urban levels on annual basis.

The specific objectives of the survey are to:

- Ø collect statistical data on the potential manpower and those who are available to take part in various socio-economic activities;
- Ø update the data and determine the size and distribution of the labour force participation and the status of economic activity for different sub-groups of the population at different levels of the country; and also to study the socioeconomic and demographic characteristics of these groups;
- Ø identify the size, distribution and characteristics of employed population i.e. working in the formal or informal employment sector of the economy and earnings from paid employees and its distribution by occupation and Industry...etc;
- Ø provide data on the size, characteristics and distribution of unemployed population and rate of unemployment;
- Ø provide data that can be used to assess the situation of women's employment or the participation of women in the labour force;
- generated time series data to trace changes over time;

1.3. Content and Organization of Survey Questionnaire

The survey questionnaire is organized into seven sections;

Section - 1 Area identification of the selected household: consists of area identification of respondents such as region, zone, wereda, kebele, EAs... etc.

Section - 2 Particulars of household members: consists of questions on the general socio-demographic characteristics of the household members such as age, sex, relationship to the head of household, literacy status, educational levels, status and types of training and marital status.

Section - 3 Economic activity status during the last seven days: this section contains questions to identify whether persons were engaged in productive activities during the last seven days prior to date of interview. It comprises question on the status and characteristics of employed persons such as occupation, industry, status in employment, hours of work, formal and informal sector and earnings from paid employment... etc.

Section - 4 Unemployment rate and characteristics of unemployed persons: comprises questions on seeking and availability for work, reasons for not being activity and type of work looking for and problems encountered to establish own business and duration of unemployment.

Section - 5 Economic activity status the population during the last six months: this section contains questions to gather Information about the economic activity status of the population during the last six months such as number of weeks on employment, unemployment and not being active and reasons for not being active.

Section - 6 Employment in the informal sector of Employment. It includes question whether the enterprise that a person engaged has book account and license, number of workers and whether the product

is mainly produced for sale.

Section - 7

Economic activity of children aged 5-17 years: this section consists of information on the participation of children aged 5-17 years in the economic activities, whether attending education, reason for not attending education, types of injuries whether using protective wear while working and frequency of working periods and for whom they are working.

A structured questionnaire was used to solicit the required data in the survey. The draft questionnaire was tested by undertaking a pretest in selected kebeles (lower administrative unit) in Addis Ababa. Based on the pretest, the content, logical flow, layout and presentation of the questionnaire was amended. The questionnaire used in the field for data collection was prepared in Amharic language. Most questions have pre coded answers and column numbers were assigned for each question. A copy of the questionnaire translated in English language is attached as Annex I of this report for reference.

1.4. Training of Field Staff

The training program of enumerators, supervisors and other field office staff was conducted in two stages. The first stage training of trainers was conducted at the head quarters of the Central Statistical Agency (CSA), in Addis Ababa for five days including half day field practice. The participants were selected from professionals and associate professionals with long time experiences, branch office statisticians and coordinators who were supposed to train enumerators and supervisors during the second stage of training conducted at the branch statistical offices of CSA. The training was guided by an enumerator and supervisor manuals, which consists of detailed explanation of concepts, ideas and instructions on how to fill each entry in the questionnaire.

The second stage training was undertaken at the CSA branch offices. The trainers conducted similar training for enumerators and field supervisors for ten days. The

training consisted of theoretical classroom discussions on concepts, definitions and techniques of completing the questionnaire as well as mock interview. The class room presentation and discussions were intended to exchange experiences among participants and pinpoint the areas of the survey questions that need more care and attention. At the end of the training, one day field practice was carried out to familiarize on the contents and structure of the questionnaire through interviewing of household members. The objectives of mock and practical interviewing of households were twofold. First, it enabled to assess how well the classroom theoretical discussions were understood by all participants to convey the same message at all levels. The second objective was to examine the difficulty, which would likely be encountered during actual field work.

1.5. Organization of the Fieldwork

In order to carry out this survey with the desired level of quality, there was a need to organize a large staff that performs the various survey activities. The 25 Branch Statistical Offices of the Agency were responsible to carry out the data collection operation. These offices have permanent and contract enumerators stationed in the selected enumeration areas. The data collection operation of the survey involved a total of about 696 enumerators and 168 field supervisors with an average supervisor to enumerator ratio of 1: 4.

The interviews have been done by canvassing from house-to-house. Information was collected directly from the head of the household or from other responsible household member. The enumerators were assigned to one sampled enumeration area, where he/she could easily converse with the respondent's dialect. However, in some cases whenever enumerator faces difficulty to converse with the respondent's dialect interpreters were used to facilitate the data collection. Data collection from the field took place from 12 March – 27 March 2012.

During the survey data collection operation, close and regular supervision was undertaken at various levels. Immediately after the commencement of the data

collection exercise the supervisors had made spot checking, re-interviewing and a thorough scrutiny of filled-in questionnaires to ensure that the data collection activities are taking place according to the given instruction. The branch statistical office heads, coordinators and statisticians were also made supervision of the data collection operation. Moreover, senior staff members from the CSA head office made field supervision activities in seven branches for a week at different towns.

1.6. Data Processing

The filled-in questionnaires that were retrieved from the field were first subjected to manual editing and coding. During the fieldwork, field supervisors and statisticians of the head and branch statistical offices have checked the filled-in questionnaires and carried out some editing. However, the major editing and coding operation was carried out at the head office. All the edited questionnaires were again fully verified and checked for consistency before they were submitted to the data entry by the subject matter experts.

Using the computer edit specifications prepared earlier for this purpose, the entered data were checked for consistencies and then computer editing or data cleaning was made by referring back to the filled-in questionnaire. This is an important part of data processing operation to maintain the quality of the data. Consistency checks and rechecks were also made based on frequency and tabulation results. This was done by senior programmers using CSPro software in collaboration with the senior subject matter experts from Manpower Statistics Team of the CSA.

1.7. Definitions of Concepts and Terms

Some of the major operational definitions of basic concepts are given below. Moreover, other concepts and definitions related to the economic activity status are provided under respective chapters.

<u>Urban Center:</u> is defined as a locality with 2000 or more inhabitants. In this survey, however, for practical purposes an urban center includes the following regardless of the number of inhabitants.

- a) All administrative capitals
 - i) Regional capitals
 - ii) Zonal capitals not included in (i)
 - iii) Wereda capitals not included in (i) and (ii)
 - iv) Localities with Urban Dweller's Association (UDAs) not included in (I-iii).
- b) Municipal town not included in item (a) above.
- c) All localities which are not included either in item (a) or (b) above, having a population of 1000 or more persons, and whose inhabitants are primarily engaged in non-agricultural activities. Note that localities with population less than 1000 persons should be considered as rural.

<u>Major Urban Centers</u>: For the purpose of this study, major urban center include those selected urban centers with population 100,000 and above. It also includes region capitals irrespective of their population size.

<u>Urban Kebele</u>: is the lowest administrative unit in an urban center with its own jurisdiction. It is an association of urban dwellers (commonly known as kebele) formed by the inhabitants.

Enumeration Area (EA): is a unit of land delineated for the purpose of enumerating population and housing units without omission and duplication. An EA in rural areas usually consists of 150-200 households. On the other hand, an EA in urban areas constitutes 150-200 housing units.

<u>Household:</u> Consists of a person or group of persons, irrespective of whether related or not, who normally live together in the same household and housing units and have common cooking and eating arrangements.

<u>Housing Unit</u>: is defined as a separate and independent place of abode either intended for habitation or not intended for habitation but was occupied as a living quarter by a household during the survey period. Although intended for habitation by one household, a housing unit may, at the survey be occupied by one or more households or may be used partly for living and partly for establishment.

Head of Household: is a person who provides economic supports or manages the households. The head of the household is selected by household members for some reasons of his age or respect regardless of their sex.

<u>Usual Member of a Household</u>: a person is considered as usual members of a household if he or she is:-

- a) a person who continuously live/reside at least for six months and have a common cooking and eating arrangements with the household; or
- b) a person who is temporarily absent from the household at the time of the survey but his absence has not violated the six months criterion.
- c) House maids, guards, baby sitters, etc...with no other dwelling and who were staying with the household at the time of the survey.
- d) Persons who plan to live more than six months due to searching for job and transfer from job ...etc.

1.8. Organization of the Report

This statistical report contains seven chapters. The first chapter covers background and objectives of the survey, where the objectives and the contents of the survey questionnaire, training of field staff, organization of field work, concepts and definitions of terms and concepts. Chapter II presents methodology of the survey including scope, coverage, sample size, covered households and response rate. Chapter III describes briefly the socio-demographic characteristics of the urban population such as the size and distribution of urban population by age group and sex. Chapter IV presents data on the economic activity status of the population aged ten years and above using the current status approach. Chapter V explores the size, characteristics and distribution of the employed population aged ten years and above. Chapter VI presents the size, characteristics and distribution of the unemployed population and unemployment rate aged ten years and above.

Annex I provides survey questionnaire English version, while estimation procedures of total, ratio and sampling errors; and estimates and CV's are also presented in Annex II and Annex III, respectively.

1.9 ABBEREVATIONS

CSA Central Statistical Agency

EA Enumeration Area

HH Households

ICSE International classification of Status in Employment

ILO International Labour Organization

CSPro Census and Survey Processing System

ISCO International Standard Classification of Occupation

ISIC International Standard Industrial Classification of All Economic Activities

ICLS International Conference on Labour Statistics

KILM Key Indicators of Labour Market
MDGs Millennium Development Goals

NIHSP National Integrated Household Survey Program

NLFS National Labour Force Survey

NS Not Stated

NOIC National Occupation and Industrial Classification

PASDEP Plan for Accelerated and Sustained Development to End Poverty

PSU Primary Sampling Unit

RLFS Rural Labour Force Survey
SNA System of National Account

SNNPR Southern Nations, Nationalities and Peoples Region

TVET Technical Vocational and Educational Training
UEUS Urban Employment and Unemployment Survey

UBEUS Urban Bi-annual Employment and Unemployment Survey

CHAPTER II

SURVEY METHODOLOGY

2.1 COVERAGE

The 2012 Urban Employment and Unemployment Survey (UEUS) covered all urban parts of the country except three zones of Afar, Six zones of Somali, where the residents are pastoralists. This survey follows household approach and covers households residing in conventional households, thus population residing in the collective quarters such as universities/colleges, hotel/hostel, monasteries and homeless population etc are not covered by this survey.

It was initially planned to cover 660 EAs and 19,800 households in the survey, but ultimately 100% of EAs and 99.86% of households were successfully covered.

2.2 SAMPLING FRAME

The list of households obtained from the 2007 population and housing census is used to select EAs. A fresh list of households from each EA was prepared at the beginning of the survey period. The list was then used as a frame in order to select 30 households from sample EAs.

2.3 SAMPLE DESIGN

For the purpose of this survey, the country was divided into two broad categories. That is major urban center and other urban center categories.

Category I:- Major urban centers:- In this category all regional capitals and five other major urban centers that have had a high population size as compared to others were included. Each urban center in this category was considered as a reporting level. The category has a total of 16 reporting levels. In this category, in order to select the sample, a stratified two-stage cluster sample design was implemented. The primary sampling units were EAs of each reporting level. From each sample EA 30 households were then selected as a Second Stage Unit (SSU).

Category II: - Other urban centers: Urban centers in the country other than those under category I were grouped into this category. A domain of other urban centers is formed for each region. Consequently 8 reporting levels were formed in this category. Harari, Addis Ababa and Dire Dawa do not have urban centers other than that grouped in category I. Hence, no domain was formed for these regions under this category.

A stratified three stage cluster sample design was also adopted to select samples from this category. The primary sampling units were urban centers and the second stage sampling units were EAs. From each EA 30 households were selected at the third stage and the survey questionnaires administered for all of them.

2.4 SAMPLE SIZE AND SELECTION SCHEME

Category I: - In this category 393 EAs and 11,790 households were selected from 16 major urban centers. Sample EAs from each reporting level in this category were selected using probability proportional to size /PPS/ with systematic sampling method; size being number of households obtained from the 2007 population and housing census. From the fresh list of households prepared at the beginning of the survey, 30 households per EA were selected using systematic random sampling techniques /SRS/.

Category II:- consists of 86 urban centers, 267 EAs and 8,010 households were selected in this category. Urban centers from each domain and EAs from each urban center were selected using probability proportional to size /PPS/ with systematic sampling method; size being number of households obtained from the 2007 Population and housing census. From the listing of households in each EA then 30 households were selected using systematic random sampling techniques /SRS/ and the survey questionnaire was administered for the 30 sample households.

The distribution of planned and covered EAs and households and the Estimation procedures are given in the appendix.

Appendix I: Number of planned and actually covered sampling units

(EAs and households) of the 2012 (2004 E.C) Urban Employment and Unemployment Survey (UEUS).

Region	Stratum	Enumera	tion Areas	Households		
	Major Urban/	Planned	Covered	Planned	Covered	
	Other Urban					
Tigray	Mekele	23	23	690	690	
	Tigray other urban	28	28	840	840	
Affar	Asayita	15	15	450	450	
	Affar other urban	14	14	420	420	
Amhara	Bahir dar	23	23	690	690	
	Gonder	23	23	690	689	
	Dessie	23	23	690	685	
	Amhara other urban	54	54	1620	1619	
Oromiya	Debrezeit/Bishoftu/	22	22	660	660	
	Nazreth/Adama/	23	23	690	690	
	Jimma	23	23	690	690	
	Shashmene	23	23	690	688	
	Oromiya other urban	83	83	2490	2486	
Somalie	Jijiga	20	20	600	599	
	Somalie other urban	15	15	450	450	
Benishangul -	Asosa	15	15	450	450	
Gumuz	Benishangul -Gumuz	13	13	390	390	
	other urban					
S.N.N.P.	Awassa	23	23	690	690	
	S.N.N.P other urban	51	51	1530	1530	
Gambela	Gambella	14	14	420	447	
	Gambella other urban	9	9	270	239	
Harari	Hareri	23	23	690	688	
Addis Ababa	Addis ababa	75	75	2250	2247	
Dire Dawa	Diredawa	25	25	750	745	
Total		660	660	19,800	19,772	



CHAPTER III

SOCIO-DEMOGRAPHIC CHARACTERISTICS OF URBAN POPULATION

3.1 Introduction

The figure in this chapter refers to the dejure population residing in conventional households. The dejure population comprises all persons who belong to a given area at a given time by virtue of usual residence. Thus, visitors, persons residing in collective quarters (hotel/hostel, boarding schools, prisons...etc.) as well as homeless persons are excluded.

This chapter presents the basic socio-demographic characteristics of urban population of the country which includes population size; distribution of the population by age and sex, sex ratio, literacy status, , attending formal or informal education, educational attainment, training status and type of training.

3.2. Age-Sex Structure of Urban Population and Other Socio-Demographic Data

This section portrays a set of tables that depict the main socio-demographic characteristics of the urban population as of 2012. The size and distribution of estimated total population of urban areas by age group, sex and region is shown in Table 3.1. The total urban population of the country estimated to be 14,011,269 in March 2012. Table 3.2 presents the distribution of urban population, number of households and average household size by region. The average household size is computed from the total population to that of the total households. Data on household size and distribution of persons per household can sometimes be used as a proxy measure of crowdedness of population and has a great implication to health and housing problem. The distribution of urban population by region, sex and the relationship of the respondents to the head of household is presented in Table 3.3.

Table 3.4 and Table 3.5 show the status of school attendance for those aged five years and above by age group and sex at national and regional levels. The literacy status and educational attainment of urban population aged ten years and above has shown in Table 3.6 and Table 3.7, respectively. As indicated in these tables secondary education refers to those persons who did not complete high school education, that is, grades 9 - 11 in old curriculum and grade 9 in new curriculum, while secondary education completed covers those persons who completed high school education which includes grade 12 in old curriculum and grade 10 in new curriculum.

The 2012 Urban Employment and Unemployment Survey also collected data on status and types of training of urban population aged ten years and above. Tables 3.8 demonstrate the status and type of training received by region and sex.

Besides, the distribution of urban population by five-year age group, sex and sex ratio is presented in Table 3.9. Sex ratio is defined as the number of males per 100 females in a given population. The sex composition has an effect on economic activities through changing the relative size of the working population.

Table 3.2 Distribution of Urban Population, Number of Households and Average Household Size by Region

Region, Population and					Hou	sehold Size		
Households		_						
Tiodoctiona	Total	1	2	3	4	5	6	7
COUNTRY TOTAL					1			
Households	3,840,602	661,419	676,954	694,021	620,803	473,059	324,501	183,719
Population	14,011,269	661,419	1,353,907	2,082,063	2,483,212	2,365,295	1,947,009	1,286,031
TIGRAY REGION								
Households	301,059	52,748	54,011	60,526	49,515	39,927	24,481	9,235
Population	1,043,639	52,748	108,022	181,577	198,062	199,636	146,886	64,642
AFFAR REGION								
Households	51,583	9,684	9,688	11,103	7,963	5,710	4,398	1,966
Population	172,140	9,684	19,375	33,308	31,854	28,550	26,391	13,761
AMHARA REGION								
Households	748,408	138,524	164,075	149,208	110,275	81,573	55,161	27,196
Population	2,474,466	138,524	328,149	447,625	441,099	407,865	330,968	190,369
OROMIA REGION								
Households	1,116,894	196,700	197,544	192,023	195,212	135,643	93,354	53,207
Population	4,031,092	196,700	395,088	576,070	780,848	678,217	560,124	372,452
SOMALI REGION								
Households	146,377	20,346	14,807	20,327	16,842	20,733	18,237	13,798
Population	682,683	20,346	29,613	60,981	67,366	103,667	109,423	96,586
BENISHANGUL GUMUZ								
REGION								
Households	41,986	8,532	9,079	7,103	6,593	4,001	2,859	1,753
Population	141,476	8,532	18,157	21,310	26,373	20,005	17,154	12,268
SNNP REGION								
Households	520,587	103,520	84,800	80,016	77,232	61,848	49,315	30,156
Population	1,936,997	103,520	169,600	240,049	308,930	309,242	295,891	211,094
GAMBELLA REGION								
Households	27,915	4,534	4,769	5,070	5,082	3,253	2,035	1,296
Population	104,449	4,534	9,538	15,211	20,327	16,265	12,208	9,074
HARARI REGION								
Households	33,646	7,444	6,226	6,334	5,819	4,190	2,033	877
Population	107,802	7,444	12,452	19,003	23,277	20,952	12,199	6,138
ADDIS ABABA CITY								
ADMINISTRATION								
Households	783,439	106,566	120,554	149,072	136,491	108,313	67,098	41,341
Population	3,061,404	106,566	241,107	447,215	545,964	541,565	402,585	289,384
DIRE DAWA								
ADMINISTRATION								
Households	68,708	12,820	11,402	13,238	9,778	7,866	5,530	2,895
Population	255,121	12,820	22,804	39,713	39,112	39,332	33,180	20,264



Table 3.3 Distribution of Urban Population by Region, Sex and Relationship to the Head of Household: 201

	Relationship to the Head of Household								
Region and Sex				Son /		_	Mother /	Brothe	
rtogion and cox				Daughter of	Son /	Son /	Father of	Sister	
		Head of		head and	Daughter of	Daughter of	head or	Head	
	Total	Household	Spouse	Spouse	Head	Spouse	Spouse	Spous	
COUNTRY TOTA	AL.								
Total	14,011,269	3,875,123	1,893,142	3,988,775	1,986,143	86,770	111,630	515	
Male	6,689,363	2,476,596	151,684	2,103,310	1,040,647	43,173	18,996	240	
Female	7,321,906	1,398,527	1,741,458	1,885,464	945,496	43,597	92,634	275	
TIGRAY REGION	1								
Total	1,043,639	314,451	127,715	269,841	188,618	7,732	9,469	30	
Male	470,348	178,361	6,639	129,654	94,381	3,143	1,642	13	
Female	573,291	136,091	121,075	140,187	94,237	4,590	7,827	16	
AFFAR REGION									
Total	172,140	52,318	29,125	57,373	15,904	967	1,331	3	
Male	84,334	36,885	1,458	28,961	9,311	560	180	1	
Female	87,805	15,433	27,667	28,413	6,594	407	1,152	1	
AMHARA REGIC	N								
Total	2,474,466	751,588	341,751	648,526	364,247	22,532	23,631	78	
Male	1,151,457	443,437	21,747	345,291	190,270	12,678	3,436	36	
Female	1,323,009	308,151	320,004	303,234	173,976	9,854	20,196	42	
OROMIA REGIO	N								
Total	4,031,092	1,129,529	577,998	1,198,776	527,728	18,028	28,620	121	
Male	1,994,569	785,222	28,092	649,853	277,228	9,706	4,353	57	
Female	2,036,523	344,307	549,906	548,923	250,500	8,322	24,267	64	
SOMALI REGION	N	•	•	•	,	,	•		
Total	682,683	147,059	83,769	279,448	113,679	1,317	5,433	14	
Male	344,567	94,459	10,463	151,218	59,024	403	375	7	
Female	338,115	52,599	73,305	128,229	54,655	913	5,058	7	

Table 3.3 Cont'd

u								
Relationship to the Head of Household								
			Son /			·		
			Daughter	_		Mother /	Brother /	
			of Head		,	Father of	Sister of	
			and	•	0	Head or	Head /	
		Spouse	Spouse	Head	Spouse	Spouse	Spouse	丄
141,476	42,070	22,467	41,021	16,153	1,451	1,028	4,730	
70,461	30,189	841	21,233	8,064	1,011	175	2,560	
71,015	11,882	21,627	19,788	8,089	439	852	2,170	
1,936,997	521,307	269,619	595,942	233,894	8,478	7,955	83,483	
955,050	371,936	11,162	309,224	129,408	3,776	1,206	37,043	
981,947	149,371	258,457	286,718	104,485	4,703	6,749	46,440	
ION								
104,449	27,999	12,559	26,497	23,020	458	1,102	3,775	
48,403	16,496	948	12,571	11,997	209	200	1,591	
56,046	11,503	11,611	13,926	11,023	249	901	2,184	
1								
107,802	33,748	15,828	24,861	16,587	593	1,054	3,186	
52,142	20,787	1,912	13,474	9,519	252	-	1,427	
55,660	12,961	13,916	11,387	7,068	342	1,054	1,760	
•		•	•	•		•	•	
3,061,404	786,259	379,316	783,073	445,779	23,841	28,931	160,800	
1,395,378	457,381	64,680	407,970	230,002	10,911	6,715	76,197	
1,666,026	328,878	314,637	375,102	215,777	12,930	22,217	84,602	
/INISTRATI	ON							
255,121	68,794	32,995	63,418	40,533	1,372	3,076	11,939	
122,652	41,443	3,741	33,862	21,441	523	714	6,061	
132,468	27,351	29,254	29,556	19,092	849	2,362	5,878	
	Total GUMUZ RE 141,476 70,461 71,015 1,936,997 955,050 981,947 HON 104,449 48,403 56,046 1 107,802 52,142 55,660 ITY ADMINI 3,061,404 1,395,378 1,666,026 INISTRATI 255,121 122,652	Head of Total Household GUMUZ REGION 141,476 42,070 70,461 30,189 71,015 11,882 1,936,997 521,307 955,050 371,936 981,947 149,371 HON 104,449 27,999 48,403 16,496 56,046 11,503 107,802 33,748 52,142 20,787 55,660 12,961 ITY ADMINISTRATION 3,061,404 786,259 1,395,378 457,381 1,666,026 328,878 MINISTRATION 255,121 68,794 122,652 41,443	Head of Total Household Spouse GUMUZ REGION 141,476	Head of Total Household Spouse Spouse GUMUZ REGION 141,476	Relationshi Head of Total Household Spouse Spouse Spouse Spouse Head	Relationship to the Head of Daughter of Head and Spouse Spouse Spouse Head Spouse Spouse Spouse Head Spouse	Relationship to the Head of Household	Relationship to the Head of Household

Table 3.4 Distribution of Urban Population Aged Five Years and Above by Age Group, Sex and School Attendance - COUNTRY TOTAL: 2012

	Sex and School				
		Attended Fo	rmal / Informal E	ducation	
Age group		Currently	Attended in	Never	
	Total	Attending	the Past	Attended	Not Stated
All Ages			·		
Total	12,553,784	5,002,762	5,288,023	2,262,490	509
Male	5,906,542	2,428,935	2,842,619	634,479	509
Female	6,647,242	2,573,827	2,445,404	1,628,011	-
5-9	-, ,	_,	_, ,	.,,-	
Total	1,443,131	1,142,657	35,412	264,587	475
Male	720,528	574,538	14,954	130,561	475
Female	722,603	568,119	20,458	134,026	-
10-14	722,000	000,110	20,400	104,020	
Total	1,592,162	1,460,316	72,720	59,126	_
Male	754,662	703,130	29,636	21,895	_
Female	837,500	757,185	43,084	37,231	_
15-19	037,300	737,103	45,004	37,231	_
Total	1,752,372	1,266,725	394,452	91,195	
					_
Male	754,648	571,764	161,784	21,099	-
Female	997,725	694,960	232,669	70,096	-
20-24	4 707 444	570 540	000 400	474 004	0.4
Total	1,707,114	573,548	962,469	171,064	34
Male .	743,495	267,748	438,266	37,447	34
Female	963,619	305,800	524,203	133,617	-
25-29					
Total	1,615,015	302,780	1,092,322	219,912	-
Male	776,270	157,604	571,144	47,522	-
Female	838,745	145,176	521,178	172,391	-
30-34					
Total	1,006,702	109,148	713,654	183,900	-
Male	507,720	59,288	401,965	46,467	-
Female	498,982	49,860	311,688	137,433	-
35-39					
Total	945,870	65,982	678,036	201,852	-
Male	465,729	38,924	372,367	54,439	-
Female	480,140	27,059	305,669	147,413	-
40-44					
Total	568,878	35,702	397,666	135,510	-
Male	299,009	24,450	240,005	34,554	-
Female	269,869	11,252	157,661	100,956	-
45-49	,	,	•	,	
Total	477,925	17,750	321,543	138,632	-
Male	233,108	11,850	195,338	25,920	-
Female	244,817	5,900	126,205	112,712	_
50-54	, •	0,000	0,_00	,	
Total	381,120	9,698	221,847	149,575	_
Male	176,523	6,702	135,472	34,349	_
Female	204,597	2,997	86,375	115,226	_
55-59	204,337	2,557	00,070	110,220	
Total	269,995	5,324	136,038	128,633	_
Male	118,356	3,133	90,049	25,174	_
					-
Female	151,640	2,191	45,990	103,459	-
60-64 Total	220.240	E 006	00 702	124 440	
Total	239,219	5,986	98,793	134,440	-
Male	103,804	4,071	65,201	34,532	-
Female	135,416	1,915	33,593	99,909	-
65+	EE 4 000	7 4 4 0	400.070	004 000	
Total	554,280	7,146	163,070	384,063	-
Male	252,690	5,732	126,439	120,519	-
Female	301,589	1,414	36,631	263,544	-



Table 3.5 Distribution of Urban Population Aged Five Years and above by Region, Sex, School Attendance: 2012

rtog.o.,	Jex, School		ormal / Informa	al Education	
Region and Sex		Currently	Attended in	Never	
region and ocx	Total	Attending	the Past	Attended	Not Stated
COUNTRY TOTAL	Total	Alterialing	tile i ast	Allended	1401 Olaica
Total	12,553,784	5,002,762	5,288,023	2,262,490	509
Male	5,906,542	2,428,935	2,842,619	634,479	509
Female	6,647,242	2,426,933	2,445,404	1,628,011	509
TIGRAY REGION	0,047,242	2,373,027	2,445,404	1,020,011	-
	917,333	364,092	388,110	165,130	
Total Male	401,155				-
Female	,	171,543	190,040	39,571	-
AFFAR REGION	516,178	192,549	198,069	125,559	-
	1 17 770	40 200	E4 920	44 624	
Total Male	147,773 70,472	48,308 23,246	54,830	44,634 14,862	-
			32,364		-
Female	77,301	25,063	22,467	29,772	-
AMHARA REGION	0.006.004	004.054	000 504	E27 026	
Total	2,236,891	801,351	898,504	537,036	-
Male	1,019,052	387,845	469,630	161,577	-
Female	1,217,839	413,506	428,874	375,459	-
OROMIA REGION	0.500.000	4 005 050	4 004 474	F04 700	500
Total	3,569,666	1,385,959	1,601,471	581,728	509
Male	1,737,368	693,654	881,074	162,131	509
Female	1,832,298	692,305	720,397	419,596	-
SOMALI REGION	550.074	044.070	400.040	000.050	
Total	558,971	241,078	109,240	208,653	-
Male	278,201	134,813	68,155	75,233	-
Female	280,770	106,266	41,085	133,419	-
BENISHANGUL					
GUMUZ REGION	407 504	40.404	10.010	00.440	
Total	127,521	49,491	48,619	29,412	-
Male .	63,355	24,051	28,540	10,764	-
Female	64,166	25,439	20,079	18,648	-
SNNP REGION					
Total	1,739,338	742,017	734,239	263,081	-
Male	855,181	363,435	412,131	79,616	-
Female	884,156	378,582	322,109	183,465	-
GAMBELLA REGION					
Total	89,710	46,276	23,722	19,712	-
Male	41,753	22,298	13,620	5,835	-
Female	47,958	23,978	10,103	13,878	-
HARARI REGION					
Total	97,553	33,956	49,639	13,958	-
Male	46,386	17,345	26,110	2,931	-
Female	51,167	16,611	23,529	11,027	-
ADDIS ABABA CITY					
ADMINISTRATION					
Total	2,837,461	1,202,356	1,280,316	354,789	-
Male	1,283,318	542,590	668,975	71,753	-
Female	1,554,143	659,766	611,341	283,036	-
DIRE DAWA					
ADMINISTRATION					
Total	231,567	87,877	99,334	44,356	-
Male	110,302	48,115	51,982	10,205	-
Female	121,266	39,762	47,352	34,151	-

Table 3.6 Distribution of Urban Population Aged Ten Years and Above by Region, Sex and Literacy Status:

	Literacy Status										
Region		Total			Literate	Illiterate					
	Total	Male	Female	Total	Male	Female	Total	Male			
COUNTRY TOTAL	11,110,653	5,186,014	5,924,639	8,912,226	4,636,559	4,275,667	2,197,510	549,421			
TIGRAY REGION	794,578	344,053	450,525	632,643	312,560	320,084	161,934	31,493			
AFFAR REGION	127,674	61,116	66,558	80,560	45,152	35,408	47,114	15,964			
AMHARA REGION	1,976,070	890,049	1,086,021	1,471,421	759,642	711,779	504,649	130,407			
OROMIA REGION	3,131,966	1,522,509	1,609,457	2,536,043	1,367,468	1,168,575	595,889	155,007			
SOMALI REGION	433,866	214,968	218,898	264,971	162,047	102,924	168,895	52,921			
BENISHANGUL- GUMUZ REGION	111,784	55,116	56,668	83,276	45,733	37,543	28,508	9,383			
SNNP REGION	1,531,368	743,394	787,974	1,277,087	674,361	602,726	254,281	69,033			
GAMBELLA REGION	75,144	34,340	40,804	59,447	30,477	28,969	15,697	3,863			
HARARI REGION	87,661	41,096	46,566	72,979	38,309	34,670	14,682	2,787			
ADDIS ABABA CITY ADMINISTRATION	2,636,013	1,183,148	1,452,865	2,278,116	1,115,141	1,162,975	357,014	68,007			
DIRE DAWA ADMINISTRATION	204,530	96,225	108,305	155,683	85,669	70,014	48,847	10,556			

Table 3.7 Distribution of Urban Population Aged 10 Years and above by Age group, Sex and Education COUNTRY TOTAL: 2012

COUNTRY TOTAL: 2012											
	Highest Grade Completed										
Region and Sex					Secondary						
		Non-	Pre-	Grades	not	Secondary		Certificat			
	Total	Formal	school	1-8	completed	Completed	Preparatory	Complete			
All Ages											
Total	9,113,039	138,194	71,163	4,756,708	891,056	1,444,551	426,730	213,366			
Male	4,682,386	75,011	26,076	2,291,250	478,942	780,760	216,212	88,472			
Female	4,430,653	63,184	45,088	2,465,458	412,115	663,791	210,517	124,894			
10-14											
Total	1,533,359	4,024	32,844	1,470,602	21,334	2,240	291	-			
Male	733,090	3,502	13,919	703,766	9,698	99	168	-			
Female	800,269	522	18,925	766,835	11,636	2,141	122	-			
15-19											
Total	1,661,177	2,518	6,136	965,566	293,184	231,801	117,121	26,053			
Male	733,548	1,548	1,157	412,404	140,146	108,742	53,105	8,916			
Female	927,629	970	4,979	553,161	153,038	123,059	64,016	17,137			
20-24											
Total	1,536,016	1,932	4,034	523,993	143,462	375,684	163,704	60,010			
Male	706,014	1,247	1,015	216,568	76,637	181,269	82,891	20,201			
Female	830,002	685	3,019	307,425	66,825	194,415	80,813	39,809			
25-29											
Total	1,395,103	5,779	6,177	525,834	129,704	275,026	81,916	56,495			
Male	728,748	3,708	893	259,090	67,415	144,810	41,367	20,368			
Female	666,354	2,071	5,283	266,744	62,289	130,216	40,549	36,127			
30-34											
Total	822,802	5,359	1,347	335,425	95,098	174,503	25,746	23,107			
Male	461,253	3,449	579	174,418	51,958	102,772	15,053	10,369			
Female	361,549	1,910	768	161,007	43,140	71,731	10,694	12,738			
35-39											
Total	744,018	16,822	5,133	309,290	82,381	149,565	14,431	16,847			
Male	411,291	8,853	2,387	159,577	48,334	81,041	7,969	7,125			
Female	332,728	7,969	2,745	149,713	34,046	68,524	6,462	9,721			
40-44	, -	•	,	, -	•	,	•	•			
Total	433,368	12,929	1,747	166,788	44,904	92,520	8,542	9,733			
Male	264,455	6,420	37	90,397	27,247	58,390	6,028	5,759			
Female	168,913	6,509	1,710	76,391	17,657	34,130	2,515	3,974			
1 Officio	100,010	0,000	1,7.13	70,001	17,007	01,100	2,010	0,014			

Table 3.7 Cont'd

	Highest Grade Completed											
Region and Sex					Secondary	l						
		Non-	Pre-	Grades	not	Secondary	I	Certificate	Di			
	Total	Formal	school	1-8	completed	Completed	Preparatory	Completed	Cor			
45-49												
Total	339,293	15,056	2,928	141,251	35,272	61,475	6,573	6,818	43			
Male	207,188	6,742	1,113	74,755	22,588	41,478	3,335	4,168	29			
Female	132,105	8,314	1,815	66,495	12,684	19,997	3,238	2,650	13			
50-54												
Total	231,546	17,239	3,133	102,680	22,491	31,349	2,801	5,514	24			
Male	142,174	6,279	1,134	54,612	16,171	22,395	1,314	4,202	17			
Female	89,371	10,960	1,999	48,069	6,320	8,954	1,487	1,312	7			
55-59												
Total	141,362	12,472	1,302	71,430	7,757	22,641	3,101	4,322	10			
Male	93,181	5,322	525	41,540	7,115	18,000	2,755	2,944	8			
Female	48,181	7,150	777	29,891	643	4,641	346	1,378	2			
60-64												
Total	104,779	13,881	1,173	55,404	6,330	11,674	751	1,817	9			
Male	69,272	7,253	893	34,228	4,161	8,708	751	1,770	7			
Female	35,507	6,628	280	21,176	2,169	2,966	-	46	1			
65+												
Total	170,217	30,183	5,212	88,446	9,140	16,073	1,753	2,651	9			
Male	132,171	20,688	2,424	69,894	7,473	13,055	1,477	2,651	9			
Female	38,045	9,495	2,787	18,552	1,668	3,017	277	-				

Table 3.8 Distribution of Urban Population Aged Ten Years and Above by Region, Sex, Training Status and Ty

Training status and		Region									
Types of Training							Benishangul -				
Types of Trailing	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela		
All Persons											
Total	11,110,653	794,578	127 674	1,976,070	3,131,966	433,866	111,784	1,531,368	75,144		
Male	5,186,014	344,053	61,116	890,049	1,522,509	214,968	55,116	743,394	34,340		
Female	5,924,639	450,525	66,558	1,086,021	1,609,457	218,898	56,668	787,974	40,804		
Not Trained	0,024,000	400,020	00,000	1,000,021	1,000,407	210,000	30,000	707,574	40,004		
Total	8,840,548	623,416	116,663	1,619,508	2,553,481	392,923	92,616	1,239,127	64,278		
Male	3,770,739	244,285	52,919	669,140	1,153,402	184,252	41,733	553,122	26,804		
Female	5,069,809	379,131	63,744	950,368	1,400,079	208,672	50,884	686,005	37,475		
Trained	3,003,003	373,131	05,7 44	330,300	1,700,073	200,012	30,004	000,000	31, 4 13		
Total	2,269,772	171,162	10,867	356,407	578,450	40,943	19,167	292,241	10,866		
Male	1,415,131	99,768	8,054	220,909	369,107	30,716	13,384	190,272	7,537		
Female	854,641	71,394	2,813	135,499	209,343	10,226	5,784	101,969	3,329		
Not Stated	001,011	11,001	2,010	100,100	200,010	.0,220	0,701	101,000	0,020		
Total	333	-	144	154	35	-	-	-	-		
Male	144	-	144	-	_	-	-	-	-		
Female	189	-	-	154	35	-	-	-	-		
Types of Training											
Agriculture	9										
Total	25,873	1,391	615	4,214	6,697	240	520	6,051	191		
Male	22,181	1,068	415	3,371	6,509	240	520	5,108	147		
Female	3,692	323	200	842	189	-	-	943	44		
Water and Soil Conserv				•							
Total	3,752	251	-	780	1,123	50	105	847	249		
Male	3,093	251	-	511	1,047	-	86	604	249		
Female	659	-	-	269	76	50	20	244	_		
Forestry Science											
Total	1,807	-	-	450	437	-	-	871	-		
Male	1,357	-	-	450	437	-	-	421	-		
Female	450	-	-	-	-	-	-	450	-		
General medicine, Ophi	thalmology Path	nology, Inte	ernal med	icine, Gyned	cology, Surg	ery, Anest	hesiology				
Total	6,697	327	176	1,118	443	365	55	1,186	31		
Male	5,463	218	176	965	384	314	55	670	31		
Female	1,234	109	-	154	59	51	-	516	-		

Table 3.8 Cont'd

Training Status and		Region								
Training Status and Types of Training							Benishangul -			
Types of Trailing	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela	
Medical Laboratory/Rac	liology									
Total	5,928	332	24	1,308	745	-	-	1,110	-	
Male	4,111	332	24	1,181	580	-	-	624	-	
Female	1,816	-	-	126	165	-	-	487	-	
Pharmacy/Pharmacy Te										
Total	12,406	433	344	707	1,910	157	49	1,456	27	
Male	7,978	324	344	502	1,606	157	49	1,178	27	
Female	4,428	109	-	205	304	-	-	277	-	
Sanitary Science										
Total	1,088	-	-	1,037	-	52	-	-	-	
Male	472	-	-	420	-	52	-	-	-	
Female	617	-	-	617	-	-	-	-	-	
Clinical Science/Nursing										
Total	25,887	1,537	24	5,676	5,084	517	433	3,902	66	
Male	12,447	254	12	3,898	3,793	153	268	2,000	66	
Female	13,440	1,283	12	1,778	1,291	364	165	1,902	-	
Public health Nurse, He										
Total	12,720	888	12	4,400	2,332	101	314	2,548	44	
Male	7,674	888	12	2,203	1,567	-	196	1,870	44	
Female	5,046	-	-	2,197	765	101	117	678	-	
Personal business man	•									
Total	79,951	5,737	251	9,643	18,525	938	1,161	13,103	230	
Male	59,422	3,340	251	7,253	15,823	887	949	10,165	114	
Female	20,529	2,398	-	2,390	2,701	50	211	2,938	116	
Economics										
Total	17,321	714	168	2,889	3,448	101	180	2,253	-	
Male	12,582	463	168	2,458	3,210	101	118	1,587	-	
Female	4,739	251	-	431	238	-	62	667	-	
Book Keeping/Accounting	-									
Total	91,718	5,656	131	7,450	20,375	616	511	16,517	232	
Male	51,175	2,483	72	4,951	13,142	465	464	9,313	144	
Female	40,542	3,173	60	2,500	7,233	151	47	7,204	87	

Table 3.8 Cont'd

Training Status and						R	egion		
Types of Training							Benishangul -		
Typoo or Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Secretarial Science									
Total	18,510	900	-	2,123	2,798	50	23	1,690	-
Male	2,512	341	-	75	481	-	-	289	-
Female	15,998	559	-	2,048	2,318	50	23	1,400	-
Banking and Finance									
Total	9,257	-	-	2,354	1,175	-	89	1,305	-
Male	6,195	-	-	1,859	1,085	-	89	1,305	-
Female	3,062	-	-	495	90	-	-	-	-
Law									
Total	28,922	2,518	36	4,162	7,641	159	66	4,857	245
Male	23,943	1,992	36	3,849	7,210	159	66	4,021	217
Female	4,979	526	-	314	430	-	-	836	28
Language and Literature	(Ethiopian an	d Foreign)							
Total	26,559	333	179	7,558	8,856	1,145	239	1,861	80
Male	16,570	224	168	5,588	6,276	416	216	1,043	80
Female	9,989	109	12	1,970	2,579	730	24	817	-
Theatrical Art									
Total	5,997	1,619	-	799	739	161	24	153	29
Male	3,980	1,079	-	162	739	161	24	-	29
Female	2,018	540	-	637	-	-	-	153	-
Educational Pedagogy, T	eaching curri	culum instr	uction Ed	ucational ad	ministration	n, Adults ed	ducation		
Total	91,325	5,884	535	21,886	18,677	2,120	414	18,001	504
Male	58,137	2,961	392	14,975	11,444	1,340	289	12,353	446
Female	33,188	2,922	144	6,910	7,233	780	124	5,648	58
Geography									
Total	27,255	1,693	87	6,378	7,762	1,099	114	5,861	248
Male	21,980	1,462	87	5,175	5,285	784	114	5,239	214
Female	5,275	231	-	1,203	2,477	314	-	622	33
History									
Total	11,791	384	176	2,235	4,832	51	83	1,305	225
Male	10,024	153	176	2,160	3,942	51	83	1,095	225
Female	1,767	231	-	75	891	-	-	210	-

Table 3.8 Cont'd

Training Status and						R	egion		
Types of Training							Benishangul -		
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Political science									
Total	2,497	-	-	640	369	-	-	599	148
Male	1,888	-	-	640	316	-	-	389	148
Female	609	-	-	-	53	-	-	210	-
Sociology and Anthropolo	gy								
Total	9,508	251	-	634	2,762	50	24	1,554	-
Male	4,976	251	-	189	1,287	50	24	1,062	-
Female	4,532	-	-	445	1,475	-	-	492	-
Psychology									
Total	3,956	504	-	363	515	51	-	1,225	66
Male	3,140	504	-	181	476	51	-	631	66
Female	816	-	-	182	39	-	-	595	-
Civil									
Engineering/Building									
Total	14,882	946	214	375	823	101	59	3,329	66
Male	10,437	946	214	375	663	101	26	2,838	66
Female	4,445	-	-	-	160	-	33	491	-
Mechanical Engineering									
Total	6,073	1,516	-	232	389	210	-	437	-
Male	5,630	1,407	-	232	144	210	-	437	-
Female	443	109	-	-	245	-	-	-	-
Electrical Engineering									
Total	8,933	1,295	-	269	1,636	50	-	1,107	83
Male	7,635	872	-	269	1,241	-	-	1,107	-
Female	1,298	424	-	-	395	50	-	-	83
Industrial Engineering									
Total	988	-	-	80	177	-	-	385	-
Male	866	-	-	80	55	-	-	385	-
Female	122	-	-	-	122	-	-	-	-
Municipality Engineering/	Architecture								
Total	4,242	-	-	588	1,005	-	-	854	-
Male	2,612	-	-	491	128	-	-	554	-
Female	1,630	-	-	97	877	-	-	299	-

Table 3.8 Cont'd

Training Status and	Region									
Types of Training							Benishangul -			
Typoo or Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela	
Other types of Engineering	ng Dryland	Rio-divers	sity Hydra	ulic Engine	erina Irriaati	on Engine	erina			
Total	4.478	682	- -	284	470	314	24	330	_	
Male	3,557	682	_	284	470	314	24	101	_	
Female	921	-	_	-	-	-	-	229	_	
Automotive technology	321	_	_	_	_	_		223	_	
Total	6,879	1,376	_	2,105	1,138	_	_	693	28	
Male	4,781	1,106	_	1,214	411	_	_	483	28	
Female	2,098	270	_	892	726	_	_	210	-	
Construction technology		210		002	720			210		
Total	9,805	941	_	682	2,160	102	-	972	_	
Male	7,989	832	_	579	2,051	102	-	670	_	
Female	1,816	109	_	103	109	-	-	302	_	
Drafting technology	.,0.0									
Total	2,461	176	_	663	143	_	-	-	_	
Male	1,829	176	_	462	-	_	-	-	_	
Female	632	-	-	201	143	-	-	-	-	
Manufacturing technolog										
Total	2,063	341	-	-	46	218	-	420	-	
Male	1,722	-	-	-	46	218	-	420	-	
Female	341	341	-	-	-	-	-	-	-	
Metal technology										
Total	3,165	646	-	99	818	121	-	490	-	
Male	2,009	323	-	99	401	50	-	490	-	
Female	1,155	323	-	-	416	70	-	-	-	
Veterinary Science, Anim	nal husbandry									
Total	5,327	109	-	1,035	1,703	50	159	841	-	
Male	4,062	109	-	293	1,180	50	159	841	-	
Female	1,265	-	-	742	522	-	-	-	-	
Communication										
Total	1,764	203	-	630	247	50	-	-	-	
Male	1,318	203	-	581	247	-	-	-	-	
Female	446	-	-	50	-	50	-	-	-	

Table 3.8 Cont'd

Training Status and						R	egion		
Types of Training							Benishangul -		
71 0	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Power, Energy and	l others like Instru	umentation	, Electrica	al Power Me	chanical an	d Manufac	turing Power		
Total	5,466	990	36	1,438	386	-	20	168	-
Male	5,466	990	36	1,438	386	-	20	168	-
Female	-	-	-	-	-	-	-	-	-
Biology									
Total	11,856	330	-	1,250	5,202	50	227	2,265	41
Male	8,228	103	-	786	3,647	-	157	1,398	41
Female	3,627	226	-	464	1,555	50	70	866	-
Chemistry									
Total	10,681	538	24	1,074	3,423	-	152	2,421	80
Male	8,225	538	24	1,074	2,878	-	83	1,648	80
Female	2,456	-	-	-	545	-	69	773	-
Geology									
Total	1,034	-	-	177	58	-	-	-	-
Male	677	-	-	177	58	-	-	-	-
Female	357	-	-	-	-	-	-	-	-
Physics									
Total	6,677	109	180	942	2,444	365	95	1,509	27
Male	6,195	109	180	942	1,963	365	95	1,509	27
Female	481	-	-	-	481	-	-	-	-
Mathematics									
Total	15,292	940	37	2,875	4,848	314	288	3,892	29
Male	12,940	741	37	2,799	4,254	314	288	2,845	29
Female	2,353	199	-	76	594	-	-	1,048	-
Statistics									
Total	2,075	109	-	54	780	-	-	370	-
Male	1,304	109	-	-	696	-	-	153	-
Female	771	-	-	54	85	-	-	217	-
Demography									
Total	357	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-
Female	357	-	-	-	-	-	-	-	-

Table 3.8 Cont'd

Training Status and						R	egion		
Types of Training							Benishangul -		
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Computer Science/Inform	ation Science	9							
Total	27,732	436	188	4,381	5,184	101	59	3,028	-
Male	16,106	218	-	2,058	2,957	101	26	2,149	_
Female	11,626	218	188	2,323	2,227	-	33	880	_
Health and Physical educ			.00	_,=_=	_,			000	
Total	6,245	323	-	572	3,124	_	124	210	_
Male	4,919	-	_	357	2,682	_	124	210	-
Female	1,326	323	_	215	442	_	-		_
Meteorological Science	1,020	020		2.0					
Total	769	_	_	_	470	_	-	_	-
Male	769	_	_	_	470	_	-	_	-
Female	-	_	_	_	-	_	-	_	_
Chemical Engineering									
Total	1,469	109	_	_	209	_	-	85	27
Male	1,469	109	_	_	209	_	_	85	27
Female	-	-	_	_	-	_	-	-	-
Textile Engineering									
Total	1,057	226	_	_	139	_	-	_	_
Male	485		_	_	139	_	-	_	_
Female	572	226	_	_	-	_	_	_	_
Purchasing supply	012	220							
Total	6,247	226	_	983	754	_	76	_	86
Male	2,782	-		403	574		76 76		58
Female	3,466	226	-	580	180	_	-	-	28
Software Engineering	3,400	220	-	360	100	-	-	-	20
Total	1,612	_	_	477	97			_	
Male	1,012	-	-	477 477	97	-	-	-	-
Female	346	-				-	-	-	-
Home economics	346	-	-	-	-	-	-	-	-
	40.000	4.000		0.505	0.504		400	070	
Total	18,969	1,982	-	6,585	2,501	-	103	273	-
Male	3,983	425	-	1,795	971	-	76	-	-
Female	14,987	1,557	-	4,790	1,530	-	27	273	-
Leather handicraft techno	••	4 005		400	400				
Total	3,950	1,233	-	103	193	-	28	-	-

Table 3.8 Cont'd

Training Status and						R	egion		
Types of Training							Benishangul -		
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Male	1,109	217	-	-	193	-	28	-	-
Female	2,841	1,015	-	103	-	-	-	-	-
Decorative work training									
Total	37,983	3,677	-	3,820	4,211	-	40	2,756	-
Male	4,140	109	-	79	150	-	20	100	-
Female	33,843	3,568	-	3,741	4,061	-	20	2,656	-
Video photographic trainii	ng								
Total	4,956	499	12	153	861	50	-	134	-
Male	3,391	499	12	153	500	50	-	-	-
Female	1,564	-	-	-	361	-	-	134	-
Pottery									
Total	3,094	360	-	498	565	-	-	622	-
Male	1,060	255	-	-	-	-	-	101	-
Female	2,034	104	-	498	565	-	-	521	-
Weavery									
Total	6,883	323	-	4,751	-	-	-	719	-
Male	4,441	323	-	3,281	-	-	-	439	-
Female	2,442	-	-	1,470	-	-	-	280	-
Wood work/carpentry									
Total	22,904	1,804	243	3,609	5,279	210	140	1,306	68
Male	21,429	1,804	243	3,351	4,864	210	140	1,138	68
Female	1,475	-	-	258	416	-	-	168	-
Electrical work									
Total	31,917	1,730	110	5,170	5,997	88	172	1,750	-
Male	24,154	1,506	110	2,136	4,696	-	109	1,540	-
Female	7,764	224	-	3,034	1,302	88	63	210	-
Metal welding and binding	g								
Total	18,666	3,791	156	2,031	3,853	101	20	1,140	27
Male	17,839	3,791	156	1,710	3,796	101	20	1,140	27
Female	827	-	-	322	57	-	-	-	-
Auto Mechanics including	motorcycle,	Aircraft ma	iintenanc	е					
Total	37,907	3,153	583	3,554	6,981	153	104	1,517	80
Male	34,285	2,448	548	3,285	6,041	153	104	1,517	80
Female	3,622	705	36	269	940	-	-	-	-
Mason									
Total	24,426	2,747	494	1,899	6,053	1,196	100	2,792	-

Table 3.8 Cont'd

Training Status and						R	tegion		
Types of Training							Benishangul -		
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Male	21,677	2,180	494	1,759	5,539	1,196	100	2,302	-
Female	2,750	567	-	141	514	-	-	491	-
Typing									
Total	20,087	1,337	96	1,971	5,072	50	53	978	27
Male	358	-	-	-	-	-	-	-	-
Female	19,729	1,337	96	1,971	5,072	50	53	978	27
Tailoring									
Total	36,837	6,818	24	5,682	8,508	576	-	3,519	27
Male	11,792	1,724	-	2,497	2,439	210	-	2,266	27
Female	25,045	5,093	24	3,186	6,069	367	-	1,253	-
Driver									
Total	254,103	15,104	1,362	26,870	66,910	5,453	1,552	23,720	939
Male	245,489	14,567	1,362	26,415	66,666	5,383	1,552	23,534	537
Female	8,614	537	-	455	243	70	-	187	403
Development Agent									
Total	4,160	109	37	638	1,917	51	112	257	-
Male	2,719	109	37	638	1,238	51	43	257	-
Female	1,440	-	-	-	679	-	69	-	-
Traditional Birth Attendar	nt								
Total	1,634	109	-	744	226	210	-	-	-
Male	816	-	-	744	72	-	-	-	-
Female	818	109	-	-	154	210	-	-	-
Computer Applications a	nd Usage								
Total	45,599	3,388	273	6,805	8,304	1,955	181	4,014	218
Male	9,423	191	188	1,609	2,514	618	-	414	56
Female	36,176	3,197	85	5,196	5,790	1,337	181	3,600	162
Lower Scale training of A	Agriculture (fro	m Agarfa,	Ardaeta a	ınd Baco)					
Total	4,083	223	-	2,795	787	-	54	-	-
Male	2,997	223	-	1,888	608	-	54	-	-
Female	1,086	-	-	907	179	-	-	-	-
Hotels and Tourism									
Total	120,619	8,782	290	19,088	36,980	730	1,271	15,910	611
Male	42,779	2,857	155	4,011	13,342	314	282	7,162	282
Female	77,840	5,925	134	15,077	23,638	415	988	8,748	329
Police	•	•		•	•			•	
Total	46,793	3,798	263	7,892	10,839	2,579	552	8,581	591

Table 3.8 Cont'd

Training Status and			1			R	egion		
Types of Training							Benishangul -		
J. 0	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Male	41,921	3,332	263	7,257	10,519	2,143	431	7,269	451
Female	4,872	466	-	635	319	436	121	1,312	140
Military science									
Total	33,004	2,872	36	8,165	6,124	373	304	4,889	108
Male	30,154	2,549	-	8,118	6,081	373	304	4,398	81
Female	2,850	323	36	47	42	-	-	491	27
Cobblestone									
Total	21,251	3,806	-	8,073	4,311	101	27	1,325	-
Male	13,495	2,497	-	5,299	2,070	50	27	1,197	-
Female	7,757	1,310	-	2,774	2,241	50	-	128	-
Embroidery									
Total	12,286	1,325	-	1,512	4,448	210	167	309	-
Male	1,192	-	-	-	-	-	-	101	-
Female	11,094	1,325	-	1,512	4,448	210	167	207	-
Plumber									
Total	5,249	109	62	800	1,348	-	-	771	-
Male	4,522	109	62	420	1,280	-	-	490	-
Female	727	-	-	380	67	-	-	280	-
Other Senior and Junio	r Professions/	Specified							
Total	592,343	37,880	2,920	93,448	196,261	13,222	7,788	87,181	4,943
Male	313,101	19,865	1,159	50,557	102,706	10,027	4,947	46,902	3,295
Female	279,242	18,015	1,761	42,891	93,555	3,195	2,841	40,279	1,648
Other Skilled and Techn	nical Profession	ons/Specifi	ied						
Total	164,895	19,746	326	29,061	32,788	3,565	713	14,980	144
Male	88,271	9,704	300	15,291	17,139	2,630	459	8,129	28
Female	76,624	10,042	26	13,771	15,649	934	254	6,851	116
Not Stated for type of tra	aining								
Total	5,153	540	287	778	2,051	-	-	739	-
Male	2,165	323	287	137	998	-	-	420	-
Female	2,988	218	-	641	1,052	-	-	319	-

Table 3.9 Distribution of Urban Population by Five Years Age Group, Sex and Sex Ratio - COUNTRY TOTAL: 2012

Age Group		Sex		Sex Ratio
Age Group	Total	Male	Female	Sex Ratio
All Ages	14,011,269	6,689,363	7,321,906	91.4
0-4	1,457,485	782,821	674,664	116.0
5-9	1,443,131	720,528	722,603	99.7
10-14	1,592,162	754,662	837,500	90.1
15-19	1,752,372	754,648	997,725	75.6
20-24	1,707,114	743,495	963,619	77.2
25-29	1,615,015	776,270	838,745	92.6
30-34	1,006,702	507,720	498,982	101.8
35-39	945,870	465,729	480,140	97.0
40-44	568,878	299,009	269,869	110.8
45-49	477,925	233,108	244,817	95.2
50-54	381,120	176,523	204,597	86.3
55-59	269,995	118,356	151,640	78.1
60-64	239,219	103,804	135,416	76.7
65+	554,280	252,690	301,589	83.8



CHAPTER IV

ECONOMIC ACTIVITY STATUS

4.1 Introduction

Statistical data on the economic activity status of a population is essential for planning and formulating policies for better utilization of human resources of a country. The economic activity status indicates the size and distribution of the work force of economically active and not active population during a given reference period at a specified area.

A wide range of information was collected on economic activity status of all persons aged five years and above in the 2012 UEUS. For the purpose of gathering information about the economic activities of children, the lower age limit was fixed at five years and hence data were collected from persons aged five years and above. However, for measuring the economic activity status of the population those persons aged ten years and above were included in the analysis.

The 2012 Urban Employment and Unemployment Survey administered detailed labour force questions to measure the economic activity status of persons in urban areas of the country. These includes:-

- a) whether persons were engaged in economic / productive activities in the last seven days prior to the survey date or reasons for not being engaged;
- whether persons were looking for work or will be available for work in the coming one month or reason for not being available during the reference period;
- c) whether persons were engaged in economic / productive activities during most of the last six months or reasons for not being engaged.

Every person aged ten years and above in the sampled household was asked if he/she was engaged in any productive activities during the last seven days, and in the last six months prior to the survey date. Based on the response obtained from respondents, the economic activity status of the population was determined by using the standard definitions of economic and non economic activities.

The 2012 Urban Employment and Unemployment Survey followed the ILO international standard definitions of economic and non-economic activities. The concept of economic activity as adopted by the 13th International Conference on Labour Statistics (ICLS,1982) is defined as the production of goods and/or services that falls within the United Nations System of National Accounts (SNA) production boundary (ILO, 1990). In this survey, economic/ productive activity is defined as work, which involves the production of goods and /or services for sale or exchange. In addition, production of goods and services for own consumption or own uses are also considered as economic activities. These include production of primary products (agriculture, hunting, fishing, forestry and logging, mining and quarrying), for own consumption; processing of primary products by the producers themselves; production of other commodities where part of it is sold on the market; and own account construction and fixed asset formation (expected life use of one year or more). Such economic activities could be performed for an individual, family or private enterprise, government establishment or social organization. The remuneration may be on daily, weekly, monthly, yearly or contract basis. The practical activities of apprentices are also considered as economic activities. On the other hand, unpaid household chores such as preparing food, cleaning the house, taking care of children are not considered as economic activities. Similarly, unpaid community and volunteer services are classified as non-economic activities.

Based on the ILO definitions of economic and non economic activities, data on economic activity status of persons aged ten years and above was collected in the survey. Persons who were engaged in productive work /employed/ or available to be engaged /unemployed/ during the reference period are considered as economically active persons. While those persons who were neither engaged nor available to be engaged in

the production of economic goods and services during the reference period are considered as economically not active or inactive population.

4.2 Data Collection Approaches of the Economic Activity Status of Urban Population

the 2012 Urban Employment and Unemployment Survey, two approaches were used in the collection of data about the economic activity status of urban population. These are current and usual activity status approaches. The main difference between these two approaches is the length of the reference period.

4.2.1 Current Activity Status Approach

The current activity status approach measures the economic activity status of a person based on a short reference period of seven days before the date of interview. In this approach, all persons aged ten years and above who were engaged in productive work or available to be engaged during the reference week were considered as currently economically active while those who were neither engaged nor available to be engaged in productive work were taken as currently economically not active.

4.2.2 Usual Activity Status Approach

The usual activity status approach measures the economic activity status of a person based on a long reference period of the six months prior to the date of interview. In the usual activity status approach, all persons aged ten years and above who were engaged in productive works or available to be engaged in the production of economic goods and/or services during most of the six months were classified as usually economically active. Those who were neither employed nor unemployed in most of the six months categorized as economically not active.

The choice of appropriate economic activity status measurement approach depends on the extent of variability of economic activities of the population during a specified period of time in the study area. In the rural areas seasonal variation of activities is common. In order to collect economic activity status of the rural population including the slack and peak periods of agricultural activities, a longer reference period (usual activity status approach) is found to be more appropriate. On the other hand, in urban areas, activities are relatively less affected by seasonal variations and hence shorter reference period or current activity status approach is found to be more convenient. But in this survey, both approaches were used to collect data on economic activity status of urban population.

This chapter presents the economic activity status, size and distribution of economically active and not active population, activity rate, and reasons for being inactive of urban population aged ten years and above using only the current status approach at national, regional and major town level.

4.3 Current Activity Status

As mentioned above, current activity status approach measures the economic activity status of a person in a short reference period, that is, the seven days prior to the date of the interview. In this approach a series of inquiries were administered to determine the economic activity status of the population during the reference week such as engagement in economic activity, attachment to jobs, seeking and availability to work, reason for not seeking or available to work, etc., These questions were presented to those persons aged five years and above. But the responses of persons aged ten years and above were used to classify the population into three mutually exclusive categories: employed, unemployed, and not in the labour force (non-active). The employed and the unemployed population together constitute the labour force or the currently economically active population. The third category, those who were not in the labour force are called currently not active population.

As part of the economically active population, employed population consists of persons aged ten years and above who engaged in a productive activity or work at least for four hours during the seven days prior to the date of the interview. In addition, persons who had regular jobs, or business, or holdings to return to but who were temporarily absent from work (not at work or worked less than four hours) for various reasons such as illness or injury, holiday or vacation, strike or lockout, and seasonality of work, annual leave, temporary closure of establishment were also considered as employed. For a person to be considered as absent from work, he/she must have formal attachment to the job.

The currently unemployed population also comprises those persons without work but looking for work or available and ready to work if any job is found during the reference period of the coming one month. The reference period of the coming one month refers to the survey week plus the next three weeks. If a person who is looking for work but engaged in productive activity during the reference week is recorded as employed but not as unemployed.

4.4 Economically Active and Not active Population and Activity Rate

Based on the above definition of economic activity status, the number of economically active population including employed and unemployed persons, population not active and activity rates of urban population aged ten years and above at national and regional level is presented by sex and age group in Table 4.1 and Table 4.1.1- 4.1.11 using the current status approach, respectively. Besides, the economic activity status and activity rate of selected major towns' population aged ten years and above is also presented in Table 4.2.

The extent of the economic participation of the population is measured using economic activity rates. Economic activity rate or economic participation rate is computed as the percentage of the economically active population over the total of active plus not active

population aged ten years and above. The national estimate shows that 62.5 percent of the population aged ten years and above were economically active (Table 4.1). With respect to sex, the activity rate of male is 69.7 percent and the corresponding female activity rate is 56.2 percent. The age distributions of activity rate illustrates that it is low up to age 20, but steadily rise and reaches its peak in age 34, and then after declines (Table 4.1).

As far as the economic activity status of major towns' population is concerned, 62.4 percent of the populations were economically active while the remaining 37.6 percent were found economically not active. In terms of sex, the activity rate of male was higher by 14.3 percent than female. Among the major towns, the highest activity rate was recorded in Bahir Dar town (70.4 percent) while the lowest was in Gambella town (50.0 percent) (Table 4.2).

The size of economically active and not active population has its own implication on the development of the local as well as the national economy of a region. The high activity rate of a population implies the prevalence of huge potential of the human resource or human labour that foster sustainable economic development. On the other hand, the low economic activity rate also implies the lack of adequate and trained human labour or the existence of large number of economically non-productive population which has no contribution for development at the time. Hence, the economic activity rate of urban population of the Ethiopia can be viewed from these perspectives.

The distribution of economically not active population of urban areas at national and regional level by age group, sex and reason for not being economically active (for those who were neither engaged nor available and/ or seeking work) during the last seven days are presented in Table 4.3 and Table 4.4, respectively. The reasons for not looking for or available to work include; student/training, home making / domestic work, pregnancy, injury / disability, illness, too young, old age /pensioned... etc. The result in Table 4.3 shows that the lion share of the population not active were students (66.4 percent). The remaining not active population (33.6 percent) comprises home makers, old aged, youngsters, and injured / disabled, remitted, and other persons.

Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- COUNTRY TOTAL: 2012 Table 4.1

	Approach - COONTRY TOTAL. 2012 Activity Status									
Age Group	All	F	conomically Activ		Faanamiaallu	Activity				
and Sex	Persons	Employed	Unemployed	Total Active	Economically Not Active	Rate				
All Ages		Limpleyea	Champioyea	101017101170						
	11,110,653	5,726,116	1,218,574	6,944,691	4,165,962	62.5				
Male	5,186,014	3,202,042	410,381	3,612,423	1,573,590	69.7				
Female	5,924,639	2,524,074	808,193	3,332,267	2,592,372	56.2				
10-14	0,021,000	2,021,011	000,100	0,002,207	2,002,072	00.2				
Total	1,592,162	195,567	24,618	220,185	1,371,977	13.8				
Male	754,662	94,187	11,787	105,974	648,689	14.0				
Female	837,500	101,380	12,831	114,211	723,289	13.6				
15-19	,	,	•	,	•					
Total	1,752,372	492,071	136,069	628,140	1,124,233	35.8				
Male	754,648	205,515	51,396	256,911	497,736	34.0				
Female	997,725	286,556	84,673	371,228	626,496	37.2				
20-24										
Total	1,707,114	902,304	361,492	1,263,796	443,318	74.0				
Male	743,495	457,454	127,407	584,861	158,634	78.7				
Female	963,619	444,850	234,085	678,935	284,684	70.5				
25-29										
Total	1,615,015	1,128,487	268,787	1,397,273	217,742	86.5				
Male	776,270	653,525	79,619	733,144	43,126	94.4				
Female	838,745	474,962	189,168	664,129	174,616	79.2				
30-34										
Total	1,006,702	767,262	128,378	895,639	111,063	89.0				
Male	507,720	448,530	39,484	488,015	19,705	96.1				
Female	498,982	318,731	88,893	407,624	91,357	81.7				
35-39										
Total	945,870	742,001	98,500	840,501	105,369	88.9				
Male .	465,729	423,955	26,203	450,158	15,571	96.7				
Female	480,140	318,046	72,297	390,343	89,797	81.3				
40-44	500.070	445.055	50.070	500 700	05.445	00.5				
Total	568,878	445,655	58,078	503,733	65,145	88.5				
Male	299,009	273,476	16,589	290,065	8,944	97.0				
Female 45-49	269,869	172,179	41,489	213,668	56,201	79.2				
Total	477,925	361,687	12 106	404 972	72.052	84.7				
Male	233,108	211,592	43,186 12,716	404,873 224,308	73,052 8,799	96.2				
Female	244,817	150,094	30,470	180,565	64,252	73.8				
50-54	244,017	150,054	30,470	100,505	04,202	73.0				
Total	381,120	266,025	36,225	302,251	78,870	79.3				
Male	176,523	153,598	12,971	166,568	9,955	94.4				
Female	204,597	112,428	23,254	135,682	68,915	66.3				
55-59	_0 .,00.	, 0	_0,_0 :	.00,002	00,010	00.0				
Total	269,995	152,584	24,471	177,055	92,940	65.6				
Male	118,356	96,469	9,677	106,145	12,210	89.7				
Female	151,640	56,115	14,794	70,910	80,730	46.8				
60-64										
Total	239,219	113,615	18,566	132,181	107,038	55.3				
Male	103,804	75,221	8,023	83,244	20,560	80.2				
Female	135,416	38,393	10,544	48,937	86,479	36.1				
65+										
Total	554,280	158,859	20,205	179,064	375,215	32.3				
Male	252,690	108,519	14,510	123,030	129,661	48.7				
Female	301,589	50,340	5,694	56,035	245,555	18.6				

Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- TIGRAY REGION: 2012 **Table 4.1.1**

	- 1-1	Activity Status						
Age Group	All	F.				Activity		
and Sex	Persons	i	conomically Activ		Economically	Rate		
A II A		Employed	Unemployed	Total Active	Not Active			
All Ages		224 224	0.4.770	400.04=	007.004	04.0		
Total	794,578	391,891	94,756	486,647	307,931	61.2		
Male	344,053	204,483	25,996	230,478	113,574	67.0		
Female	450,525	187,408	68,761	256,169	194,356	56.9		
10-14								
Total	107,902	7,272	603	7,875	100,027	7.3		
Male	52,790	3,165	281	3,445	49,345	6.5		
Female	55,112	4,107	323	4,430	50,682	8.0		
15-19								
Total	139,083	36,408	12,014	48,422	90,662	34.8		
Male	56,861	13,795	4,398	18,193	38,667	32.0		
Female	82,223	22,613	7,615	30,229	51,994	36.8		
20-24	,	,-,-	,,,,,	,				
Total	125,551	65,371	30,545	95,916	29,635	76.4		
Male	46,449	28,440	9,254	37,694	8,756	81.2		
Female	79,102	36,931	21,291	58,222	20,880	73.6		
25-29	79,102	30,931	21,291	30,222	20,000	75.0		
	111701	79,187	22 202	102 500	12,121	89.4		
Total	114,701		23,393	102,580				
Male	48,065	41,064	5,335	46,399	1,666	96.5		
Female	66,636	38,123	18,059	56,181	10,455	84.3		
30-34								
Total	67,877	53,581	8,462	62,043	5,834	91.4		
Male	34,111	31,791	1,365	33,157	955	97.2		
Female	33,766	21,790	7,097	28,887	4,879	85.5		
35-39								
Total	60,067	47,215	6,513	53,728	6,339	89.4		
Male	28,032	26,320	1,480	27,800	232	99.2		
Female	32,036	20,895	5,033	25,928	6,107	80.9		
40-44								
Total	34,911	27,380	4,051	31,431	3,481	90.0		
Male	18,240	16,618	1,298	17,916	324	98.2		
Female	16,672	10,762	2,753	13,515	3,157	81.1		
45-49	•	•	,	,	,			
Total	30,600	21,678	3,210	24,888	5,712	81.3		
Male	12,166	10,994	630	11,624	542	95.5		
Female	18,434	10,684	2,580	13,264	5,170	72.0		
50-54	10,101	10,001	2,000	10,201	0,170	72.0		
Total	25,692	18,603	1,941	20,545	5,147	80.0		
Male	11,299	10,003	536	10,689	610			
Female						94.6		
	14,393	8,450	1,406	9,856	4,537	68.5		
55-59 	40.000	44.444	4.000	40.404	7.050	00.4		
Total	19,663	11,114	1,290	12,404	7,258	63.1		
Male	6,742	6,098	254	6,352	389	94.2		
Female	12,921	5,016	1,036	6,052	6,869	46.8		
60-64								
Total	21,192	11,191	1,506	12,696	8,496	59.9		
Male	8,015	7,009	434	7,443	573	92.9		
Female	13,177	4,182	1,072	5,254	7,923	39.9		
65+								
Total	47,337	12,892	1,227	14,119	33,218	29.8		
Male	21,284	9,035	732	9,768	11,516	45.9		
Female	26,053	3,856	495	4,351	21,702	16.7		
-	, -	, -		· · · · · · · · · · · · · · · · · · ·	,			

Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- AFFAR REGION: 2012 **Table 4.1.2**

Approach - AFFAR REGION: 2012 Activity Status									
Age Group	All	-				Activity			
and Sex	Persons	Employed	conomically Activ Unemployed	Total Active	Economically Not Active	Rate			
All Ages		Employed	Oriempioyeu	Total Active	NOT ACTIVE				
Total	127,674	77,068	9,846	86,914	40,759	68.1			
Male	61,116	43,657	2,425	46,082	15,034	75.4			
Female	66,557	33,411	7,421	40,832	25,726	61.3			
10-14	00,557	33,411	7,421	40,032	25,720	01.5			
Total	19,920	4,132	227	4,358	15,561	21.9			
Male	9,600	1,607	221	1,607	7,993	16.7			
Female	10,319	2,524	227	2,751	7,568	26.7			
15-19	10,515	2,024	221	2,701	7,500	20.7			
Total	15,019	5,365	684	6,049	8,971	40.3			
Male	6,525	2,749	311	3,060	3,466	46.9			
Female	8,494	2,616	372	2,989	5,505	35.2			
20-24	0,101	2,0.0	0.2	2,000	0,000	30.2			
Total	15,774	8,848	2,455	11,302	4,471	71.7			
Male	6,978	5,305	506	5,811	1,167	83.3			
Female	8,796	3,542	1,949	5,491	3,305	62.4			
25-29	0,700	0,0 12	1,010	0,101	0,000	02.1			
Total	19,941	14,868	2,266	17,134	2,806	85.9			
Male	8,868	8,055	559	8,613	254	97.1			
Female	11,073	6,814	1,707	8,521	2,552	77.0			
30-34	11,070	0,011	1,707	0,021	2,002	77.0			
Total	13,870	10,723	1,173	11,896	1,974	85.8			
Male	6,672	5,868	409	6,277	395	94.1			
Female	7,197	4,856	763	5,619	1,578	78.1			
35-39	.,	.,555	. 00	0,0.0	.,0.0				
Total	12,365	10,217	1,203	11,420	944	92.4			
Male	6,305	6,090	48	6,138	168	97.3			
Female	6,059	4,127	1,156	5,282	777	87.2			
40-44	,	•	•	•					
Total	10,893	8,664	856	9,520	1,373	87.4			
Male	5,096	4,753	343	5,096	-	100.0			
Female	5,797	3,911	513	4,424	1,373	76.3			
45-49									
Total	6,757	5,664	541	6,205	552	91.8			
Male	3,608	3,549	12	3,561	48	98.7			
Female	3,149	2,115	529	2,645	504	84.0			
50-54									
Total	4,766	3,636	406	4,042	724	84.8			
Male	2,711	2,376	201	2,578	133	95.1			
Female	2,055	1,259	204	1,464	591	71.2			
55 - 59									
Total	2,813	1,838	12	1,850	962	65.8			
Male	1,163	984	12	996	168	85.6			
Female	1,649	855	-	855	795	51.8			
60-64									
Total	2,373	1,976	12	1,988	385	83.8			
Male	1,378	1,256	12	1,268	110	92.0			
Female	995	720	-	720	275	72.3			
65+									
Total	3,185	1,137	12	1,149	2,036	36.1			
Male	2,210	1,066	12	1,077	1,133	48.8			
Female	975	72	=	72	903	7.4			
			-						

Table 4.1.3 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- AMHARA REGION: 2012

	1-1	(Activity Status			
Age Group	All	Fo	conomically Acti	V/Q	Economically	Activity
and Sex	Persons	LC		VC	Not Active	Rate
and Cox	1 0100110	Employed	Unemployed	Total Active	11017101110	rato
All Ages		Linployed	Onemployed	101017101170]	
Total	1,976,070	1,063,958	209,533	1,273,491	702,578	64.4
Male	890,049	562,815	68,229		259,005	70.9
Female	1,086,021	501,144	141,304	642,448	443,573	59.2
10-14	000 450	40.004	4 = 0 =	=4.000	004.0==	400
Total	282,456	46,894	4,505		231,057	18.2
Male	127,488	20,712	3,900		102,877	19.3
Female	154,967	26,182	605	26,787	128,180	17.3
15-19						
Total	318,539	106,615	27,257	133,872	184,667	42.0
Male	145,969	50,539	9,267	59,806	86,164	41.0
Female	172,570	56,076	17,990	74,067	98,503	42.9
20-24						
Total	318,284	171,226	75,607	246,832	71,452	77.6
Male	128,423	80,185	25,276	105,461	22,962	82.1
Female	189,861	91,040	50,331	141,371	48,490	74.5
25-29	.00,00.	0.,0.0	00,00.	,	.0,.00	
Total	277,108	201,112	42,132	243,244	33,865	87.8
Male	128,579	111,705	12,342		4,532	96.5
Female	148,529	89,407	29,790		29,333	80.3
30-34	140,529	09,407	29,790	119,197	29,333	00.3
	151 670	104 150	14 202	120 452	12 220	04.2
Total	151,672	124,150	14,303		13,220	91.3
Male	75,368	68,031	4,394		2,944	96.1
Female	76,304	56,119	9,909	66,028	10,276	86.5
35-39						
Total	158,480	122,684	15,404		20,392	87.1
Male	73,725	65,070	3,790		4,865	93.4
Female	84,756	57,615	11,614	69,229	15,527	81.7
40-44						
Total	87,769	67,780	8,239	76,018	11,751	86.6
Male	44,189	40,336	2,165	42,501	1,688	96.2
Female	43,580	27,444	6,074	33,517	10,063	76.9
45-49						
Total	92,767	72,389	8,004	80,393	12,373	86.7
Male	42,951	39,013	2,388		1,550	96.4
Female	49,815	33,376	5,617		10,823	78.3
50-54	10,010	,	-,	,	,	
Total	74,978	53,685	4,470	58,155	16,823	77.6
Male	29,271	26,813	849		1,608	94.5
Female	45,707	26,871	3,621	30,492	15,215	66.7
55-59	45,707	20,071	3,021	30,492	15,215	00.7
	F0 700	22.000	4 700	20,000	40.040	CO F
Total	53,720	32,020	4,782		16,918	68.5
Male .	20,891	17,775	1,438		1,679	92.0
Female	32,828	14,245	3,344	17,589	15,239	53.6
60-64						
Total	43,478	24,872	1,192		17,414	59.9
Male	21,274	16,883	520	17,403	3,870	81.8
Female	22,204	7,989	672	8,661	13,543	39.0
65+						
Total	116,818	40,532	3,639	44,171	72,647	37.8
Male	51,920	25,753	1,901	27,654	24,266	53.3
Female	64,897	14,779	1,738		48,381	25.4
- I OIIIUIO	U-1,UU1	17,110	1,700	10,010	70,001	20.7

Table 4.1.4 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- OROMIA REGION: 2012

	1.	,	A 41 11	0		
A O	A II	_	Activity			A -4: .: 4
Age Group	All	Е	conomically Activ	/e	Faanamiaally	Activity
and Sex	Persons	Consider to al	l la sasals cad	Tatal A ativa	Economically	Rate
A 11 A		Employed	Unemployed	Total Active	Not Active	
All Ages						
Total	3,131,966	1,643,077	330,723	1,973,800	1,158,165	63.0
Male	1,522,509	963,543	115,759	1,079,302	443,207	70.9
Female	1,609,457	679,534	214,964	894,498	714,959	55.6
10-14		·	·	,	·	
Total	499,611	62,670	11,148	73,818	425,793	14.8
Male	244,631	35,683	4,801	40,484	204,146	16.5
Female	254,981	26,987	6,347	33,334	221,647	13.1
	234,961	20,907	0,347	33,334	221,041	13.1
15-19	407.405	400 755	40.470	404.000	045 477	00.0
Total	497,405	133,755	48,172	181,928	315,477	36.6
Male	226,258	68,607	18,526	87,134	139,124	38.5
Female	271,147	65,148	29,646	94,794	176,353	35.0
20-24						
Total	502,385	283,346	104,073	387,419	114,966	77.1
Male	234,894	159,428	37,409	196,837	38,057	83.8
Female	267,491	123,918	66,664	190,582	76,909	71.2
25-29	_0.,.0.	0,0.0	00,00	.00,002	. 0,000	–
Total	444,847	316,724	71,304	388,028	56,819	87.2
	228,652	197,210	20,499			
Male	•	•		217,709	10,942	95.2
Female	216,196	119,514	50,804	170,318	45,877	78.8
30-34						
Total	273,528	217,268	27,749	245,017	28,511	89.6
Male	134,872	122,380	7,981	130,361	4,511	96.7
Female	138,656	94,888	19,768	114,656	24,001	82.7
35-39						
Total	275,890	226,653	25,353	252,006	23,883	91.3
Male	142,624	131,824	7,634	139,458	3,166	97.8
Female	133,266	94,829	17,720	112,549	20,717	84.5
40-44	100,200	34,023	17,720	112,040	20,717	04.0
Total	150 776	122 100	10 715	125 014	14 969	90.1
	150,776	123,199	12,715	135,914	14,862	
Male .	79,002	72,172	4,353	76,525	2,477	96.9
Female	71,774	51,027	8,362	59,389	12,385	82.7
45-49						
Total	127,777	103,373	6,515	109,887	17,889	86.0
Male	69,919	65,656	2,268	67,925	1,995	97.1
Female	57,857	37,716	4,246	41,963	15,894	72.5
50-54						
Total	92,024	64,589	8,771	73,360	18,664	79.7
Male	41,791	36,871	3,292	40,164	1,628	96.1
Female	50,232	27,718	5,479	33,196	17,036	66.1
	30,232	21,110	5,479	33,190	17,030	00.1
55-59 	70 754	10.710	5 570	40.000	04 470	00.7
Total	70,754	43,710	5,573	49,283	21,470	69.7
Male	32,993	28,330	1,953	30,283	2,710	91.8
Female	37,761	15,380	3,620	19,000	18,761	50.3
60-64						
Total	61,110	27,111	4,515	31,626	29,485	51.8
Male	28,702	19,414	3,166	22,581	6,121	78.7
Female	32,409	7,697	1,348	9,045	23,364	27.9
65+	3_,.30	. ,551	.,510	2,210	_0,00.	
Total	135,860	40,679	4,835	45,514	90,346	33.5
Male	58,172	25,966	3,876	29,842	28,330	51.3
Female	77,688	14,713	960	15,672	62,016	20.2

Table 4.1.5 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- SOMALI REGION: 2012

		,				
Age Group	All	E	Activity conomically Activ			Activity
and Sex	Persons	Canalay and	l la anamia de la d	Tatal Astina	Economically	Rate
All Agos		Employed	Unemployed	Total Active	Not Active	
All Ages	422.066	202 5 47	24.427	226 074	106 000	E4.6
Total Male	433,866 214,968	202,547 111,869	34,427 12,124	236,974 123,994	196,892 90,974	54.6 57.7
Female	214,900	90,677	22,302	112,980	105,918	57.7 51.6
10-14	210,090	90,077	22,302	112,900	105,916	31.0
Total	92,168	12,450	1,683	14,132	78,036	15.3
Male	52,100	6,972	680	7,652	44,634	14.6
Female	39,882	5,478	1,003	6,481	33,402	16.2
15-19	00,002	0,170	1,000	0,101	00,102	10.2
Total	72,665	17,659	4,557	22,217	50,448	30.6
Male	37,256	7,161	2,056	9,216	28,040	24.7
Female	35,409	10,499	2,502	13,000	22,408	36.7
20-24	•	•	,	,	,	
Total	50,357	23,737	7,604	31,341	19,016	62.2
Male	22,021	12,792	1,741	14,533	7,487	66.0
Female	28,337	10,945	5,863	16,808	11,529	59.3
25-29						
Total	47,996	28,029	6,470	34,499	13,496	71.9
Male	20,569	15,908	1,775	17,684	2,885	86.0
Female	27,427	12,121	4,695	16,816	10,611	61.3
30-34						
Total	39,534	28,916	3,417	32,334	7,200	81.8
Male	16,223	14,810	875	15,686	537	96.7
Female	23,311	14,106	2,542	16,648	6,663	71.4
35-39	00.740	05.407	0.044	00.007	4.004	05.4
Total Male	33,748	25,187	3,641	28,827	4,921 346	85.4
rviale Female	16,019 17,730	14,303 10,883	1,369 2,272	15,672 13,155	4,575	97.8 74.2
40-44	17,730	10,003	2,212	13,133	4,373	74.2
Total	31,822	24,280	2,510	26,791	5,032	84.2
Male	15,441	13,340	1,153	14,493	948	93.9
Female	16,381	10,940	1,357	12,298	4,083	75.1
45-49	. 0,00	. 0,0 . 0	.,	,	.,000	
Total	15,432	12,559	1,075	13,634	1,798	88.3
Male	8,720	8,141	372	8,512	208	97.6
Female	6,712	4,418	704	5,122	1,590	76.3
50-54						
Total	16,384	10,959	2,008	12,967	3,417	79.1
Male	8,324	6,255	1,203	7,457	867	89.6
Female	8,060	4,704	805	5,509	2,551	68.4
55 - 59						
Total	6,637	4,475	543	5,019	1,618	75.6
Male	3,802	3,049	229	3,278	523	86.2
Female	2,835	1,426	314	1,740	1,095	61.4
60-64	0.700	0.570	000	7 400	0.000	70.0
Total	9,736	6,570	830	7,400	2,336	76.0
Male	5,343	4,128	584	4,712	631	88.2
Female 65+	4,393	2,442	246	2,688	1,705	61.2
Total	17,386	7,724	88	7,812	9,574	44.9
Male	8,964	5,009	88	5,097	3,867	56.9
Female	8,422	2,715	-	2,715	5,707	32.2
Torridio	0,722	۷,1 ال		۷,1 ال	0,101	02.2

Table 4.1.6 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- BENISHANGUL- GUMUZ REGION: 2012

	Арргоаст	I)- BENISHA	Activity	Status	1012	
Age Group	All	E	conomically Activ			Activity
and Sex	Persons		continually Activ		Economically	Rate
		Employed	Unemployed	Total Active	Not Active	
All Ages						
Total	111,784	72,518	5,505	78,024	33,760	69.8
Male	55,116	39,854	1,897	41,751	13,365	75.8
Female	56,668	32,664	3,609	36,273	20,395	64.0
10-14						
Total	16,948	4,905	-	4,905	12,043	28.9
Male	8,487	2,252	-	2,252	6,235	26.5
Female	8,461	2,653	-	2,653	5,808	31.4
15-19						
Total	21,192	9,040	901	9,941	11,251	46.9
Male	9,435	3,944	562	4,507	4,928	47.8
Female	11,758	5,096	339	5,435	6,323	46.2
20-24						
Total	16,117	10,926	1,846	12,772	3,345	79.2
Male	6,818	5,274	568	5,842	976	85.7
Female	9,299	5,652	1,278	6,930	2,369	74.5
25-29						
Total	15,425	13,202	891	14,093	1,332	91.4
Male	8,363	7,909	340	8,249	114	98.6
Female	7,062	5,293	551	5,843	1,219	82.7
30-34						
Total	9,605	8,311	442	8,753	852	91.1
Male	4,251	4,051	147	4,197	54	98.7
Female	5,354	4,260	295	4,555	798	85.1
35-39						
Total	9,626	8,322	614	8,937	689	92.8
Male	4,686	4,620	66	4,686	-	100.0
Female	4,940	3,702	548	4,250	689	86.1
40-44						
Total	6,375	5,608	459	6,067	307	95.2
Male	4,231	4,050	103	4,153	77	98.2
Female	2,144	1,558	356	1,914	230	89.3
45-49						
Total	4,542	4,220	63	4,283	258	94.3
Male	2,754	2,730	24	2,754	-	100.0
Female	1,788	1,490	40	1,530	258	85.6
50-54						
Total	2,868	2,542	107	2,649	219	92.4
Male	1,269	1,269	-	1,269	-	100.0
Female	1,599	1,273	107	1,380	219	86.3
55-59						
Total	2,041	1,789	33	1,822	219	89.3
Male	1,230	1,197	33	1,230	-	100.0
Female	810	592	-	592	219	73.0
60-64		4 000			0.40	
Total	2,214	1,306	95	1,401	813	63.3
Male	840	811	-	811	28	96.6
Female	1,375	495	95	590	784	42.9
65+	4.00:	0.046		0.400	0.40=	
Total	4,831	2,346	53	2,400	2,432	49.7
Male	2,753	1,746	53	1,799	954	65.4
Female	2,078	600	-	600	1,478	28.9

Table 4.1.7 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- SNNP REGION: 2012

Approach)- SNNP REGION: 2012											
Age Group	All	_	Activity			Activity					
and Sex	Persons	Е	conomically Activ	/e	Economically	Rate					
and Sex	1 6130113	Employed	Unemployed	Total Active	Not Active	ivaic					
All Ages		Linployea	Griompioyou	101017101170							
Total	1,531,368	837,386	119,176	956,562	574,806	62.5					
Male	743,394	466,330	40,404	506,734	236,660	68.2					
Female	787,974	371,057	78,771	449,828	338,146	57.1					
10-14	707,071	071,007	70,771	1 10,020	000,110	07.1					
Total	263,019	42,829	3,566	46,395	216,624	17.6					
Male	120,319	19,185	628	19,812	100,506	16.5					
Female	142,701	23,644	2,938	26,582	116,118	18.6					
15-19											
Total	272,312	80,799	12,899	93,698	178,615	34.4					
Male	115,837	33,118	5,855	38,973	76,864	33.6					
Female	156,475	47,681	7,044	54,724	101,751	35.0					
20-24											
Total	226,337	131,432	36,312	167,744	58,592	74.1					
Male	112,224	70,414	13,773	84,187	28,037	75.0					
Female	114,113	61,018	22,540	83,558	30,555	73.2					
25-29											
Total	209,680	160,140	27,120	187,260	22,419	89.3					
Male	110,082	96,043	7,899	103,942	6,141	94.4					
Female	99,597	64,097	19,221	83,319	16,279	83.7					
30-34											
Total	136,023	108,830	13,033	121,863	14,160	89.6					
Male	72,101	65,815	4,344	70,160	1,942	97.3					
Female	63,922	43,015	8,688	51,703	12,218	80.9					
35-39											
Total	125,953	104,541	8,881	113,422	12,531	90.1					
Male	56,987	54,629	1,698	56,327	660	98.8					
Female	68,966	49,912	7,182	57,095	11,871	82.8					
40-44											
Total	82,153	70,376	4,519	74,895	7,258	91.2					
Male	45,930	44,177	1,232	45,409	521	98.9					
Female	36,223	26,199	3,287	29,486	6,737	81.4					
45-49											
Total	59,963	49,333	3,177	52,509	7,453	87.6					
Male	29,788	27,665	1,007	28,672	1,117	96.3					
Female	30,174	21,667	2,170	23,838	6,337	79.0					
50-54											
Total	43,200	36,324	3,479	39,803	3,397	92.1					
Male	25,912	23,720	1,491	25,212	700	97.3					
Female	17,288	12,604	1,988	14,591	2,697	84.4					
55-59											
Total	26,231	17,952	1,916	19,868	6,364	75.7					
Male	12,426	10,996	616	11,612	814	93.5					
Female	13,806	6,956	1,300	8,256	5,550	59.8					
60-64											
Total	25,259	13,973	1,815	15,789	9,470	62.5					
Male	8,696	6,628	601	7,229	1,467	83.1					
Female	16,563	7,345	1,215	8,560	8,003	51.7					
65+											
Total	61,238	20,857	2,460	23,317	37,921	38.1					
Male	33,091	13,939	1,262	15,200	17,891	45.9					
Female	28,147	6,919	1,198	8,117	20,030	28.8					

Table 4.1.8 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- GAMBELLA REGION: 2012

	- 4- 4		Activity	Status		
Age Group	All					Activity
and Sex	Persons	E	conomically Activ	/e	Economically	Rate
and Cox	. 0.000	Employed	Unemployed	Total Active	Not Active	
All Ages						
Total	75,144	35,656	3,020	38,676	36,468	51.5
Male	34,340	18,797	484	19,281	15,059	56.1
Female	40,804	16,859	2,536	19,395	21,409	47.5
10-14	4.4.000	0.45		400	40.050	
Total	14,388	315	115	429	13,959	3.0
Male	7,363	163	27	190	7,173	2.6
Female	7,025	152	88	239	6,786	3.4
15-19						
Total	11,575	1,954	417	2,372	9,203	20.5
Male	5,183	742	27	769	4,414	14.8
Female	6,392	1,212	390	1,603	4,789	25.1
20-24						
Total	11,385	6,301	814	7,115	4,270	62.5
Male	3,883	2,589	55	2,644	1,240	68.1
Female	7,501	3,712	759	4,471	3,030	59.6
25-29	,	-,		,	-,	
Total	12,534	8,852	858	9,710	2,823	77.5
Male	5,776	4,828	143	4,972	804	86.1
Female	6,758	4,023	715	4,738	2,019	70.1
30-34	0,700	1,020	710	1,700	2,010	70.1
Total	7,126	5,252	406	5,658	1,469	79.4
Male	3,544	3,114	152	3,266	279	92.1
Female	3,582	2,138	254	2,392	1,190	66.8
35-39	3,302	2,130	204	2,592	1,190	00.0
Total	6,713	5,442	221	5,663	1,049	84.4
Male	3,248	3,442	221	3,157	91	97.2
Female	3,465	2,284	221	2,506	959	72.3
40-44	3,403	2,204	221	2,500	939	12.3
Total	2.466	2 900	188	2 000	478	86.2
	3,466	2,800		2,989		
Male	1,727	1,551	80	1,631	97	94.4
Female	1,739	1,249	109	1,358	381	78.1
45-49	0.000	0.000		0.000	004	07.7
Total	2,629	2,306	-	2,306	324	87.7
Male	1,091	1,036	-	1,036	55	95.0
Female	1,538	1,270	-	1,270	268	82.5
50-54						
Total	2,624	1,537	-	1,537	1,087	58.6
Male	1,345	1,025	-	1,025	320	76.2
Female	1,279	511	-	511	768	40.0
55-59						
Total	802	382	-	382	420	47.7
Male	386	274	-	274	112	71.1
Female	416	108	-	108	308	26.0
60-64						
Total	1,048	266	-	266	782	25.4
Male	372	129	-	129	243	34.6
Female	676	137	-	137	539	20.3
65+						
Total	855	250	-	250	605	29.2
Male	421	188	-	188	233	44.8
Female	434	62		62	372	14.2

Table 4.1.9 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- HARARI REGION: 2012

	Арргоас	ii, iiAiAiXi	Activity			
Age Group	All		Activity			Activity
and Sex	Persons	E	conomically Activ	/e	Economically	Rate
and ook	1 0100110	Employed	Unemployed	Total Active	Not Active	rato
All Ages						
Total	87,661	49,969	6,973	56,942	30,719	65.0
Male	41,096	26,901	2,086	28,987	12,109	70.5
Female	46,566	23,068	4,887	27,955	18,610	60.0
10-14	+0,500	25,000	4,007	21,300	10,010	00.0
Total	11,438	557	_	557	10,881	4.9
Male	5,782	325		325	5,457	5.6
Female	5,656	231	_	231	5,425	4.1
15-19	5,050	231	-	231	5,425	4.1
Total	11,174	2.970	E10	2 /10	7 756	30.6
		2,870	548	3,418	7,756	
Male	4,885	1,058	151	1,209	3,676	24.7
Female	6,290	1,812	398	2,210	4,080	35.1
20-24	40.040	0.400	4 504	0.000	0.400	70.0
Total	10,243	6,489	1,591	8,080	2,163	78.9
Male	4,385	3,054	692	3,746	640	85.4
Female	5,858	3,435	899	4,334	1,524	74.0
25-29						
Total	11,207	8,966	1,218	10,184	1,023	90.9
Male	5,722	5,127	248	5,375	348	93.9
Female	5,485	3,839	970	4,809	676	87.7
30-34						
Total	9,509	7,415	1,152	8,567	942	90.1
Male	4,300	3,905	245	4,151	150	96.5
Female	5,209	3,510	907	4,417	792	84.8
35-39						
Total	8,241	6,850	948	7,798	443	94.6
Male	4,545	4,093	205	4,298	247	94.6
Female	3,696	2,757	743	3,500	196	94.7
40-44						
Total	5,006	4,377	240	4,617	389	92.2
Male	2,519	2,322	44	2,367	152	94.0
Female	2,487	2,055	196	2,250	237	90.5
45-49						
Total	5,047	3,823	392	4,216	831	83.5
Male	2,403	2,216	49	2,265	138	94.2
Female	2,644	1,607	343	1,951	693	73.8
50-54	_,-,-	1,001		,,,,,,		
Total	4,171	3,129	196	3,325	846	79.7
Male	1,865	1,661	49	1,709	156	91.6
Female	2,306	1,469	147	1,616	690	70.1
55-59	2,000	1,100		1,010	000	70.1
Total	3,943	2,672	369	3,040	903	77.1
Male	1,604	1,382	222	1,604	-	100.0
Female	2,339	1,290	147	1,436	903	61.4
60-64	2,339	1,290	147	1,430	903	01.4
Total	1,872	1,104	49	1,153	719	61.6
			49			
Male	862	723	40	723	138	83.9
Female	1,010	381	49	430	581	42.5
65+ Tatal	F 000	4 740	071	4 00=	0.000	24.2
Total	5,809	1,716	271	1,987	3,822	34.2
Male	2,222	1,033	182	1,215	1,008	54.7
Female	3,587	683	89	772	2,815	21.5

Table 4.1.10 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- ADDIS ABABA CITY ADMINISTRATION: 2012

	1 1 1 1 1 1 1 1					
Age Group	All		Activity conomically Activ			Activity
and Sex	Persons		CONDINICALLY ACTIV	/e	Economically	Rate
		Employed	Unemployed	Total Active	Not Active	
All Ages	'				<u>, </u>	
Total	2,636,013	1,250,842	374,626	1,625,468	1,010,545	61.7
Male	1,183,148	709,206	130,686	839,892	343,256	71.0
Female	1,452,865	541,635	243,940			54.1
	1,432,003	541,635	243,940	785,576	667,289	34.1
10-14			0.400	40.000	0.40.400	
Total	255,780	9,877	2,423	12,300	243,480	4.8
Male	110,897	2,198	1,384	3,582	107,315	3.2
Female	144,883	7,679	1,038	8,718	136,165	6.0
15-19						
Total	362,076	88,588	25,488	114,076	248,000	31.5
Male	132,116	20,908	8,923	29,832	102,284	22.6
Female	229,960	67,680	16,564	84,244	145,716	36.6
20-24	•	•	•	,	,	
Total	399,932	179,999	93,482	273,481	126,451	68.4
Male	162,924	81,956	35,314	117,270	45,654	72.0
Female	237,008	98,043	58,168	156,211	80,797	65.9
25-29	237,000	90,043	50,100	150,211	00,797	05.9
	400.000	070.005	07.505	005.040	07.050	04.5
Total	432,893	278,305	87,535	365,840	67,053	84.5
Male	198,252	154,679	28,830	183,509	14,742	92.6
Female	234,641	123,625	58,705	182,330	52,311	77.7
30-34						
Total	280,394	190,385	54,507	244,892	35,502	87.3
Male	147,522	121,817	18,026	139,844	7,678	94.8
Female	132,872	68,567	36,481	105,048	27,824	79.1
35-39	,	,	,	•	•	
Total	237,113	171,092	33,520	204,612	32,501	86.3
Male	121,145	106,224	9,473	115,698	5,447	95.5
Female	115,968	64,868	24,046	88,914	27,054	76.7
40-44	110,000	04,000	24,040	00,514	27,004	70.7
Total	146 255	104 512	22 442	126.025	10 421	86.7
	146,355	104,512	22,412	126,925	19,431	
Male	78,559	70,707	5,192	75,899	2,660	96.6
Female	67,796	33,805	17,220	51,025	16,771	75.3
45-49						
Total	119,931	77,829	17,560	95,390	24,541	79.5
Male	54,238	45,823	5,445	51,268	2,971	94.5
Female	65,692	32,006	12,115	44,122	21,571	67.2
50-54						
Total	105,520	65,169	13,738	78,907	26,613	74.8
Male	48,899	40,139	5,176	45,315	3,584	92.7
Female	56,622	25,029	8,562	33,592	23,030	59.3
55 - 59	00,022	20,020	0,002	00,002	20,000	00.0
Total	77,396	32,963	8,835	41,798	35,598	54.0
			4,331			
Male	33,874	24,076		28,407	5,467	83.9
Female	43,522	8,888	4,504	13,392	30,131	30.8
60-64	00.00=	co == :		64 =6:	0= 05=	
Total	66,907	23,754	7,770	31,524	35,382	47.1
Male	26,572	17,448	2,273	19,721	6,852	74.2
Female	40,334	6,306	5,497	11,804	28,531	29.3
65+						
Total	151,716	28,367	7,356	35,723	115,993	23.5
Male	68,150	23,230	6,318	29,547	38,603	43.4
Female	83,566	5,137	1,038	6,176	77,390	7.4
	20,000	5,.51	.,000	5,	,550	

Table 4.1.11 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- DIRE DAWA ADMINISTRATION: 2012

		,				
Age Group	All	E	Activity conomically Activ			Activity
and Sex	Persons		•		Economically	Rate
A II A a a a		Employed	Unemployed	Total Active	Not Active	
All Ages	204 520	101 201	20.000	121 102	70 000	64.4
Total Male	204,530 96,225	101,204 54,588	29,988 10,291	131,192 64,878	73,338 31,347	64.1 67.4
Female	108,305	46,617	19,697	66,314	41,990	61.2
10-14	100,303	40,017	19,097	00,314	41,990	01.2
Total	28,533	3,668	348	4,016	24,517	14.1
Male	15,019	1,925	87	2,012	13,007	13.4
Female	13,514	1,743	262	2,005	11,509	14.8
15-19	10,011	1,7 10	202	2,000	11,000	11.0
Total	31,331	9,016	3,132	12,148	19,183	38.8
Male	14,323	2,894	1,320	4,213	10,109	29.4
Female	17,008	6,122	1,812	7,934	9,074	46.7
20-24	,	-,	,-	,	-,-	
Total	30,749	14,629	7,164	21,793	8,956	70.9
Male	14,495	8,016	2,820	10,836	3,659	74.8
Female	16,254	6,613	4,345	10,957	5,297	67.4
25-29						
Total	28,683	19,103	5,599	24,701	3,982	86.1
Male	13,343	10,996	1,649	12,645	698	94.8
Female	15,341	8,106	3,950	12,056	3,284	78.6
30-34						
Total	17,563	12,430	3,735	16,164	1,399	92.0
Male	8,754	6,947	1,545	8,493	262	97.0
Female	8,809	5,482	2,189	7,671	1,137	87.1
35-39 Total	17 675	12 700	2,201	15 000	1,675	90.5
Male	17,675 8,414	13,798 7,625	2,201 440	15,999 8,065	350	95.8
Female	9,260	6,173	1,762	7,935	1,325	95.6 85.7
40-44	3,200	0,170	1,702	7,555	1,020	00.7
Total	9,350	6,678	1,888	8,566	784	91.6
Male	4,075	3,449	626	4,075	-	100.0
Female	5,275	3,229	1,262	4,491	784	85.1
45-49	•	,	•	•		
Total	12,481	8,513	2,648	11,161	1,320	89.4
Male	5,468	4,769	523	5,292	176	96.8
Female	7,013	3,744	2,125	5,869	1,144	83.7
50-54						
Total	8,894	5,853	1,109	6,962	1,932	78.3
Male	3,837	3,314	173	3,487	350	90.9
Female	5,057	2,539	935	3,475	1,582	68.7
55-59						 0.0
Total	5,996	3,668	1,118	4,785	1,211	79.8
Male	3,245	2,307	588	2,895	350	89.2
Female 60-64	2,751	1,361	529	1,890	861	68.7
Total	4,030	1,491	783	2,274	1,756	56.4
Male	1,750	791	434	1,224	525	70.0
Female	2,280	700	350	1,049	1,231	46.0
65+	2,200	, 50	230	1,0 70	1,201	10.0
Total	9,244	2,359	263	2,622	6,622	28.4
Male	3,502	1,554	87	1,641	1,861	46.9
Female	5,742	805	176	981	4,761	17.1

Table 4.2 Populations of Major Towns Aged Ten Years and Above by Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach): 2012

and Rate	during me	Last Seven Days (Current Status Approach): 2012					
	Δ.II		Activi	ty Status		A ativity	
Major Towns and Sex	All Persons	E.	conomically Activ	10	Economically	Activity Rate	
_	Persons		Unemployed		Not Active	Nate	
MA IOD TOWNS TOTAL		Employed	Unemployed	Total Active			
MAJOR TOWNS TOTAL	4 500 070	0.040.000	000 000	0.044.000	4 005 000	00.4	
Total	4,509,970	2,212,903	602,000	2,814,902	1,695,068	62.4	
Male	2,049,041	1,240,813	197,663	1,438,476	610,565	70.2	
Female	2,460,930	972,089	404,337	1,376,426	1,084,503	55.9	
MEKELE TOWN	0.44.000	101 110	00.044	454.000	00.070	00.0	
Total	241,332	121,116	29,944	151,060	90,272	62.6	
Male	105,549	65,727	7,951	73,678	31,872	69.8	
Female	135,783	55,389	21,993	77,382	58,401	57.0	
ASAYITA TOWN	10.010	0.000	0.40	0.040	4.004	07.7	
Total	13,612	8,369	849	9,218	4,394	67.7	
Male .	6,540	4,624	286	4,910	1,630	75.1	
Female	7,072	3,745	564	4,309	2,764	60.9	
GONDER TOWN							
Total	181,710	97,868	18,392	116,260	65,450	64.0	
Male	81,795	51,748	5,780	57,528	24,268	70.3	
Female	99,915	46,121	12,612	58,732	41,182	58.8	
DESSIE TOWN							
Total	102,107	47,543	13,755	61,298	40,809	60.0	
Male	45,608	26,410	4,136	30,546	15,062	67.0	
Female	56,499	21,133	9,618	30,752	25,747	54.4	
BAHIR DAR TOWN							
Total	163,850	93,781	21,606	115,388	48,462	70.4	
Male	71,488	46,806	5,672	52,479	19,009	73.4	
Female	92,362	46,975	15,934	62,909	29,453	68.1	
BISHOFTU TOWN							
Total	84,864	41,284	15,052	56,336	28,528	66.4	
Male	39,579	23,681	4,901	28,582	10,997	72.2	
Female	45,285	17,602	10,151	27,753	17,531	61.3	
ADAMA TOWN							
Total	191,245	92,946	29,899	122,846	68,399	64.2	
Male	90,429	55,237	8,097	63,334	27,095	70.0	
Female	100,816	37,709	21,802	59,511	41,304	59.0	
JIMMA TOWN							
Total	106,414	59,065	11,711	70,776	35,638	66.5	
Male	50,853	33,607	3,449	37,056	13,797	72.9	
Female	55,561	25,458	8,263	33,720	21,841	60.7	
SHASHEMENE TOWN							
Total	99,036	45,485	14,606	60,090	38,945	60.7	
Male	47,497	28,269	4,098	32,367	15,130	68.1	
Female	51,539	17,216	10,508	27,723	23,816	53.8	
JIJIGA TOWN							
Total	97,536	40,942	9,476	50,418	47,118	51.7	
Male	45,624	23,990	2,558	26,548	19,075	58.2	
Female	51,912	16,952	6,918	23,870	28,042	46.0	
ASOSA TOWN							
Total	30,002	17,931	2,189	20,119	9,883	67.1	
Male	14,734	10,159	803	10,962	3,772	74.4	
Female	15,268	7,772	1,385	9,157	6,111	60.0	

Table 4.2 Cont'd

	ΛII		Activity			
Major Towns and Sex	All Persons	Ed	conomically Activ	ve	Economically	Rate
	1 0130113	Employed	Unemployed	Total Active	Not Active	Nate
HAWASSA TOWN						
Total	232,077	127,071	21,428	148,499	83,578	64.0
Male	110,817	70,534	6,539	77,073	33,744	69.5
Female	121,260	56,537	14,889	71,426	49,835	58.9
GAMBELLA TOWN						
Total	37,981	17,486	1,506	18,992	18,989	50.0
Male	18,059	9,326	330	9,656	8,402	53.5
Female	19,923	8,160	1,176	9,336	10,587	46.9
HARAR TOWN						
Total	87,661	49,969	6,973	56,942	30,719	65.0
Male	41,096	26,901	2,086	28,987	12,109	70.5
Female	46,566	23,068	4,887	27,955	18,610	60.0
ADDIS ABABA CITY						
ADMINISTRATION						
Total	2,636,013	1,250,842	374,626	1,625,468	1,010,545	61.7
Male	1,183,148	709,206	130,686	839,892	343,256	71.0
Female	1,452,865	541,635	243,940	785,576	667,289	54.1
DIRE DAWA TOWN						
Total	204,530	101,204	29,988	131,192	73,338	64.1
Male	96,225	54,588	10,291	64,878	31,347	67.4
Female	108,305	46,617	19,697	66,314	41,990	61.2

Table 4.3 Economically Not Active Population of Urban Areas Aged Ten Years and Above by Age Grou Being Active During the Last Seven Days (Current Status Approach) - COUNTRY TOTAL: 201

	Reasons for Not Being Active									
Age Group and				Re		Being Activ	1			
sex	Economically	Home		_	Injury/		Too			
	Not Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Remittance		
All Ages										
Total	4,165,962	421,945	135,069	2,768,241		236,712	43,283			
Male	1,573,590	11,959	-	1,296,597	16,573		20,484			
Female	2,592,372	409,986	135,069	1,471,644	13,149	159,703	22,799	44,913		
10-14										
Total	1,371,977	16,140	-	1,299,633	3,027		43,283	,		
Male	648,689	1,223	-	619,005	2,468	3,247	20,484			
Female	723,289	14,917	-	680,628	559	2,403	22,799	357		
15-19										
Total	1,124,233	32,873	18,191	1,042,688	2,736		-	3,204		
Male	497,736	2,827	-	483,478	1,267	4,956	-	1,089		
Female	626,496	30,046	18,191	559,211	1,469	8,995	-	2,115		
20-24										
Total	443,318	66,164	42,890	302,637	2,441	15,859	-	3,902		
Male	158,634	1,859	-	145,463	965	6,347	-	635		
Female	284,684	64,305	42,890	157,175	1,476	9,513	-	3,266		
25-29										
Total	217,742	68,234	43,493	66,450	3,294	17,250	-	7,739		
Male	43,126	1,115	-	30,801	2,004	5,408	-	1,454		
Female	174,616	67,119	43,493	35,649	1,290	11,843	-	6,285		
30-34										
Total	111,063	49,188	16,553	14,374	1,189	17,751	-	4,021		
Male	19,705	1,153	· -	5,411	458	8,802	-	1,219		
Female	91,357	48,035	16,553	8,963	731	8,949	-	2,801		
35-39	,	,	,	,		,		,		
Total	105,369	51,575	9,931	7,795	2,783	19,697	_	7,759		
Male	15,571	1,458	-	2,149	2,272		-	1,062		
Female	89,797	50,117	9,931	5,647	511	13,619	-	6,697		
40-44	22,. 0.	55,	2,201	5,5	3	. 5,5 . 6		2,301		
Total	65,145	29,638	3,690	4,802	1,782	15,884	_	4,176		
Male	8,944	346	-	1,375	979	4,263	_	688		
Female	56,201	29,292	3,690	3,427	803	11,620	_	3,488		
remale	30,201	23,232	5,090	5,421	003	11,020	_	5,400		

Table 4.3 Cont'd

Table 4.3 Cont of	4							
Ago Group and				Re	easons for Not l	Being Acti	ve	
Age Group and sex	Economically Not Active	Home Maker	Pregnancy	Student	Injury/ Disabled	Illness	Too Young	Remittance
45-49								!
Total	73,052	32,392	322	6,681	1,936	22,227	-	4,902
Male	8,799	906	-	1,088	550	5,435	-	- 517
Female	64,252	31,486	322	5,593	1,386	16,792	-	4,385
50-54								!
Total	78,870	28,923	_	3,022	1,776	18,813	-	- 3,102
Male	9,955	962		523				- 275
Female	68,915	27,961		2,499	167	•		- 2,828
55-59	,	•		•		•		· !
Total	92,940	20,371	-	4,437	2,377	23,237	-	- 2,088
Male	12,210	-	_	-	272			- 725
Female	80,730	20,371	-	4,437	2,105	•		- 1,363
60-64	,	• •		•	•	- ,		, ,
Total	107,038	12,341	-	2,596	1,988	26,616	-	- 5,830
Male	20,560	-	_	613	•			- 1,039
Female	86,479	12,341	-	1,983		,		- 4,791
65+	,	,		•	•	- ,		•
Total	375,215	14,106	, <u> </u>	13,124	4,391	39,776	-	- 10,742
Male	129,661	109		6,691	3,013			4,204
Female	245,555	13,997		6,433		22,127		- 6,537

Table 4.4 Economically Not Active population of Urban Areas Aged Ten Years and Above by Region, Sex at for Not Being Active during the Last Seven Days (Current Status Approach): 2012

				F	Reason for N	Not Being /	Active		
Region and Sex		Home			lnjury/		Too		Old ag
	Total	Maker	Pregnancy	Student	Disabled	Illness	Young	Remittance	Pensione
COUNTRY TOTAL									
Total	4,165,962	421,945	135,069	2,767,894	29,723	236,712	45,216	58,921	401,64
Male	1,573,590	11,959	-	1,296,251	16,573	77,009	21,908	14,008	116,61
Female	2,592,372	409,986	135,069	1,471,644	13,149	159,703	23,307	44,913	285,03
TIGRAY REGION									
Total	307,931	31,382	9,387	201,239	1,136	24,290	1,899	3,319	32,03
Male	113,574	389	-	94,522	606	6,008	1,091	743	9,87
Female	194,356	30,993	9,387	106,717	530	18,282	808	2,577	22,16
AFFAR REGION									
Total	40,759	6,200	2,921	25,418	371	2,618	50	985	1,82
Male	15,034	37	-	12,254	59	1,556	38	144	79
Female	25,726	6,162	2,921	13,164	311	1,062	12	841	1,03
AMHARA REGION									
Total	702,578	78,293	21,309	451,137	8,997	51,677	3,535	9,482	68,64
Male	259,005	684	-	209,443	4,176	17,085	1,102	1,530	19,98
Female	443,573	77,609	21,309	241,694	4,820	34,592	2,432	7,952	48,65
OROMIA REGION									
Total	1,158,165	114,973	40,751	793,554	8,411	62,747	16,589	13,127	94,64
Male	443,207	3,604	· -	373,065	4,794	23,078	8,713	2,693	23,85
Female	714,959	111,368	40,751	420,489	3,616	39,669	7,875	10,434	70,79
SOMALI REGION	,	,	•	·	•	,	·		·
Total	196,892	30,156	6,246	132,034	1,227	7,657	3,591	3,488	10,22
Male	90,974	749	-	78,064	916	3,246	2,594	1,515	2,41
Female	105,918	29,406	6,246	53,970	310	4,411	997	1,973	7,81
BENISHANGUL	•	,	,	,		,		•	,
GUMUZ REGION									
Total	33,760	2,237	1,426	25,694	53	1,296	169	407	2,29
Male	13,365	-	-	11,964	-	447	169	31	68
Female	20,395	2,237	1,426	13,730	53	850	-	375	1,61
SNNP REGION									
Total	574,806	46,771	14,632	429,687	4,223	20,859	10,196	7,410	35,85
Male	236,660	1,241	-	203,539	3,081	7,704	4,329	1,643	14,08
Female	338,146	45,531	14,632	226,148	1,142	13,155	5,868	5,767	21,77
	,	- ,	,	-,	, -=	-,	- ,	-,	,

Table 4.4 Cont'd

	Reason for Not Being Active									
Region and Sex		Home			lnjury/		Too		Old age	
	Total	Maker	Pregnancy	Student	Disabled	Illness	Young	Remittance	Pensione	
GAMBELLA REGION			·						<u>.</u>	
Total	36,468	3,495	1,264	26,941	208	957	130	1,833	1,44	
Male	15,059	127	-	13,343	27	451	35	589	45	
Female	21,409	3,367	1,264	13,598	181	505	96	1,244	98	
HARARI REGION										
Total	30,719	3,122	528	19,804	299	2,910	98	503	3,19	
Male	12,109	89	-	9,874	54	1,145	-	49	73	
Female	18,610	3,033	528	9,930	244	1,765	98	454	2,46	
ADDIS ABABA CITY										
ADMINISTRATION										
Total	1,010,545	99,377	32,973	614,920	4,357	57,040	7,475	16,286	145,26	
Male	343,256	4,950	-	264,370	2,685	14,437	3,229	4,725	41,96	
Female	667,289	94,427	32,973	350,550	1,672	42,603	4,246	11,562	103,29	
DIRE DAWA										
ADMINISTRATION										
Total	73,338	5,939	3,632	47,466	444	4,660	1,483	2,081	6,22	
Male	31,347	87	-	25,813	173	1,851	608	347	1,76	
Female	41,990	5,853	3,632	21,654	270	2,809	875	1,734	4,45	

CHAPTER V

CHARACTERISTICS OF THE CURRENTLY URBAN EMPLOYED POPULATION

5. 1. Introduction

The previous chapter has defined the economically active population to be the sum of the employed and the unemployed population. This chapter presents statistical data on the size, distribution and characteristics of the currently employed population based on the definition given in chapter IV. The survey has collected the following information regarding the characteristics of the currently employed population:

- a) Main type of occupation;
- b) Main product or service of the establishment or industry;
- c) Employment status of the main activity and terms of employment;
- d) Reasons for absence from work for those employed persons who worked less than four hours or does not work at all but have formal job attachment during the reference period;
- e) Amount of earnings from paid employees;
- f) Hours worked per week and
- g) Whether employed persons were available or seeking to work additional hours.
- h) Sector of the economy (Formal or informal employment sector);

5.2. Employment to Population Ratio

According to the 18 Key Indicators of Labour Market (KILM) used by the ILO, employment to population ratio is calculated as a percentage of total employed persons to that of the working age population (ILO, 1999). The distribution of employment to population ratio of urban areas of each region by age group and sex is shown in Table 5.1. The data for employment to population ratio each selected major towns by age group and sex is presented in Table 5.2.

5.3. Occupation and Industry

In this survey, the currently employed persons were asked about the type of main activity or occupation and major product or service of the establishment or industry in which they were engaged during the survey reference period. The questions used to identify the type of occupation and industry, were left open ended so that they will be filled in with the fullest description and its code in the field. For those employed persons engaged in multiple activities, the activity that took the largest share of the respondents' time was taken as the main type of activity.

The responses of the descriptions and codes for the type of occupation and industry filled in by enumerators were further checked and edited at the head quarters using the National Occupational and Industrial Classifications (NOIC) code-book. The NOIC codes were adopted from the International Standard Classification of Occupation (ISCO-88) and International Standard Industrial Classification (ISIC, 1993) by taking into account the prevailing national socio-economic conditions.

The NOIC uses a 3-digit coding systems corresponding to 3 level classifications in successively finer detail. In the cases of occupation and industry, the levels are ordered as major divisions, sub major divisions, and minor divisions. The occupational classification was categorized into 9 major groups, 28 sub major groups and 113 minor groups. The industrial classification contains, 13, 60 and 159 major, sub major and minor divisions, respectively.

The distribution of employed population of urban areas by major occupational groups, age group and sex for the country total and each region is presented in Table 5.3 and 5.4, respectively. Table 5.5 presents the distribution of employed population by major occupation, sex for each selected major towns. The distribution of employed population by minor occupational groups for each region classified by sex appears in Table 5.6. One should note that in Table 5.6 and 5.10 major and sub major occupation and industrial classification are written in capital letter, while minor occupation and

industrial classification are written in small letter, respectively. Analogous information, which deals with the industrial divisions are also shown in Table 5.7 to 5.10.

5.4. Employment Status

Employment status of a person indicates the level of involvement and degree of decision-making in his main activity. In this survey employed persons were asked about the employment status of their main activity which was classified into: employer, paid employees, self-employed, unpaid family worker, apprentice, and member's of producers, cooperative and others categories. Paid employees were further disaggregated into employees-government, employees government parastatal, employees NGO's, employees private organization, employees domestic and other paid employees.

Table 5.11 and Table 5.12 present the distribution of status in employment by age group and sex for urban areas of country and each region. The distribution of status in employment for major towns classified by sex is presented in Tables 5.13. Table 5.14 contains data on major occupational groups cross classified by major industrial divisions and sex for urban areas of the country total. The status in employment by major industrial divisions and sex for a country total is shown in Table 5.15. Also, the figures for status in employment classified by major occupation and sex for country total are presented in Table 5.16.

5.5. Number of Hours worked

The absence of standard working hours and irregular nature of working days in the informal sectors or in the self employment, data collection as well as getting accurate data on hours of work were found to be difficult. In addition, the reliability of the data collection is likely to be affected due to memory lapse and lack of knowledge or information about the concepts of time on the part of the respondent. Thus, considering the inherent data collection problem on hours of work in developing

countries, the figures presented in these tables should be regarded as indicative and not the true levels of intensity of work.

The 2012 Urban Employment and Unemployment Survey included questions on the number of hours worked for two purposes. First, the response on the number of hours worked during the seven days prior to the date of the interview is used to classify the respondents as employed or not on the basis of the given criteria of the minimum required hours of work. As described in Chapter IV, persons who worked at least four hours or more, and those who were not worked or worked less than 4 hours but had a job to return to were considered as employed. The rest of persons were then subjected to other filtering questions for unemployment and inactivity. The second objective of including questions on number of hours worked was to gather data on intensity of work among the employed population.

In this survey, all persons who were engaged in productive activities during the seven days prior to the date of interview were asked to provide the number of hours they worked in all of their productive activities. The response was recorded for each of the days in a separate notebook so as to get the total actual hours of work during the reference week. Then the hours so obtained were added together for the week and finally filled in the main questionnaire.

According to the resolution concerning statistics on hours actually worked that adopted by the tenth ICLS in 1962 (ILO, 1976), in this survey, the hours actually worked included:

- a) Hours actually worked during normal periods of work and time spent for waiting for the market;
- b) Time worked in addition to hours worked during normal periods of work, and generally paid at higher rates than normal rates (overtime);
- c) Time spent at place of work on activities such as the preparation of the work place; repairing maintenance, preparing and cleaning of tools and others;
- d) Time spent at the place of work waiting or standing by for such reasons as lack of supply of work, break down of machinery, or accidents, or time spent at the

- place of wok during which no work is done but for which payment is made under a guaranteed employment contract and
- e) Time corresponding to short rest periods at the work place including tea and coffee breaks:

According to the resolution in recording the number of hours worked, care was taken to exclude hours paid for but not worked, such as paid annual leave, paid public holidays or paid sick leave. Also excluded are meal breaks, time spent on travel from home to work and vice versa /for those who have specified place of work/, and hours spent on household activities that were not considered as productive. For employed persons who were not at work during the seven days prior to the date of interview, the number of hours of work is recorded as zero. On the other hand, any time that employed persons have spent in productive activity in places other than work site is considered as working hour.

Table 5.17 presents the distribution of the number of hours worked during the last seven days by age group and sex at country urban level. Similarly, the distribution for each region classified by sex appears in Table 5.18. The distribution of number of hours worked by major industrial divisions, major occupation, and status in employment for country total disaggregated by sex are shown in Table 5.19, 5.20 and 5.21, in that order.

5.6. Earning from Paid Employment

Paid employment jobs are those jobs where the employees hold explicit (written or oral) or implicit employment contracts, which give them a basic remuneration. Some or all of the tools, capitals equipment, information systems and/or premises used by the employees may be owned by others, and the employees may work under direct supervision or according to the strict guidelines set by the owner(s) or persons in the owners' employment. Persons in "paid employment" jobs are typically remunerated by wages and salaries, but may be paid by commission from sales, by piece rates, bonuses or in kind payments such as food, housing or clothing. In this survey,

earnings for paid employees refer to gross remuneration and include bonus, overtime, allowances and other benefits that are obtained only from the main job.

Table 5.22 shows the distribution of paid employees disaggregated by types of payment and sex for urban areas of each region. The distribution of paid employees in urban areas by sex and period of payment and sex at country level is presented in Table 5.23. The distribution of paid employees in urban areas of the country classified by amount and period of payment and sex is presented in table 5.24. Paid employees of urban areas of the country by major industrial divisions, sex and amount of payment per month has also shown in Table 5.25.

5.7. Absence from Work

Employed persons that were considered to be absent from work were those who were not engaged in any productive activity or who work less than four hours during the seven days prior to the date of the interview but have formal job attachment. These persons were asked their main reason for being absent from work. Reasons for absence from work was classified as illness, holiday /vacation, Training/education, off-season, reduction in economic activity and "others" categories. The data presented in Table 5.26 indicates the distribution of employed population for their reasons for being absent from work by region and sex.

5.8. Availability to Work Additional Hours

This survey collected data on time related under employment in an attempt to capture the insufficiency of the hours of work of employed persons in relation to an alternative employment situation in which persons are willing and available to be engaged in. The response of employed persons depending on their availability and readiness or not to work for additional hours was coded as "yes" or "no". The response for additional hours available could be for another job (jobs) in addition to the current job; replacing

any of current jobs by another job with increased hours of work or longer time on the current job.

The distribution of employed population whether they were available or ready to work additional hours for urban areas of the country by age group and sex is shown in Table 5.27. Table 5.28 shows distribution of employed persons who sought or available to work additional hours together with the actual number of hours worked by age group and sex for the total urban areas of the country. Table 5.29 present the distribution of urban employed population of each region who were sought or available to work additional hours classified by type of work sought or available for and sex.

5.9. Formal and Informal Employment Sectors of the Economic Activity

For statistical purposes, the informal sector is regarded as a group of production units, which according to the definitions and classification provided in the United Nation System of National Accounts (SNA Rev.4), form part of the household sector as household enterprises or, equivalently, unincorporated enterprises owned by households. The informal sector is defined irrespective of the kind of work place where the productive activities are carried out, the extent of fixed capital used, the duration of the enterprise and its operation as main or secondary activity of the owner.

The figures here were not referring to the whole employed population. In this survey, according to the 15th ICLS recommendations, those employed persons who were engaged in subsistence farming and work in private households were exempted from the analysis of the formal and informal sectors of the economic activity.

This survey includes questions to identify the employment sector of the economy and some of the characteristics in which the employed persons are engaged in their main activity. However, employed persons whose employment status were classified as employee-government, employee-government parasitatal, employee-non-governmental organization, or members of producers' cooperative were not

necessary to ask such questions related to the activities of their sectors because they were considered as formal and later classified as being working in the formal sector. On the other hand, other employed persons whose employment status were employer, employee of private organization, self employed, and apprentice were asked the following questions whether the business /enterprise they were engaged in,

- has a book of account that shows the monthly income statement and balance sheet; or
- b) has business/enterprise license.

Based on the response to these three questions classification on sector of economy was made as: formal, informal or "not – identified". Employed persons who satisfy at least one of the above conditions were considered as working in the formal sector. For those who respond "no" for all of the three questions, the activity was taken as informal. For those who don't know the situation about their main activity/business/enterprise with respect to the above questions was labeled as "not – identified".

Table 5.30 presents the distribution of employed population of urban areas of the country in relation to the sector of the economy classified by age group and sex. The distribution of employed population by sector of economy, region and sex is presented in 5.31. Similar information for selected major town is shown in Table 5.32. Table 5.33-5.37 presents the cross classification of the informal sector by sex and region by some of the selected variables, such as major occupation and branches of economic sector, educational attainment, size of workers and type of working place. The distribution of informal sector by type of difficulties they faced during their work at national level is also given in Table 5.38. The last Table 5.39 has shown the distribution of total employed population by Mean number of hours worked with major branches of economic sector in each region and sex.

Table 5.1 Urban Population Aged Ten Years and Above by Region, Age Group, Sex and Employment to Population Ratio - 2012

Region and	То	tal Population	on .	Total E	mployed po	pulation	Employm	ent to Po	pulation
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
COUNTRY TO	OTAL								
All Ages	11,110,653	5,186,014	5,924,639	5,726,116	3,202,042	2,524,074	51.5	61.7	42.6
10-14	1,592,162	754,662	837,500	195,567	94,187	101,380	12.3	12.5	12.1
15-19	1,752,372	754,648	997,725	492,071	205,515	286,556	28.1	27.2	28.7
20-24	1,707,114	743,495	963,619	902,304	457,454	444,850	52.9	61.5	46.2
25-29	1,615,015	776,270	838,745	1,128,487	653,525	474,962	69.9	84.2	56.6
30-34	1,006,702	507,720	498,982	767,262	448,530	318,731	76.2	88.3	63.9
35-39	945,870	465,729	480,140	742,001	423,955	318,046	78.4	91.0	66.2
40-44	568,878	299,009	269,869	445,655	273,476	172,179	78.3	91.5	63.8
45-49	477,925	233,108	244,817	361,687	211,592	150,094	75.7	90.8	61.3
50-54	381,120	176,523	204,597	266,025	153,598	112,428	69.8	87.0	55.0
55-59	269,995	118,356	151,640	152,584	96,469	56,115	56.5	81.5	37.0
60-64	239,219	103,804	135,416	113,615	75,221	38,393	47.5	72.5	28.4
65+	554,280	252,690	301,589	158,859	108,519	50,340	28.7	42.9	16.7
TIGRAY REG	ION								
All Ages	794,578	344,053	450,525	391,891	204,483	187,408	49.3	59.4	41.6
10-14	107,902	52,790	55,112	7,272	3,165	4,107	6.7	6.0	7.5
15-19	139,083	56,861	82,223	36,408	13,795	22,613	26.2	24.3	27.5
20-24	125,551	46,449	79,102	65,371	28,440	36,931	52.1	61.2	46.7
25-29	114,701	48,065	66,636	79,187	41,064	38,123	69.0	85.4	57.2
30-34	67,877	34,111	33,766	53,581	31,791	21,790	78.9	93.2	64.5
35-39	60,067	28,032	32,036	47,215	26,320	20,895	78.6	93.9	65.2
40-44	34,911	18,240	16,672	27,380	16,618	10,762	78.4	91.1	64.5
45-49	30,600	12,166	18,434	21,678	10,994	10,684	70.8	90.4	58.0
50-54	25,692	11,299	14,393	18,603	10,153	8,450	72.4	89.9	58.7
55-59	19,663	6,742	12,921	11,114	6,098	5,016	56.5	90.5	38.8
60-64	21,192	8,015	13,177	11,191	7,009	4,182	52.8	87.4	31.7
65+	47,337	21,284	26,053	12,892	9,035	3,856	27.2	42.5	14.8
AFFAR REGI	ON								
All Ages	127,674	61,116	66,558	77,068	43,657	33,411	60.4	71.4	50.2
10-14	19,920	9,600	10,319	4,132	1,607	2,524	20.7	16.7	24.5
15-19	15,019	6,525	8,494	5,365	2,749	2,616	35.7	42.1	30.8
20-24	15,774	6,978	8,796	8,848	5,305	3,542	56.1	76.0	40.3
25-29	19,941	8,868	11,073	14,868	8,055	6,814	74.6	90.8	61.5
30-34	13,870	6,672	7,197	10,723	5,868	4,856	77.3	87.9	67.5
35-39	12,365	6,305	6,059	10,217	6,090	4,127	82.6	96.6	68.1
40-44	10,893	5,096	5,797	8,664	4,753	3,911	79.5	93.3	67.5
45-49	6,757	3,608	3,149	5,664	3,549	2,115	83.8	98.3	67.2
50-54	4,766	2,711	2,055	3,636	2,376	1,259	76.3	87.7	61.3
55-59	2,813	1,163	1,649	1,838	984	855	65.4	84.6	51.8
60-64	2,373	1,378	995	1,976	1,256	720	83.3	91.2	72.3
65+	3,185	2,210	975	1,137	1,066	72	35.7	48.2	7.4

Table 5.1 Cont'd

Region and	То	tal Population	on	Total Em	ployed por	oulation		oloymen ulation F	
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
					,	<u>.</u>			<u>, </u>
AMHARA REC		000 040	4 000 004	4 000 050	500.045	504.444	50.0	00.0	40.4
All Ages	1,976,070	890,049	1,086,021	1,063,958	562,815	501,144	53.8	63.2	46.1
10-14	282,456	127,488	154,967	46,894	20,712	26,182	16.6	16.2	16.9
15-19	318,539	145,969	172,570	106,615	50,539	56,076	33.5	34.6	32.5
20-24 25-29	318,284	128,423 128,579	189,861 148,529	171,226 201,112	80,185 111,705	91,040 89,407	53.8	62.4 86.9	48.0 60.2
30-34	277,108		76,304			56,119	72.6	90.3	73.5
35-39	151,672	75,368		124,150 122,684	68,031	57,615	81.9	88.3	68.0
40-44	158,480 87,769	73,725 44,189	84,756 43,580	•	65,070 40,336	•	77.4	91.3	63.0
	•			67,780	•	27,444	77.2		
45-49 50-54	92,767	42,951	49,815	72,389	39,013	33,376	78.0	90.8	67.0
50-54	74,978	29,271	45,707	53,685	26,813	26,871	71.6	91.6	58.8
55-59	53,720 43,478	20,891 21,274	32,828	32,020 24,872	17,775	14,245	59.6 57.2	85.1 79.4	43.4 36.0
60-64			22,204	40,532	16,883	7,989			
65+	116,818	51,920	64,897	40,532	25,753	14,779	34.7	49.6	22.8
OROMIA REG	ION								
All Ages	3,131,966	1,522,509	1,609,457	1,643,077	963,543	679,534	52.5	63.3	42.2
10-14	499,611	244,631	254,981	62,670	35,683	26,987	12.5	14.6	10.6
15-19	497,405	226,258	271,147	133,755	68,607	65,148	26.9	30.3	24.0
20-24	502,385	234,894	267,491	283,346	159,428	123,918	56.4	67.9	46.3
25-29	444,847	228,652	216,196	316,724	197,210	119,514	71.2	86.2	55.3
30-34	273,528	134,872	138,656	217,268	122,380	94,888	79.4	90.7	68.4
35-39	275,890	142,624	133,266	226,653	131,824	94,829	82.2	92.4	71.2
40-44	150,776	79,002	71,774	123,199	72,172	51,027	81.7	91.4	71.1
45-49	127,776	69,919	57,857	103,373	65,656	37,716	80.9	93.9	65.2
50-54	92,024	41,791	50,232	64,589	36,871	27,718	70.2	88.2	55.2
55-59	70,754	32,993	37,761	43,710	28,330	15,380	61.8	85.9	40.7
60-64	61,110	28,702	32,409	27,111	19,414	7,697	44.4	67.6	23.7
65+	135,860	58,172	77,688	40,679	25,966	14,713	29.9	44.6	18.9
SOMALI REG	ION								
All Ages	433,866	214,968	218,898	202,547	111,869	90,677	46.7	52.0	41.4
10-14	92,168	52,286	39,883	12,450	6,972	5,478	13.5	13.3	13.7
15-19	72,665	37,256	35,409	17,659	7,161	10,499	24.3	19.2	29.6
20-24	50,357	22,021	28,337	23,737	12,792	10,945	47.1	58.1	38.6
25-29	47,996	20,569	27,427	28,029	15,908	12,121	58.4	77.3	44.2
30-34	39,534	16,223	23,311	28,916	14,810	14,106	73.1	91.3	60.5
35-39	33,748	16,019	17,730	25,187	14,303	10,883	74.6	89.3	61.4
40-44	31,822	15,441	16,381	24,280	13,340	10,940	76.3	86.4	66.8
45-49	15,432	8,720	6,712	12,559	8,141	4,418	81.4	93.4	65.8
50-54	16,384	8,324	8,060	10,959	6,255	4,704	66.9	75.1	58.4
55-59	6,637	3,802	2,835	4,475	3,049	1,426	67.4	80.2	50.3
60-64	9,736	5,343	4,393	6,570	4,128	2,442	67.5	77.3	55.6
65+	17,386	8,964	8,422	7,724	5,009	2,715	44.4	55.9	32.2

Table 5.1 Cont'd

Region and Region
BENISHANGUL – GUMUZ REGION All Ages 111,784 55,116 56,668 72,518 39,854 32,664 64.9 72.3 57.8 10-14 16,948 8,487 8,461 4,905 2,252 2,653 28.9 26.5 31.4 15-19 21,192 9,435 11,758 9,040 3,944 5,096 42.7 41.8 43.3 20-24 16,117 6,818 9,299 10,926 5,274 5,652 67.8 77.4 60.8 25-29 15,425 8,363 7,062 13,202 7,909 5,293 85.6 94.6 74.3 30-34 9,605 4,251 5,354 8,311 4,051 4,260 86.5 95.3 79.8 35-39 9,626 4,686 4,940 8,322 4,620 3,702 86.5 98.6 75.0 40-44 6,375 4,231 2,144 5,608 4,050 1,558 88.0 95.7 72.3 45-49 4,542 2,754 1,788 4,220 2,730 1,490 92.9 99.1 83.3 50-54 2,868 1,269 1,599 2,542 1,269 1,273 88.6 100.0 79.6 55-59 2,041 1,230 810 1,789 1,197 592 87.7 97.3 73.0 60-64 2,214 840 1,375 1,306 811 495 59.0 96.6 36.0 65+ 4,831 2,753 2,078 2,346 1,746 600 48.6 63.4 28.8 SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.7 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.2 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.2 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 114,113 131,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 134,141 31,432 70,414 61,018 58.1 62.7 53.3 20-24 226,337 12,224 134,141 331,432 70,414 61,018 58.1 62.7 53.3 50.9 12,955 56.9 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.6 60-64 25,259 8,696 16,563 13,907 6,628 7,345 55.3 76.2 4
All Ages 111,784 55,116 56,668 72,518 39,854 32,664 64.9 72.3 57.6 10-14 16,948 8,487 8,461 4,905 2,252 2,653 28.9 26.5 31.4 15-19 21,192 9,435 11,758 9,040 3,944 5,096 42.7 41.8 43.2 20-24 16,117 6,818 9,299 10,926 5,274 5,652 67.8 77.4 60.8 25-29 15,425 8,363 7,062 13,202 7,909 5,293 85.6 94.6 74.9 30-34 9,605 4,251 5,354 8,311 4,051 4,260 86.5 95.3 79.6 35-39 9,626 4,686 4,940 8,322 4,620 3,702 86.5 98.6 75.6 40-44 6,375 4,231 2,144 5,608 4,050 1,558 88.0 95.7 72.4 45-49 4,542 2,754 1,788 4,220 2,730 1,490 92.9 99.1 83.5 50-54 2,868 1,269 1,599 2,542 1,269 1,273 88.6 100.0 79.6 55-59 2,041 1,230 810 1,789 1,197 592 87.7 97.3 73.6 60-64 2,214 840 1,375 1,306 811 495 59.0 96.6 36.6 65+ 4,831 2,753 2,078 2,346 1,746 600 48.6 63.4 28.5 SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.1 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.5 15.9 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.5 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.5 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.5 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.4 30.34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.4 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.6 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.5 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.6 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.5 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+ 61,238 33,091 28,1
10-14
15-19
20-24
25-29
30-34 9,605 4,251 5,354 8,311 4,051 4,260 86.5 95.3 79.6 35-39 9,626 4,686 4,940 8,322 4,620 3,702 86.5 98.6 75.0 40-44 6,375 4,231 2,144 5,608 4,050 1,558 88.0 95.7 72.7 45-49 4,542 2,754 1,788 4,220 2,730 1,490 92.9 99.1 83.3 50-54 2,868 1,269 1,599 2,542 1,269 1,273 88.6 100.0 79.6 55-59 2,041 1,230 810 1,789 1,197 592 87.7 97.3 73.0 60-64 2,214 840 1,375 1,306 811 495 59.0 96.6 36.0 65+ 4,831 2,753 2,078 2,346 1,746 600 48.6 63.4 28.9 SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.7 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.2 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.8 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.4 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.8 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.5 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.4 66+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
35-39 9,626 4,686 4,940 8,322 4,620 3,702 86.5 98.6 75.0 40-44 6,375 4,231 2,144 5,608 4,050 1,558 88.0 95.7 72.7 45-49 4,542 2,754 1,788 4,220 2,730 1,490 92.9 99.1 83.3 50-54 2,868 1,269 1,599 2,542 1,269 1,273 88.6 100.0 79.6 55-59 2,041 1,230 810 1,789 1,197 592 87.7 97.3 73.0 60-64 2,214 840 1,375 1,306 811 495 59.0 96.6 36.0 65+ 4,831 2,753 2,078 2,346 1,746 600 48.6 63.4 28.9 SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.1
40-44 6,375 4,231 2,144 5,608 4,050 1,558 88.0 95.7 72.3 45-49 4,542 2,754 1,788 4,220 2,730 1,490 92.9 99.1 83.3 50-54 2,868 1,269 1,599 2,542 1,269 1,273 88.6 100.0 79.6 55-59 2,041 1,230 810 1,789 1,197 592 87.7 97.3 73.0 60-64 2,214 840 1,375 1,306 811 495 59.0 96.6 36.0 65+ 4,831 2,753 2,078 2,346 1,746 600 48.6 63.4 28.9 SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.7 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.8 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.8 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.2 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.8 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.5 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.6 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
45-49
50-54 2,868 1,269 1,599 2,542 1,269 1,273 88.6 100.0 79.6 55-59 2,041 1,230 810 1,789 1,197 592 87.7 97.3 73.6 60-64 2,214 840 1,375 1,306 811 495 59.0 96.6 36.6 65+ 4,831 2,753 2,078 2,346 1,746 600 48.6 63.4 28.9 SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.1 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.5 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.5
55-59 2,041 1,230 810 1,789 1,197 592 87.7 97.3 73.0 60-64 2,214 840 1,375 1,306 811 495 59.0 96.6 36.0 65+ 4,831 2,753 2,078 2,346 1,746 600 48.6 63.4 28.5 SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.7 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.8 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.8 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2
60-64
SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.7 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.5 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.5 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.4 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7
SNNP REGION All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.7 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.5 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.5 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.4 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.5 45-49 59,963 29,788 30,174 </td
All Ages 1,531,368 743,394 787,974 837,386 466,330 371,057 54.7 62.7 47.7 10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.5 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.5 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.2 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.2 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.8 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.5 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.4 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6 65+
10-14 263,019 120,319 142,701 42,829 19,185 23,644 16.3 15.9 16.6 15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.5 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.5 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.4 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.5 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.6 50-54 43,200 25,912 17,288 36,324 23,720 12,604
15-19 272,312 115,837 156,475 80,799 33,118 47,681 29.7 28.6 30.5 20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.5 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.4 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.6 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.5 55-59 26,231 12,426 13,806 17,952 10,996 6,956
20-24 226,337 112,224 114,113 131,432 70,414 61,018 58.1 62.7 53.8 25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.2 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.8 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.9 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.2 60-64 25,259 8,696 16,563 13,973 6,628 7,345 <td< td=""></td<>
25-29 209,680 110,082 99,597 160,140 96,043 64,097 76.4 87.2 64.2 30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.6 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.5 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.2 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 </td
30-34 136,023 72,101 63,922 108,830 65,815 43,015 80.0 91.3 67.3 35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.6 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.9 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.4 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
35-39 125,953 56,987 68,966 104,541 54,629 49,912 83.0 95.9 72.4 40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.8 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.9 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.4 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
40-44 82,153 45,930 36,223 70,376 44,177 26,199 85.7 96.2 72.3 45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.8 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.5 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.4 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
45-49 59,963 29,788 30,174 49,333 27,665 21,667 82.3 92.9 71.8 50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.9 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.4 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
50-54 43,200 25,912 17,288 36,324 23,720 12,604 84.1 91.5 72.5 55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.4 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
55-59 26,231 12,426 13,806 17,952 10,996 6,956 68.4 88.5 50.4 60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
60-64 25,259 8,696 16,563 13,973 6,628 7,345 55.3 76.2 44.3 65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
65+ 61,238 33,091 28,147 20,857 13,939 6,919 34.1 42.1 24.6
GAMBELLA REGION
All Ages 75,144 34,340 40,804 35,656 18,797 16,859 47.4 54.7 41.3
10-14 14,388 7,363 7,025 315 163 152 2.2 2.2 2.2
15-19 11,575 5,183 6,392 1,954 742 1,212 16.9 14.3 19.0
20-24 11,385 3,883 7,501 6,301 2,589 3,712 55.3 66.7 49.5
25-29 12,534 5,776 6,758 8,852 4,828 4,023 70.6 83.6 59.5
30-34 7,126 3,544 3,582 5,252 3,114 2,138 73.7 87.9 59.7
35-39 6,713 3,248 3,465 5,442 3,157 2,284 81.1 97.2 65.9
40-44 3,466 1,727 1,739 2,800 1,551 1,249 80.8 89.8 71.9
45-49 2,629 1,091 1,538 2,306 1,036 1,270 87.7 95.0 82.5
50-54 2,624 1,345 1,279 1,537 1,025 511 58.6 76.2 40.0
55-59 802 386 416 382 274 108 47.7 71.1 26.0
60-64 1,048 372 676 266 129 137 25.4 34.6 20.3
65+ 855 421 434 250 188 62 29.2 44.8 14.2

Table 5.1 Cont'd

Region and	То	tal Population	on	Total Em	ployed pop	oulation		oloymer ulation F	
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
	1011								
HARARI REG		44.000	40.500	40.000	00 004	00.000	57. 0	05.5	40.5
All Ages	87,661	41,096	46,566	49,969	26,901	23,068	57.0	65.5	49.5
10-14	11,438	5,782	5,656	557	325	231	4.9	5.6	4.1
15-19	11,174	4,885	6,290	2,870	1,058	1,812	25.7	21.7	28.8
20-24	10,243	4,385	5,858	6,489	3,054	3,435	63.4	69.6	58.6
25-29	11,207	5,722	5,485	8,966	5,127	3,839	80.0	89.6	70.0
30-34	9,509	4,300	5,209	7,415	3,905	3,510	78.0	90.8	67.4
35-39	8,241	4,545	3,696	6,850	4,093	2,757	83.1	90.1	74.6
40-44	5,006	2,519	2,487	4,377	2,322	2,055	87.4	92.2	82.6
45-49	5,047	2,403	2,644	3,823	2,216	1,607	75.8	92.2	60.8
50-54	4,171	1,865	2,306	3,129	1,661	1,469	75.0	89.0	63.7
55-59	3,943	1,604	2,339	2,672	1,382	1,290	67.8	86.2	55.1
60-64	1,872	862	1,010	1,104	723	381	59.0	83.9	37.7
65+	5,809	2,222	3,587	1,716	1,033	683	29.5	46.5	19.0
ADDIS ABAB ADMINISTRA									
All Ages		1,183,148	1,452,865	1,250,842	709,206	541,635	47.5	59.9	37.3
10-14	255,780	110,897	144,883	9,877	2,198	7,679	3.9	2.0	5.3
15-19	362,076	132,116	229,960	88,588	20,908	67,680	24.5	15.8	29.4
20-24	399,932	162,924	237,008	179,999	81,956	98.043	45.0	50.3	41.4
25 - 29	432,893	198,252	234,641	278,305	154,679	123,625	64.3	78.0	52.7
30-34	280,394	147,522	132,872	190,385	121,817	68,567	67.9	82.6	52. <i>1</i> 51.6
35-39	237,113	121,145	115,968	171,092		64,868	72.2	87.7	55.9
40-44	146,355	78,559	67,796	171,092	106,224 70,707	33,805	72.2	90.0	49.9
45-49	119,931	54,238	65,692	77,829	45,823	32,006	64.9	84.5	48.7
50-54		48,899					61.8	82.1	44.2
	105,520		56,622	65,169	40,139	25,029			
55-59	77,396	33,874	43,522	32,963	24,076	8,888	42.6	71.1	20.4
60-64 65+	66,907 151,716	26,572 68,150	40,334 83,566	23,754 28,367	17,448 23,230	6,306 5,137	35.5 18.7	65.7 34.1	15.6 6.1
001	101,710	00,100	00,000	20,007	20,200	3,137	10.7	04.1	0.1
DIRE DAWA									
All Ages	204,530	96,225	108,305	101,204	54,588	46,617	49.5	56.7	43.0
10-14	28,533	15,019	13,514	3,668	1,925	1,743	12.9	12.8	12.9
15-19	31,331	14,323	17,008	9,016	2,894	6,122	28.8	20.2	36.0
20-24	30,749	14,495	16,254	14,629	8,016	6,613	47.6	55.3	40.7
25-29	28,683	13,343	15,341	19,103	10,996	8,106	66.6	82.4	52.8
30-34	17,563	8,754	8,809	12,430	6,947	5,482	70.8	79.4	62.2
35-39	17,675	8,414	9,260	13,798	7,625	6,173	78.1	90.6	66.7
40-44	9,350	4,075	5,275	6,678	3,449	3,229	71.4	84.6	61.2
45-49	12,481	5,468	7,013	8,513	4,769	3,744	68.2	87.2	53.4
50-54	8,894	3,837	5,057	5,853	3,314	2,539	65.8	86.4	50.2
55-59	5,996	3,245	2,751	3,668	2,307	1,361	61.2	71.1	49.5
60-64	4,030	1,750	2,280	1,491	791	700	37.0	45.2	30.7
65+	9,244	3,502	5,742	2,359	1,554	805	25.5	44.4	14.0

Table 5.2 Population of Major Towns Aged Ten Years and Above by Age Group, Sex and Employment to Population Ratio: 2012

–		_					Employm	ent to Por	oulation
Major Towns and	Тс	otal Populati	on	I otal Em	nployed Pop	ulation		Ratio	
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
TOTAL MAJOR	TOWNS								
All Ages		2,049,041	2,460,930	2,212,903	1,240,813	972,089	49.1	60.6	39.5
10-14	516,248	235,476	280,772	35,832	12,838	22,994	6.9	5.5	8.2
15-19	665,663	257,590	408,073	175,031	51,416	123,615	26.3	20.0	30.3
20-24	699,159	290,398	408,761	340,249	161,132	179,116	48.7	55.5	43.8
25-29	722,327	331,677	390,650	477,031	266,654	210,377	66.0	80.4	53.9
30-34	444,395	231,983	212,412	314,119	196,791	117,328	70.7	84.8	55.2
35-39	394,588	201,299	193,290	295,793	179,606	116,187	75.0	89.2	60.1
40-44	243,657	130,504	113,153	180,491	117,873	62,619	74.1	90.3	55.3
45-49	201,756	95,196	106,560	138,459	83,231	55,228	68.6	87.4	51.8
50-54	167,019	76,066	90,954	106,614	64,308	42,306	63.8	84.5	46.5
55-59	117,048	52,214	64,833	56,364	39,099	17,265	48.2	74.9	26.6
60-64	105,188	43,240	61,948	41,448	29,314	12,135	39.4	67.8	19.6
65+	232,923	103,399	129,524	51,470	38,550	12,920	22.1	37.3	10.0
MEKELE TOWN									
All Ages	241,332	105,549	135,783	121,116	65,727	55,389	50.2	62.3	40.8
10-14	33,092	15,108	17,984	2,150	436	1,714	6.5	2.9	9.5
15-19	40,610	16,169	24,441	11,406	3,199	8,207	28.1	19.8	33.6
20-24	37,359	12,823	24,536	19,140	8,151	10,989	51.2	63.6	44.8
25-29	39,697	15,018	24,679	24,908	12,232	12,676	62.7	81.5	51.4
30-34	20,375	10,885	9,491	15,890	10,347	5,543	78.0	95.1	58.4
35-39	21,990	12,657	9,334	17,921	11,791	6,130	81.5	93.2	65.7
40-44	13,956	8,704	5,252	12,166	8,272	3,894	87.2	95.0	74.1
45-49	8,219	3,417	4,802	5,498	3,312	2,186	66.9	96.9	45.5
50-54	6,829	3,173	3,656	4,876	3,062	1,813	71.4	96.5	49.6
55-59	3,779	1,435	2,345	2,181	1,326	856	57.7	92.4	36.5
60-64	4,748	1,823	2,925	2,038	1,505	533	42.9	82.6	18.2
65+	10,678	4,340	6,338	2,941	2,093	848	27.5	48.2	13.4
ASAYITA TOWN									
All Ages	13,612	6,540	7,072	8,369	4,624	3,745	61.5	70.7	53.0
10-14	1,423	669	754	251	143	108	17.6	21.4	14.3
15-19	1,979	865	1,114	648	265	384	32.8	30.6	34.4
20-24	2,229	910	1,319	1,223	622	601	54.9	68.4	45.5
25-29	2,341	1,245	1,096	1,956	1,112	844	83.6	89.3	77.0
30-34	1,340	611	729	1,040	574	466	77.6	94.0	64.0
35-39	1,224	613	610	973	542	431	79.5	88.3	70.7
40-44	731	370	361	588	347	240	80.4	93.8	66.6
45-49	718	384	334	563	360	203	78.4	93.8	60.7
50-54	528	192	337	397	168	229	75.1	87.5	68.0
55-59	347	180	167	240	144	96	69.0	80.1	57.1
60-64	252	156	96	216	144	72	85.7	92.3	75.0
65+	500	345	155	274	202	72	54.8	58.7	46.2

Table 5.2 Cont'd

Major Towns		Total Population			ployed Pop	ulation	Employme		oulation
and					· · ·			Ratio	
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
GONDER TOV	VN								
All Ages	181,710	81,795	99,915	97,868	51,748	46,121	53.9	63.3	46.2
10-14	29,886	14,056	15,830	7,924	3,908	4,015	26.5	27.8	25.4
15-19	29,160	12,023	17,137	9,838	4,075	5,763	33.7	33.9	33.6
20-24	27,936	10,476	17,460	15,476	6,702	8,775	55.4	64.0	50.3
25-29	25,247	13,445	11,803	17,499	10,272	7,228	69.3	76.4	61.2
30-34	13,321	6,672	6,649	10,259	5,356	4,903	77.0	80.3	73.7
35-39	12,815	5,911	6,904	10,310	5,635	4,675	80.5	95.3	67.7
40-44	9,737	4,442	5,295	6,729	4,149	2,580	69.1	93.4	48.7
45-49	8,225	3,927	4,298	6,367	3,357	3,010	77.4	85.5	70.0
50-54	5,556	2,276	3,280	3,739	2,153	1,586	67.3	94.6	48.4
55-59	4,870	1,728	3,142	2,747	1,553	1,193	56.4	89.9	38.0
60-64	4,839	2,065	2,775	3,094	1,838	1,257	63.9	89.0	45.3
65+	10,116	4,774	5,342	3,885	2,750	1,135	38.4	57.6	21.2
DESSIE TOW									
All Ages	102,107	45,608	56,499	47,543	26,410	21,133	46.6	57.9	37.4
10-14	12,131	5,709	6,422	704	54	650	5.8	0.9	10.1
15-19	18,638	7,850	10,788	4,235	1,889	2,345	22.7	24.1	21.7
20-24	16,161	7,588	8,573	8,198	4,659	3,539	50.7	61.4	41.3
25-29	12,735	5,907	6,828	8,749	5,041	3,709	68.7	85.3	54.3
30-34	7,238	3,928	3,310	5,638	3,400	2,238	77.9	86.5	67.6
35-39	8,958	3,985	4,973	6,493	3,458	3,035	72.5	86.8	61.0
40-44	4,915	2,419	2,495	3,370	2,166	1,205	68.6	89.5	48.3
45-49	4,026	1,839	2,186	2,727	1,626	1,102	67.7	88.4	50.4
50-54	4,372	1,124	3,248	2,619	1,070	1,550	59.9	95.2	47.7
55-59	2,488	1,008	1,479	1,309	810	499	52.6	80.3	33.8
60-64	3,158	1,471	1,687	1,613	1,209	404	51.1	82.2	24.0
65+	7,288	2,778	4,510	1,888	1,030	858	25.9	37.1	19.0
BAHIR DAR T	_								
All Ages	163,850	71,488	92,362	93,781	46,806	46,975	57.2	65.5	50.9
10-14	16,691	7,989	8,703	1,612	755	858	9.7	9.4	9.9
15-19	27,214	9,586	17,628	10,189	2,309	7,880	37.4	24.1	44.7
20-24	34,767	11,556	23,211	19,809	6,841	12,968	57.0	59.2	55.9
25-29	28,491	14,395	14,096	21,225	13,012	8,213	74.5	90.4	58.3
30-34	13,613	6,251	7,363	10,818	5,965	4,853	79.5	95.4	65.9
35-39	12,530	6,321	6,209	11,050	5,812	5,238	88.2	91.9	84.4
40-44	7,093	4,313	2,780	6,066	3,909	2,157	85.5	90.6	77.6
45-49	6,157	3,050	3,107	4,861	2,899	1,962	79.0	95.1	63.1
50-54	4,917	2,470	2,447	3,505	2,090	1,415	71.3	84.6	57.8
55-59	3,723	1,538	2,185	2,057	1,362	695	55.2	88.5	31.8
60-64	2,936	1,318	1,618	1,271	902	369	43.3	68.4	22.8
65+	5,716	2,701	3,015	1,318	951	367	23.1	35.2	12.2

Table 5.2 Cont'd

Major Towns and		al Populatio	n	Total Emp	oloyed Pop	ulation	Employme	ent to Pop Ratio	oulation
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
BISHOFTU TO	OWN								
All Ages	84,864	39,579	45,285	41,284	23,681	17,602	48.6	59.8	38.9
10-14	10,014	4,805	5,209	413	240	173	4.1	5.0	3.3
15-19	12,106	4,541	7,565	2,454	940	1,514	20.3	20.7	20.0
20-24	13,334	6,614	6,720	7,162	4,056	3,107	53.7	61.3	46.2
25-29	14,165	6,769	7,397	8,941	5,298	3,643	63.1	78.3	49.3
30-34	7,780	4,745	3,035	6,197	4,241	1,956	79.7	89.4	64.4
35-39	7,702	3,640	4,062	5,855	3,165	2,691	76.0	87.0	66.2
40-44	3,729	1,705	2,024	2,891	1,513	1,378	77.5	88.88	68.1
45-49	3,318	1,441	1,876	2,429	1,328	1,101	73.2	92.1	58.7
50-54	3,226	1,071	2,155	1,801	829	972	55.8	77.3	45.1
55-59	1,835	884	951	929	665	264	50.7	75.3	27.8
60-64	2,369	900	1,469	860	498	361	36.3	55.4	24.6
65+	5,285	2,465	2,821	1,351	908	443	25.6	36.8	15.7
ADAMA TOW									
All Ages	191,245	90,429	100,816	92,946	55,237	37,709	48.6	61.1	37.4
10-14	24,629	12,107	12,522	725	177	548	2.9	1.5	4.4
15-19	31,831	15,172	16,658	8,782	3,959	4,823	27.6	26.1	29.0
20-24	28,180	12,215	15,964	15,660	8,808	6,852	55.6	72.1	42.9
25-29	33,229	13,747	19,483	21,071	11,606	9,465	63.4	84.4	48.6
30-34	18,155	9,337	8,819	13,135	8,319	4,816	72.3	89.1	54.6
35-39	16,370	9,236	7,134	12,520	8,298	4,222	76.5	89.8	59.2
40-44	9,766	5,168	4,598	6,850	4,360	2,490	70.1	84.4	54.1
45-49	8,394	4,955	3,439	6,212	4,558	1,654	74.0	92.0	48.1
50-54	5,139	2,181	2,958	3,181	1,889	1,291	61.9	86.6	43.6
55-59	3,322	1,267	2,055	1,528	1,000	528	46.0	79.0	25.7
60-64	4,048	1,318	2,730	1,388	842	547	34.3	63.9	20.0
65+	8,181	3,726	4,456	1,894	1,420	474	23.2	38.1	10.6
JIMMA TOWN									
All Ages	106,414	50,853	55,561	59,065	33,607	25,458	55.5	66.1	45.8
10-14	14,612	6,477	8,135	2,228	764	1,464	15.2	11.8	18.0
15-19	15,866	6,250	9,616	5,245	2,016	3,228	33.1	32.3	33.6
20-24	18,152	8,544	9,608	10,760	5,747	5,013	59.3	67.3	52.2
25-29	15,919	8,164	7,756	11,860	7,093	4,767	74.5	86.9	61.5
30-34	10,015	4,925	5,090	7,246	4,517	2,729	72.4	91.7	53.6
35-39	9,458	4,998	4,460	7,825	4,662	3,163	82.7	93.3	70.9
40-44	5,128	2,784	2,344	4,139	2,486	1,654	80.7	89.3	70.6
45-49	4,292	2,463	1,829	3,541	2,363	1,179	82.5	95.9	64.4
50-54	3,887	1,883	2,005	2,489	1,632	857	64.0	86.7	42.8
55-59	2,743	1,415	1,329	1,564	1,105	458	57.0	78.2	34.5
60-64	2,016	870	1,146	935	640	295	46.4	73.5	25.8
65+	4,325	2,081	2,244	1,232	581	651	28.5	27.9	29.0

Table 5.2 Cont'd

Major Towns and	Tota	al Populatio	n	Total Emp	oloyed Pop	ulation	Employme	ent to Pop Ratio	oulation
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
SHASHEMENI	E TOWN								
All Ages	99,036	47,497	51,539	45,485	28,269	17,216	45.9	59.5	33.4
10-14	16,042	7,205	8,836	739	352	387	4.6	4.9	4.4
15-19	15,852	7,126	8,726	3,553	1,875	1,678	22.4	26.3	19.2
20-24	14,601	7,099	7,502	7,381	4,541	2,840	50.6	64.0	37.9
25-29	16,549	7,446	9,103	10,326	6,113	4,213	62.4	82.1	46.3
30-34	10,212	5,103	5,109	7,307	4,640	2,668	71.6	90.9	52.2
35-39	7,969	4,415	3,554	5,790	3,708	2,082	72.7	84.0	58.6
40-44	5,606	2,985	2,621	4,167	2,805	1,362	74.3	94.0	52.0
45-49	3,385	1,787	1,598	2,487	1,622	865	73.5	90.8	54.1
50-54	2,714	1,221	1,493	1,495	954	541	55.1	78.1	36.2
55-59	1,284	726	558	729	590	139	56.8	81.3	24.9
60-64	1,817	1,059	758	777	547	230	42.8	51.7	30.3
65+	3,004	1,324	1,680	734	522	212	24.4	39.4	12.6
JIJIGA TOWN									
All Ages	97,536	45,624	51,912	40,942	23,990	16,952	42.0	52.6	32.7
10-14	18,192	9,713	8,479	1,130	369	761	6.2	3.8	9.0
15-19	13,559	6,247	7,311	2,918	1,113	1,805	21.5	17.8	24.7
20-24	13,107	5,445	7,662	4,788	2,397	2,391	36.5	44.0	31.2
25-29	13,446	5,259	8,187	7,819	4,356	3,464	58.2	82.8	42.3
30-34	9,503	4,358	5,145	5,908	3,577	2,331	62.2	82.1	45.3
35-39	8,902	4,263	4,639	5,703	3,679	2,023	64.1	86.3	43.6
40-44	6,272	3,465	2,808	4,387	3,093	1,294	69.9	89.3	46.1
45-49	4,736	2,826	1,910	3,512	2,564	948	74.2	90.7	49.6
50-54	3,107	1,357	1,749	1,948	1,042	906	62.7	76.8	51.8
55-59	1,683	1,101	581	1,275	845	430	75.8	76.7	74.0
60-64	2,084	700	1,384	694	453	241	33.3	64.7	17.4
65+	2,946	889	2,057	859	502	357	29.2	56.4	17.4
ASOSSA TOW	/N								
All Ages	30,002	14,734	15,268	17,931	10,159	7,772	59.8	68.9	50.9
10-14	4,382	2,252	2,131	744	345	398	17.0	15.3	18.7
15-19	5,345	1,824	3,520	1,845	500	1,346	34.5	27.4	38.2
20-24	5,344	2,204	3,140	3,089	1,597	1,491	57.8	72.5	47.5
25-29	4,715	2,453	2,262	3,840	2,175	1,665	81.4	88.7	73.6
30-34	2,978	1,597	1,381	2,608	1,575	1,034	87.6	98.6	74.9
35-39	2,768	1,767	1,001	2,533	1,701	832	91.5	96.3	83.1
40-44	1,563	1,107	457	1,346	1,006	340	86.1	90.9	74.5
45-49	921	640	281	716	617	99	77.7	96.3	35.3
50-54	586	274	311	475	274	201	81.1	100.0	64.4
55-59	476	199	277	366	166	200	76.8	83.2	72.2
60-64	298	144	155	195	115	80	65.3	80.2	51.4
65+	626	273	354	175	88	87	28.0	32.4	24.5
331	020	210	JU- 1	175	00	01	20.0	JZ. T	27.0

Table 5.2 Cont'd

Table 5.2 Co	nta								
Major Towns and	Tot	al Populatio	n	Total Emp	oloyed Pop	ulation	Employme	nt to Po _l Ratio	oulation
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
HAWASSA TO	OWN								
All Ages	232,077	110,817	121,260	127,071	70,534	56,537	54.8	63.6	46.6
10-14	33,063	14,194	18,870	2,915	762	2,153	8.8	5.4	11.4
15-19	42,321	15,535	26,786	12,591	4,193	8,398	29.8	27.0	31.4
20-24	41,354	21,109	20,246	23,475	12,974	10,501	56.8	61.5	51.9
25-29	36,677	17,431	19,246	27,900	14,979	12,921	76.1	85.9	67.1
30-34	19,046	11,200	7,847	15,419	10,135	5,284	81.0	90.5	67.3
35-39	17,323	7,645	9,678	14,396	7,559	6,836	83.1	98.9	70.6
40-44	12,676	7,107	5,568	10,815	6,535	4,280	85.3	92.0	76.9
45-49	10,436	5,635	4,801	8,234	5,151	3,083	78.9	91.4	64.2
50-54	6,553	3,695	2,858	5,376	3,618	1,758	82.0	97.9	61.5
55-59	2,628	1,772	856	1,823	1,562	261	69.4	88.1	30.5
60-64	3,325	2,087	1,238	1,900	1,566	333	57.1	75.1	26.9
65+	6,674	3,407	3,266	2,227	1,497	730	33.4	43.9	22.3
00+	0,074	3,407	3,200	2,221	1,497	730	33. 4	43.9	22.3
GAMBELLA T	OWN								
All Ages	37,981	18,059	19,923	17,486	9,326	8,160	46.0	51.6	41.0
10-14	6,340	3,495	2,845	194	83	111	3.1	2.4	3.9
15-19	6,602	3,079	3,523	853	223	630	12.9	7.2	17.9
20-24	5,709	2,010	3,699	2,969	1,010	1,958	52.0	50.3	52.9
25-29	6,332	3,083	3,249	4,563	2,565	1,999	72.1	83.2	61.5
30-34	3,352	1,795	1,557	2,425	1,475	950	72.3	82.1	61.0
35-39	3,550	1,743	1,807	2,683	1,653	1,030	75.6	94.8	57.0
40-44	1,773	781	992	1,409	753	656	79.5	96.4	66.1
45-49	1,469	720	748	1,145	665	480	78.0	92.4	64.1
50-54	1,018	548	470	563	414	149	55.2	75.5	31.7
55-59	533	237	296	314	206	108	58.8	86.6	36.5
60-64	489	146	343	119	92	27	24.3	63.0	7.9
65+	814	421	393	250	188	62	30.7	44.8	15.7
HARAR TOW	N								
All Ages	87,661	41,096	46,566	49,969	26,901	23,068	57.0	65.5	49.5
10-14	11,438	5,782	5,656	557	325	231	4.9	5.6	4.1
15-19	11,174	4,885	6,290	2,870	1,058	1,812	25.7	21.7	28.8
20-24	10,243	4,385	5,858	6,489	3,054	3,435	63.4	69.6	58.6
25-29	11,207	5,722	5,485	8,966	5,127	3,839	80.0	89.6	70.0
30-34	9,509	4,300	5,209	7,415	3,905	3,510	78.0	90.8	67.4
35-39	8,241	4,545	3,696	6,850	4,093	2,757	83.1	90.1	74.6
40-44	5,006	2,519	2,487	4,377	2,322	2,055	87.4	92.2	82.6
45-49	5,047	2,403	2,644	3,823	2,216	1,607	75.8	92.2	60.8
50-54	4,171	1,865	2,306	3,129	1,661	1,469	75.0	89.0	63.7
55-59	3,943	1,604	2,339	2,672	1,382	1,290	67.8	86.2	55.1
60-64	1,872	862	1,010	1,104	723	381	59.0	83.9	37.7
65+	5,809	2,222	3,587	1,716	1,033	683	29.5	46.5	19.0
	•	•	,	•	,			-	_

Table 5.2 Cont'd

Major Towns and	То	tal Populati	on	Total Em	oloyed Pop	ulation	Employme	ent to Pop Ratio	oulation
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
ADDIS ABAB	A CITY								
All Ages	2,636,013	1,183,148	1,452,865	1,250,842	709,206	541,635	47.5	59.9	37.3
10-14	255,780	110,897	144,883	9,877	2,198	7,679	3.9	2.0	5.3
15-19	362,076	132,116	229,960	88,588	20,908	67,680	24.5	15.8	29.4
20-24	399,932	162,924	237,008	179,999	81,956	98,043	45.0	50.3	41.4
25-29	432,893	198,252	234,641	278,305	154,679	123,625	64.3	78.0	52.7
30-34	280,394	147,522	132,872	190,385	121,817	68,567	67.9	82.6	51.6
35-39	237,113	121,145	115,968	171,092	106,224	64,868	72.2	87.7	55.9
40-44	146,355	78,559	67,796	104,512	70,707	33,805	71.4	90.0	49.9
45-49	119,931	54,238	65,692	77,829	45,823	32,006	64.9	84.5	48.7
50-54	105,520	48,899	56,622	65,169	40,139	25,029	61.8	82.1	44.2
55-59	77,396	33,874	43,522	32,963	24,076	8,888	42.6	71.1	20.4
60-64	66,907	26,572	40,334	23,754	17,448	6,306	35.5	65.7	15.6
65+	151,716	68,150	83,566	28,367	23,230	5,137	18.7	34.1	6.1
DIRE DAWA 1	OWN								
All Ages	204,530	96,225	108,305	101,204	54,588	46,617	49.5	56.7	43.0
10-14	28,533	15,019	13,514	3,668	1,925	1,743	12.9	12.8	12.9
15-19	31,331	14,323	17,008	9,016	2,894	6,122	28.8	20.2	36.0
20-24	30,749	14,495	16,254	14,629	8,016	6,613	47.6	55.3	40.7
25-29	28,683	13,343	15,341	19,103	10,996	8,106	66.6	82.4	52.8
30-34	17,563	8,754	8,809	12,430	6,947	5,482	70.8	79.4	62.2
35-39	17,675	8,414	9,260	13,798	7,625	6,173	78.1	90.6	66.7
40-44	9,350	4,075	5,275	6,678	3,449	3,229	71.4	84.6	61.2
45-49	12,481	5,468	7,013	8,513	4,769	3,744	68.2	87.2	53.4
50-54	8,894	3,837	5,057	5,853	3,314	2,539	65.8	86.4	50.2
55-59	5,996	3,245	2,751	3,668	2,307	1,361	61.2	71.1	49.5
60-64	4,030	1,750	2,280	1,491	791	700	37.0	45.2	30.7
65+	9,244	3,502	5,742	2,359	1,554	805	25.5	44.4	14.0

Table 5.3 Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, 9 Groups – COUNTRY TOTAL: 2012

Major Occupational Groups Service, Age Group Legislator Shop & and Sex Skilled Crafts and Total Senior Technical and Market **Employed** Officials & Associate sales agricultural related Person Managers Professional Professionals Clerks Workers and Fishery trades All Ages Total 165,216 343,555 1,048,027 5,726,116 376,286 359,728 316,379 1,516,219 Male 3,202,042 128,932 259,132 225,287 123,190 646,234 246,151 616,303 134,441 2,524,074 36,284 117,154 869,985 97,404 431,723 Female 193,190 10-14 Total 195,567 2,984 61,572 25,335 16,215 24,360 20,776 6,574 Male 94,187 516 101,380 37,212 4,559 9,641 Female 2,468 15-19 492,071 9,652 154,084 28,524 71,097 Total 1,394 1,496 8,458 604 42,652 Male 205,515 524 3,159 4,006 53,670 17,811 Female 286,556 870 892 5,298 5,646 100,414 10,713 28,445 20-24 902,304 11,482 56,515 56,572 62,038 239,947 29,753 161,094 Total Male 457,454 7,062 30,894 25,655 21,751 99,283 23,620 103,670 444,850 Female 4,421 25,621 30,917 40,287 140,664 6,133 57,424 25-29 31,673 112,177 Total 1,128,487 93,521 293,850 38,190 199,081 79,136 Male 653,525 22,592 80,185 51,721 29,647 129,882 27,281 130,621 474,962 9,081 31,992 41,800 49,489 163,969 10,909 68,460 Female 30-34 Total 767,262 24,462 59,577 55,757 42,598 210,723 37,570 145,359 18,907 39,033 37,062 15,563 90,844 448,530 91,846 24,156 Male Female 318,731 5,555 20,544 18,694 27,035 118,878 13,414 54,515 35-39 48,654 Total 742,001 27,668 54,174 44,874 198,415 40,496 149,026 Male 423,955 21,683 33,807 37,074 14,837 87,592 26,425 85,156 14,847 17,101 318,046 5,986 30,038 110,823 14,070 63,870 Female 40-44 445,655 26,574 33,084 78,953 25,862 34,186 33,786 113,204 Total 273,476 25,002 24,562 25,011 46,779 Male 22,269 9,822 52,514 Female 172,179 3,593 9,184 9,224 16,753 60,691 8,073 32,174

Table 5.3 Cont'd

					Major Oc	cupational C	Groups	
						Service,		
Age Group		Legislator				Shop &		
and Sex	Total	Senior		Technical and		Market	Skilled	Crafts and
	Employed	Officials &		Associate		sales	agricultural	related
	Person	Managers	Professional	Professionals	Clerks	Workers	and Fishery	trades
45-49								
Total	361,687	20,751	27,609	23,111	21,221	89,940	24,911	68,446
Male	211,592	15,822	18,943	17,676	10,072	42,905	16,069	32,862
Female	150,094	4,928	8,665	5,435	11,149	47,035	8,841	35,584
50-54								
Total	266,025	11,834	21,838	17,504	11,934	60,074	19,765	55,401
Male	153,598	10,450	17,387	12,984	6,422	22,417	13,846	22,814
Female	112,428	1,384	4,452	4,520	5,512	37,657	5,920	32,587
55-59								
Total	152,584	4,426	7,885	9,228	7,810	34,583	16,043	35,166
Male	96,469	4,390	7,620	8,179	5,423	11,879	10,174	20,239
Female	56,115	36	264	1,049	2,387	22,704	5,868	14,927
60-64								
Total	113,615	3,178	2,536	3,805	4,056	22,440	18,783	28,748
Male	75,221	2,987	2,536	3,805	2,744	10,512	15,205	13,553
Female	38,393	191	-	-	1,313	11,928	3,578	15,195
65+								
Total	158,859	2,485	3,814	3,813	3,502	37,388	31,102	39,440
Male	108,519	2,246	3,122	3,411	2,387	19,375	25,776	20,541
Female	50,340	239	692	403	1,115	18,013	5,326	18,900

Table 5.4 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Major O

Table 5.4 Currently Li	pioyeu i o	Palation O	CIDAII AICAS	rigod Toll Tee		cupational G		a major O
					,	Service.		
Region and		Legislator				Shop &		
Sex	Total	Senior		Technical and		Market	Skilled	Crafts and
	Employed	Officials &		Associate		sales	agricultural	related
	Persons	Managers	Professional	Professionals	Clerks	Workers	and Fishery	trades
COUNTRY TOTAL								
Total	5,726,116	165,216	376,286	359,728	316,379	1,516,219	343,555	1,048,027
Male	3,202,042	128,932	259,132	225,287	123,190	646,234	246,151	616,303
Female	2,524,074	36,284	117,154	134,441	193,190	869,985	97,404	431,723
TIGRAY REGION								
Total	391,891	13,563	24,773	20,141	24,632	101,721	17,642	90,632
Male	204,483	10,562	17,213	13,607	8,362	33,751	12,166	47,142
Female	187,408	3,000	7,560	6,534	16,270	67,970	5,476	43,490
AFFAR REGION								
Total	77,068	1,992	1,942	3,259	3,404	22,155	8,142	9,556
Male	43,657	1,732	1,441	2,383	2,300	5,750	7,343	6,076
Female	33,411	260	500	875	1,104	16,405	800	3,480
AMHARA REGION								
Total	1,063,958	27,602	57,403	72,654	45,543	260,922	88,420	218,176
Male	562,815	21,608	42,381	45,109	17,789	90,272	70,611	113,948
Female	501,144	5,994	15,022	27,545	27,755	170,650	17,809	
OROMIA REGION								
Total	1,643,077	42,584	87,713	97,248	72,830	443,850	115,490	313,892
Male	963,543	34,523	60,011	61,054	35,708	202,112	78,227	184,268
Female	679,534	8,061	27,702	36,194	37,122	241,738	37,263	129,625
SOMALI REGION								
Total	202,547	4,824	9,096	12,972	5,039	87,662	30,538	12,245
Male	111,869	4,773	5,916	9,887	2,933	32,657	21,514	8,551
Female	90,677	51	3,180	3,085	2,106	55,006	9,024	
BENISHANGUL GUMUZ	•		,	,	,	•	,	•
REGION								
Total	72,518	2,735	3,949	5,070	3,415	14,967	6,773	16,949
Male	39,854	2,400	3,066	3,536	1,049	6,738	5,071	8,077
Female	32,664	335	883	1,534	2,366	8,229	1,702	8,872
SNNP REGION								
Total	837,386	25,460	65,657	62,410	43,238	233,481	63,304	146,655
Male	466,330	20,859	45,587	39,820	19,288	105,781	41,173	77,920
Female	371,057	4,601	20,071	22,590	23,950	127,700	22,131	68,735

Table 5.4 Cont'd

					Major Occ	cupational Gr	oups	
Region and Sex	Total Employed	Legislator Senior Officials &		Technical and Associate		Service, Shop & Market sales	Skilled agricultural	Crafts and related
	Persons	Managers	Professional	Professionals	Clerks	Workers	and Fishery	
GAMBELLA REGION								
Total	35,656	855	2,882	4,058	1,593	8,729	3,840	8,048
Male	18,797	640	2,407	2,863	787	4,084	3,179	2,057
Female	16,859	215	475	1,196	807	4,645	661	5,991
HARARI REGION								
Total	49,969	1,499	2,628	2,952	3,225	16,703	440	8,703
Male	26,901	1,267	1,885	1,357	873	5,485	440	- , -
Female ADDIS ABABA CITY ADMINISTRATION	23,068	231	742	1,595	2,352	11,219	-	1,861
Total	1,250,842	41,197	115,979	73,917	108,202	295,161	6,863	208,306
Male	709,206	28,011	76,413	41,949	31,770	146,825	5,552	153,026
Female DIRE DAWA ADMINISTRATION	541,635	13,186	39,566	31,967	76,432	148,336	1,311	55,280
Total	101,204	2,906	4,264	5,047	5,257	30,867	2,102	14,864
Male	54,588	2,556	2,811	3,723	2,331	12,779	874	8,396
Female	46,617	350	1,454	1,325	2,925	18,088	1,228	6,469

Table 5.5 Currently Employed Population of Major Towns Aged Ten Years and above by Sex and Major Company Compa

Table 5.5 Cur	rently Emp	loyed Popu	ulation of Ma	jor Towns Age	<u>∌d Ten Ye</u>	ears and abo	ve by Sex a	<u>nd Major C</u>
					Major Occ	cupational Grou	ups	
Major Towns and		Legislator	1			Service,		Crafts
Sex	Total	Senior	1	Technical and		Shop &	Skilled	and
	Employed	Officials &	1	Associate	'	Market sales	agricultural	related
	Persons	Managers	Professional	Professionals	Clerks	Workers	and Fishery	trades
TOTAL MAJOR T								<u></u> .
Total	2,212,903		179,079	127,231	162,443			388,021
Male	1,240,813		118,967	74,900	49,476			273,107
Female	972,089	17,853	60,113	52,331	112,967	292,164	13,902	114,914
MEKELE TOWN								I
Total	121,116		7,451	4,359	6,307		1,183	25,191
Male	65,727		4,551	2,687	1,473		970	17,667
Female	55,389	1,174	2,900	1,672	4,834	20,174	213	7,524
ASAYITA TOWN								I
Total	8,369		348	598	485		1,086	1,147
Male	4,624		275	372	216	1,041	775	839
Female	3,745	48	73	227	269	1,703	311	308
GONDER TOWN								
Total	97,868		3,646	,	4,136			14,895
Male	51,748		2,415	3,089	1,346			8,167
Female	46,121	242	1,230	1,923	2,790	11,027	3,723	6,729
DESSIE TOWN								
Total	47,543		3,266		2,858		654	9,143
Male	26,410		2,286	1,653	1,066		366	6,263
Female	21,133	50	980	1,179	1,792	7,048	288	2,880
BAHIR DAR								
TOWN	00 704	4 4	0.000	4.00-	0.400	40.044	007	24.000
Total	93,781	1,774	8,329	4,837	6,420	,	887	21,036
Male	46,806		5,755		1,824		384	12,283
Female	46,975	77	2,574	2,490	4,596	14,562	503	8,753
BISHOFTU								
TOWN	41 204	1 171	2.007	2 660	4 075	7.062	2 202	44.050
Total	41,284	,	2,097	2,669	1,875	7,963	2,283	11,052
Male	23,681	1,085	1,273	1,510	738	,	1,200	7,556
Female	17,602	389	824	1,159	1,137	4,743	1,083	3,496
ADAMA TOWN	00.040	0.047	F 404	2.040	4.550	04.005	0.507	10.701
Total	92,946	,	5,494	3,619	4,558	21,865	2,507	19,761
Male .	55,237		3,727	2,582	984		1,887	14,291
Female	37,709	435	1,767	1,037	3,575	11,815	620	5,470
JIMMA TOWN	50.005	2 200	F 070	0.400	0.000	10.010	704	44 404
Total	59,065	,	5,873	3,432	3,638		761	11,434
Male	33,607		4,299	1,973	1,481	5,766	528	7,393
Female	25,458	373	1,574	1,458	2,156	6,876	233	4,041

Table 5.5 Cont'd

	Major Occupational Groups											
Major Tourns and	1			<u> </u>	Major Occ	•	ups					
Major Towns and	.	Legislator		-		Service,	01.31	Crafts				
Sex	Total	Senior		Technical and		Shop &	Skilled	and				
	Employed	Officials &	Professional	Associate Professionals	Clerks	Market sales Workers	agricultural	related trades				
SHASHEMENE	Persons	Managers	Professional	Professionals	Cierks	vvoikeis	and Fishery	trades				
TOWN												
Total	45,485	1,723	2,013	2,668	1,892	14,573	2,425	8,424				
Male	28,269	1,550	1,552	1,740	874	6,949	1,861	5,955				
Female	17,216	173	461	929	1,018	7,624	564	2,468				
JIJIGA TOWN	•				,	,		•				
Total	40,942	998	2,839	2,245	2,209	13,616	1,086	5,411				
Male	23,990	948	1,807	1,801	1,046	5,303	447	4,389				
Female	16,952	51	1,031	443	1,163	8,312	639	1,022				
ASOSA TOWN	•		·		·	·		·				
Total	17,931	727	1,524	1,690	1,783	4,332	911	2,926				
Male	10,159	626	1,105	1,027	554	1,953	634	1,705				
Female	7,772	102	419	663	1,229	2,379	277	1,221				
HAWASSA TOWN	l											
Total	127,071	5,457	11,955	9,057	8,304	25,677	4,427	22,281				
Male	70,534	4,625	7,717	5,513	2,346	10,222	1,644	16,794				
Female	56,537	832	4,238	3,544	5,958	15,456	2,782	5,488				
GAMBELLA TOW	N											
Total	17,486	511	1,375	2,296	1,293	4,924	479	3,446				
Male	9,326	370	1,096	1,577	553	2,122	353	1,541				
Female	8,160	141	279	719	740	2,802	126	1,905				
HARAR TOWN												
Total	49,969	1,499	2,628	2,952	3,225	16,703	440	8,703				
Male	26,901	1,267	1,885	1,357	873	5,485	440	6,841				
Female	23,068	231	742	1,595	2,352	11,219	-	1,861				
ADDIS ABABA CI	TY											
Total	1,250,842	41,197	115,979	73,917	108,202	295,161	6,863	208,306				
Male	709,206	28,011	76,413	41,949	31,770	146,825	5,552	153,026				
Female	541,635	13,186	39,566	31,967	76,432	148,336	1,311	55,280				
DIRE DAWA TOW	'N											
Total	101,204	2,906	4,264	5,047	5,257	30,867	2,102	14,864				
Male	54,588	2,556	2,811	3,723	2,331	12,779	874	8,396				
Female	46,617	350	1,454	1,325	2,925	18,088	1,228	6,469				

Table 5.6 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region Occupational Groups: 2012

		Region							
Minor Occupational						110	Benishangul-		
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie		SNNPR	Gambe
Total Employed Persons		-97			2.2		_ =	1	
Total	5,726,116	391,891	77,068	1,063,958	1,643,077	202,547	72,518	837,386	35,6
Male	3,202,042	204,483	43,657	562,815	963,543	111,869	39,854		
Female	2,524,074		33,411	501,144	679,534	90,677	32,664		
MAJOR GROUP 1: LEGISLA						, -	,	, , , ,	-,-
Total	165,216	13,563	1,992	27,602	42,584	4,824	2,735	25,460	8
Male	128,932	10,562	1,732	21,608	34,523	4,773	2,400	20,859	
Female	36,284	3,000	260	5,994	8,061	51	335	4,601	2
LEGISLATORS AND SENIO	OR OFFICIAL	S							
Total	37,658	2,876	795	6,725	9,862	1,900	488	5,092	1
Male	29,560	2,129	795	6,077	8,435	1,900	456	3,724	1
Female	8,098	747	-	648	1,427	-	31	1,367	
Legislators									
Total	2,024	-	-	323	-	171	-	107	
Male	1,604	-	-	323	-	171	-	107	
Female	420	-	-	-	-	-	-	-	
Senior government officials									
Total	16,173	2,016	-	4,794	2,801	889	311	2,310	
Male	13,688	1,693	-	4,145	2,656	889	311	2,029	
Female	2,485	323	-	648	145	-	-	280	
Traditional chiefs and heads	•								
Total	8,029	111	475	213	2,244	770	151	389	
Male	6,057	111	475	213	2,200	770	119	179	
Female	1,972	-	-	-	44	-	31	210	
Senior officials of special-inte	•	ations							
Total	11,433	749	320	1,396	4,817	70	26	2,285	
Male	8,212	325	320	1,396	3,579	70	26	1,409	
Female	3,221	424	-	-	1,238	-	-	877	
CORPORATE MANAGERS									
Total	55,229	5,740	768	8,462	13,055	623	334	-,	
Male	44,123	4,369	756	6,910	11,042	573	314	7,362	
Female	11,106	1,372	12	1,551	2,013	51	20	1,426	
Directors and chief executive									
Total	11,638	856	553	478	2,770	51	-	2,732	
Male	10,406	856	553	478	2,728	-	-	2,732	
Female	1,232	-	-	-	42	51	-	-	

Table 5.6 Cont'd

. abio dia Gont a						Da	egion		
Minor Occupational						RE	Benishangul		
Group and Sex	Total	Tiarass	۸٤٠٠	A b	0	Camalia	-Gumuz	CNINIDD	O a ma la a
Draduation and appretions	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Guilluz	SNNPR	Gambe
Production and operations		•		0.40	007		40	704	
Total Male	6,541	635	-	843	337	-	46	701	
	5,420	635	-	621	130	-	46	701	
Female	1,122	-	-	223	207	-	-	-	
Other department managers									
Total	37,050	4,250	215	7,140	9,948	573	289	5,356	18
Male .	28,297	2,878	203	5,812	8,185	573	269	3,929	1
Female	8,753	1,372	12	1,329	1,764	-	20	1,426	
GENERAL MANAGERS									
Total	72,329	4,947	429	12,416	19,666	2,300	1,913	11,580	4:
Male	55,249	4,064	181	8,621	15,046	2,300	1,629	9,773	30
Female	17,080	882	248	3,794	4,621	-	284	1,807	1
General managers									
Total	72,329	4,947	429	12,416	19,666	2,300	1,913	11,580	4:
Male	55,249	4,064	181	8,621	15,046	2,300	1,629	9,773	3
Female	17,080	882	248	3,794	4,621	-	284	1,807	1
MAJOR GROUP 2:PROFE									
Total	376,286	24,773	1,942	57,403	87,713	9,096	3,949	65,657	2,88
Male	259,132	17,213	1,441	42,381	60,011	5,916	3,066	45,587	2,40
Female	117,154	7,560	500	15,022	27,702	3,180	883	20,071	47
PHYSICAL, MATHEMATIC	AL AND EN	GINEERI	NG SCII	ENCE PRO	FESSIONA	\LS			
Total	32,502	1,998	424	3,686	4,601	467	191	2,699	14
Male	25,968	1,465	212	3,101	3,690	153	191	2,545	14
Female	6,533	533	212	585	911	314	-	153	
Physicists, chemists and r	elated profe	ssionals							
Total	3,018	-	-	177	624	52	24	115	
Male	2,286	-	-	177	584	52	24	115	
Female	732	_	-	-	40	_	_	_	
Mathematicians, statisticia	ans and relat	ted profes	sionals						
Total	3,350	155	60	1,336	938	-	47	110	
Male	2,145	155	24	1,012	451	-	47	110	
Female	1,205	-	36	324	488	-	-	-	
Computing professionals	.,_00			5					
Total	6,690	_	176	1,130	1,614	416	_	496	
Male	5,225	_	-	959	1,443	101	_	496	
Female	1,465	_	176	170	171	314	_	-	
· Siliaio	1, 100			.,,	.,,	017			

Table 5.6 Cont'd

Table did dont a						Do	aion		
Minor Occupational						Re	egion		1
Group and Sex	T	- :	A.C.	A 1	•	0	Benishangul	ONINIDD	0
,	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Architects, engineers and									
Total	19,443	1,843	188	1,042	1,424	-	120		
Male	16,313	1,310	188	952	1,212	-	120	1,824	
Female	3,130	533	-	91	212	-	-	153	
LIFE SCIENCE AND HEA									
Total	56,409	3,029	475	8,915	14,654	1,998	997	,	
Male	31,378	1,769	416	5,525	8,609	904	805		3
Female	25,031	1,261	60	3,389	6,045	1,093	192	4,350	
Life science professionals									
Total	7,759	109	12	1,921	2,073	52	329	1,297	1
Male	5,891	109	-	1,698	1,432	-	329	1,087	1
Female	1,867	-	12	223	641	52	-	210	
Health professionals/exce	pt nursing/								
Total	26,287	1,194	380	5,155	4,780	522	209	4,454	1:
Male	17,345	1,194	380	2,415	3,538	472	209	3,027	
Female	8,942	· -	-	2,739	1,243	51	-	1,427	
Nursing and midwifery pro				,	,			,	
Total	22,364	1,726	84	1,839	7,800	1,424	459	3,882	1:
Male	8,142	465	36	1,412	3,639	433	268		
Female	14,222	1,261	48	427	4,161	991	192		
TEACHING PROFESSIO		, -			, -			,	
Total	180,256	12,576	635	33,446	44,024	4,955	1,664	36,866	1,8
Male	128,511	9,464	598	25,858	30,459	3,505	1,363		
Female	51,746	3,112	37	7,588	13,565	1,450	301	10,281	3
College, university and high				.,	,	.,		,	
professionals	,		-9						
Total	31,590	4,514	36	5,408	6,814	252	205	6,842	
Male	26,953	3,911	36	5,128	6,123	151	205	5,736	
Female	4,636	603	-	280	691	101	-	1,107	
Secondary education tead	,			_00				.,	
Total	83,241	3,380	301	15,192	20,561	2,607	898	15,930	1,4
Male	68,318	2,951	289	13,499	15,956	2,139	762		
Female	14,923	429	12	1,693	4,604	468	136	3,585	
Primary and pre-primary e					1,004	100	100	0,000	
Total	56,987	4,023	285	11,292	13,129	1,729	464	13,393	3
Male	25,572	2,266	273	5,676	4,895	848	353	7,804	
Female	31,415	1,757	12	5,615	8,234	881	111	5,589	
i Gillale	51,415	1,737	12	5,015	0,234	001	111	5,509	2

Table 5.6 Cont'd

Table 5.0 Cont u						Re	egion		
Minor Occupational						110	Benishangul		
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamba
Special education teaching				u.u	- I SIIIIG				
Total	721	-	_	_	375	_	_	_	
Male	375	_	_	_	375	_	_	_	
Female	346	_	_	_	-	_	_	_	
Other teaching profession									
Total	7,718	658	13	1,555	3,145	367	97	701	
Male	7,710	335	-	1,555	3,110	367	43	701	
Female	425	323	13	1,000	36	-	54	-	
OTHER PROFESSIONAL		020	10		30		54		
Total	107,119	7,170	407	11,356	24,434	1,676	1,098	16,459	4
Male	73,275	4,516	215	7,897	17,253	1,354	707	11,173	
Female	33,844	2,654	191	3,459	7,181	322	391	5,286	
Business professionals	30,044	2,004		0, 100	7,101	022	331	0,200	
Total	53,568	3,895	60	5,384	13,001	748	459	9,169	1
Male	31,339	2,091	12	3,224	7,579	577	306	5,364	
Female	22,228	1,804	48	2,160	5,422	171	154	3,805	
Legal professionals	,0	.,551	.0	2,.50	0, .22		101	0,000	
Total	21,407	1,259	60	1,895	5,989	524	282	2,332	1.
Male	15,616	726	60	1,683	5,173	473	166	1,504	
Female	5,791	534	-	211	815	50	115	828	
Archivists, librarians and r	,		ofessiona		3.0			0_0	
Total	5,248	410	144	1,039	1,770	151	43	648	
Male	3,370	202	-	944	1,123	51	24	418	
Female	1,878	208	144	95	646	101	20	231	
Social science and related				30	2.10		20	_3.	
Total	10,400	613	_	517	1,636	50	270	2,976	
Male	9,208	613	-	294	1,386	50	188	2,653	
Female	1,193	-	_	223	250	-	82	323	
Writers and creative or pe	,	ists							
Total	9,617	208	_	1,790	782	101	43	521	
Male	6,862	99	-	1,020	736	101	24	421	
Female	2,754	109	-	770	47	-	20	100	
Religious professionals	,			_					
Total	6,879	785	144	731	1,256	101	_	812	
	0.013								
Male	6,879	785	144	731	1,256	101	-	812	

Table 5.6 Cont'd

- Labit Cit Goile G								-	
Minor Occupational		T			T.	Re	egion	1	
Group and Sex							Benishangul		
'	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MAJOR GROUP 3:TECHN									
Total	359,728	20,141	3,259	72,654	97,248	12,972		,	
Male	225,287	13,607	2,383	45,109	61,054	9,887			
Female	134,441	6,534	875	27,545	36,194	3,085	1,534	22,590	1,1
PHYSICAL AND ENGINE									
Total	44,362	3,250	346	5,255	10,092	935	452		
Male	34,388	2,604	334	4,021	7,814	935	379		
Female	9,974	645	12	1,234	2,278	-	73	2,377	
Physical and engineering									
Total	26,850	2,582	250	2,906	6,249	260	279		
Male	22,770	2,147	250	2,522	5,457	260	226		
Female	4,080	434	-	384	792	-	53	387	
Computer associate profe									
Total	5,368	-	-	587	551	310	46	,	
Male	3,336	-	-	72	551	310	26		
Female	2,032	-	-	515	-	-	20	864	
Optical and electronic equ		rators							
Total	4,762	-	60	1,006	540	50	73		
Male	3,203	-	48	1,006	391	50	73		
Female	1,560	-	12	-	150	-	-	706	
Ship and aircraft controller		icians							
Total	2,121	-	-	-	215	-	-	420	
Male	1,577	-	-	-	91	-	-		
Female	545	-	-	-	125	-	-	420	
Safety and quality inspect									
Total	5,260	668	36	756	2,537	314			
Male	3,503	457	36	421	1,326	314	54	369	
Female	1,758	211	-	335	1,211	-	-	-	
LIFE SCIENCE AND HEA									
Total	80,038	2,829	916	16,928	20,546	3,306	1,795		
Male	42,815	910	651	10,234	10,999	1,913			
Female	37,223	1,919	265	6,694	9,546	1,393	980	6,578	3
Life science technicians ar	nd related a	ssociate p	orofessio	nals					
Total	17,909	307	529	3,305	4,966	819	412	6,353	
Male	16,225	307	467	3,305	4,863	768	318		
Female	1,683	-	62	-	103	51	94		

Table 5.6 Cont'd

Table 3.0 Cont u									
Minor Occupational			——г				gion		
Group and Sex		- .			•		Benishangul	01.11.12	
•	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Modern health associate pro				4 = -			_		
Total	37,120	1,275	363	6,752	11,014	1,132	679	,	
Male	15,557	322	172	2,535	3,968	994	212	,	
Female	21,563	953	191	4,217	7,047	139	467	3,137	1
Nursing and midwifery ass					_				
Total	21,893	1,247	24	5,927	3,297	1,145	609		
Male	8,495	281	12	3,451	988	151	189	,	
Female	13,398	966	12	2,477	2,308	994	420	2,194	1:
Traditional medicine practi		faith heal	ers						
Total	3,116	-	-	944	1,269	210	95	599	
Male	2,538	-	-	944	1,180	-	95	319	
Female	578	-	-	-	88	210	-	280	
TEACHING ASSOCIATE I									
Total	111,168	7,845	1,266	25,620	35,690	5,082	551	19,134	
Male	57,616	4,587	740	12,731	18,504	3,881	250		
Female	53,552	3,258	526	12,889	17,186	1,202	301	8,065	3
Primary education teachin									
Total	80,574	5,455	1,230	19,454	24,486	2,482	551	17,029	
Male	45,742	3,295	704	10,279	14,201	2,168	250	,	4
Female	34,833	2,159	526	9,176	10,285	314	301	6,528	3
Pre-primary education tea									
Total	19,806	670	24	3,312	8,115	1,690	-	1,327	
Male	5,258	111	24	614	2,930	960	-	-	14
Female	14,547	559	-	2,698	5,186	730	-	1,327	
Special education teaching	g associate		nals						
Total	3,910	1,349	-	483	1,768	229	-	-	
Male	2,489	810	-	483	888	229	-	-	
Female	1,421	540	-	-	881	-	-	-	
Other teaching Associate									
Total	6,878	371	12	2,371	1,321	681	-	779	
Male	4,126	371	12	1,355	485	524	-	568	
Female	2,752	-	-	1,016	835	157	-	210	
OTHER ASSOCIATE PRO	OFESSION A	\LS							
Total	124,160	6,217	731	24,852	30,920	3,648	2,272	17,450	1,4
Male	90,468	5,505	659	18,123	23,736	3,157	2,092		9:
Female	33,692	712	72	6,729	7,184	491	180	5,569	5

Table 5.6 Cont'd

						Re	gion		
Minor Occupational						110	Benishangul		
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Finance and sales associat			,		3.5	20		, 3	20
Total	22,120	660	192	3,600	4,514	-	308	1,610	2
Male	13,322	210	192	3,081	3,990	-	246		
Female	8,798	450	-	[,] 519	524	-	62		
Business services agents		rokers						,	
Total	10,108	749	144	183	3,086	445	92	1,559	
Male	8,606	749	144	183	2,242	266	92	1,426	
Female	1,502	-	-	-	844	179	-	134	
Administrative associate p	rofessionals	3							
Total	40,751	1,860	276	10,946	13,859	292	1,046	6,579	8
Male	26,324	1,752	240	6,842	9,721	292	975	3,491	
Female	14,427	109	36	4,104	4,139	-	71	3,087	
Customs, tax and related	government								
associate professionals	_								
Total	5,874	542	12	1,935	1,012	730	297	995	
Male	3,580	389	12	843	393	679	278	895	
Female	2,295	153	-	1,092	619	50	20	100	
Police inspectors and dete	ectives								
Total	10,097	770	72	2,693	1,683	1,279	145	1,256	
Male	7,887	770	36	1,781	1,683	1,069	145	1,256	
Female	2,210	-	36	912	-	210	-	-	
Social work associate prof	essionals								
Total	1,933	-	12	103	-	-	26	340	
Male	594	-	12	-	-	-	26	153	
Female	1,339	-	-	103	-	-	-	187	
Artistic, entertainment and			fessiona						
Total	9,532	744	-	684	1,318	123	140	,	
Male	7,525	744	-	684	900	70	140	,	
Female	2,007	-	-	-	417	52	-	488	
Religious associate profes									
Total	23,340	892	24	4,708	5,448	466	219		
Male	22,316	892	24	4,708	4,806	466	192		1
Female	1,023	-	-	-	642	-	27	275	
Associat professionals not		classified							
Total	404	-	-	-	-	314	-	-	
Male	314	-	-	-	-	314	-	-	
Female	90	-	-	-	-	-	-	-	

Table 5.6 Cont'd

. a.c. c.c com a									
Minor Occupational		г				Re	gion	1	
Group and Sex							Benishangul		
•	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
MAJOR GROUP 4: CLER									
Total	316,379	24,632	3,404	45,543	72,830	5,039	3,415	43,238	
Male	123,190	8,362	2,300	17,789	35,708	2,933	1,049		
Female	193,190	16,270	1,104	27,755	37,122	2,106	2,366	23,950	8
OFFICE CLERKS									
Total	229,927	17,556	2,798	30,541	52,939	3,481	1,849	,	1,3
Male	78,955	6,154	2,097	9,836	21,716	1,841	547	13,463	
Female	150,972	11,402	701	20,705	31,223	1,640	1,302	19,988	7
Secretaries and keyboard	operating c	lerks							
Total	100,075	7,160	675	12,906	24,851	973	1,129	11,922	5
Male	15,824	884	144	1,097	5,876	365	76	3,027	
Female	84,251	6,276	531	11,810	18,975	608	1,053	8,895	4
Numerical clerks									
Total	63,554	3,352	634	7,093	11,728	1,070	459	11,907	
Male	30,080	1,897	610	3,434	6,282	604	284	6,442	2
Female	33,475	1,454	24	3,659	5,446	466	175	5,465	
Material-recording and tra	nsport clerk	S							
Total	29,503	3,656	168	4,707	6,826	101	67	2,237	
Male	14,866	1,700	133	2,243	4,087	-	47	1,449	
Female	14,637	1,956	36	2,464	2,739	101	20	787	
Library, maill and related of	clerks								
Total	25,129	2,361	169	3,725	6,290	884	175	5,798	
Male	9,288	1,007	169	1,502	3,080	469	121	1,744	
Female	15,841	1,353	-	2,223	3,209	415	54	4,054	1
Other office clerks									
Total	11,666	1,028	1,152	2,110	3,244	453	20	1,587	
Male	8,898	666	1,042	1,561	2,390	402	20	801	
Female	2,768	362	110	549	854	50	-	786	
CUSTOMER SERVICES	CLERKS								
Total	86,453	7,076	607	15,002	19,892	1,558	1,566	9,787	
Male	44,235	2,208	203	7,953	13,992	1,092	502	5,825	
Female	42,218	4,868	403	7,050	5,899	466	1,064		
Cashiers, tellers and relate	ed clerks	•		•	-		•	•	
Total	75,817	5,761	607	14,212	18,352	1,507	1,439	8,944	1
Male	38,648	2,005	203	7,700	12,744	1,041	375	5,067	
Female	37,169	3,756	403	6,512	5,609	466	1,064	3,877	
	- ,	-,		- ,	-,	, ,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	- ,	

Table 5.6 Cont'd

						Re	egion		
Minor Occupational							Benishangul		
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Client information clerks		<u> </u>					1	J.	1
Total	1,516,219	101,721	22,155	260,922	443,850	87,662	14,967	233,481	8,7
Male	646,234	33,751	5,750	90,272	202,112	32,657	6,738	105,781	4,0
Female	869,985	67,970	16,405	170,650	241,738	55,006	8,229	127,700	
MAJOR GROUP 5: SERVICE				MARKET SA					
Total	549,602	44,823	8,752	86,277	155,138	24,303	4,537	92,043	3,5
Male	219,624	12,762	1,308	25,190	77,987	11,334	1,529	37,547	1,3
Female	329,979	32,061	7,444	61,087	77,151	12,969	3,009	54,497	
PERSONAL AND PROTECT	IVE SERVICE	ES WORK	ERS						
Total	12,327	1,081	250	532	3,321	786	116	1,800	
Male	9,819	1,081	23	532	3,225	786	83	1,386	
Female	2,508	-	227	-	96	-	33	415	
Travel attendants and related	workers								
Total	291,101	23,496	3,225	59,395	82,247	8,006	3,138	49,027	2,0
Male	78,687	2,628	629	9,450	29,369	2,335	933	18,012	
Female	212,414	20,869	2,595	49,945	52,877	5,672	2,206	31,015	1,6
Housekeeping and restauran		rkers							
Total	67,721	931	4,536	5,821	14,029	8,855	192	16,162	
Male	15,031	-	213	1,420	2,427	2,830	-	5,130	
Female	52,690	931	4,323	4,401	11,602	6,025	192	11,032	
Personal care and related wo									
Total	101,632	8,609	332	10,550	33,297	3,582	402	10,735	
Male	59,733	3,510	332	6,451	24,368	2,800	236	3,954	
Female	41,899	5,099	-	4,099	8,930	782	165	6,781	1
Other personal services work									
Total	1,405	254	-	-	969	-	-	133	
Male	1,356	254	-	-	969	-	-	133	
Female	49	-	-	-	-	-	-	-	
Astrologers, fortune-Tellers a									
Total	61,576	5,969	109	8,473	18,358	2,916	460	10,357	
Male	54,997	5,289	109	7,337	17,629	2,583	277	8,931	7
Female	6,578	679	-	1,136	729	333	183	1,426	
Protective services workers				1,516,219	101,721	22,155	260,922	443,850	
Total	646,234	33,751	5,750	90,272	202,112	32,657	6,738	105,781	4,0
Male	869,985	67,970	16,405	170,650	241,738	55,006	8,229	127,700	4,6
Female									

Table 5.6 Cont'd

						Da	gion		
Minor Occupational		1		П		Ke	egion	I	
Group and Sex	-	-				0 "	Benishangul	ON IN IDD	
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Prostitutes.	40.044	4 400	000	4 500	0.04=	4==	200		4.
Total	13,841	4,483	299	1,506	2,917	157	229	3,828	1:
Male	-		-				-	-	
Female	13,841	4,483	299	1,506	2,917	157	229	3,828	1:
Protactive services/bodyg	urd,patrol no	ot else wh	ere clas	sified.					
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
PERSONAL AND PROTE									
Total	966,617	56,898	13,403	174,646	288,712	63,359	,	141,438	,
Male	426,611	20,988	4,442	65,082	124,125	21,323	5,209	68,235	2,7
Female	540,006	35,909	8,961	109,564	164,587	42,037	5,221	73,203	2,3
Fashion and other models	3								
Total	4,809	927	-	134	541	51	250	396	
Male	2,787	552	-	134	106	51	77	396	
Female	2,022	375	-	-	436	-	173	-	
Shop salespersons and de	emonstrator	S							
Total	451,062	30,035	4,267	75,933	125,098	23,469	5,207	54,946	1,8
Male	260,863	14,585	2,165	35,119	75,666	11,234	3,130	37,206	1,3
Female	190,200	15,450	2,102	40,815	49,433	12,235	2,077	17,741	4
Sellers in ?gulit? & open n	narket any it	ems inclu	iding trad	ders of					
Total	331,643	18,624	3,126	75,820	91,006	23,956	3,856	58,326	2,4
Male	107,857	4,251	986	22,553	28,484	5,623	1,744	20,437	
Female	223,786	14,373	2,140	53,267	62,521	18,333	2,112	,	1,5
"Katikala" retailers	•	,	,	,	,	,	,	,	,
Total	29,131	1,697	443	8,255	11,292	365	54	5,272	19
Male	1,887	· -	12	401	282	-	-	366	
Female	27,244	1,697	431	7,855	11,010	365	54	4,907	
Related workers not stated	•	•		,	,			,	
Total	149,971	5,614	5,568	14,503	60,775	15,518	1,064	22,498	6
Male	53,217	1,600	1,280	6,875	19,587	4,415	258	9,831	2
Female	96,755	4,014	4,288	7,627	41,188	11,104	806	12,667	3
MAJOR GROUP 6:SKILLE						, , , ,	200	,	
Total	343,555	17,642	8,142	88,420	115,490	30,538	6,773	63,304	3,8
Male	246,151	12,166	7,343	70,611	78,227	21,514	5,071	41,173	
Female	97,404	5,476	800	17,809	37,263	9,024	1,702	22,131	6
	0.,.0.	٥, ٥		,550	0.,200	0,0= .	.,. 02	,	•

Table 5.6 Cont'd

. a.c. c c c c c c c c c c c c c c c c c									
Minor Occupational		г			1	Re	egion		
Group and Sex							Benishangul		
'	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MARKET-ORIENTED SKIL									
Total	219,956	16,395	7,475	52,663	63,502	26,136	4,010	,	
Male	160,869	11,122	6,855	42,208	44,153	17,322	3,196		
Female	59,088	5,274	620	10,455	19,348	8,815	813	11,604	4
Market gardeners and cro									
Total	126,739	7,695	1,339	34,174	35,767	14,235	2,992		
Male	99,351	5,087	1,184	28,013	29,199	10,087	2,403		
Female	27,388	2,607	156	6,161	6,568	4,149	589	6,552	1
Market-oriented animal pro									
Total	63,790	7,162	1,633	12,545	18,989	8,655	562	,	
Male	37,853	4,605	1,168	9,247	9,336	4,882	456	,	
Female	25,937	2,557	465	3,299	9,653	3,772	106	4,621	2
Market-oriented crop and									
Total	16,820	1,319	12	3,519	7,227	1,674	210		
Male	12,349	1,319	12	3,190	4,237	781	210		
Female	4,471	-	-	329	2,990	894	-	258	
Forestry and related worker									
Total	10,528	220	4,204	2,323	1,190	1,572	246		
Male	9,410	111	4,204	1,656	1,053	1,572	127		
Female	1,119	109	-	667	137	-	118	-	
Fishery workers, hunters a		;							
Total	2,079	-	287	102	328	-	-	727	
Male	1,906	-	287	102	328	-	-	000	
Female	173	-	-	-	-	-	-	173	
SUBSISTENCE AGRICUL									
Total	123,599	1,247	667	35,757	51,988	4,402	2,763		
Male	85,282	1,044	488	28,403	34,074	4,192	1,875		
Female	38,316	203	180	7,354	17,914	210	888	10,526	1
Subsistence agricultural-c		n vegetab	les prod	lucers					
Total	51,059	737	488	14,106	27,192	1,467	1,929		
Male	38,377	534	488	11,146	19,140	1,467	1,228	3,666	
Female	12,682	203	-	2,960	8,053	-	701	678	
Subsistence livestock prod	ducers								
Total .	28,818	510	144	2,881	12,689	210	111	11,857	1
Male	12,368	510	-	1,318	6,132	210	62		
Female	16,449	-	144	1,563	6,557	-	49	7,876	
	•			*	•			•	

Knowledgeable subsistence crop & animal producers

Table 5.6 Cont'd

Minor Occupational	Т		·	Т		Re	egion		
Group and Sex							Benishangul		
·	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Total	41,064	-	-	18,250	11,113	2,725	723	5,346	48
Male	33,832	-	-	15,679	8,356	2,515	585	4,482	48
Female	7,232	-	-	2,571	2,756	210	138	864	
Subsistence forestry and r	elated prod	uct worke	rs						
Total	2,138	-	36	-	994	-	-	1,108	
Male	446	-	-	-	446	-	-	-	
Female	1,692	-	36	-	549	-	-	1,108	
Fishery workers and hunte	ers for own o	consumpt	ion						
Total	520	-	-	520	-	-	-	-	
Male	260	-	-	260	-	-	-	-	
Female	260	-	-	260	-	-	-	-	
MAJOR GROUP 7: CRAFT	AND REL	ATED TR	ADES W	ORKERS					
Total	1,048,027	90,632	9,556	218,176	313,892	12,245	16,949	146,655	8,04
Male	616,303	47,142	6,076	113,948	184,268	8,551	8,077	77,920	2,0
Female	431,723	43,490	3,480	104,227	129,625	3,694	8,872	68,735	5,99
EXTRACTION AND BUILD	DING TRAD	ES WOR	KERS						
Total	284,997	18,323	4,582	59,594	77,619	5,709	6,973	39,168	1,2
Male	253,861	15,745	3,869	49,560	71,072	5,547	5,423	37,010	1,04
Female	31,136	2,578	713	10,033	6,547	162	1,550	2,158	16
Miners, shotfirers, stone cu	utters and c	arvers							
Total	36,347	3,831	906	12,297	6,740	50	356	2,613	
Male	30,913	3,435	586	9,912	6,241	50	356	2,613	
Female	5,434	396	320	2,385	499	-	-	· -	
Building frame and related	trades wor	kers		,					
Total	195,161	10,121	3,387	34,893	54,345	4,889	6,233	31,012	9
Male	176,688	9,011	3,055	30,226	49,516	4,726	4,683	29,028	79
Female	18,474	1,110	332	4,667	4,829	162	1,550	1,984	1.
Building finiishers and rela	,			,	,		,	,	
Total	39,553	2,988	133	9,613	13,044	719	290	3,965	20
Male	35,098	2,459	72	7,415	11,903	719	290	3,965	22
Female	4,455	529	61	2,197	1,141	-	-	, -	
Painters, building structure					,				
Total	13,936	1,384	156	2,791	3,490	52	94	1,578	
Male	11,162	840	156	2,007	3,412	52	94	1,404	
Male									

METAL, MACHINERY AND RELATED TRADES WORKERS

Table 5.6 Cont'd

Table 3.0 Cont u						Re	egion		
Minor Occupational		Benishangul							
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Total	131,335	10,291	1,011	16,967	28,469	1,247	822	15,254	4
Male	125,732	9,677	976	16,326	28,220	1,247	822	15,062	40
Female	5,603	614	36	641	249	-	-	192	
Metal moulders, weldeers									
Total	46,755	5,657	426	7,286	12,760	353	276	5,836	1
Male	44,864	5,246	390	7,180	12,725	353	276	5,721	1:
Female	1,892	412	36	107	35	-	-	115	
Blacksmiths, tool-makers	and related		rkers						
Total	6,713	892	-	760	802	51	23	177	
Male	5,971	892	-	760	691	51	23	100	
Female	742	-	-	-	112	-	-	77	
Machinery mechanics and									
Total	43,725	1,927	440	5,076	6,706	614	294		2
Male	42,193	1,927	440	4,634	6,655	614	294	4,890	2
Female	1,531	-	-	443	50	-	-	-	
Electrical and electronic e	quipment m		and fitte	rs					
Total	34,143	1,815	146	3,844	8,201	229	230	4,351	
Male	32,704	1,612	146	3,752	8,148	229	230	4,351	
Female	1,439	203	-	92	53	-	-	-	
PRECISION, HANDICRAI									
Total	72,856	5,707	311	13,619	24,496	103	1,007	6,194	10
Male	33,242	3,718	60	6,311	9,914	103	379	3,187	
Female	39,614	1,990	251	7,308	14,582	-	628	3,007	10
Precision workers in meta			S						
Total	8,802	1,378	-	987	3,264	52	46	973	
Male	8,385	1,378	-	987	3,225	52	46	973	
Female	417	-	-	-	39	-	-	-	
Potters, glass-makers and									
Total	36,656	2,100	287	6,794	14,404	-	554	2,332	
Male	3,671	111	36	67	967	-	-	100	
Female	32,985	1,990	251	6,726	13,437	-	554	2,232	
Handicraft workers in woo									
Total	16,181	1,642	24	3,430	4,088	-	301	2,047	
Male	15,465	1,642	24	3,354	4,036	-	227	1,878	
Female	716	-	-	75	52	-	74	168	

Printing and related trades workers

Table 5.6 Cont'd

Minor Occupational						Re	egion			
Minor Occupational Group and Sex							Benishangul			
Gloup and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb	
Total	11,216	587	-	2,409	2,740	51	106	842		
Male	5,721	587	-	1,902	1,686	51	106	236		
Female	5,495	-	-	506	1,054	-	-	606		
OTHER CRAFT AND REI										
Total	558,838	56,311	3,652	127,996	183,308	5,186		86,039	6,2	
Male	203,468	18,003	1,172	41,751	75,062	1,654		22,660	5	
Female	355,370	38,308	2,480	86,245	108,246	3,532	6,694	63,378	5,6	
Food processing and relat	ted trades w	orkers								
Total	291,338	19,953	2,985	51,021	111,053	2,792	6,459	61,586	5,7	
Male	47,423	2,349	837	4,302	25,806	472	411	5,990	1	
Female	243,916	17,605	2,148	46,719	85,247	2,320	6,048	55,595	5,5	
Wood treaters, cabinet-makers and related trades workers										
Total	23,678	1,034	-	6,356	7,939	298	397	2,739	1	
Male	20,279	712	-	5,484	7,089	298	397	2,335	1	
Female	3,399	323	-	872	850	-	-	404		
Textile, garment and relat										
Total	226,364	33,662	523	69,023	61,843	1,992	1,292	20,733	3	
Male	124,537	14,384	192	30,539	40,192	780	646	13,635	2	
Female	101,827	19,278	332	38,484	21,651	1,212	647	7,098	1	
Pelt, leather and shoemal										
Total	17,458	1,661	144	1,596	2,472	105	-	980		
Male	11,230	559	144	1,426	1,974	105	-	700		
Female	6,228	1,102	-	171	498	-	-	280		
MAJOR GROUP 8:PLANT										
Total	305,437	21,182	2,181	35,896	80,011	4,619	1,445	36,141	4	
Male	286,619	18,406	2,169	33,470	77,034	4,467	,	33,529	4	
Female	18,818	2,776	12	2,426	2,977	151	33	2,612		
STATIONARY-PLANT AN										
Total	15,092	1,100	12	1,693	4,844	155	60	1,239	1	
Male	11,843	533	12	1,693	4,154	105	60	881	1	
Female	3,249	567	-	-	691	50	-	358		
Mining- and mineral-proce			5							
Total	1,028	255	-	483	42	-	60	134		
Male	772	-	-	483	42	-	60	134		
Female	255	255	-	-	-	-	-	-		

Metal-processing-plant operators

Table 5.6 Cont'd

Minor Conventional Region										
	Minor Occupational							Benishangul		
	Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
	Total	3,454	-	-	100	1,620	-	_	279	
	Male	2,501	-	-	100	1,620	-	-	100	
	Female	953	-	-	-	-	-	-	179	
	Glass, ceramics and relate	ed plant ope	rators							
	Total	910	218	-	-	-	-	-	-	
	Male	455	109	-	-	-	-	-	-	
	Female	455	109	-	-	-	-	-	-	
	Wood-processing- and pa	permaking-p	olant oper	ators						
	Total	1,480	-	-	-	96	-	-	-	
	Male	1,480	-	-	-	96	-	-	-	
	Female	-	-	-	-	-	-	-	-	
	Chemical-processing-plan	t operators								
	Total	2,585	203	-	-	1,616	-	-	420	
	Male	1,345	-	-	-	925	-	-	420	
	Female	1,240	203	-	-	691	-	-	-	
Power-production and related plant operators										
	Total	4,847	-	12	1,034	1,470	155	-	407	1
	Male	4,502	-	12	1,034	1,470	105	-	228	1
	Female	346	-	-	-	-	50	-	179	
	Automated-assembly-line			perators						
	Total	788	424	-	77	-	-	-	-	
	Male	788	424	-	77	-	-	-	-	
	Female	-	-	-	-	-	-	-	-	
MACHINE OPERATORS AND ASSEMBLERS										
	Total	76,319	5,458	191	11,146	25,568	101	152		
	Male	62,240	3,571	180	8,898	23,479	-	119	7,183	
	Female	14,079	1,887	12	2,247	2,089	101	33	2,254	
	Metal- and mineral-produc			;						
	Total	7,612	1,992	-	975	1,402	-	28		
	Male	6,815	1,510	-	745	1,317	-	28	257	
	Female	797	482	-	231	85	-	-	-	
	Chemical-prpdicts machin	•								
	Total	1,469	-	12	183	55	-	-	527	
	Male	814	-	-	183	55	-	-	230	
	Female	655	-	12	-	-	-	-	297	

Rubber- and plastic-products machine operators

Table 5.6 Cont'd

Minor Occupational						Re	egion		
Minor Occupational							Benishangul		
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Total	5,643	155	-	47	3,333	-	-	-	
Male	3,559	155	-	-	2,607	-	-	-	
Female	2,084	-	-	47	727	-	-	-	
Wood-products machine of	perators								
Total	3,205	-	-	1,178	1,064	-	33	584	
Male	2,985	-	-	1,178	1,064	-	-	397	
Female	220	-	-	-	-	-	33	187	
Printing- binding- and pap		machine	operator						
Total	2,322	109	-	69	409	-	-	268	
Male	2,105	109	-	69	409	-	-	100	
Female	217	-	-	-	-	-	-	168	
Textile- fur- and leather-pr		hine oper	ators						
Total	7,206	546	-	1,486	728	-	-	550	
Male	2,261	109	-	411	432	-	-	153	
Female	4,945	437	-	1,075	296	-	-	397	
Food and related products									
Total	26,222	1,663	36	4,081	12,061	101	37	4,952	
Male	21,604	695	36	3,283	11,154	-	37	3,832	
Female	4,618	968	-	799	907	101	-	1,120	
Assemblers									
Total	21,443	884	144	2,830	6,035	-	53	2,299	
Male	20,900	884	144	2,734	5,961	-	53	2,213	
Female	543	-	-	96	74	-	-	85	
Machines products assem		se where	classified						
Total	1,198	109	-	298	482	-	-	-	
Male	1,198	109	-	298	482	-	-	-	
Female	-	-	-	-	-	-	-	-	
DRIVERS AND MOBILE-I		RATORS							
Total	214,026	14,625	1,978	23,057	49,598	4,362	1,232		33
Male	212,537	14,302	1,978	22,878	49,401	4,362	1,232	25,464	33
Female	1,490	323	-	179	197	-	-	-	
Locomotive-engine drivers		d workers							
Total	2,025	-	-	58	901	-	-	336	
Male	2,025	-	-	58	901	-	-	336	
Female	-	-	-	-	-	-	-	-	

Motor-vehicle drivers

Table 5.6 Cont'd

Mi O of the						Re	egion		
Minor Occupational							Benishangul		
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Total	208,557	14,092	1,547	22,454	48,247	4,292	1,199	24,575	
Male	207,067	13,769	1,547	22,275	48,049	4,292	1,199		2
Female	1,490	323	-	179	197	-	-	-	
Agricultural and other mob	ile plant ope	ertors							
Total	3,265	533	431	467	451	70	33	453	
Male	3,265	533	431	467	451	70	33	453	
Female	-	-	-	-	-	-	-	-	
Ships deck crews and rela	ated workers								
Total	180	-	-	79	-	-	-	101	
Male	180	-	-	79	-	-	-	101	
Female	-	-	-	-	-	-	-	-	
MAJOR GROUP 9: ELEME									
Total	1,293,289	77,405	24,437	257,251	388,989	35,434	17,214	160,682	
Male	669,003	43,274	14,461	127,536	230,488	21,113	8,505	82,194	
Female	624,286	34,131	9,976	129,715	158,501	14,321	8,709	78,489	2,8
SALES AND SERVICES E	ELEMENTAI	RY OCCL	JPATION	NS .					
Total	940,563	55,546	12,000	153,985	272,912	24,507	10,481	125,265	
Male	443,770	26,202	6,751	68,168	159,133	12,594	4,565	59,270	
Female	496,793	29,344	5,249	85,817	113,779	11,913	5,917	65,995	1,9
Street vendors and related	d workers								
Total	88,041	3,258	316	11,933	37,758	2,519	211	11,889	
Male	49,684	2,074	63	4,992	24,274	858	107	,	1.
Female	38,357	1,184	254	6,940	13,484	1,660	104	3,348	
Shoe cleaning and other s									
Total	25,893	1,153	596	3,228	12,339	155	166		
Male	24,896	1,041	596	3,052	12,339	155	166		
Female	997	111	-	176	-	-	-	709	
Domestic and related help									
Total	336,520	22,201	3,562	51,669	64,216	6,191	2,682		
Male	19,344	1,217	417	4,855	4,057	312	166	,	
Female	317,176	20,985	3,144	46,813	60,158	5,879	2,516	42,012	1,1
Building caretakers, helpe									
Total	13,477	1,093	12	819	4,372	101	47	1,037	
Male	4,944	105	-	54	2,288	-	47	580	
Female	8,533	988	12	765	2,083	101	-	456	

Messengers, porters, doorkeepers and related workers

Table 5.6 Cont'd

Miner Occupational						Re	gion		
Minor Occupational			1				Benishangul	Γ	
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambe
Total	460,882	27,151	7,514	81,110	151,592	15,016	7,376	59,486	2,3
Male	338,092	21,656	5,675	52,463	114,776	10,794	4,079	41,277	1,6
Female	122,789	5,495	1,840	28,646	36,817	4,222	3,297	18,208	
Garbage collectors and re	,		, -	, = · ·	, =	,	,	, , , ,	-
Total	15,750	690	-	5,227	2,635	526	-	2,485	
Male	6,810	109	-	2,751	1,398	475	-	1,224	
Female	8,939	581	-	2,476	1,237	51	-	1,261	
AGRICULTURAL, FISHER	RY AND REI	LATED LA		ERS	,			•	
Total	171,794		10,362	65,764	51,917	7,523	5,347	17,369	
Male	83,583	5,534	5,792	31,437	20,929	5,322	2,866	9,159	1:
Female	88,212	2,882	4,570	34,326	30,988	2,201	2,481	8,209	
Agricultural, fishery anf rel			, -	, = - =	, = = =	, ===	,	, , , ,	
Total	171,794		10,362	65,764	51,917	7,523	5,347	17,369	
Male	83,583	5,534	5,792	31,437	20,929	5,322	2,866	9,159	12
Female	88,212	2,882	4,570	34,326	30,988	2,201	2,481	8,209	
LABOURERS IN MINING,							,		
Total	180,932	13,443	2,074	37,503	64,160	3,404	1,386	18,048	
Male	141,651	11,537	1,919	27,931	50,426	3,196	1,074	13,764	
Female	39,281	1,906	156	9,571	13,734	208	² 312	4,284	
Mining and construction la		,		•	,		_	•	
Total	80,147	7,056	274	20,640	21,017	1,785	461	8,796	
Male	53,604	5,580	262	12,520	15,214	1,578	177	5,003	
Female	26,543	1,476	12	8,120	5,802	208	285	3,793	
Manufacturing labourers	ŕ	,		•	,		-	•	
Total	26,596	1,570	467	1,921	17,754	101	-	359	
Male	15,625	1,140	323	952	10,188	101	-	359	
Female	10,972	430	144	969	7,566	-	-	-	
Transport labourers and fr									
Total	74,189	4,816	1,334	14,942	25,389	1,518	925	8,893	
Male	72,422	4,816	1,334	14,459	25,024	1,518	898	8,403	22
Female	1,767	-	-	483	365	-	27	491	
NOT STATED	•								
Total	1,979	199	-	91	470	116	-	358	
Male	1,190	-	-	91	118	58	_	179	
Female	788	199	-	-	352	58	-	179	

Table 5.7 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Age Industrial Divisions - Country Total: 2012

Major Industrial			,			Age	Group 10)		
Divisions							•			
and Sex	All Ages	10-14	15-19	20-24	25-29	30-34	35-39	40-44	45-49	50-54
Total Employed	Population									
Total	5,726,116				1,128,487	767,262	742,001	445,655	361,687	266,025
Male	3,202,042	94,187	205,515	457,454	653,525	448,530	423,955	273,476	211,592	153,598
Female	2,524,074			444,850	474,962	318,731	318,046	172,179	150,094	112,428
Agriculture, Hun	iting, Forset	ery & Fish								
Total	469,377	58,078	47,410	44,845	49,953	47,639	48,323	39,842	30,703	24,484
Male	315,988	45,379	33,366	30,192	30,177	29,151	28,612	27,546	16,800	16,320
Female	153,389	12,699	14,044	14,653	19,777	18,488	19,711	12,296	13,902	8,164
Mining & Quarry										
Total	21,374	-	371	2,598	5,923	4,578	2,886	1,073	1,303	1,434
Male	16,486	-	371	1,863	3,542	3,542	2,886	1,073	1,303	732
Female	4,888	-	-	735	2,380	1,036	-	-	-	702
Manufacturing										
Total	757,055	14,039	57,471	127,307	149,375	83,841	96,982	54,345	54,688	41,777
Male	419,831	8,163	35,242	78,429	89,882	44,020	46,322	27,687	26,262	17,592
Female	337,224	5,876	22,229	48,878	59,493	39,820	50,660	26,658	28,426	24,185
Electricity, Gas										
Total	53,226	483	3,499	7,368	10,285	6,763	7,130	5,489	3,490	2,330
Male	38,176	-	2,111	5,141	7,927	3,940	5,046	3,703	2,866	1,838
Female	15,050	483	1,387	2,227	2,358	2,823	2,084	1,786	624	492
Construction										
Total	395,129	1,508	22,046	71,740	84,673	64,525	62,878	29,305	22,830	14,591
Male	324,893	657	14,576	50,076	71,952	56,557	53,755	25,745	18,838	12,639
Female	70,235	851	7,470	21,664	12,721	7,968	9,123	3,560	3,992	1,952
Whole Sale and										
Total	1,235,258	34,173	89,936	,	231,138	190,628	186,531	102,762	83,337	56,694
Male	625,873	18,595	46,955	83,936	119,025	97,819	89,863	51,514	42,423	25,011
Female	609,386	15,577	42,981	81,359	112,113	92,808	96,669	51,248	40,913	31,682
Hotels and Rest										
Total	457,810	15,263	63,681	83,181	87,151	50,949	51,049	28,740	24,244	19,917
Male .	112,237	2,234	15,822	22,984	23,276	10,492	13,072	8,111	5,029	3,678
Female	345,572	13,029	47,859	60,197	63,874	40,457	37,977	20,629	19,215	16,239
Transport, Stora										
Total	260,370	732	9,964	39,278	62,376	48,830	35,272	24,011	16,656	11,517
Male .	246,962	732	9,310	36,763	57,671	46,375	34,050	23,072	15,945	11,467
Female	13,407	-	653	2,515	4,705	2,455	1,222	940	712	50

Table 5.7 Cont'd

Table 5.7 Con	t u						<u> </u>			
Major Industrial			í	ı		Age	Group 10	·	ı	
Divisions										
and Sex	All Ages	10-14	15-19	20-24	25-29	30-34	35-39	40-44	45-49	50-54
Financial Interme										
Total	115,655	-	345	16,710	29,800	21,644	18,882	9,681	8,735	5,142
Male	69,717	-	-	5,234	20,649	12,705	11,300	6,853	5,718	3,300
Female	45,938	-	345	11,476	9,152	8,938	7,582	2,828	3,017	1,842
Real Estate, Rer										
Total	119,259	341	3,382	25,840	31,013	17,895	14,526	9,471	7,076	3,180
Male	74,806	-	2,151	13,949	19,194	11,947	8,470	5,816	4,864	2,483
Female	44,453	341	1,231	11,892	11,819	5,947	6,055	3,655	2,212	697
PUBLIC ADMIN	ISTRATION	AND DE	FENCE							
Total	381,757	-	3,094	52,973	80,947	58,690	58,220	51,056	33,220	24,238
Male	251,433	-	846	27,980	47,533	36,291	36,938	37,558	27,512	20,009
Female	130,324	-	2,248	24,993	33,414	22,399	21,283	13,498	5,708	4,229
Education										
Total	404,111	-	5,561	66,815	108,808	64,128	52,172	33,752	30,304	25,191
Male	239,864	-	1,582	31,358	66,447	38,772	31,446	18,850	18,016	18,840
Female	164,247	-	3,978	35,457	42,361	25,356	20,726	14,902	12,288	6,350
Health and Socia	al Work									
Total	182,330	-	4,106	31,094	50,898	23,929	27,491	18,547	9,594	7,171
Male	91,421	-	1,090	9,498	22,350	12,474	14,378	12,322	6,154	4,735
Female	90,909	-	3,016	21,595	28,547	11,455	13,113	6,226	3,441	2,436
Other Communit	ty, Social ar	nd Person	al Service	;						
Activities										
Total	566,625	45,030	74,806	98,083	106,691	68,003	59,429	27,382	28,012	24,477
Male	330,050	16,404	37,948	55,340	66,447	39,097	40,456	18,603	17,843	13,692
Female	236,574	28,626	36,858	42,743	40,244	28,905	18,973	8,779	10,169	10,785
Private Househo	olds with Em	ployed Pe								
Total	269,616		105,606	66,098	31,096	10,824	11,037	4,379	5,678	2,639
Male	20,520	2,023	4,143	3,039	2,591	2,629	1,171	750	897	389
Female	249,095		101,463	63,059	28,505	8,195	9,865	3,629	4,781	2,251
Extra-Territorial				•	•	•	•	•	*	•
Total	34,003	-	793	2,265	6,816	4,307	8,986	5,600	1,528	1,243
Male	21,696	-	-	1,095	3,619	2,629	6,191	4,097	1,121	873
Female	12,307	-	793	1,170	3,197	1,678	2,795	1,503	408	370
Not Stated	•				•	•	•	•		
Total	3,162	-	-	815	1,544	91	207	218	287	-
Male	2,087	_	_	578	1,242	91	-	176	-	-
Female	1,076	-	-	237	301	-	207	42	287	-

Table 5.8 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Major -COUNTRY TOTAL: 2012

-COUNTRI	TOTAL: 20	12							
Major Industrial				TI.	II.	Reg			
Divisions and Sex		- .					Benishangul-	01110	
	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Total Employed Pop									
Total	5,726,116	391,891	77,068	1,063,958	1,643,077	202,547	72,518	837,386	35,656
Male	3,202,042	204,483	43,657	562,815	963,543	111,869	39,854	466,330	18,797
Female	2,524,074	187,408	33,411	501,144	679,534	90,677	32,664	371,057	16,859
Agriculture, Hunting	•	•							
Total	469,377	22,364	21,391	138,534	147,139	39,539	11,248	74,269	5,602
Male	315,988	15,316	15,592	93,209	99,548	28,027	7,121	47,356	3,352
Female	153,389	7,048	5,799	45,325	47,590	11,511	4,128	26,913	2,250
Mining & Quarrying									
Total	21,374	5,111	260	5,416	1,092	171	77	3,098	28
Male	16,486	4,360	224	4,708	781	171	77	1,838	-
Female	4,888	751	36	708	311	-	-	1,260	28
Manufacturing									
Total	757,055	70,706	7,464	151,510	247,198	5,029	7,561	87,083	1,687
Male	419,831	36,355	4,302	76,413	146,747	2,361	2,984	39,035	706
Female	337,224	34,351	3,162	75,098	100,451	2,668	4,577	48,048	981
Electricity, Gas and	Water Supply	1							
Total	53,226	4,130	2,235	10,690	14,537	3,471	537	4,188	83
Male	38,176	2,912	1,755	6,264	10,649	2,266	511	2,596	55
Female	15,050	1,218	480	4,427	3,888	1,205	26	1,591	28
Construction									
Total	395,129	27,780	5,037	77,790	95,075	7,511	9,318	51,559	1,728
Male	324,893	23,472	3,854	58,622	82,534	7,127	6,501	44,066	1,377
Female	70,235	4,309	1,183	19,168	12,541	384	2,817	7,493	351
Wholesale and Reta	ail Trade								
Total	1,235,258	70,372	12,953	201,437	385,964	68,517	11,281	168,955	5,440
Male	625,873	31,862	5,116	87,026	193,652	25,144	6,250	94,180	3,052
Female	609,386	38,510	7,837	114,410	192,312	43,373	5,031	74,776	2,388
Hotels and Restaura	ants								
Total	457,810	42,973	5,798	93,341	137,416	8,710	6,506	73,563	6,334
Male	112,237	4,228	1,434	12,320	39,899	2,494	1,233	21,349	543
Female	345,572	38,746	4,364	81,021	97,517	6,216	5,274	52,213	5,791
Transport, Storage	and Communi	cations	•	-	•	•		•	
Total	260,370	16,431	2,232	30,135	67,455	6,129	1,651	30,520	478
Male	246,962	15,396	2,032	29,104	65,798	6,078	1,526	29,208	424
Female	13,407	1,036	200	1,030	1,657	50	125	1,312	54

Table 5.8 Cont'd

Major Industrial						Reg	ion		
Major Industrial Divisions and Sex						<u> </u>	Benishangul-		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNP	Gambela
Financial Intermedia	ition								
Total	115,655	8,492	302	12,932	30,893	2,027	519	15,919	324
Male	69,717	5,431	253	9,443	20,812	1,089	368	9,216	240
Female	45,938	3,061	49	3,489	10,081	937	151	6,703	84
Real Estate, Renting									
Total	119,259	4,161	979	15,090	27,117	1,818	1,133	17,447	371
Male	74,806	2,703	622	10,921	18,737	1,504	794	11,693	252
Female	44,453	1,457	357	4,169	8,379	314	338	5,754	118
PUBLIC ADMINISTI	RATION AND	DEFENCE							
Total	381,757	26,163	5,399	68,125	105,834	14,891	6,109	66,025	6,298
Male	251,433	16,705	3,524	47,392	76,003	10,773	4,440	44,437	4,160
Female	130,324	9,458	1,875	20,733	29,831	4,118	1,669	21,588	2,138
Education									
Total	404,111	28,992	1,438	76,701	108,393	12,064	3,544	80,524	3,190
Male	239,864	18,384	959	46,109	63,939	9,077	2,593	49,807	2,381
Female	164,247	10,608	479	30,592	44,455	2,987	951	30,717	809
Health and Social W	/ork								
Total	182,330	11,355	1,287	28,520	47,973	6,128	3,144	33,325	1,057
Male	91,421	5,214	750	15,196	23,355	3,123	1,680	18,376	476
Female	90,909	6,141	538	13,324	24,619	3,005	1,465	14,950	581
Other Community, S	Social and Per	sonal Servic	e Activitie						
Total	566,625	37,803	7,745	112,322	170,206	21,708	7,957	94,420	2,213
Male	330,050	19,566	2,648	60,314	111,319	12,059	3,476	48,346	1,455
Female	236,574	18,237	5,097	52,009	58,887	9,648	4,481	46,074	758
Private Households	with Employe	d Persons							
Total	269,616	11,984	2,229	37,869	48,448	3,665	1,453	32,937	470
Male	20,520	409	417	3,573	3,762	-	54	2,835	-
Female	249,095	11,575	1,812	34,296	44,686	3,665	1,399	30,102	470
Extra-Territorial Org	anizations and	d NGOs							
Total	34,003	2,874	320	3,455	6,972	1,054	480	3,195	354
Male	21,696	2,171	176	2,110	4,993	516	247	1,812	325
Female	12,307	703	144	1,345	1,978	538	233	1,383	29
Not Stated	•			•				•	
Total	3,162	199	-	91	1,366	116	-	358	-
Male	2,087	-	-	91	1,014	58	-	179	-
Female	1,076	199	-	-	352	58	-	179	-

Table 5.9 Currently Employed Population of Major Towns Aged Ten Years and Above by Sex Divisions: 2012

	ופועום	JIIS. 20	14										
								Ma	ajor Industrial I	Divisions			
Major Towns and Sex	Total Employed Population	Agriculture, Hunting, Forestry & Fishing	Mining & Quarrying	Manufact uring	Electricity, Gas and Water Supply	Construction	Whole Sale and Retail Trade	Hotels and Restaurants	Transport, Storage and Communica tions	Financial Intermedia tion	Real Estate, Renting and Business Activities	Public Administration And Defense	Edu
MAJOR TOWNS TO)T A I												
Total	2,212,903	52,658	8,585	284,019	22,712	197,243	473.171	147,616	147,320	55,668	66,363	140,971	14
Male	1,240,813	34,582	6,763	173,944	16,487		262,096	44,685	136,723	30,910	37,893	82,160	
Female	972,089	18,075	1,823	110,075	6,225		211,075	102,931	10,597	24,758	28,471	58,811	ļ
MEKELE TOWN													l
Total	121,116	1,805	1,286	21,846	1,328	12,516	22,179	9,381	6,854	1,723	1,769	7,530	l
Male	65,727	1,591	1,286	13,489	1,108	10,479	10,109	1,198	6,636	1,404	1,248	4,858	
Female	55,389	213	-	8,356	220	2,037	12,070	8,183	218	319	522	2,673	
ASAYITA TOWN													
Total	8,369	1,254	12	393	108	950	1,995	800	370	13	48	1,009	l
Male	4,624	931	12	73	60	804	896	182	346	-	12	623	
Female	3,745	323	-	320	49	146	1,099	618	24	13	37	386	
GONDER TOWN													l
Total	97,868	22,288	163	10,483	2,400	7,166	14,204	7,288	2,326	758	1,239	4,966	
Male	51,748	15,495	163	5,407	280	5,598	8,027	822	2,096	510	659	4,056	l
Female	46,121	6,794	-	5,076	2,120	1,568	6,177	6,466	230	249	580	910	
DESSIE TOWN													
Total	47,543	573	60	5,406	311	4,278	11,863	2,552	1,951	398	643	4,229	
Male	26,410	241	60	3,274	265	3,510	6,026	473	1,808	290	396	2,502	
Female	21,133	333	-	2,132	46	768	5,837	2,080	142	108	247	1,727	
BAHIR DAR TOWN													
Total	93,781	2,410	207	12,558	675	13,137	14,597	9,983	4,065	1,528	2,586	8,140	
Male	46,806	1,156	207	6,336	571	8,862	6,818	1,783	3,622	1,144	2,052	5,259	l
Female BISHOFTU TOWN	46,975	1,254	-	6,222	103	4,275	7,779	8,200	443	384	534	2,881	
Total	41,284	2,946	197	10,382	245	5,013	6,226	2,810	2,537	672	696	2,413	
Male	23,681	1,400	197	6,626	245	4,282	2,925	602	2,288	326	453	1,612	l
Female	17,602	1,546	-	3,755	-	732	3,300	2,209	249	346	243	801	
ADAMA TOWN													
Total	92,946	2,202	235	16,025	778	6,466	24,160	7,586	8,485	1,298	2,333	2,903	
Male	55,237	1,579	235	10,338	665	6,002	12,076	2,861	8,032	886	1,515	2,086	
Female	37,709	623	-	5,687	113	464	12,084	4,726	453	412	818	816	,

Table 5.9 Cont'd

								Ma	ajor Industrial	Divisions			
Major Towns and Sex	Total Employed Population	Agriculture, Hunting, Forestry & Fishing	Mining & Quarrying	Manufact uring	Electricity, Gas and Water Supply	Constru ction	Whole Sale and Retail Trade	Hotels and Restaurants	Transport, Storage and Communica tions	Financial Intermedia tion	Real Estate, Renting and Business Activities	Public Administration and Defence	ı Ed
JIMMA TOWN													
Total	59,065	602	81	4,749	380	7,364	11,769	3,744	2,731	669	1,194	3,463	}
Male	33,607	439	81	2,649	245	5,549	6,120	1,526	2,597	437	763		
Female	25,458	163	-	2,100	135	1,815	5,650	2,218	134	232	431	1,012)
SHASHEMENE TOV	WN												
Total	45,485	2,817		5,104	193	2.797	14,405	4.094	3,439	676	793	1,491	
Male	28,269	2,072	_	3,772	193	2,458	7,928	1,407	3,091	521	459		
Female	17,216	745	-	1,332	-	338	6,477	2,686	347	155	333	295	
JIJIGA TOWN	,=10	0		.,		230		=,=30			230		
Total	40,942	968	171	2,335	531	3,584	12,408	1,478	2,251	455	665	3,713	}
Male	23,990	671	171	1,261	380	3,200	4,927	607	2,200	303	665	2,687	
Female	16,952	296	-	1,074	152	384	7,481	871	50	151	-	1,026	
ASOSA TOWN	.0,702	270		.,071	.52	301	,,.51	371	00	.01		.,020	
Total	17,931	1.071	24	1.394	223	1.788	3,247	1.707	966	132	755	2,128	}
Male	10,159	678	24	712	197	1,414	1,701	360	917	89	486	1,446	
Female HAWASSA	7,772	393	-	682	26	373	1,545	1,346	49	43	269	682	
TOWN													
Total	127,071	3,762	-	13,425	2,102	11,380	23,740	10,060	5,646	2,775	2,266	11,926)
Male	70,534	1,446	-	8,573	1,057	9,597	13,821	3,698	5,386	1,895	1,460	6,885	j
Female	56,537	2,316	-	4,852	1,045	1,783	9,920	6,362	261	880	806	5,041	ı
GAMBELLA TOWN													
Total	17,486	667	28	1,104	83	1,474	2,038	2,964	360	324	230	4,147	!
Male	9,326	416	-	507	55	1,211	1,132	428	306	240	145	2,499	,
Female	8,160	251	28	597	28	263	906	2,536	54	84	85	1,648	}
HARAR TOWN													
Total	49,969	538	565	6,007	770	3,894	16,349	1,361	1,504	734	692	5,184	Ĺ
Male	26,901	538	512	3,624	565	3,357	6,189	285	1,455	338	487	3,117	
Female	23,068	-	53	2,383	205	537	10,160	1,076	49	397	205	2,067	1
ADDIS ABABA CITY	'ADMINISTRAT	ION											
Total	1,250,842	5,904	4,934	162,016	11,175	108,369	262,548	76,648	97,060	42,626	49,114	72,212	,
Male	709,206	4,520	3,192	101,765	9,445	88,369	159,385	27,427	89,581	21,819	26,032	36,330)
Female	541,635	1,384	1,742	60,251	1,729		103,163	49,221	7,479	20,808	23,082	35,881	
DIRE DAWA TOWN													
Total	101,204	2,850	622	10,794	1,409	7,068	31,443	5,159	6,775	888	1,340	5,518	}
Male	54,588	1,408	622	5,539	1,156	5,615	14,016	1,025	6,360	709	1,062		
Female	46,617	1,441	-	5,254	253	1,453	17,426	4,134	415	179	278		



Table 5.10 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region Industrial Divisions -COUNTRY TOTAL: 2012

ווומטאנו מו טואואנע	112 -000	NIKII	UIAL.	2012					
Minor Industrial Divisions		1	1	1	1	Re	egion		
and Sex							Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
TOTAL EMPLOYED PERSON									
Total				1,063,958				837,386	
Male	3,202,042			562,815		111,869		466,330	
Female	2,524,074			501,144	679,534	90,677	32,664	371,057	16,8
MAJOR GROUP 1:AGRICUL		,	FOREST		ISHING				
Total	469,377	,	,	138,534	147,139	39,539			5,6
Male	315,988		15,592	93,209	99,548	,		,	
Female	153,389	7,048	5,799	45,325	47,590	11,511	4,128	26,913	2,2
AGRICULTURE, HUNTING A									
Total	441,247			131,551	140,336	38,220			
Male	299,221		11,259	90,753	95,012				
Female	142,026	6,707	5,082	40,798	45,324	11,450	3,696	24,998	1,4
Growing of crops; market gard									
Total	215,428		12,484	58,030	69,967	19,036		35,765	
Male	151,183	6,080	8,773	36,573	52,197			25,221	2,0
Female	64,245	3,158	3,711	21,457	17,770	4,463	2,057	10,544	3
Farming of animals									
Total	112,952	11,140	3,201	25,200	33,825	13,912			
Male	61,734	7,817	2,150	16,096	16,560	7,134			
Female	51,217	3,323	1,051	9,105	17,265	6,778	776	10,884	1,1
Growing of crops combined w	ith farming o	of animals	(mixed	farming)					
Total	76,267	253	467	41,519	15,135	4,573			
Male	60,789	253	323	33,055	12,009	4,364			2
Female	15,478	-	144	8,464	3,126	210	863	1,891	
Agricultural and animal husba									
Total	36,028	742	189	6,802	21,409	699			
Male	25,168	742	13	5,030	14,246	699	37	,	
Female	10,859	-	176	1,772	7,163	-	-	1,680	
Hunting, trapping and game p	ropagation i	including i	related s	ervice activ	ities				
Total	572	226	-	-	-	-	-	-	
Male	346	-	-	-	-	-	-	-	
Female	226	226	-	-	-	-	-	-	
FORESTRY, LOGGING AND	RELATED	SERVICE	ACTIVI	TIES					
Total	26,336	765	4,763	6,881	6,803	1,319		3,373	3
Male	15,318	424	4,046	2,354	4,536	1,258		1,458	
Female	11,018	341	717	4,527	2,266	61	431	1,915	7

Table 5.10 Cont'd

Minor Industrial Divisions						Re	gion		
and Sex							Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Forestry, logging and related s		rities						,	
Total	26,336	765	4,763	6,881	6,803	1,319	1,362	3,373	8
Male	15,318	424	4,046	2,354	4,536	1,258	931	1,458	•
Female	11,018	341	717	4,527	2,266	61	431	1,915	-
FISHING									
Total	1,795	-	287	102	-	-	-	424	
Male	1,449	-	287	102	-	-	-	424	(
Female	346	-	-	-	-	-	-	-	
Fishing, operation of fish hatch	neries and fi	ish farms;	service	activities in	cidental to				
Fishing									
Total	1,795	-	287	102	-	-	-	424	(
Male	1,449	-	287	102	-	-	-	424	(
Female	346	-	-	-	-	-	-	-	
MAJOR GROUP 2: MINING A									
Total	21,374	5,111	260	5,416	1,092	171	77	3,098	
Male	16,486	4,360	224	4,708	781	171	77	1,838	
Female	4,888	751	36	708	311	-	-	1,260	
MINING OF COAL AND LIGN		ACTION C	OF PEAT						
Total	142	-	-	103	39	-	-	-	
Male	142	-	-	103	39	-	-	-	
Female	-	-	-	-	-	-	-	-	
Mining and agglomeration of h									
Total	142	-	-	103	39	-	-	-	
Male	142	-	-	103	39	-	-	-	
Female	-	-	-	-	-	-	-	-	
Mining and agglomeration of li	ignite								
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Extraction and agglomeration	of peat								
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	

Table 5.10 Cont'd

Missaula dustrial Divisions						Re	gion		
Minor Industrial Divisions			, , , , , , , , , , , , , , , , , , ,] _[]		Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
EXTRACTION OF CRUDE PE								·	
Total	1,284	277	-	-	-	-	-	210	
Male	938	277	_	-	-	-	-	210	
Female	346	-	_	-	-	-	-	-	
Extraction of crude petroleum		gas							
Total	346	•	-	-	-	-	-	-	
Male	346	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Service activities incidental to	oil and gas	extraction	excludi	ng					
surveying	_			-					
Total	938	277	-	-	-	-	-	210	
Male	592	277	-	-	-	-	-	210	
Female	346	-	-	-	-	-	-	-	
MINING OF GOLD, COPPER	AND AS W	ELL AS U	IRANIUN	M AND THO	RIUM ORE	S			
Total	612	612	-	-	-	-	-	-	
Male	612	612	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Mining of gold, copper and as v			horium c	ores					
Total	612	612	-	-	-	-	-	-	
Male	612	612	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
MINING OF METAL ORES									
Total	4,896	2,532	-	-	311	-	54	210	
Male	3,467	2,176	-	-	-	-	54	210	
Female	1,429	356	-	-	311	-	-	-	
Mining of iron ores									
Total	455	109	-	-	-	-	-	-	
Male	455	109	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Mining of non-ferrous metal or	es, except u		nd thoriu	ım ores					
Total	4,441	2,423	-	-	311	-	54	210	
Male	3,012	2,067	-	-	-	-	54	210	
Female	1,429	356	-	-	311	-	-	-	
OTHER MINING AND QUARF	RYING								
Total	14,439	1,690	260	5,313	742	171	24	2,677	
Male	11,326	1,294	224	4,605	742	171	24	1,418	
Female	3,113	396	36	708	-	-	-	1,260	

Table 5.10 Cont'd

Min - u la de staint Divinio						Re	egion		
Minor Industrial Divisions							Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Quarrying of stone, sand and	clay	<u> </u>		'			'	/	<u>'</u>
Total	12,808	1,294	188	5,231	475	171	24	2,677	
Male	10,414	1,294	188	4,523	475	171	24	1,418	
Female	2,394	-	-	708	-	_	-	1,260	
Mining and quarrying n.e.c.	,							,	
Total	1,631	396	72	82	267	-	-	-	
Male	912	-	36	82	267	-	-	-	
Female	719	396	36	-	-	-	-	-	
MAJOR GROUP 3: MANUFA	ACTURING								
Total	757,055	70,706	7,464	151,510	247,198	5,029	7,561	87,083	1,6
Male	419,831	36,355	4,302	76,413	146,747	2,361	2,984	39,035	7
Female	337,224	34,351	3,162	75,098	100,451	2,668	4,577	48,048	
MANUFACTURE OF FOOD I			/ERAGE	S					
Total	215,162	10,695	5,337	29,931	95,052	2,276	785	30,932	7
Male	101,232	4,466	2,913	13,017	51,882	838	106	9,049	
Female	113,930	6,229	2,424	16,915	43,169	1,438	680	21,884	6
Production, processing and p	reservation of	of meat, fi	sh, fruit,	vegetables,	oils and fats	3			
Total	14,206	841	287	2,013	5,949	419	28	1,398	
Male	9,844	403	287	1,153	3,931	314	28	977	
Female	4,361	438	-	860	2,018	105	-	421	
Manufacture of dairy products	3								
Total	2,467	-	-	1,642	264	139	-	77	
Male	1,219	-	-	703	170	-	-	-	
Female	1,248	-	-	939	94	139	-	77	
Manufacture of grain mill prod	ducts, starch	es and sta	arch prod	ducts, and p	repared an	imal			
feeds									
Total	72,441	5,375	466	8,429	44,295	271	131	5,663	
Male	50,995	2,112	454	6,028	34,308	210	23	3,344	
Female	21,446	3,263	12	2,401	9,987	61	108	2,320	2
Manufacture of other food pro									
Total	95,666	3,927	4,584	11,387	31,811	1,398	626	20,338	3
Male	23,981	1,614	2,172	2,281	6,849	314	54	4,021	
Female	71,684	2,313	2,412	9,107	24,962	1,083	572	16,317	3
Manufacture of beverages									
Total	30,382	552	-	6,460	12,732	51	-	3,457	
Male	15,192	337	-	2,852	6,623	-	-	707	
Female	15,190	215	-	3,608	6,109	51	-	2,750	

Table 5.10 Cont'd

Table 3.10 Cont u						PΔ	gion		
Minor Industrial Divisions						rie	Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MANUFACTURE OF TOBAC				Aiiiiaia	Oloma	Sumane	-Guilluz	SININFIX	Gairib
PRODUCTS	CO AND LO	CALBEV	ENAGE						
Total	31,299	323		4,482	8,546		2,967	14,216	
Male	1,365	323	-	4,462	498	-	460	14,210	
Female	29,935	323	-	4,075	8,048	-	2,507	- 14,216	
Manufacture of tobacco produ		323	-	4,075	0,040	-	2,507	14,210	
Total	2,482			1,221	915				
Male	407	-	-	407	913	-	-	_	
Female	2,075	-	-	814	915	-	-	-	
	•	-	-	014	915	-	-	-	
Manufacture and Wholesale o		323			1 101			280	
Total Male	2,037	323	-	-	1,434	-	-	200	
	2.027	323	-	-	1 121	-	-	280	
Female Manufacture and Wholesale of	2,037	323	-	-	1,434	-	-	200	
	447				166			200	
Total Male		-	-	-	166 112	-	-	280	
Female	112 335	-	-	-	55	-	-	200	
Manufacture and Wholesale o		-	-	-	55	-	-	280	
Total	26,333	(eia/		3,261	6,031		2.067	13,655	
Male	20,333	-	_	3,201	386	-	2,967 460	13,000	
Female	25,487	-	-	2 264	5,644	-		10 CEE	
MANUFACTURE OF TEXTIL		-	-	3,261	5,644	-	2,507	13,655	
		OF 640	AEE.	62.062	44,264	664	442	10.046	4
Total Male	176,353	25,619	455 144	63,063		664	413	10,946	,
	79,859	9,319		24,206	22,380	298	138	3,994	4
Female	96,494	16,300	311	38,857	21,884	367	276	6,951	1
Spinning, weaving and finishir Total	144,798	24,671	431	53,519	33,141	419	62	8,727	
Male	70,669	9,220	144	20,827	19,163	210	62	3,841	
Female	70,669	15,451	287	32,692	13,978	210	02	4,886	
Manufacture of other textiles	74,129	15,451	201	32,092	13,970	210	-	4,000	
	22.270	200	24	0.444	0.070	245	76	958	
Total	23,279	208		8,414	8,072	245	76 76		
Male	8,450	200	-	3,379	2,921	88	76	153	
Female Manufacture of knitted and cre	14,829	208	24	5,035	5,151	157	-	804	
			licies	1 100	2 OE4		076	1 264	
Total	8,276	740	-	1,129	3,051	-	276	1,261	
Male	741 7.536	99	-	1 100	296	-	-	1.004	
Female	7,536	642	-	1,129	2,755	-	276	1,261	

Table 5.10 Cont'd

Table 5.10 Cont d									
Minor Industrial Divisions						Re	gion		
and Sex						0 "	Benishangul	011155	
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MANUFACTURE OF WEARIN						a=-			_
Total	41,514	6,345	356	6,585	12,480	871	536	6,149	3
Male .	29,245	3,422	180	4,415	10,466	213	420	5,787	3
Female	12,269	2,923	176	2,171	2,015	659	116	362	
Manufacture of wearing appar					40.0-	a - :		0 1 1 -	_
Total	41,305	6,345	356	6,585	12,271	871	536	6,149	3
Male .	29,245	3,422	180	4,415	10,466	213	420	5,787	3
Female	12,060	2,923	176	2,171	1,806	659	116	362	
Dressing and dyeing of fur; ma		of articles	ot fur						
Total	209	-	-	-	209	-	-	-	
Male .	-	-	-	-	-	-	-	-	
Female	209	-	- 	-	209	-	-	-	
TANNING AND DRESSING C HANDBAGS, SADDLER, HAF				RE OF LUGO	GAGE,				
Total	18,059	1,260	-	900	5,234	105	107	795	
Male	9,546	481	-	237	4,495	105	107	515	
Female	8,513	779	-	663	739	-	-	280	
Tanning and dressing of leath			ggage, h			arness			
Total	4,832	583	-	853	689	-	107	680	
Male	2,384	217	-	237	635	-	107	399	
Female	2,449	366	-	616	55	-	-	280	
Manufacture of footwear									
Total	13,226	677	-	47	4,544	105	-	115	
Male	7,162	264	-	-	3,860	105	-	115	
Female	6,064	414	-	47	684	-	-	-	
MANUFACTURE OF WOOD									
FURNITURE; MANUFACTUR			STRAW			RIALS			
Total	40,449	4,418	-	8,855	13,298	-	455	7,193	
Male	34,898	3,800	-	6,593	12,153	-	422	6,738	
Female	5,551	618	-	2,262	1,144	-	33	455	
Saw milling and planning of w									
Total	4,751	-	-	1,849	1,044	-	-	474	
Male	2,878	-	-	525	941	-	-	374	
Female	1,873	-	-	1,325	102	-	-	100	
Manufacture of products of wo		raw and p	laiting m						
Total	35,697	4,418	-	7,005	12,254	-	455	6,719	
Male	32,019	3,800	-	6,068	11,212	-	422	6,364	

Table 5.10 Cont'd

Table 5.10 Contro	·								
Minor Industrial Divisions				1		Re	gion	1	
and Sex							Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	
Female	3,678	618	-	937	1,042	-	33	355	
MANUFACTURE OF PAPER		R PRODL	JCTS						
Total	2,384	-	-	-	296	-	-	-	
Male	1,939	-	-	-	209	-	-	-	
Female	445	-	-	-	87	-	-	-	
Manufacture of paper and paper	per products	Manufact	ture of o	ther articles	of paper				
and paperboard									
Total	2,384	-	-	-	296	-	-	-	
Male	1,939	-	-	-	209	-	-	-	
Female	445	-	-	-	87	-	-	-	
PUBLISHING, PRINTING AN	D REPROD	UCTION (OF REC	ORDED ME	DIA				
Total	12,804	300	-	845	2,299	101	211	832	
Male	5,180	300	-	466	853	-	188	200	
Female	7,624	-	-	379	1,446	101	23	631	
Publishing									
Total	8,350	300	-	560	1,649	101	92	463	
Male	3,185	300	-	237	397	-	69	100	
Female	5,165	-	-	323	1,253	101	23	363	
Printing and service activities	related to pr	inting							
Total	4,233	-	-	285	549	-	-	368	
Male	1,775	-	-	228	356	-	-	100	
Female	2,458	-	-	56	193	-	-	268	
Reproduction of recorded med	dia								
Total	220	-	-	-	101	-	119	-	
Male	220	-	-	-	101	-	119	-	
Female	-	-	-	-	-	-	-	-	
MANUFACTURE OF COKE, I	REFINED P	ETROLE	JM PRO	DUCTS AN	D				
NUCLEAR FUEL									
Total	2,555	-	-	-	465	-	-	1,051	
Male	1,879	-	-	-	-	-	-	841	
Female	676	-	-	-	465	-	-	210	
Manufacture of coke oven pro									
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	

Table 5.10 Cont'd

Min and advertisal Division a						Re	gion		
Minor Industrial Divisions							Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Manufacture of refined petrole	eum product			'.				,	
Total	1,744	-	-	-	-	-	-	1,051	
Male	1,533	-	-	-	-	-	-	841	
Female	210	-	-	-	-	-	-	210	
Processing of nuclear fuel									
Total	811	-	-	-	465	-	-	-	
Male	346	-	-	-	-	-	-	-	
Female	465	-	-	-	465	-	-	-	
MANUFACTURE OF CHEMIC	CALS AND (CHEMICA	L PROD	UCTS					
Total	11,084	406	-	-	4,903	52	-	987	
Male	7,915	203	-	-	4,569	-	-	777	
Female	3,169	203	-	-	334	52	-	210	
Manufacture of basic chemica	ıls Manufact	ure of bas	ic chem	icals, excep	t fertilizers				
and nitrogen compounds									
Total	1,670	-	-	-	768	-	-	210	
Male	1,257	-	-	-	565	-	-	-	
Female	413	-	-	-	202	-	-	210	
Manufacture of other chemica	I products a	nd of plas	tics in p	rimery forms	3				
Total	9,414	406	-	-	4,135	52	-	777	
Male	6,658	203	-	-	4,004	-	-	777	
Female	2,756	203	-	-	131	52	-	-	
Manufacture of man-made fib	res								
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
MANUFACTURE OF RUBBE			RODUCT						
Total	16,136	541	-	55	6,172	-	62	134	
Male	9,297	155	-	55	2,540	-	-	134	
Female	6,839	386	-	-	3,632	-	62	-	
Manufacture of rubber produc									
Total	3,226	277	-	55	128	-	62	134	
Male	2,334	-	-	55	128	-	-	134	
Female	892	277	-	-	-	-	62	-	
Manufacture of plastic produc									
Total	12,909	264	-	-	6,045	-	-	-	
Male	6,963	155	-	-	2,413	-	-	-	
Female	5,946	109	-	-	3,632	-	-	-	

Table 5.10 Cont'd

Table 5.10 Cont d									
Minor Industrial Divisions						Re	gion	1	
and Sex	.	T.	A (Λ 1	•	ا ا	Benishangul		
	Total	Tigray	Afar	Amhara		Somalie	-Gumuz	SNNPR	Gamb
MANUFACTURE OF OTHER							225	0 =00	
Total	37,431	4,343	598	11,601	10,022	51	362	2,709	
Male	29,855	2,687	598	8,955	8,287	51	362	2,357	
Female	7,575	1,656	-	2,647	1,735	-	-	352	
Manufacture of other non-met				44.55	40.0==	= -		<u> </u>	
Total	37,431	4,343	598	11,601	10,022	51	362	2,709	
Male .	29,855	2,687	598	8,955	8,287	51	362	2,357	
Female	7,575	1,656	-	2,647	1,735	-	-	352	
MANUFACTURE OF BASIC N		.						<u>.</u>	
Total	20,091	2,917	144	1,116	5,669	101	162	922	
Male .	17,508	2,058	144	1,116	5,669	101	162	922	
Female	2,584	860	-	-	-	-	-	-	
Manufacture of basic iron and				_	_				
Total	3,912	206	-	802	688	50	20	-	
Male .	3,167	206	-	802	688	50	20	-	
Female	745	-		-	-	-	-	-	
Manufacture of basic precious			als						
Total	5,577	721	-	-	1,875	-	23	421	
Male	4,629	464	-	-	1,875	-	23	421	
Female	949	257	-	-	-	-	-	-	
Manufacture of molded/shape					_				
Total	10,602	1,990	144	314	3,106	50	119	501	
Male	9,712	1,387	144	314	3,106	50	119	501	
Female	890	603	-	-	-	-	-	-	
MANUFACTURE OF FABRIC	ATED MET	AL PROD	UCTS, E	EXCEPT MA	CHINERY	AND			
EQUIPMENT									
Total	23,936	3,380	13	6,316	6,286	204	194	2,294	
Male	22,628	3,380	13	6,018	6,286	204	99	2,178	
Female	1,308	-	-	298	-	-	94	115	
Manufacture of structural meta									
Total	20,148	2,909	13	4,292	5,608	151	194	1,996	
Male	19,110	2,909	13	4,217	5,608	151	99	1,881	
Female	1,038	-	-	75	-	-	94	115	
Manufacture of other fabricate			tal worki						
Total	3,787	471	-	2,023	678	52	-	297	
Male	3,517	471	-	1,801	678	52	-	297	
Female	270	-	-	223	-	-	-	-	

Table 5.10 Cont'd

Table 3.10 Golff G						Re	gion		
Minor Industrial Divisions						110	Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MANUFACTURE OF MACHIN		FOUIPME			O TOTTING	Comano	Gamaz	Oran II	Carris
Total	6,355	118	-	808	2,381	51	74	420	
Male	3,996	-	_	808	943	51	37		
Female	2,359	118	_	-	1,438	-	37	420	
Manufacture of general purpo					1,100		O1	120	
Total	2,138	118	_	413	791	51	_	420	
Male	946	-	_	413	136	51	_	-	
Female	1,192	118	_	-	655	-	_	420	
Manufacture of special purpos					000				
Total	3,741	-	_	_	1,590	_	74	_	
Male	2,575	_	_	_	807	_	37	_	
Female	1,166	-	_	_	783	_	37	_	
Manufacture of domestic appl									
Total	476	-	-	395	_	_	_	-	
Male	476	-	-	395	-	_	-	-	
Female	_	-	-	_	-	_	-	-	
MANUFACTURE OF OFFICE	, ACCOUNT	TING AND	COMP	UTING MAC	CHINERY				
Total	1,854	-	-	-	-	50	-	420	
Male	420	-	-	-	-	-	-	420	
Female	1,435	-	-	-	-	50	-	-	
Manufacture of office, accoun	ting and con	nputing m	achinery	•					
Total	1,854	· -	-	-	-	50	-	420	
Male	420	-	-	-	-	-	-	420	
Female	1,435	-	-	-	-	50	-	-	
MANUFACTURE OF ELECT	RICAL MACI	HINERY A	AND APF	PARATUS N	N.E.C.				
Total	3,660	607	-	-	696	52	-	-	
Male	3,449	396	-	-	696	52	-	-	
Female	211	211	-	-	-	-	-	-	
Manufacture of electric motors		s and tran	sformers	6					
Total	1,032	-	-	-	-	52	-	-	
Male	1,032	-	-	-	-	52	-	-	
Female	-	-	-	-	-	-	-	-	
Manufacture of electricity distr			paratus						
Total	2,063	607	-	-	418	-	-	-	
Male	1,852	396	-	-	418	-	-	-	
Female	211	211	-	-	-	-	-	-	

Table 5.10 Cont'd

Table 3.10 Gont d						Da	gion		
Minor Industrial Divisions]	Re		1	T
and Sex	T-4-1	T:	۸ ۲	A I	0	0 1!-	Benishangul		0
Manufacture of insulted wine	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Manufacture of insulted wire a	ind cable								
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Manufacture of accumulators,		is and prir	mary bai	tteries					
Total	287	-	-	-	-	-	-	-	
Male	287	-	-	-	-	-	-	-	
Female	-		-	-	-	-	-	-	
Manufacture of electric lamps	and lighting	equipme	nt						
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Manufacture of other electrica		n.e.c.			070				
Total	278	-	-	-	278	-	-	-	
Male .	278	-	-	-	278	-	-	-	
Female	- TELEVIOLO	-	-	-	-	-	-	-	
MANUFACTURE OF RADIO, EQUIPMENT AND APPARAT		N AND C	OMMUN	NICATION					
Total	2,202			100	768			299	
Male	2,202	-	_	100	682	-	-	299	
Female	2,116	-	-	100	86	-	-	299	
Manufacture of electronic valv		- ond oth	or alaatr	- onic compo		-	-	-	
Total	817	s and our	ei eiecti	onic compo	470				
Male	817	-	_	-	470	-	-	-	
Female	017	-	_	-	470	-	-	-	
Manufacture of television and	radio trancr	nittore and	- d annar	- atus for line	tolophopya	nd -	-	-	
linetelegraphy	Taulo II al ISI	iiilleis aiil	и аррага	alus ioi iiile	teleprioriya	iiu			
Total	134		_	_	101	_	_	_	
Male	91	-	_	-	58	-	-	-	
Female	42	-	_	_	42	-	_	-	
Manufacture of television and		ore equin	d or vida	- oo rocordina		- cina anna	ratus and	-	
associated goods	radio receiv	cis, souli	u oi vide	eo recording	j or reprodu	ciriy appa	iiaius, aiiu		
Total	1,252			100	197			299	
Male	1,252	-	-	100	154	-	-	299 299	
	1,208	-	-	100	44	-	-	299	
Female	44	-	-	-	44	-	-	-	

Table 5.10 Cont'd

Minor Industrial Divisions						Re	gion		
and Sex							Benishangul		[
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MANUFACTURE OF MEDICA	L, PRECISI	ON AND	OPTIC/	LINSTRUM	IENTS,	·		,	
WATCHES AND CLOCKS									
Total	1,751	203	-	-	509	-	-	-	
Male	1,147	203	-	-	252	-	-	-	
Female	604	-	-	-	258	-	-	-	
Manufacture of medical applia						,			
checking, testing, navigating a	nd other pu	rposes, e	xcept of	tical Instrun	nent				
Total	1,059	203	-	-	509	-	-	-	
Male	801	203	-	-	252	-	-	-	
Female	258	-	-	-	258	-	-	-	
Manufacture of optical instrum	ents and ph	otograph	ic equipi	ment					
Total	692	-	-	-	-	-	-	-	
Male	346	-	-	-	-	-	-	-	
Female	346	-	-	-	-	-	-	-	
Manufacture of watches and c	ocks								
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
MANUFACTURE OF MOTOR	VEHICLES	, TRAILE	RS AND	SEMI-TRA	ILERS				
Total	10,301	2,234	-	1,060	2,054	50	-	-	
Male	9,182	1,696	-	662	1,872	50	-	-	
Female	1,119	538	-	398	183	-	-	-	
Manufacture of motor vehicles									
Total	734	218	-	-	170	-	-	-	
Male	625	109	-	-	170	-	-	-	
Female	109	109	-	-	-	-	-	-	
Manufacture of bodies (coach)	vork) for mo	otor vehicl	es; mar	ufacture of					
trailers and semi- trailers									
Total	9,567	2,016	-	1,060	1,884	50	-	-	
Male	8,557	1,587	-	662	1,701	50	-	-	
Female	1,010	429	-	398	183	-	-	-	
MANUFACTURE OF OTHER		RT EQUIF	PMENT						
Total	3,152	104	-	795	1,215	50	-	332	
Male	2,808	104	-	795	1,034	50	-	168	
Female	344	_	_	_	181		_	163	

Table 5.10 Cont'd

Miner Industrial Divisions						Re	gion		
Minor Industrial Divisions							Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Building and repairing of ships	s and boats/i		of railwa	y, aircraft,m	anufacture		/	<u>!</u>	
Total	882	-	-	573	-	-	-	-	
Male	882	-	-	573	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Manufacture of railway and tra	amway locor	notives ar	nd rolling	stock					ļ
Total	152	-	-	-	152	-	-	-	
Male	54	-	-	-	54	-	-	-	
Female	98	-	-	-	98	-	-	-	
Manufacture of aircraft and sp	acecraft								
Total	170	-	-	-	170	-	-	-	
Male	88	-	-	-	88	-	-	-	
Female	82	-	-	-	82	-	-	-	
Manufacture of transport equipment	pment n.e.c								
Total	1,948	104	-	223	892	50	-	332	
Male	1,784	104	-	223	892	50	-	168	
Female	163	-	-	-	-	-	-	163	
MANUFACTURE OF FURNIT			RING N.E	E.C.					
Total	75,976	6,616	562	14,998	23,847	349	1,209	6,118	2
Male	41,821	3,409	311	8,564	10,239	349	460	4,319	
Female	34,156	3,207	251	6,434	13,608	-	749	1,798	1
Manufacture of household and									
Total	75,976	6,616	562	14,998	23,847	349	1,209	6,118	2
Male	41,821	3,409	311	8,564	10,239	349	460	4,319	
Female	34,156	3,207	251	6,434	13,608	-	749	1,798	1
RICYCLING									
Total	2,546	277	-	-	742	-	24	336	
Male	2,546	277	-	-	742	-	24	336	
Female	-	-	-	-	-	-	-	-	
Recycling of metal waste and									
Total	1,456	-	-	-	742	-	-	336	
Male	1,456	-	-	-	742	-	-	336	
Female	-	-	-	-	-	-	-	-	
Recycling of non-metal waste									
Total	1,090	277	-	-	-	-	24	-	
Male	1,090	277	-	-	-	-	24	-	
Female	-	-	-	-	-	-	-	-	ļ

Table 5.10 Cont'd

Table 3.10 Gont a									
Minor Industrial Divisions						Re	egion	1	
and Sex				_			Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MAJOR GROUP 4: ELECTR									
Total	53,226	4,130	2,235	10,690	14,537	3,471	537	4,188	
Male	38,176	2,912	1,755	6,264	10,649		511	2,596	
Female	15,050	1,218	480	4,427	3,888	1,205	26	1,591	
ELECTRICITY, GAS, STEAM									
Total	22,024	2,242	323	3,398	6,324	151	173	1,844	
Male	19,956	2,023	299	3,248	5,765	101	173	1,499	
Female	2,068	220	24	150	559	50	-	345	
Production, collection and dis-									
Total	19,555	2,119	323	3,248	5,973	151	153	1,844	
Male	17,686	1,900	299	3,248	5,413	101	153	1,499	
Female	1,869	220	24	-	559	50	-	345	
Manufacture of gas; distribution			rough m						
Total	2,019	123	-	46	351	-	20	-	
Male	1,924	123	-	-	351	-	20	-	
Female	96	-	-	46	-	-	-	-	
Steam and hot water supply									
Total	450	-	-	103	-	-	-	-	
Male	346	-	-	-	-	-	-	-	
Female	103	-	-	103	-	-	-	-	
COLLECTION, PURIFICATION	ON AND DIS	TRIBUTIO		/ATER					
Total	31,202	1,888	1,912	7,292	8,213	3,320	364	2,343	
Male	18,220	890	1,456	3,015	4,884	2,165	338	1,097	
Female	12,982	998	456	4,277	3,328	1,155	26	1,246	
Collection, purification and dis									
Total	31,202	1,888	1,912	7,292	8,213	3,320	364	2,343	
Male	18,220	890	1,456	3,015	4,884	2,165	338	1,097	
Female	12,982	998	456	4,277	3,328	1,155	26	1,246	
MAJOR GROUP 5: CONSTR	UCTION								
Total	395,129	27,780	5,037	77,790	95,075	7,511	9,318	51,559	1,7
Male	324,893	23,472	3,854	58,622	82,534	7,127	6,501	44,066	1,3
Female	70,235	4,309	1,183	19,168	12,541	384	2,817	7,493	(
CONSTRUCTION									
Total	395,129	27,780	5,037	77,790	95,075		9,318	51,559	
Male	324,893	23,472	3,854	58,622	82,534	7,127	6,501	44,066	1,3
Female	70,235	4,309	1,183	19,168	12,541	384	2,817	7,493	(

Table 5.10 Cont'd

Table 5.10 Contro	-								
Minor Industrial Divisions			,	,		Re	gion		
and Sex							Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Site preparation									
Total	49,577	4,316	12	8,322	7,815	2,043	235	3,545	
Male	41,216	3,598	12	5,954	7,191	1,973	125	3,079	
Female	8,361	718	-	2,367	624	70	109	466	
Building complete construction	n or parts the	ereof; civil	l engine	ering					
Total	283,251	18,740	4,833	60,178	71,046	4,085	8,780	40,108	1,3
Male	229,639	16,146	3,650	45,066	61,404	3,923	6,072	33,693	1,1
Female	53,612	2,594	1,183	15,112	9,642	162	2,708	6,415	2
Building installation	•	•	•	•	•		•	•	
Total	15,339	612	48	2,953	3,448	101	246	1,548	1
Male	13,952	612	48	2,881	2,530	51	246	1,548	1
Female	1,387	-	-	['] 72	918	51	-	, <u>-</u>	
Building completion	ŕ								
Total	43,995	4,113	144	6,338	12,058	1,281	57	6,358	1
Male	37,812	3,116	144	4,721	10,701	1,180	57	5,746	1
Female	6,183	997	_	1,616	1,357	101	-	612	
Renting of construction or der			h operat		,				
Total	2,966	' <u>-</u>	· -	-	707	_	-	_	
Male	2,273	-	_	-	707	_	-	_	
Female	692	-	_	-	-	_	-	_	
MAJOR GROUP 6: WHOLE		RETAIL T	RADE:	REPAIR OF	MOTOR \	/EHICLES	S.MOTORCY	CLES	
AND PERSONAL AND HOU			,				,		
Total	1,235,258	70,372	12.953	201,437	385,964	68,517	11.281	168,955	5,4
Male	625,873	31,862	,	87,026	193,652	25,144	6,250	94,180	3,0
Female	609,386		7,837	114,410	192,312		5,031	74,776	2,3
WHOLESALE AND RETAIL T								, 0	_,c
PERSONAL AND HOUSEHO				,		/			
Total	58,493	2,478	287	3,997	10,351	618	165	7,907	1
Male	51,684	2,227	275	3,856	10,086	618	165	7,806	1
Female	6,808	251	12	140	265	-	-	101	•
Sale of motor vehicles	2,230	_5.		. 10	_55				
Total	1,618	_	_	_	_	_	_	133	
Male	569	_	_	_	_	_	_	133	
Female	1,050	_	_	_	_	_	_	-	
· cindio	1,000								

Table 5.10 Cont'd

Minor Industrial Divisions and Sex				,		1.0	gion		
							Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Maintenance and repair of mo			ng and g						
Total	37,879	1,628	48	2,754	6,007	560	43	5,127	1
Male	35,445	1,378	48	2,709	5,957	560	43	5,127	1
Female	2,434	251	-	45	50	-	-	-	
Sale of motor vehicle parts an	nd accessorie								
Total	6,799	356	-	583	559	58	46	479	
Male	5,072	356	-	488	433	58	46	479	
Female	1,728	-	-	95	127	-	-	-	
Sale, maintenance and repair	of motorcyc	les and re	elated pa	arts and acc	essories				
Total	4,854	-	72	519	1,117	-	76	757	
Male	4,508	-	72	519	1,117	-	76	757	
Female	346	-	-	-	-	-	-	-	
Retail sale of automotive fuel									
Total	7,341	494	168	140	2,666	-	-	1,410	
Male	6,090	494	156	140	2,579	-	-	1,310	
Female	1,251	-	12	-	88	-	-	101	
WHOLESALE AGENCIES, IM	IPORTERS	AND EXP	ORTER	S, DISTRIB	UTORS,				
BROKERS AND TENDER									
Total	181,180	9,802	2,457	24,580	54,777	4,524	882	27,283	3
Male	109,933	5,482	1,657	13,997	35,174	2,837	672	15,349	1
Female	71,247	4,320	800	10,583	19,602	1,687	211	11,933	1
Wholesale agencies, importers	s and export	ers,ditribu	ters,bro	kers and					
tender									
Total	44,456	844	391	700	14,283	917	114	9,676	2
Male	30,300	746	331	643	11,505	657	114	4,913	
Female	14,156	99	60	56	2,778	260	-	4,763	1
Wholesale of agricultural raw	materials, liv	e animals	s, food, b	everages a	ind tobacco				
Total	77,137	3,998	2,018	16,526	26,649	2,600	768	10,561	1
Male	38,986	1,588	1,290	9,239	12,294	1,648	557	4,942	1
Female	38,151	2,410	728	7,287	14,355	951	211	5,619	
Wholesale of household good		,		•	•			•	
Total	46,852	4,072	12	7,066	9,684	1,007	-	6,613	
Male	30,952	2,416	_	4,009	7,250	532	-	5,061	
Female	15,900	1,657	12	3,057	2,434	475	-	1,551	
Wholesale of non-agricultural					,	_		,	
Total	7,231	255	36	289	1,769	-	-	433	
Male	5,419	255	36	107	1,769	-	-	433	

Table 5.10 Cont'd

Table 5.10 Cont u									
Minor Industrial Divisions				1		Re	egion		
and Sex							Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Female	1,812	-	-	182	-	-	-	-	
Wholesale of machinery, equp		upplies							
Total	5,503	633	-	-	2,391	-	-	-	
Male	4,275	478	-	-	2,356	-	-	-	
Female	1,228	155	-	-	35	-	-	-	
Other wholesale									
Total	-	-	_	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
RETAIL TRADE, OF MOTOR	VEHICLES	AND MO	TORCY	CLES; REPA	AIR OF PE	RSONAL	AND HOUSE	HOLD GO	OODS
Total	995,586	58,092	10,208	172,860	320,837	63,375	10,233	133,766	4,9
Male	464,256	24,152	3,183	69,173	148,392	21,689	5,413	71,024	
Female	531,330	33,939	7,025	103,687	172,445	41,687	4,820	62,742	2,2
Non-specialized retail trade in	,	,	,	,	,	,	,	,	,
Total	182,268	17,165	1,955	25,094	57,952	14,570	1,976	17,752	3
Male	95,129	6,583	1,013	7,810	33,103	7,193	1,268	11,682	6
Female	87,139	10,583	943	17,284	24,850	7,376		6,070	
Retail sale of food, beverages			alized st		•	,		,	
Total	217,740	7,704	1,840	40,883	87,037	10,409	2,678	39,551	ç
Male	85,584	3,268	48	17,863	31,562	3,056	1,577	16,447	4
Female	132,156	4,436	1,792	23,020	55,475	7,353	1,101	23,104	4
Other retail trade of new good	,	,		,	,	,	,	,	
Total	148,450	10,176	142	21,128	42,664	4,582	1,483	13,814	
Male	96,778	7,329	107	12,762	29,646	3,398		8,594	
Female	51,671	2,847	35	8,366	13,018	1,184		5,220	
Retail sale of second-hand go		,		-,	-,-	, -		-, -	
Total	13,325	449	_	1,315	5,214	600	76	2,521	
Male	10,778	340	_	921	4,067	234		2,338	
Female	2,547	109	_	395	1,147	367		184	
Retail trade not in stores	,-				,				
Total	387,120	21,844	4,990	78,996	115,593	31,676	3,780	48,475	2,4
Male	146,867	6,306	1,636	26,019	43,527	6,738		24,753	
Female	240,253	15,538	3,354	52,977	72,065	24,937		23,722	
Repair of personal and house		. 5,550	٠,٠٠٠.	0_,0	,550	,551	_,_ 10	,	٠,
Total	46,683	754	1,281	5,443	12,377	1,539	240	11,654	4
Male	29,119	327	380	3,798	6,488	1,069	77	7,211	2
Female	17,564	427	901	1,645	5,889	470	163	4,442	_
1 GITIGIO	17,504	741	301	1,043	5,003	710	100	¬,¬¬∠	

Table 5.10 Cont'd

Table 3.10 Cont u									
Minor Industrial Divisions				· · ·		Re	gion	1	1
and Sex							Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MAJOR GROUP 7: HOTELS									
Total	457,810	42,973	5,798	93,341	137,416	8,710	6,506	73,563	6,3
Male	112,237	4,228	1,434	12,320	39,899	2,494	1,233	21,349	5
Female	345,572	38,746	4,364	81,021	97,517	6,216	5,274	52,213	5,7
HOTELS AND RESTAURAN									
Total	457,810	42,973	5,798	93,341	137,416	8,710	6,506	73,563	6,3
Male	112,237	4,228	1,434	12,320	39,899	2,494	1,233	21,349	5
Female	345,572	38,746	4,364	81,021	97,517	6,216	5,274	52,213	5,7
Hotels; camping sites and oth	er provision	of short-s	stay						
accommodation									
Total	98,950	10,016	1,093	12,243	21,522	469	591	11,491	•
Male	38,876	2,351	705	3,748	10,952	418	199	3,876	
Female	60,074	7,665	388	8,495	10,570	51	392	7,615	•
Restaurants, bars and cantee	ns								
Total	358,859	32,957	4,705	81,098	115,894	8,241	5,916	62,071	6,1
Male	73,361	1,877	729	8,573	28,947	2,075	1,034	17,473	5
Female	285,498	31,080	3,976	72,526	86,947	6,166	4,882	44,598	5,6
MAJOR GROUP 8:TRANSPO		AGE AND	COMM	UNICATION	NS				
Total	260,370	16,431	2,232	30,135	67,455	6,129	1,651	30,520	4
Male	246,962	15,396	2,032	29,104	65,798	6,078	1,526	29,208	4
Female	13,407	1,036	200	1,030	1,657	50	125	1,312	
LAND TRANSPORT; TRANS	PORT VIA F	PIPELINE	S						
Total	215,229	14,017	2,074	25,773	58,929	5,226	1,164	25,784	3
Male	210,730	13,740	1,886	25,594	57,754	5,226	1,114	25,293	3
Female	4,499	277	188	179	1,175	-	49	491	
Transport via railways									
Total	448	-	-	-	102	-	-	-	
Male	448	-	-	-	102	-	-	-	
Female	-	-	-	-	-	-	-	-	
Other land transport									
Total	214,781	14,017	2,074	25,773	58,827	5,226	1,164	25,784	3
Male	210,282	13,740	1,886	25,594	57,652	5,226	1,114	25,293	(
Female	4,499	277	188	179	1,175	-	49	491	
WATER TRANSPORT									
Total	1,399	-	36	384	-	-	53	926	
Male	1,050	-	36	315	-	-	53	646	
Female	349	-	-	69	-	-	-	280	

Table 5.10 Cont'd

Minor Industrial Divisions	1								
and Sex							Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Sea and coastal water transpo									
Total	1,176	-	36	161	-	-	53	926	
Male	827	-	36	92	-	-	53	646	
Female	349	-	-	69	-	-	-	280	
Inland water transport									
Total	223	-	-	223	-	-	-	-	
Male	223	-	-	223	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
AIR TRANSPORT									
Total	7,573	838	-	217	658	51	24	257	
Male	5,114	475	-	117	614	51	24	257	
Female	2,460	363	-	100	44	-	-	-	
Scheduled air transport									
Total	6,589	838	-	72	658	-	24	257	
Male	4,230	475	-	72	614	-	24	257	
Female	2,359	363	-	-	44	-	-	-	
Non-scheduled air transport									
Total .	984	-	-	145	-	51	-	-	
Male	884	-	-	45	-	51	-	-	
Female	100	-	-	100	-	-	-	-	
SUPPORTING AND AUXILIA	RY TRANSF	PORT AC	TIVITIES	S; ACTIVITI	ES OF				
TRAVEL AGENCIES									
Total	16,041	616	-	1,469	2,128	782	89	1,114	
Male	13,183	330	-	1,469	1,909	732	89	981	
Female	2,858	286	-	-	219	50	-	133	
Supporting and auxiliary trans	port activitie	s; activitie	es of trav	el agencies	6				
Total	16,041	616	-	1,469	2,128	782	89	1,114	
Male	13,183	330	-	1,469	1,909	732	89	981	
Female	2,858	286	-	· -	219	50	-	133	
POST AND TELECOMMUNIC									
Total	19,840	961	122	2,292	5,740	70	321	2,438	
Male	16,598	851	110	1,609	5,521	70	246	2,030	
Female	3,242	109	12	683	220	-	76	408	
Post and courier activities	,								
Total	2,854	-	-	339	1,855	-	-	-	
Male	2,376	-	_	211	1,855	_	-	_	
Female	478	-	-	127	,	-	-	_	

Table 5.10 Cont'd

Mineral Industrial Divisions						Re	gion		
Minor Industrial Divisions							Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Telecommunications			·					,	
Total	16,986	961	122	1,953	3,885	70	321	2,438	
Male	14,222	851	110	1,398	3,666	70	246	2,030	
Female	2,764	109	12	555	220	-	76	408	
MAJOR GROUP 9: FINANCI	AL INTERM	IDIATION	i						
Total	115,655	8,492	302	12,932	30,893	2,027	519	15,919	3
Male	69,717	5,431	253	9,443	20,812	1,089	368	9,216	2
Female	45,938	3,061	49	3,489	10,081	937	151	6,703	
FINANCIAL INTERMEDIATION)N, EXCEPT		NCE AN	1D					
PENSION FUNDING									
Total	87,612	7,425	-	7,981	23,094	1,145	398	11,610	1
Male	53,662	4,898	-	6,529	16,282	359	321	6,111	1
Female	33,949	2,527	-	1,452	6,812	786	77	5,498	
Monetary intermediation	•	•		•	•			•	
Total	56,934	4,851	-	4,151	13,851	466	140	5,297	
Male	37,428	3,029	-	3,087	11,461	309	140	3,238	
Female	19,506	1,822	-	1,064	2,390	157	-	2,059	
Other financial intermediation		•		•	•			•	
Total	30,677	2,575	-	3,830	9,242	680	259	6,312	
Male	16,234	1,869	-	3,442	4,821	51	182	2,873	
Female	14,444	705	-	388	4,421	629	77	3,439	
INSURANCE AND PENSION		EXCEPT	COMPU	JLSORY SO				•	
Total	3,139	109	-	-	1,547	-	-	75	
Male	807	-	-	-	386	-	-	75	
Female	2,332	109	_	-	1,161	-	-	-	
Insurance and pension fundin	g, except co	mpulsory	social s	ecurity	,				
Total .	3,139	109	-	-	1,547	-	-	75	
Male	807	-	-	-	386	-	-	75	
Female	2,332	109	_	-	1,161	-	-	-	
ACTIVITIES AUXILIARY TO 			DIATIO	N	,				
Total	24,905	957	302	4,951	6,252	881	120	4,234	1
Male	15,248	533	253	2,914	4,144	730	47	3,029	1
Female	9,657	425	49	2,036	2,108	151	73	1,205	
Activities auxiliary to financial								•	
Total	23,541	957	302	4,951	6,214	881	120	4,234	1
Male	14,230	533	253	2,914	4,106	730	47	3,029	1
Female	9,311	425	49	2,036	2,108	151	73	1,205	
	•	_	_	,	,	-	_	,	

Table 5.10 Cont'd

Activities auxiliary to insurance and pension funding Total 1,364 39 Male 1,018 39 Female 346 MAJOR GROUP 10: REAL ESTATE, RENTING AND BUSINESS ACTIVITIES Total 119,259 4,161 979 15,090 27,117 1,818 1,133 17,4 Male 74,806 2,703 622 10,921 18,737 1,504 794 11,6	
Activities auxiliary to insurance and pension funding Total 1,364 39 Male 1,018 39 Female 346 39 MAJOR GROUP 10: REAL ESTATE, RENTING AND BUSINESS ACTIVITIES Total 119,259 4,161 979 15,090 27,117 1,818 1,133 17,4 Male 74,806 2,703 622 10,921 18,737 1,504 794 11,6	- - - 147 3
Activities auxiliary to insurance and pension funding Total 1,364 39 Male 1,018 39 Female 346 MAJOR GROUP 10: REAL ESTATE, RENTING AND BUSINESS ACTIVITIES Total 119,259 4,161 979 15,090 27,117 1,818 1,133 17,4 Male 74,806 2,703 622 10,921 18,737 1,504 794 11,6	- - - 147 3
Total 1,364 39 Male 1,018 39 39 39	93 2
Male 1,018 - - - 39 - - Female 346 - - - - - - - - MAJOR GROUP 10: REAL ESTATE, RENTING AND BUSINESS ACTIVITIES Total 119,259 4,161 979 15,090 27,117 1,818 1,133 17,4 Male 74,806 2,703 622 10,921 18,737 1,504 794 11,6	93 2
Female 346 -	93 2
MAJOR GROUP 10: REAL ESTATE, RENTING AND BUSINESS ACTIVITIES Total 119,259 4,161 979 15,090 27,117 1,818 1,133 17,4 Male 74,806 2,703 622 10,921 18,737 1,504 794 11,6	93 2
Total 119,259 4,161 979 15,090 27,117 1,818 1,133 17,4 Male 74,806 2,703 622 10,921 18,737 1,504 794 11,6	93 2
Male 74,806 2,703 622 10,921 18,737 1,504 794 11,6	93 2
Edmald /// /63 1 /6/ 36/ / // 360 93/0 31/ 330 E	′ 54 1
REAL ESTATE ACTIVITIES	
Total 1,554 157 39	-
Male 906 157 39	-
Female 648	-
Real estate activities with own or leased property	
Total 1,169 157	-
Male 867 157	-
Female 302	-
Real estate activities on a fee or contract basis	
Total 385 39	-
Male 39 39	-
Female 346	-
RENTING OF MACHINERY AND EQUIPMENT WITHOUT	
OPERATOR AND OF PERSONAL AND HOUSEHOLD GOODS	
	80
	85
Female 2,254 155 - 95 109 - 27	96
Renting of transport equipment	
	35
	' 40
Female 191 95	96
Renting of other machinery ant equipment	
,	981
	981
Female 529 155 34	-
Renting of personal and household goods n.e.c.	
	365
	365
Female 1,533 75 - 27	-

Table 5.10 Cont'd

Table 5.10 Cont o						Re	gion		
Minor Industrial Divisions							Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
COMPUTER AND RELATED						I			
Total	24,480	109	189	3,331	6,677	310	160	2,861	
Male	11,788	109	-	877	3,125	310	80	1,700	
Female	12,692	-	189	2,453	3,553	-	81	1,161	
Hardware consultancy service	,			,	,			,	
Total	1,148	-	-	-	400	-	-	287	
Male	1,148	-	-	-	400	-	-	287	
Female	-	-	-	-	-	-	-	-	
Software consultancy service									
Total	6,394	-	13	617	2,416	-	57	928	
Male	1,773	-	-	223	994	-	-	210	
Female	4,621	-	13	394	1,422	-	57	717	
Data processing/analysis/ ser	vice								
Total	6,822	-	176	1,181	2,316	210	24	849	
Male	3,673	-	-	-	1,181	210	-	849	
Female	3,149	-	176	1,181	1,134	-	24	-	
Data base activities									
Total	4,231	-	-	180	1,033	-	54	633	
Male	1,773	-	-	180	193	-	54	353	
Female	2,459	-	-	-	840	-	-	280	
Manitenance and repair of off	ice, account	ing and co	omputing)					
machinery									
Total	2,808	109	-	351	315	-	26	-	
Male	2,808	109	-	351	315	-	26	-	
Female	-	-	-	-	-	-	-	-	
Other computer related activit									
Total	3,077	-	-	1,002	198	101	-	163	
Male	613	-	-	124	42	101	-	-	
Female	2,463	-	-	878	156	-	-	163	
RESEARCH AND DEVELOP									
Total	6,726	109	586	311	2,542	211	318	1,508	
Male	3,854	109	431	311	1,506	211	298	620	
Female	2,872	-	156	-	1,036	-	20	888	
Research and experimental d				es and engi					
Total	5,652	109	431	45	2,472	211	221	1,432	
Male	3,169	109	287	45	1,436	211	201	544	
Female	2,483	-	144	-	1,036	-	20	888	

Table 5.10 Cont'd

Table 5.10 Cont d						Re	gion		
Minor Industrial Divisions						110	Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Research and experimental d	evelopment			s and huma	nities	II.			<u>" </u>
(SSH)	·								
Total	1,075	-	156	266	70	-	96	75	
Male	685	-	144	266	70	-	96	75	
Female	389	-	12	-	-	-	-	-	
OTHER BUSINESS ACTIVITI	IES								
Total	70,176	3,026	192	10,712	15,455	1,297	628	10,398	1
Male	44,216	1,723	180	9,092	11,773	982	417	6,789	1
Female	25,960	1,303	12	1,620	3,682	314	211	3,609	
Legal accounting, book-keepi	ng and audit	ing activit	ies; tax o	consultancy	; market res	search an	d public		
opinion polling; business and	managemer	nt consulta	ancy	-					
Total	34,915	997	12	4,551	5,853	242	461	7,073	1
Male	18,760	733	-	3,630	4,438	242	269	4,233	
Female	16,155	264	12	921	1,415	-	191	2,841	
Architectural, engineering and	dother techn	ical activi	ties						
Total	4,584	363	-	1,220	135	-	-	1,279	
Male	3,998	363	-	1,119	84	-	-	1,204	
Female	586	-	-	102	51	-	-	75	
Advertising									
Total	6,056	362	-	161	2,950	-	47	388	
Male	3,343	109	-	161	2,288	-	47	107	
Female	2,714	253	-	-	662	-	-	280	
Business activities n.e.c.									
Total	24,622	1,304	180	4,780	6,516	1,055	120	1,658	
Male	18,116	518	180	4,182	4,963	740	100	1,245	
Female	6,505	786	-	598	1,554	314	20	413	
MAJOR GROUP 11: PUBLIC		RATION A	AND DE	FENCE;					
COMPULSORY SOCIAL SEC	CURITY								
Total	381,757	26,163	5,399	68,125	105,834	14,891	6,109	66,025	6,2
Male	251,433	16,705	3,524	47,392	76,003	10,773	4,440	44,437	
Female	130,324	9,458	1,875	20,733	29,831	4,118	1,669	21,588	2,1
PUBLIC ADMINISTRATION A	AND DEFEN	ICE; COM	IPULSO	RY SOCIAL	. SECURIT	Y			
Total	381,757	26,163	5,399	68,125	105,834	14,891	6,109	66,025	6,2
Male	251,433	16,705	3,524	47,392	76,003	10,773	4,440	44,437	4,1
Female	130,324	9,458	1,875	20,733	29,831	4,118	1,669	21,588	2,1

Table 5.10 Cont'd

Table 3.10 Cont a							. •		
Minor Industrial Divisions		1				Re	egion	1	
and Sex							Benishangul	0111:55	
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Administration of the state and									
Total	233,595	12,087	4,583	44,740	65,667	8,803	4,313	43,572	
Male	142,363	5,899	2,879	30,240	43,045	5,740		27,787	
Female	91,232	6,189	1,704	14,500	22,622	3,063	1,344	15,785	1,3
Provision of services to the co	,								
Total	142,086	13,408	816	23,024	37,839	5,949	1,773	21,778	
Male	106,074	10,806	645	16,940	31,433	4,945	1,448	16,077	1,3
Female	36,012	2,602	171	6,084	6,406	1,004	325	5,701	8
Compulsory social security ac	tivities and p		ecurity						
Total	6,076	667	-	361	2,329	138	23	675	
Male	2,996	-	-	213	1,526	88	23	574	
Female	3,080	667	-	148	803	50	-	101	
MAJOR GROUP 12: EDUCA									
Total	404,111	28,992	1,438	76,701	108,393	12,064	3,544	80,524	
Male	239,864	18,384	959	46,109	63,939	9,077	2,593	49,807	
Female	164,247	10,608	479	30,592	44,455	2,987	951	30,717	3
EDUCATION									
Total	404,111	28,992	1,438	76,701	108,393	12,064	3,544	80,524	
Male	239,864	18,384	959	46,109	63,939	9,077	2,593	49,807	
Female	164,247	10,608	479	30,592	44,455	2,987	951	30,717	3
Primary education									
Total	188,576	12,809	992	36,837	54,035	6,034	1,617	35,950	
Male	89,144	6,840	561	16,203	26,653	4,335	1,104	20,106	3
Female	99,432	5,969	431	20,634	27,382	1,699	513	15,844	4
Secondary education									
Total	105,660	5,178	313	21,103	23,855	5,012	943	17,374	1,5
Male	82,154	4,342	301	17,508	17,262	4,280	831	13,694	1,4
Female	23,506	835	12	3,595	6,593	732	112	3,680	•
Technical and vocational, drivi	ng and typin	g							
Total	27,470	2,096	49	6,785	9,943	310	98	4,677	
Male	19,868	1,987	37	5,014	7,652	210	43	3,061	
Female	7,602	109	12	1,771	2,291	101	55	1,617	
Higher education/college and	university ed	ducation/							
Total	82,405	8,909	85	11,976	20,561	708	886	22,523	2
Male	48,698	5,215	60	7,384	12,372	252	615	12,946	
Female	33,707	3,694	25	4,592	8,189	455	271	9,577	
	, -	•	_	, -	,			,	

Table 5.10 Cont'd

Tubic circ cont a									
Minor Industrial Divisions						Re	gion		
and Sex					_		Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
MAJOR GROUP 13: HEALTH									
Total	182,330	11,355	1,287	28,520	47,973	6,128	3,144		
Male	91,421	5,214	750	15,196	23,355	3,123	1,680		
Female	90,909	6,141	538	13,324	24,619	3,005	1,465	14,950	5
HEALTH AND SOCIAL WORK									
Total	182,330	11,355	1,287	28,520	47,973	6,128	3,144		
Male	91,421	5,214	750	15,196	23,355		1,680		
Female	90,909	6,141	538	13,324	24,619	3,005	1,465	14,950	5
Human health activities									
Total	150,770	10,550	1,189	25,831	36,015	4,699	2,543		
Male	71,436	4,789	676	13,931	16,778		1,168		
Female	79,334	5,762	514	11,900	19,237	3,005	1,375	13,921	5
Veterinary activities									
Total	11,500	307	62	206	6,081	629	523	2,379	
Male	9,234	307	50	206	4,216	629	434	2,210	
Female	2,266	-	12	-	1,865	-	89	168	
Social work activities									
Total	20,060	497	36	2,483	5,877	800	78		
Male	10,751	118	24	1,059	2,361	800	78		
Female	9,308	379	12	1,424	3,516	-	-	861	
MAJOR GROUP 14: OTHER	COMMUNIT	TY, SOCI	al and	PERSONA	L				
SERVICE ACTIVITIES									
Total	566,625	37,803	7,745	112,322	170,206		7,957		
Male	330,050	19,566	2,648	60,314	111,319	12,059	3,476		
Female	236,574	18,237	5,097	52,009	58,887	9,648	4,481	46,074	. 7
SEWAGE AND REFUSE DISI									
Total	15,954	903	168	6,260	2,369	421	-	2,461	
Male	7,529	217	144	3,108	1,232	161	-	1,527	
Female	8,425	686	24	3,152	1,137	260	-	934	
Sewage and Refuse Disposal	, Sanitation	and Simila	ar Activit	ies					
Total	15,954	903	168	6,260	2,369	421	-	2,461	
Male	7,529	217	144	3,108	1,232	161	-	1,527	
Female	8,425	686	24	3,152	1,137	260	-	934	
ACTIVITIES OF MEMBERSH	IP ORGANI	ZATIONS	N.E.C.						
Total	75,239	5,060	668	11,511	24,355	1,194	554	9,807	. 6
Male	58,380	3,861	524	9,682	19,212	778	351	6,923	
Female	16,858	1,199	144	1,829	5,143	415	203	2,885	

Table 5.10 Cont'd

Table 5.10 Cont d	1								
Minor Industrial Divisions						Re	egion	1	
and Sex							Benishangul		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Other Activities of business ar			_						
Total	12,873	1,545	13	2,367	4,701	681	-	300	
Male	9,621	1,276	13	1,335	4,620	316		210	
Female	3,252	270	-	1,032	81	365	-	90	
Activities of trade unions									
Total	787	-	-	359	428	-	-	-	
Male	787	-	-	359	428	-	-	-	
Female	-	-	-	-	-	-	-	-	
Activities of other membership	organizatio	ns							
Total	53,771	3,514	655	8,298	16,984	513	411	7,709	Ę
Male	47,328	2,585	511	7,766	14,108	462	351	6,712	4
Female	6,443	929	144	532	2,876	50	60	996	
RECREATIONAL, CULTURA	L AND SPO	RTING A	CTIVITIE	S					
Total	40,094	2,578	368	8,317	6,504	689	472	6,386	•
Male	28,351	1,830	368	5,790	4,948	374	249	4,655	•
Female	11,743	748	-	2,527	1,557	314	223	1,731	
Motion picture, radio, televisio	n and other	entertainr	ment act	ivities					
Total	14,768	1,262	188	1,755	1,862	202	221	3,275	
Male	11,815	1,044	188	1,273	1,399	202	72	2,416	
Female	2,952	218	-	482	462	-	149	858	
News agency activities									
Total	2,711	317	-	886	465	121	-	421	
Male	1,583	208	-	663	41	121	-	421	
Female	1,128	109	-	223	424	-	-	-	
Library, archives, museums a	nd other cult	tural activi	ities						
Total	4,240	-	-	1,583	273	-	74	757	
Male	1,613	-	-	718	109	-	-	337	
Female	2,627	-	-	865	164	-	74	421	
Sporting and other recreations	al activities								
Total	18,375	999	180	4,093	3,905	366	177	1,933	
Male	13,339	578	180	3,135	3,398	52	177	1,482	
Female	5,036	421	-	957	507	314	-	452	
OTHER SERVICE ACTIVITIE	S								
Total	435,338	29,262	6,542	86,235	136,978	19,404	6,931	75,767	1,3
Male	235,791	13,658	1,613	41,734	85,928	10,746	2,877	35,242	
Female	199,547	15,604	4,929	44,501	51,050	8,658	4,055	40,525	

Table 5.10 Cont'd

						Re	gion		
Minor Industrial Divisions						110	Benishangul		
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gamb
Repairing of old cloths	104.	ı ığıay	7 11 (41	, armara	Oronna	Comano	Jamaz	01111111	Juine
Total	91,551	9,218	452	11,377	29,340	1,712	571	10,747	1
Male	39,023	2,535	133	5,524	16,433	1,400	259	3,392	•
Female	52,528	6,683	320	5,853	12,907	313	312	7,355	1
Other service activities	02,020	0,000	020	0,000	12,001	010	012	7,000	
Total	343,787	20,043	6.090	74,857	107,638	17,692	6,360	65,019	1,1
Male	196,768	11,122	1,480	36,210	69,495	9,346	2,618	31,850	.,.
Female	147,020	8,921	4,609	38,647	38,142	8,346	3,743	33,169	F
MAJOR GROUP 15: PRIVAT						0,0 .0	0,1 10	00,100	
Total	269,616	11,984	2,229	37,869	48,448	3,665	1,453	32,937	4
Male	20,520	409	417	3,573	3,762	-	54	2,835	•
Female	249,095	11,575	1,812	34,296	44,686	3,665	1,399	30,102	4
PRIVATE HOUSEHOLDS WIT	,		•	0 .,=00	,000	0,000	.,000	00,.02	
Total	269,616	11,984	2,229	37,869	48,448	3,665	1,453	32,937	4
Male	20,520	409	417	3,573	3,762	-,	54	2,835	
Female	249,095	11,575	1,812	34,296	44,686	3,665	1,399	30,102	4
Private households with emplo			.,	- 1,=	,	-,	1,200	,	
Total	269,616	11,984	2,229	37,869	48,448	3,665	1,453	32,937	4
Male	20,520	409	417	3,573	3,762	-	54	2,835	
Female	249,095	11,575	1,812	34,296	44,686	3,665	1,399	30,102	4
MAJOR GROUP 16:EXTRA-1					IGOS?	,	,	,	
Total	34,003	2,874	320	3,455	6,972	1,054	480	3,195	3
Male	21,696	2,171	176	2,110	4,993	516	247	1,812	3
Female	12,307	703	144	1,345	1,978	538	233	1,383	
EXTRA-TERRITORIAL ORGA	ANIZATIONS	3							
Total	34,003	2,874	320	3,455	6,972	1,054	480	3,195	3
Male	21,696	2,171	176	2,110	4,993	516	247	1,812	3
Female	12,307	703	144	1,345	1,978	538	233	1,383	
Extra-territorial organizations									
Total	34,003	2,874	320	3,455	6,972	1,054	480	3,195	3
Male	21,696	2,171	176	2,110	4,993	516	247	1,812	3
Female	12,307	703	144	1,345	1,978	538	233	1,383	
Not stated									
Total	3,162	199	-	91	1,366	116	-	358	
Male	2,087	-	-	91	1,014	58	-	179	
Female	1,076	199	-	-	352	58	-	179	

Table 5.11 Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Statu Country-Total: 2012

	intry-10ta	al. 2012									
		1				Status	in Employm	ent			
				Pa	id employees						
Age Group and Sex	Total Employed					Employee -NGO				Una alab	
	Person	Total paid Employees	Employee -Government	Employee- Government Parasitatal	Employee -Private organization	(including International organization)	Employee Domestic	Other Employee	Self Employed	Unpaid Family Worker	Employer A
All Ages	J.				,	,	U.		, ,		. , , ,
Total	5,726,116	2,867,785	876,306	297,994	1,297,136	64,029	267,461	64,860	2,215,946	493,930	39,445
Male	3,202,042	1,682,365	541,563	206,729	822,436	38,077	19,940	53,620	1,218,230	206,822	29,577
emale	2,524,074	1,185,420	334,742	91,265	474,700	25,952	247,520	11,240	997,715	287,108	9,869
10-14											
Total	195,567	38,620	-	1,308	11,396	-	25,500	415	20,693	132,539	-
Male	94,187	8,907	-	679	5,789	-	2,023	415	13,118	70,534	-
emale 15-19	101,380	29,713	-	629	5,607	-	23,477	-	7,575	62,006	-
Total	492,071	235,653	7,428	5,695	111,089	826	105,356	5,258	112,936	135,484	492
Male	205,515	67,300	2,402	3,641	53,082	-	4,143	4,032	67,597	66,213	39
emale 20-24	286,556	168,352	5,025	2,054	58,008	826	101,212	1,227	45,339	69,271	453
Total	902,304	542,346	127,352	42,164	295,822	6,037	65,518	5,454	254,904	75,209	3,533
Male	457,454	259,685	61,335	23,485	165,504	2,339	3,039	3,983	144,480	36,178	2,501
emale	444,850	282,661	66,017	18,678	130,318	3,698	62,479	1,471	110,424	39,031	1,033
25-29	444,030	202,001	00,017	10,070	130,310	5,070	02,477	1,771	110,424	37,031	1,033
Total	1,128,487	657,158	216,957	62,465	325,380	12,475	30,192	9,689	393,264	47,288	8,150
Male	653,525	395,376	127,403	43,340	207,590	6,690	2,011	8,341	221,229	15,302	5,629
emale 30-34	474,962	261,782	89,553	19,125	117,790	5,785	28,181	1,348	172,035	31,986	2,521
Total	767,262	380,487	133,388	45,043	174,303	7,927	10,824	9,002	340,164	31,105	5,612
Male	448,530	243,162	81,506	28,524	117,976	4,884	2,629	7,645	189,409	6,389	4,300
emale	318,731	137,325	51,882	16,520	56,328	3,043	8,195	1,357	150,755	24,717	1,312
35-39											
Total	742,001	356,911	124,583	41,847	152,635	16,990	11,037	9,820	335,037	28,976	7,703
Male	423,955	225,291	75,602	28,625	101,313	10,516	1,171	8,064	177,060	4,580	6,807
emale	318,046	131,620	48,981	13,222	51,322	6,473	9,865	1,757	157,976	24,396	896
40-44											
Total	445,655	218,490	99,953	26,805	73,949	7,860	4,379	5,543	201,351	13,798	4,109
Male	273,476	152,396	66,019	19,881	55,171	5,652	750	4,923	112,000	1,972	3,338
emale	172,179	66,094	33,934	6,925	18,778	2,208	3,629	620	89,351	11,826	770

Table 5.11 Cont'd

1 able 5.11	Conta										
						Emplo	oyment Stati	us			
				Р	aid employee						
Age Group and Sex	Total Employed Persons	Total paid Employees	Employee -Government	Employee- Government Parasitatal	Employee -Private organization	Employee -NGO (including International organization)	Employee Domestic		Self Employed	Unpaid Family Worker	Employer
45-49											
Total	361,687	172,087	71,162	26,422	58,178	3,878	5,678	6,769	168,222	11,824	3,570
Male	211,592	120,120	. ,			1,904	897	5,561	84,784	965	, -
emale	150,094	51,967	19,828	6,813	17,364	1,973	4,781	1,209	83,438	10,859	869
50-54											
Total	266,025	122,180	55,961	19,552	37,107	4,222	2,639	2,700	131,790	5,754	
Male	153,598	88,584		,		2,711	389	1,331	59,808	1,835	
emale 55-59	112,428	33,596	13,146	5,589	9,733	1,510	2,251	1,368	71,982	3,919	767
Total	152,584	66,658	24,896	13,048	22,858	1,221	1,143	3,493	79,384	3,142	1,293
Male	96,469	56,172	20,472	12,725	18,033	829	664	3,449	38,555	542	649
emale	56,115	10,486	4,424	323	4,825	392	479	44	40,829	2,600	644
60-64											
Total	113,615	38,565	8,659	8,006	12,999	2,594	2,854	3,453	69,329	3,459	854
Male	75,221	32,876	7,550	7,235	11,203	2,551	1,696	2,641	40,421	747	577
emale	38,393	5,689	1,108	771	1,796	43	1,158	812	28,908	2,711	277
65+											
Total	158,859	38,629	5,968	5,638	21,420	-	2,341	3,263	108,872	5,350	2,370
Male	108,519	32,497	5,124	5,021	18,588	-	528	3,236	69,769	1,565	2,043
emale	50 340	6 133	844	617	2 833	_	1 812	27	39 103	3 785	327

Table 5.12 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region Status in Employment: 2012

	- Otal	.us III L	pioyii	Hent. 201								
							Status in Emp	ployment		·		
					Pa	aid employees				<u> </u>		
Region and Sex	Total Employed persons	Total Employed Person	Total paid Employees	Employee Government	Employee- Gov't Parasitatal	Employee- Private organization	Employee- NGO (including International organization)	Employee Domestic	Other Employee	Self Employed	Unpaid Family Worker	Employer
COUNTRY TOTAL	 L					1 - 1	1				•	
Total	5,726,116	5,726,116	2,867,785	876,306	297,994	1,297,136	64,029	267,461	64,860	2,215,946	493,930	39,445
Male	3,202,042	3,202,042	1,682,365	541,563	206,729	822,436	38,077	19,940	53,620	1,218,230	206,822	29,577
Female	2,524,074	2,524,074	1,185,420	334,742	91,265	474,700	25,952	247,520	11,240	997,715	287,108	9,869
TIGRAY REGION												
Total	391,891	391,891	189,677	64,502	17,399	86,620	5,385	11,984	3,788	148,442	34,603	8,969
Male	204,483	204,483	107,939	39,498	10,737	50,457	3,407	409	3,431	71,719	12,706	7,111
Female	187,408	187,408	81,738	25,003	6,661	36,163	1,978	11,575	357	76,722	21,897	1,858
AFFAR REGION												
Total	77,068	77,068	38,124	8,725	7,851	17,641	428	2,229	1,250	28,680	9,266	876
Male	43,657	43,657	25,538	5,868	5,847	12,679	260	417	467	14,317	3,009	721
Female	33,411	33,411	12,586	2,857	2,004	4,961	168	1,812	783	14,363	6,257	155
AMHARA REGION	١											
Total	1,063,958										152,045	8,231
Male	562,815	562,815	257,734	105,571	28,759			2,993	10,407	222,046	58,040	6,152
Female	501,144	501,144	177,511	62,198	10,544	66,838	2,314	33,716	1,901	211,119	94,005	2,079
OROMIA REGION												
Total	1,643,077						17,443					5,930
Male	963,543				52,229						66,955	3,769
Female	679,534	679,534	281,856	81,799	18,238	125,948	7,909	44,435	3,526	325,529	62,132	2,161
SOMALI REGION												
Total	202,547											952
Male	111,869			20,625					.,		10,719	830
Female	90,677		20,169	9,509	4,151	1,892	588	3,665	365	50,995	18,965	122
BENISHANGUL G												
Total	72,518											974
Male	39,854				2,729						6,315	584
Female	32,664	32,664	8,473	3,864	691	2,205	266	1,399	47	11,807	11,363	391

Table 5.12 Cont'd

Table 5.12	Cont a										
							Status in Emp	Joyment			_,
	'	'			Pai	id employees					
Region and Sex	Total Employed persons	Total Employe d Person	Total		Employee-	Employee-	Employee -NGO (including				Uni
	F======		paid	Employee	Gov't	Private	International	Employee	Other	Self	Far
CAUND DECICAL	'	<u></u> '	Employees	Government	Parasitatal	organization	organization)	Domestic	Employee	Employed	Wo
SNNP REGION	007.00/	007.007	201.077	1/0.004	24.000	100.071	7.010	20 517		250 (04	00
Total	837,386				34,090		7,919				/
Male	466,330		,	. ,	23,474		., -				32,
Female	371,057	371,057	151,067	61,571	10,615	45,395	3,102	29,682	702	163,997	51,
GAMBELLA REG											
Total	35,656	35,656	,	10,631	828	3,862	498	470		- 1 -	
Male	18,797	18,797	10,877	7,100	550	,		-	333	,	
Female	16,859	16,859	5,851	3,531	277	1,407	58	470	108	9,813	1,
HARARI REGION	I.										
Total	49,969	49,969	26,144	10,828	1,805	11,060	634	1,670	147	20,061	1,
Male	26,901	26,901	15,197	5,950	1,406	7,188	449	107	98	9,812	
Female	23,068	23,068	10,947	4,878	399	3,873	186	1,563	49	10,249	1
ADDISABAB CITY ADMINISTRATION	•										
Total	1,250,842	1,250,842	878,145	156,428	107,681	451,525	21,298	123,036	18,177	316,991	23
Male	709,206	709,206	463,149	80,817	70,542	275,556	12,361	9,098	14,775	213,139	10
Female	541,635	541,635	414,996	75,611	37,139	175,969	8,937	113,938	3,402	103,851	12
DIRE DAWA ADM	/INISTRATION	1									
Total	101,204	101,204	49,847	11,946	2,855	27,132	1,501	5,529	884	35,773	10
Male	54,588	54,588	29,622	8,026	2,310	17,084	1,054	265	884	16,503	4
Female	46,617	46,617	20,225	3,920			447	5,265	-	40.070	

Table 5.13 Currently Employed Population of Major Towns Aged Ten Years and Above by Sex at Status in Employment: 2012

	Otatao	– p	obylitelit.	-01-							
							in Employmer	nt			
					Paid Employee						
Major Towns and Sex						Employee - NGO					
and Sex	Total			Employee-	Employee-	(including				Unpaid	
	Employed		Employee-	Government	Private	International	Employee	Other	Self	Family	
	Person	Emplyees	Government	Parasitatal	organization	organization)	Domestic	Employee	Employed	Worker	Emplo
MAJOR TOWNS TO											
Total	2,212,903	1,410,578	307,281	151,342	710,526	33,386	179,207	28,837	642,031	92,307	20,
Male	1,240,813	768,740	169,289	100,927	441,332	20,559	12,577	24,056	387,932	36,928	15,
Female	972,089	641,837	137,991	50,414	269,194	12,827	166,630	4,782	254,099	55,380	4,
MEKELE TOWN											
Total	121,116	65,647	15,766	4,870	34,494	1,243	8,106	1,168	39,040	7,567	5,
Male	65,727	36,135	8,494	2,981	22,339	1,037	218	1,067	20,199	2,372	4,
Female	55,389	29,512	7,272	1,889	12,155	206	7,888	102	18,841	5,195	
ASAYITA TOWN											
Total	8,369	3,825	1,790	195	1,333	36	434	36	3,411	824	
Male	4,624	2,390	1,116	108	1,093	12	26	36	1,697	323	
Female	3,745	1,434	674	87	241	24	408	-	1,714	501	
GONDER TOWN											
Total	97,868	41,659	11,231	3,734	20,469	156	4,648	1,421	29,223	22,242	
Male	51,748	24,006	6,964	2,542	12,664	75	340	1,421	16,892	8,222	
Female	46,121	17,653	4,268	1,192	7,805	80	4,308	_	12,331	14,020	
DESSIE TOWN											
Total	47,543	27,168	8,762	2,137	12,179	761	2,849	480	16,881	1,975	
Male	26,410	14,960	5,019	1,496	7,515	396	99	436	9,672	935	
Female	21,133	12,208	3,743	641	4,664	364	2,750	45	7,209	1,040	
BAHIR DAR TOWN											
Total	93,781	58,017	16,260	5,774	28,805	683	6,065	430	29,128	3,604	1,
Male	46,806	30,124	9,105	3,705	16,007	683	341	282	13,404	1,337	
Female	46,975	27,894	7,155	2,069	12,798	-	5,724	147	15,724	2,267	
BISHOFTU TOWN	,		,	,	•						
Total	41,284	26,422	4,353	3,729	16,282	372	1,350	337	13,276	1,000	
Male	23,681	15,573	2,462	2,432	10,094	227	176	182	7,369	319	
Female	17,602	10,849	1,890	1,297	6,189	145	1,174	155	5,907	681	
ADAMA TOWN	,502		.,570	- , / /	2,.07		.,	.00	27.07		
Total	92,946	58,610	8,236	5,375	36,255	1,297	4,498	2,949	30,152	2,019	
Male	55,237	34,232	4,949	4,036	21,423	642	419	2,764	18,200	977	
Female	37,709	24,378	3,287	1,339	14,832	655	4,079	186	11,951	1,041	
i citiale	31,107	21,070	5,201	1,557	1 1,002	000	1,017	100	11,701	1,011	

Table 5.13 Cont'd

Table 5.13 C	Jiii u					Status ir	n Employme	nt			
					Paid Employee	es	. *				
Major Towns and Sex	Total Employed Person	Total Paid Emplyees	Employee- Government	Employee- Government Parasitatal	Employee- Private organization	Employee- Ngo (including International organization)	Employee Domestic		Self Employed	Unpaid Family Worker	
JIMMA TOWN					<u></u>						
Total	59,065	38,531	13,867	2,861	15,322	1,156	4,647	678	16,499	2,775	
Male	33,607	22,178	7,815	2,328	10,033	876	765	361	9,672	879	
Female	25,458	16,353	6,051	533	5,289	281	3,882	318	6,828	1,896	
SHASHEMENE TOV	VN										
Total	45,485	19,735	3,531	1,845	10,773	726	2,559	301	22,536	2,174	
Male	28,269	12,903	2,492	1,365	8,232	542			13,208		
Female	17,216	6,832	1,039	480	2,541	184	2,489	100			
JIJIGA TOWN											
Total	40,942	18,903	7,146	1,396	7,735	625	1,778	222	17,684	2,865	
Male	23,990	13,009	4,765	861	6,839	373	_		9,067		
Female	16,952	5,893	2,381	535	896	252	1,778	51	8,617		
ASOSA TOWN			·				,			·	
Total	17,931	8,965	4,188	1,148	2,951	139	412	127	6,284	2,437	
Male	10,159	5,783	2,600	907	2,090	106	-	80	3,333	855	
Female	7,772	3,182	1,588	241	861	33	412		2,952		
HAWASSA TOWN											
Total	127,071	78,142	26,552	5,317	31,585	2,260	11,266	1,162	39,296	6,068	
Male	70,534	42,729	14,900	3,481	21,400	1,285	654	1,009	23,555	2,286	
Female	56,537	35,413	11,652	1,836	10,185	975	10,611	153			
GAMBELLA TOWN											
Total	17,486	10,817	6,396	620	2,625	498	358	319	5,796	873	
Male	9,326	6,749	3,815	426	1,777	440	-	291	2,211		
Female	8,160	4,067	2,581	194	848	58	358	28			
HARAR TOWN											
Total	49,969	26,144	10,828	1,805	11,060	634	1,670	147	20,061	1,929	
Male	26,901	15,197	5,950	1,406	7,188	449	107				
Female	23,068	10,947	4,878	399	3,873	186	1,563				
ADDIS ABABA CITY			•		•						
Total	1,250,842	878,145	156,428	107,681	451,525	21,298	123,036	18,177	316,991	23,216	-
Male	709,206	463,149	80,817	70,542	275,556	12,361	9,098		213,139		
Female	541,635	414,996	75,611	37,139	175,969	8,937	113,938		103,851		
DIRE DAWA TOWN		.,	-,	- ,	.,	-,	-,	-,	,	, - , -	
Total	101,204	49,847	11,946	2,855	27,132	1,501	5,529	884	35,773	10,741	2
Male	54,588	29,622	8,026	2,310	17,084	1,054	265		16,503		
Female	46,617	20,225	3,920	545	10,048	447	5,265				



Table 5.14 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Occupati Major Industrial Divisions - COUNTRY TOTAL: 2012

Major Industrial					Major O	ccupational Gro	ups	
Divisions and Sex	Total Employed Persons	Legislator Senior Officials & Managers	Professional	Technical and Associate Professionals	Clerks	Service, Shop & Market sales Workers	Skilled agricultural and Fishery	Crafts and related trades
Total Employed Popula	tion	<u></u>						
Total	5,726,116	165,216	376,286	359,728	316,379	1,516,219	343,555	1,048,02
Male	3,202,042	128,932	259,132	225,287	123,190	646,234	246,151	616,30
Female	2,524,074	36,284	117,154	134,441	193,190	869,985	97,404	
Agriculture, Hunting, Fo				•	•		*	•
Total	469,377	4,025	1,017	2,669	2,594	10,494	313,108	4,96
Male	315,988	3,517	963	1,864	2,185	3,544	225,555	
Female	153,389	509	55	805	409	6,950	87,553	
Mining & Quarrying	,	, , ,				- , 0	- ,	-,
Total	21,374	685	-	770	1,375	1,134	-	9,904
Male	16,486	649	-	770	-	467	-	7,81
Female	4,888	36	-	-	1,375	666	-	2,09
Manufacturing	.,500				.,5. 3	330		_,00
Total	757,055	12,715	6,374	11,159	29,916	40,415	2,441	497,03
Male	419,831	7,496	3,763	6,871	11,799	17,395	1,535	
Female	337,224	5,218	2,611	4,288	18,117	23,020	906	
Electricity, Gas and Wa		-,0	_,•	.,	~,	,0	550	,=0
Total	53,226	2,368	3,382	3,490	7,464	4,035	-	9,842
Male	38,176	2,368	2,099	2,606	2,410	1,439	-	9,842
Female	15,050	_,,,,,	1,283	885	5,055	2,596	-	2,01
Construction	. 5,555		. ,200	000	5,500	_,000		
Total	395,129	8,082	13,117	11,040	11,222	1,708	-	231,10
Male	324,893	6,908	11,432	9,226	5,173	1,584	-	207,188
Female	70,235	1,174	1,685	1,813	6,049	124	-	23,910
Whole Sale and Retail		1,114	1,000	1,010	5,545	127		20,01
Total	1,235,258	4,984	4,778	12,567	13,367	890,574	22,269	119,01
Male	625,873	3,498	2,005	7,840	5,455	415,612	15,359	
Female	609,386	1,486	2,773	4,727	7,912	474,961	6,910	
Hotels and Restaurants		1,-100	2,113	7,121	1,012	77 - 7,30 1	0,310	- 1,03
Total	457,810	10,941	3,289	2,483	18,748	280,458	_	105,964
Male	112,237	6,877	1,715	1,473	7,810	70,230	-	6,48
Female	345,572	4,064	1,575	1,010	10,938	210,228	-	99,48
Transport, Storage and			1,070	1,010	10,330	210,220	-	<i>33,</i> 40
Total	260,370	2,190	5,341	E 400	12542	10,664	682	5,02
Male	260,370 246,962	2,190	5,341 4,112	5,498 4,034	12,543 9,260	9,266	682	
		•		•	•	•	682	
Female	13,407	182	1,229	1,464	3,282	1,398	-	5

Table 5.14 Cont'd

Major Industrial					Major C	Occupational Gro	oups	
Divisions and Sex	Total Employed Persons	Legislator Senior Officials & Managers	Professional	Technical and Associate Professionals	Clerks	Service, Shop & Market sales Workers	Skilled agricultural and Fishery	Crafts and related trades
Financial Intermediation								
Total	115,655	7,677	24,314	14,146	44,162	1,220	, <u>-</u>	- 89
Male	69,717		17,063	•		•	. <u>-</u>	- 89
Female	45,938						, <u>-</u>	
Real Estate, Renting ar			•		•	•		
Total	119,259		18,817	15,523	34,732	14,269	62	11,912
Male	74,806			•		•		
Female	44,453			•		•		
Public Administration a	•	,		•	· =•	•		•
Total	381,757	71,351	43,361	64,181	65,786	66,346	1,491	4,366
Male	251,433		28,521	45,031	22,664	•	•	
Female	130,324							,
Education	- - - , -	/-	•	- •	· - ,	- ,		•
Total	404,111	13,318	184,678	112,538	26,870	14,942	912	2 4,532
Male	239,864			•				
Female	164,247							,
Health and Social Work		-,-	,-	 ,-	,			-,
Total	182,330	6,384	49,270	64,132	16,314	11,213	343	3 2,345
Male	91,421							,
Female	90,909			·		•		- 713
Other Community, Soc				• .,	,	• ,		-
Total	566,625			35,887	24,600	141,063	1,924	34,062
Male	330,050		,	,	•			
Female	236,574			•	13,439	•		
Private Households wit			1,010	0,	10, 100	02, .00	1,100	1,00
Total	269,616		_	_	89	24,053		- 7,06
Male	209,010		_	_	-			- 7,00. - 45
Female	249,095		_	_	89	·		- 45 - 6,60
Extra-Territorial Organi					00	22,071		0,00
Total	34,003		4,256	3,647	6,597	2,737	323	3 79
Male	21,696			,	3,131			
Female	12,307			·	•			- 42
Not Stated	12,301	1,002	J4∠	1,∠30	3,400	1,032	-	42
Total	3,162	-	-	-	_	896	, -	
Male	2,087		-	-	_	896		
Female	1,076		_	_	-			-

Table 5.15 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Industria Status in Employment - COUNTRY TOTAL: 2012

	Status I	n Employn	ieni - coo	NIKI IOI							
						Status in Em	ployment				
Major Industrial Divisions and Sex	Total Employed Population	Employee- Government	Employee- Gov't Parasitatal	Employee- Private organization	Employee-NGO (including International organization)	Employee Domestic	Other Paid Employee	Self Employed	Unpaid Family Worker	Employer	App
Total Employed Po	pulation										
Total	5,726,116	876,306	297,994	1,297,136	64,029	267,461	64,860	2,215,946	493,930	39,445	
Male	3,202,042	541,563	206,729	822,436	38,077	19,940	53,620	1,218,230	206,822	29,577	
Female	2,524,074	334,742	91,265	474,700	25,952	247,520	11,240	997,715	287,108	9,869	
Agriculture, Hunting											
Total	469,377	1,081	15,114	41,323	-	-	10,877	246,716	144,277	3,298	
Male	315,988	704	10,586	27,690	-	-	7,832	176,987	86,885	2,347	
Female	153,389	377	4,528	13,632	-	-	3,046	69,730	57,392	950	
Mining &											
Quarrying											
Total	21,374	39	2,977	5,326	-	-	883	9,877	215	36	
Male	16,486	39	1,579	3,531	-	-	883	8,581	215	36	
Female	4,888	-	1,399	1,795	-	-	-	1,295	-	-	
Manufacturing											
Total	757,055	-	43,894	278,009	-	-	1,236	346,123	58,987	9,073	
Male	419,831	-	27,139	184,207	-	-	466	159,689	27,866	8,331	
Female	337,224	-	16,755	93,802	-	-	771	186,434	31,121	742	
Electricity, Gas and											
Total	53,226	16,134	19,950	5,286	-	-	1,875	4,622	5,250	109	
Male	38,176	10,647	17,430	4,380	-	-	1,530	3,584	497	109	
Female	15,050	5,487	2,520	906	-	-	345	1,038	4,753	-	
Construction				404					= = 0.0		
Total	395,129	-	48,805	196,787	-	-	11,356	112,080	5,528	3,071	
Male	324,893	-	39,856	152,837	-	-	10,301	103,103	3,853	3,071	
Female	70,235	-	8,949	43,950	-	-	1,055	8,977	1,675	-	
Whole Sale and Re			10.0/0	477.075			4.070	000 507	440.005	0.405	
Total	1,235,258	-	12,069	177,065	-	-	1,079	909,526	113,935	9,135	
Male	625,873	-	7,250	107,481	-	-	602	448,428	48,236	7,023	
Female	609,386	-	4,819	69,584	-	-	477	461,098	65,699	2,112	
Hotels and Restaur				150.071			000	010.107	(4 0 5 0	0.007	
Total	457,810	-	6,993	159,371	-	-	982	213,196	64,850	9,337	
Male	112,237	-	3,189	61,296	-	-	- 000	31,351	12,113	4,081	
Female	345,572	- lootions	3,804	98,075	-	-	982	181,845	52,737	5,257	
Transport, Storage		ications	47.007	100.045			0.504	/0 / 25	0.740	1 051	
Total	260,370	-	47,036	129,045	-	-	3,501	69,635	3,740	1,351	
Male	246,962	-	41,029	123,497	-	-	3,076	68,308	3,740	1,299 51	
Female	13,407	-	6,008	5,548	-	-	425	1,327	-	51	

Table 5.15 Cont'd

Table 5.15 CC						Status ir	n Employmen	t			
Major Industrial Divisions and Sex	Total Employed Population	Employee- Government	Employee- Gov't Parasitatal	Employee- Private organization	Employee-NGO (including International organization)	Employee Domestic	Other Paid Employee	Self Employed	Unpaid Family Worker	Employer	Aį
Financial Intermedia			07.04:	07.467			450				
Total	115,655	-	87,364	27,137	-	-	153	635	-	-	
Male	69,717	-	51,413	18,039	-	-	-	264	-	-	
Female	45,938	-	35,950	9,097	-	-	153	370	-	-	
Real Estate, Rentin		ss Activities									
Total	119,259	-	4,695	70,792	-	-	890	38,462	1,891	1,344	
Male	74,806	-	2,662	37,102	-	-	287	31,935	1,018	1,035	
Female	44,453	-	2,033	33,690	-	-	603	6,527	873	310	
Public and Adminis											
Total	381,757	381,697	-	-	-	-	-	-	-	-	
Male	251,433	251,433	-	-	-	-	-	-	-	-	
Female	130,324	130,263	-	-	-	-	-	-	-	-	
Education											
Total	404,111	327,369	3,956	62,820	6,100	-	360	-	-	-	
Male	239,864	204,557	1,591	28,490	2,469	-	300	-	-	-	
Female	164,247	122,812	2,365	34,330	3,631	-	61	-	-	-	
Health and Social V	Vork										
Total	182,330	115,634	2,081	32,963	14,134	-	102	10,747	3,959	1,170	
Male	91,421	54,772	1,057	16,703	6,976	-	-	8,009	2,365	1,170	
Female	90,909	60,862	1,025	16,260	7,158	-	102	2,739	1,594	-	
Other Community, S Activities											
Total	566,625	34,353	3,060	109,851	9,881	-	31,566	252,968	91,297	1,522	
Male	330,050	19,411	1,950	56,285	6,935	-	28,345	177,200	20,033	1,075	
Female	236,574	14,941	1,110	53,565	2,946	-	3,221	75,768	71,264	447	
Private Households		ed Persons									
Total	269,616	-	-	-	-	267,461	-	-	-	-	
Male	20,520	-	-	-	-	19,940	-	-	-	-	
Female	249,095	-	-	-	-	247,520	-	-	-	-	
Extra-Territorial Org	ganizations an	d NGOs									
Total	34,003	-	-	-	33,913	-	-	-	-	-	
Male	21,696	-	-	-	21,696	-	-	-	-	-	
Female	12,307	-	-	-	12,217	-	-	-	-	-	
Not Stated											
Total	3,162	-	-	1,363	-	-	-	1,359	-	-	
Male	2,087	-	-	896	-	-	-	792	-	-	
Female	1,076	-	-	466	-	-	-	567	-	-	



Table 5.16 Currently Employed Population of Urban Areas Aged Ten Years and Above by Major Occupation Status in Employment - COUNTRY TOTAL: 2012

	1	Status III EI	iipioyiiieiit -	COUNTRI	101AL. 2012					
Major				1		Status in En	nployment	1		
Occupational					Employee-NGO					
Groups and	Total		Employee-	Employee-	(including		0.1	0 1/	Unpaid	
Sex	Employed	Employee-	Government	Private	International	Employee	Other	Self	Family	le
Tatal Cambara	Population	Government	Parasitatal	organization	organization)	Domestic	Employee	Employed	Worker	Emplo
Total Employe		070.000	000 005	4 007 400	04.000	007.404	00.400	0.045.040	400.000	00.44
Total	5,726,116	876,306	296,685	1,297,136	64,029	267,461	66,168	2,215,946	,	,
Male	3,202,042	541,563	206,049	822,436	38,077	19,940	54,299	1,218,230		
	2,524,074	334,742	90,636	474,700	25,952	247,520	11,869	997,715	287,108	9,869
Legislator Ser		-								
Total	165,216	92,593	14,663	28,668	8,570	-	1,376	13,023	1,085	4,07
Male	128,932	75,189	10,152	21,501	6,352	-	1,332	9,883	223	3,189
Female Professional	36,284	17,403	4,511	7,168	2,218	-	44	3,140	863	888
Total	376,286	237,666	37,861	69,696	9,518	-	4,501	13,494	684	1,19
Male	259,132	164,562	27,266	43,151	5,915	-	4,347	11,054	441	885
Female	117,154	73,104	10,595	26,544	3,603	-	153	2,440	242	310
Technical and	Associate F	Professionals								
Total	359,728	204,820	33,037	63,958	11,742	-	14,321	21,829	1,478	1,26
Male	225,287	118,709	22,258	35,312	7,941	-	14,008	18,585	849	1,26
Female	134,441	86,111	10,780	28,646	3,801	-	312	3,244	629	-
Clerks										
Total	316,379	110,104	59,526	101,820	8,515	89	2,314	19,686	7,550	983
Male	123,190	34,644	26,957	36,449	4,033	-	1,275	14,459	2,623	983
Female	193,190	75,459	32,569	65,371	4,482	89	1,039	5,228	4,927	-
Service, Shop	& Market sa	ales Workers								
Total	1,516,219	84,331	9,226	198,686	7,166	22,222	3,349	963,419	194,352	14,92
Male	646,234	62,187	4,373	82,850	1,492	925	1,498	414,880	62,583	8,583
Female	869,985	22,144	4,853	115,836	5,674	21,297	1,852	548,539	131,768	6,338
Skilled agricul	tural and Fis	hery								
Total	343,555	2,746	8,776	12,454	323	-	4,561	225,914	79,705	3,310
Male	246,151	1,839	5,969	7,405	323	-	3,695	174,784	46,943	2,45
Female	97,404	907	2,807	5,049	-	-	866	51,130	32,762	857
Crafts and rela	,		,	•				, -	•	
Total	1,048,027	13,206	39,012	295,192	1,089	7,065	11,597	569,852	66,159	10,88
Male	616,303	9,213	32,612	225,263	454	457	10,287	281,414	25,948	9,66
Female	431,723	3,993	6,399	69,930	635	6,607	1,310	288,438	40,211	1,22
	,	5,555	5,555	55,555	550	٥,٥٥.	.,5.0	_55,.55	,	.,

Table 5.16 Cont'd

Major						Status in Er	nployment			
Occupational					Employee-NGO					
Groups and	Total		Employee-	Employee-	(including				Unpaid	
Sex	Employed	Employee-	Government	Private	International	Employee	Other	Self	Family	
COX	Population	Government	Parasitatal	organization	organization)	Domestic	Employee	Employed	Worker	Employe
Plant ,Machine	e Operators	& Assemble	rs							
Total	305,437	17,419	32,743	174,277	3,616	479	2,854	62,452	6,951	2,2
Male	286,619	16,917	30,910	161,244	3,616	479	2,775	60,328	6,363	3 2,1
Female	18,818	502	1,833	13,033	-	-	79	2,124	588	3
Elementary O	ccupation									
Total	1,293,289	113,420	61,841	352,206	13,490	237,606	21,296	324,918	135,965	5 5
Male	669,003	58,303	45,553	209,260	7,950	18,079	15,083	232,053	60,848	3
Female	624,286	55,117	16,287	142,947	5,539	219,527	6,213	92,865	75,117	7 2
N.S										
Total	1,979	-	-	179	-	-	-	1,359		-
Male	1,190	-	-	-	-	-	-	792		-
Female	788	-	-	179	-	-	-	567		-

Table 5.17 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Age Gro Hours Worked Per Week - COUNTRY TOTAL: 2012

Age Group					Number of	Hours Wor	ked Per We	ek	
Age Group and Sex									
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57
All Ages									
Total	5,726,116	275,568	9,353	322,625					
Male	3,202,042	145,905	2,594	117,929		252,231			
Female	2,524,074	129,663	6,759	204,697	241,566	251,337	333,936	512,712	2 321,94
10-14									
Total	195,567	3,412	402	54,354					
Male	94,187	1,450	-	24,750					
Female	101,380	1,962	402	29,604	17,480	12,518	8,593	5,065	10,56
15-19									
Total	492,071	15,840	507	50,811	57,135	47,343			
Male	205,515	6,003	43	20,890	27,359	24,301	20,094	32,313	
Female	286,556	9,838	464	29,921	29,776	23,042	20,748	35,425	41,97
20-24									
Total	902,304	35,140	1,388	39,164	57,790	76,475	107,980	209,778	3 135,14
Male	457,454	17,277	577	13,204	26,624	37,658	50,181	106,287	69,32
Female	444,850	17,864	811	25,959	31,167	38,818	57,799	103,490	65,81
25-29									
Total	1,128,487	52,946	2,276	43,442	,		134,939	287,926	
Male	653,525	25,161	1,044	14,660	30,958	49,749			113,61
Female	474,962	27,785	1,232	28,781	32,732	40,907	61,859	116,510	64,00
30-34									
Total	767,262	40,498	961	30,004	46,653	61,204	99,071	182,308	119,55
Male	448,530	23,357	-	9,134	16,971	27,742	47,247	110,464	81,68
Female	318,731	17,140	961	20,870	29,682	33,463	51,824	71,845	37,86
35-39									
Total	742,001	34,269	517	30,065	46,960	56,665	98,862	181,246	107,13
Male	423,955	15,956	-	11,886	16,685	27,504	52,824	104,704	68,74
Female	318,046	18,313	517	18,178		29,161	46,038	76,542	38,38
40-44									
Total	445,655	22,892	223	17,442	28,697	38,516	63,822	107,626	65,85
Male	273,476	12,819	_	7,395					
Female	172,179	10,073	223	10,048	15,430				•

Table 5.17 Cont'd

					Number of	Hours Work	ed Per Wee	k	
Age Group and Sex									
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57
45-49									
Total	361,687	18,311	346	16,289	27,127	30,632	50,304	79,274	50,91
Male	211,592	11,008	-	4,654	10,728	14,167	26,082	49,963	34,70
Female	150,094	7,303	346	11,635	16,399	16,465	24,222	29,310	16,21
50-54									
Total	266,025	19,576	1,185	11,476	27,483	26,439	32,395	57,784	30,19
Male	153,598	10,403	367	590	10,924	11,167	19,508	35,751	20,31
Female	112,428	9,173	818	10,887	16,559	15,271	12,887	22,033	9,87
55-59									
Total	152,584	10,989	422	7,192	10,894	15,865	18,009	33,695	21,91
Male	96,469	5,799	75	2,834	3,316	9,214	10,823	26,113	14,83
Female	56,115	5,190	346	4,359	7,578	6,651	7,186	7,582	7,07
60-64									
Total	113,615	8,783	1,067	7,963	10,161	9,230	14,360	21,212	16,50
Male	75,221	7,215	428	1,956	4,693	4,732	7,659	17,221	11,45
Female	38,393	1,568	640	6,007	5,468	4,498	6,701	3,991	5,05
65+									
Total	158,859	12,913	60	14,423	17,782	22,280	15,342	24,726	19,43
Male	108,519	9,457	60	5,975	8,763	12,604	9,795	19,842	15,09
Female	50,340	3,455	-	8,448	9,019	9,676	5,547	4,884	4,34

Table 5.18 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex, Mea Number of Hours Worked Per Week: 2012

INUI	liber of flour	3 WOINEU	CI WCCI	\. 201Z						
Region and Sex				N	umber of H	ours Worke	d Per Weel	<		
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	
COUNTRY TOTAL										
Total	5,726,116	275,568	9,353	322,625	429,717	503,568	697,095	1,266,142	825,717	8
Male	3,202,042	145,905	2,594	117,929	188,151	252,231	363,160	753,429	503,771	5
Female	2,524,074	129,663	6,759	204,697	241,566	251,337	333,936	512,712	321,946	3
TIGRAY REGION										
Total	391,891	23,377	626	18,384	20,566	29,016	35,570	78,802	71,589	
Male	204,483	13,895	-	4,963	5,945	15,646	16,279	41,009	41,061	
Female	187,408	9,482	626	13,421	14,621	13,369	19,290	37,793	30,528	
AFFAR REGION										
Total	77,068	5,497	144	2,773	7,365	6,041	8,621	13,538	15,048	
Male	43,657	3,975	-	643	2,678	3,056	3,866	9,213	10,289	
Female	33,411	1,522	144	2,130	4,687	2,984	4,755	4,325	4,759	
AMHARA REGION										
Total	1,063,958	68,967	2,565	92,287	117,980	100,399	112,604	225,776	133,245	1
Male	562,815	27,645	156	31,682	54,677	50,466	61,722	138,668	80,960	
Female	501,144	41,322	2,409	60,605	63,304	49,933	50,882	87,108	52,285	
OROMIA REGION										
Total	1,643,077	76,603	2,338	98,980	135,010	187,665	194,305	352,830	232,140	2
Male	963,543	44,989	1,871	46,108	60,265	95,172	103,400	216,760	147,837	1
Female	679,534	31,614	467	52,872	74,746	92,493	90,905	136,070	84,303	
SOMALI REGION										
Total	202,547	7,112	50	15,597	20,432	19,685	35,529	35,664	29,001	
Male	111,869	5,117	-	4,200	8,025	12,372	20,348	19,975	16,871	
Female	90,677	1,995	50	11,397	12,407	7,313	15,181	15,689	12,130	
BENISHANGUL-										
GUMUZ REGION	70.710			- 040			4= 000	0 = 0.4	- 400	
Total	72,518	3,753	293	7,612	8,206	9,590	15,260	9,504	7,129	
Male .	39,854	1,974	28	2,561	3,659	4,898	9,234	5,935	4,771	
Female	32,664	1,779	264	5,052	4,547	4,692	6,027	3,569	2,358	
SNNP REGION	007.000	00.456	0.070	57 400	70.407	70.077	00.00=	000.440	400.04=	
Total	837,386	38,456	2,079	57,436	76,197	78,977	83,827	208,112	102,217	1
Male .	466,330	19,337	495	14,680	31,408	35,866	43,686	130,930	64,858	
Female	371,057	19,119	1,584	42,756	44,789	43,111	40,141	77,183	37,359	

Table 5.18 Cont'd

Table 3.10 Cont a										
Region and Sex				Nu	mber of Ho	urs Worked	d Per Week			
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	_
GAMBELLA REGION										
Total	35,656	817	-	1,196	2,617	3,423	6,950	9,475	4,764	
Male	18,797	315	-	501	715	1,302	3,833	6,097	2,496	
Female	16,859	502	-	695	1,902	2,120	3,117	3,378	2,268	
HARARI REGION										
Total	49,969	3,313	43	1,458	4,029	3,360	7,995	12,803	6,964	
Male	26,901	1,402	43	648	1,686	1,687	4,180	7,419	4,281	
Female	23,068	1,912	-	810	2,343	1,673	3,815	5,384	2,684	
ADDIS ABABA CITY ADMINISTRATION										
Total	1,250,842	43,891	1,038	19,535	31,381	57,583	186,475	300,181	207,356	2
Male	709,206	25,169	-	9,235	15,782	27,731	91,307	166,285	121,888	
Female	541,635	18,722	1,038	10,300	15,599	29,853	95,168	133,896	85,468	
DIRE DAWA										
ADMINISTRATION										
Total	101,204	3,783	176	7,367	5,933	7,829	9,960	19,456	16,264	
Male	54,588	2,088	-	2,709	3,312	4,035	5,306	11,138	8,459	
Female	46,617	1,695	176	4,658	2,621	3,795	4,654	8,318	7,804	

Table 5.19 Currently Employed Population of Urban Areas Aged Ten Years and Above by Major Indust and Number of Hours Worked Per Week - COUNTRY TOTAL: 2012

and Numi	per of Hour	s Worked	Per V						
Major Industrial Divisions				Nι	ımber of I	Hours Wo	rked Per \	Neek	
and Sex									
and cox									
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57
Total Employed Population									
Total								1,266,142	
Male	3,202,042								503,771
Female	2,524,074		6,759	204,697	241,566	251,337	333,936	512,712	321,946
Agriculture, Hunting Fores									
Total	469,377	,	839	50,435	79,524	73,915	54,931	69,867	43,207
Male	315,988	25,715	395	28,733	48,090	50,540	40,520	52,721	33,505
Female	153,389	22,178	444	21,702	31,435	23,375	14,411	17,147	9,703
Mining & Quarrying									
Total	21,374	1,811	-	1,001	132	1,856	3,487	5,657	4,365
Male	16,486	696	-	1,001	103	1,856	2,795	3,521	3,537
Female	4,888	1,115	-	-	28	-	692	2,136	828
Manufacturing									
Total	757,055	40,502	2,682	46,617	59,084	67,308	79,312	161,165	127,157
Male	419,831	17,536	549	15,468	16,188	27,858	33,848	100,037	85,203
Female	337,224	22,966	2,133	31,149	42,896	39,450	45,465	61,127	41,954
Electricity, Gas and Water	Supply								
Total	53,226	1,891	-	4,961	2,897	1,738	5,791	18,088	9,287
Male	38,176	1,611	-	1,533	248	1,273	3,533	12,831	8,929
Female	15,050	280	-	3,428	2,649	466	2,258	5,257	358
Construction									
Total	395,129	23,345	1,023	10,664	19,472	36,483	43,943	113,207	68,470
Male	324,893	21,114	1,023	8,182	16,667	29,272	35,017	90,598	54,873
Female	70,235	2,231	´ -	2,482	2,804	7,211	8,926	22,609	13,597
Wholesale and Retail Trad	,	, -		, -	,	,	-,-	,	-,
Total	1,235,258	59,521	2.091	67,671	106.813	135,357	108.946	203.986	183,398
Male	625,873	30,739	60	22,656	40,467	50,989	44,683		105,955
Female	609,386	28,782		45,016	66,346	84,368	64,263	102,275	77,443
Hotels and Restaurants	000,000	_0,. 0_	_,00.	.0,0.0	00,010	0 1,000	0 1,200	. 02,2.	,
Total	457,810	19,531	645	24,918	29,668	36,373	38,510	58,791	63,605
Male	112,237	2,170	-	2,485	3,681	4,152	6,792	11,564	13,289
Female	345,572	17,361	645	22,433	25,987	32,220		47,227	50,316
Transport, Storage and Co			0.0	22, 100	20,007	32,220	51,710	,	50,010
Total	260,370	16,519	_	3,452	6,257	10,032	18,034	46,025	37,134
Male	246,962	16,086	_	2,986	6,257	9,404	15,631	42,617	35,195
Female	13,407	433	_	465	0,237	628	2,404	3,408	1,939
Torridio	10,707	733	_	700	_	020	۷,+04	5,700	1,555

Table 5.19 Cont'd

Major Industrial Divisions				Nu	mber of H	lours Wo	ked Per V	Veek	
and Sex									
und SCX								10.10	
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57
Financial Intermediation									
Total	115,655	4,259	-	1,055	1,592	3,192	28,578	54,813	11,827
Male	69,717	1,955	-	446	713	1,785	12,390	34,377	8,791
Female	45,938	2,304	-	610	879	1,407	16,188	20,436	3,035
Real Estate, Renting and E									
Total	119,259	5,049	-	3,568	3,271	5,422	23,094	43,428	15,203
Male	74,806	3,357	-	2,777	1,710	3,475	12,103	26,295	10,735
Female	44,453	1,692	-	790	1,561	1,947	10,991	17,133	4,467
Public administration and D									
Total	381,757	16,830	96	2,167	6,503	10,083	99,095	144,009	53,278
Male	251,433	10,899	96	1,273	2,748	5,712	55,219	91,685	40,845
Female	130,324	5,931	-	895	3,754	4,370	43,876	52,324	12,433
Education									
Total	404,111	8,564	-	2,101	35,473	51,686	87,128	139,451	50,383
Male	239,864	3,384	-	1,429	20,553	28,303	49,506	83,387	31,144
Female	164,247	5,180	-	673	14,920	23,384	37,622	56,064	19,239
Health and Social Work									
Total	182,330	8,914	-	2,670	3,537	5,156	26,203	72,727	36,810
Male	91,421	3,748	-	1,498	1,462	1,484	12,019	35,973	17,070
Female	90,909	5,166	-	1,172	2,075	3,672	14,184	36,754	19,740
Other Community, Social a	nd Persona			,	,	,	,	•	,
Service Activities									
Total	566,625	16,527	1,639	94,309	61,231	52,424	51,044	85,876	69,134
Male	330,050	4,948	471	26,904	26,532	35,160	32,301	57,969	49,021
Female	236,574	11,579	1.168	67,404	34,699	17,265	18,743	27,906	20,113
Private Households with Er			,	- , -	- ,	,	-, -	,	-, -
Total	269,616	1,024	338	5,617	13,145	12,212	20,120	38,218	47,455
Male	20,520	98	_	112	1,613	767	2,041	1,616	1,994
Female	249,095	925	338	5,505	11,532	11,445	18,079	36,602	45,461
Extra-Territorial Organization		Os		,	,	,	,	,	,
Total	34,003	1,809	_	1,421	939	330	8,592	10,834	3,930
Male	21,696	838	_	447	939	203	4,764	6,527	2,789
Female	12,307	970	_	974	-	128	3,828	4,307	1,141
Not Stated	-,,	5.5		. .		3	-,0-0	.,	-,
Total	3,162	1,578	_	_	179	_	287	_	1,075
Male	2,087	1,011	_	_	179	_		_	896
Female	1,076	567	_	_	-	_	287	_	179
1 0.110.0	.,5.0	007					201		



Table 5.20 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Occup
Mean and Number of Hours Worked Per Week - COUNTRY TOTAL: 2012

Maia a Casumatia a a l	i tallibel ol	ilouis V	oi keu				rked Per \	Neek		
Major Occupational	Total									Ī
Groups and Sex	Employed									
	Persons	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58
Total Employed Persons	1				II.		1			
Total	5,726,116	275,568	9,353	322,625	429,717	503,568	697,095	1,266,142	825,717	851
Male	3,202,042								503,771	
Female	2,524,074	129,663	6,759	204,697	241,566	251,337	333,936	512,712	321,946	305
Legislator Senior Officials										
& Managers										
Total	165,228	7,027	367	1,096	3,307	3,196	33,567	67,351		
Male	128,944	5,052	367	432	2,337	2,163	24,741	54,146	,	
Female	36,284	1,975	-	664	970	1,033	8,826	13,205	4,202	2
Professional										
Total	376,286	10,921	-	3,813	20,310	,	87,703	149,032		
Male	259,132	6,633	-	3,295	13,489	20,286	59,390	99,142	,	
Female	117,154	4,289	-	518	6,821	10,951	28,313	49,890	10,897	•
Technical and Associate										
Professionals										
Total	359,728	12,943	261	5,805	21,632	,	69,572	136,617	,	
Male	225,287	6,978	171	3,708	13,972	17,584	42,223	84,951	,	
Female	134,441	5,965	90	2,098	7,660	16,217	27,349	51,666	18,150	(
Clerks										
Total	316,379	10,731	356	6,034	5,464	6,175	80,245	127,818	,	
Male	123,190	3,491	-	1,914	1,229	,	22,958	49,899	,	
Female	193,190	7,241	356	4,120	4,235	3,530	57,286	77,918	20,948	10
Service, Shop & Market										
sales Workers										
Total	1,516,208	72,051			123,821		,		224,158	
Male	646,222	28,836	-	27,080	41,493	,	46,843	,	106,545	
Female	869,985	43,215	2,895	80,834	82,328	89,943	79,595	124,114	117,613	140
Skilled agricultural and										
Fishery .										_
Total	343,555	31,489	472	35,179	55,136	,	39,858	55,300		
Male	246,151	22,613	28	20,090	33,758	41,079	29,781	43,359	,	
Female	97,404	8,876	444	15,088	21,378	14,261	10,078	11,942	5,845	. 7

Table 5.20 Cont'd

Table 0120 Collica										
Major Occupational				Nι	ımber of I	Hours Wo	rked Per V	Veek		
Groups and Sex	Total									
Groups and Sex	Employed									
	Persons	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58
Crafts and related trades									·	
Total	1,048,027	61,628	3,787	63,439	90,225	108,655	117,235	229,496	162,017	158
Male	616,303	34,926	1,376	21,950	29,084	47,020	58,300	161,074	112,296	114
Female	431,723	26,702	2,411	41,489	61,141	61,635	58,935	68,422	49,721	43
Plant ,Machine Operators										
& Assemblers										
Total	305,437	18,507	-	2,513	5,630	11,751	19,676	59,506	54,503	74
Male	286,619	16,583	-	2,073	5,581	10,520	18,731	53,744	50,657	7′
Female	18,818	1,924	-	440	49	1,231	945	5,762	3,845	2
Elementary Occupation										
Total	1,293,289	48,692	1,215	96,832	104,014	118,270	122,802	225,736	191,390	226
Male	669,003	19,784	652	37,385	47,029	65,736	60,193	115,945	100,845	134
Female	624,286	28,909	564	59,447	56,985	52,534	62,609	109,792	90,546	92
Not Stated										
Total	1,979	1,578	-	-	179	-	-	-	179	
Male	1,190	1,011	-	-	179	-	-	-		
Female	788	567	-	-	-	-	-	-	179	

Table 5.21 Currently Employed Population Urban Areas Aged Ten Years and Above by Sex, Status in Employed Hours Worked Per Week - COUNTRY TOTAL: 2012

Hours Wor	ked Per We	ek - COUI	NTRY T							
0				N	lumber of I	Hours of wo	orked per w	/eek	,	_
Status in Employment	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	
Total Employed Population										
Total	5,726,116	275,568	9,353	322,625	429,717	503,568	697,095	1,266,142	825,717	8
Male	3,202,042	145,905	2,594	117,929	188,151	252,231	363,160	753,429	503,771	5
Female	2,524,074	129,663	6,759	204,697	241,566	251,337	333,936	512,712	321,946	3
Employee-Government										
Total	876,306	31,361	96	6,069	44,595	58,067	200,695	322,598	125,247	
Male	541,563	16,265	96	3,662	24,446	32,065	115,001	196,575	83,257	
Female	334,742	15,096	-	2,407	20,149	26,002	85,694	126,023	41,990	
Employee-Government Parasitatal										
Total	297,994	12,845	223	3,239	4,512	9,156	58,683	110,181	51,329	
Male	206,729	8,973	-	1,974	2,115	5,531	31,535	76,527	39,922	
Female	91,265	3,871	223	1,265	2,397	3,625	27,149	33,654	11,407	
Employee-Private organization										
Total	1,297,136	49,003	1,464	22,332	36,881	74,424	113,805	343,574	237,568	
Male	822,436	32,742	916	12,127	19,759	39,818	59,319	208,389	150,804	1
Female	474,700	16,260	549	10,205	17,123	34,606	54,486	135,184	86,764	
Employee-NGO (including Inte organization)	rnational									
Total	64,029	3,469	90	1,920	1,583	1,844	13,424	19,323	11,037	
Male	38,077	1,502	-	514	939	1,300	7,434	10,364	7,288	
Female	25,952	1,968	90	1,407	644	544	5,990	8,959	3,749	
Employee Domestic										
Total	267,461	1,024	338	5,197	11,985	12,212	19,796	37,968	47,455	
Male	19,940	98	-	112	1,033	767	2,041	1,616	1,994	
Female	247,520	925	338	5,084	10,952	11,445	17,755	36,352	45,461	
Other Employee										
Total	64,860	3,683	-	3,240	3,717	6,705	5,789	11,533	11,985	
Male	53,620	2,293	-	2,155	2,686	6,057	3,959	9,530	9,392	
Female	11,240	1,390	-	1,086	1,031	649	1,830	2,004	2,593	
Self Employed										
Total	2,215,946	139,042	5,234	147,552	211,284	262,181	221,904	353,978	290,988	3
Male	1,218,230	73,356	1,583	53,043	85,603	125,463	113,008	214,014	181,996	
Female	997,715	65,686	3,651	94,509	125,681	136,718	108,896	139,965	108,992	

Table 5.21 Cont'd

Table 5.21 Cont u										
				N	lumber of I	lours of w	orked per	week		
Status in Employment	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74
Unpaid Family Worker		·		·	·		·	·		
Total	493,930	25,456	1,101	121,674	103,991	65,817	48,339	39	797,	28,904
Male	206,822	4,825	-	41,359	45,396	33,102	22,868	3 16	6,694	12,162
Female	287,108	20,631	1,101	80,316	58,595	32,715	25,472	2 23	3,103	16,742
Employer										
Total	39,445	2,142	-	156	752	2,035	2,392	2 7	7,924	6,444
Male	29,577	1,588	-	103	691	815	1,853	3 5	5,849	5,713
Female	9,869	554	-	54	61	1,220	540) 2	2,074	730
Apprentice										
Total	10,872	230	-	904	860	844	1,039	9 3	3,840	2,156
Male	7,327	230	-	115	346	498	769	9 3	3,160	1,557
Female	3,544	-	-	789	514	346	270)	680	599
Member of Co-Operatives										
Total	50,871	4,026	808	2,878	1,832	4,470	7,696	5 9	9,401	5,665
Male	28,239	1,880	-	1,541	1,792	2,056	2,976	5 5	5,704	4,015
Female	22,632	2,146	808	1,337	40	2,414	4,720) 3	3,697	1,650
Others										
Total	47,268	3,288	-	7,463	7,725	5,812	3,531	6	5,025	6,940
Male	29,482	2,153	-	1,223	3,345	4,759	2,397	7 .	5,008	5,670
Female	17,786	1,135	-	6,240	4,380	1,053	1,134	ļ [*]	1,017	1,269

Type of Payment	Total	Paid Emplo	yees	Emplo	yee-Gover	nment		yee-Gover Parasitatal			oloyee-Priv rganization		(includ	nployee-f ding Inter rganizati	national	-
	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Т
COUNTRY TOTAL																
Total	2,867,785	1,682,365	1,185,420	876,306	541,563	334,742	297,994	206,729	91,265	1,297,136	822,436	474,700	64,029	38,077	25,952	267,
In cash	2,766,231	1,633,779	1,132,452	861,498	533,211	328,287	289,906	201,275	88,632	1,250,677	794,722	455,955	62,407	37,950	24,457	241,
In kind	21,647	8,571	13,076	3,210	1,010	2,199	918	824	94	8,159	3,202	4,956	91	51	39	6,
Both	75,979	37,185	38,794	11,543	7,310	4,233	6,670	4,131	2,539	36,307	22,558	13,750	1,115	75	1,040	18,
Not stated	3,928	2,830	1,098	54	31	23	499	499	-	1,993	1,953	40	416	-	416	
TIGRAY REGION																
Total	189,677	107,939	81,738	64,502	39,498	25,003	17,399	10,737	6,661	86,620	50,457	36,163	5,385	3,407	1,978	11,
In cash	182,024	106,109	75,915	64,069	39,242	24,827	17,399	10,737	6,661	82,246	49,138	33,108	4,960	3,407	1,553	10,
In kind	2,686	662	2,024	-	-	-	-	-	-	2,431	407	2,024	-	-	-	
Both	4,967	1,168	3,799	433	257	176	-	-	-	1,943	912	1,031	425	-	425	1,
Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
AFFAR REGION																
Total	38,124	25,538	12,586	8,725	5,868	2,857	7,851	5,847	2,004	17,641	12,679	4,961	428	260	168	2,
In cash	37,871	25,285	12,586	8,582	5,724	2,857	7,778	5,774	2,004	17,616	12,655	4,961	416	248	168	2,
In kind	61	61	-	-	-	-	36	36	-	13	13	-	12	12	-	
Both	181	181	-	144	144	-	37	37	-	-	-	-	-	-	-	
Not stated	12	12	-	-	-	-	-	-	-	12	12	-	-	-	-	
AMHARA REGION																
Total	435,245	257,734	177,511	167,769	105,571	62,198	39,303	28,759	10,544	172,824	105,986	66,838	6,331	4,018	2,314	36,
In cash	419,259	248,532	170,727	163,485	102,605	60,880	37,454	28,022	9,432	165,701	102,299	63,402	6,331	4,018	2,314	35,
In kind	2,700	1,741	959	89	-	89	89	89	-	1,296	703	594	-	-	-	
Both	13,286	7,461	5,825	4,195	2,966	1,229	1,760	648	1,112	5,826	2,984	2,842	-	-	-	
Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
OROMIA REGION																
Total	762,056	480,200	281,856	233,690	151,891	81,799	70,467	52,229	18,238	371,357	245,409	125,948	17,443	9,534	7,909	48,
In cash	733,919	464,940	268,979	229,721	150,041	79,680	67,785	49,806	17,978	358,391	236,011	122,381	16,679	9,494	7,184	41,
In kind	6,018	2,492	3,526	59	-	59	373	279	94	2,885	954	1,932	79	39	39	1,
Both	19,895	10,960	8,935	3,909	1,849	2,060	1,810	1,644	166	8,771	7,136	1,636	269	-	269	
Not stated	2,224	1,808	416	-	-	-	499	499	-	1,309	1,309	-	416	-	416	
SOMALI REGION																
Total	64,880	44,711	20,169	30,133	20,625	9,509	12,296	8,145	4,151	15,507	13,615	1,892	1,904	1,316	588	3,
In cash	61,841	43,558	18,283	28,058	20,436	7,622	12,246	8,095	4,151	14,907	13,015	1,892	1,904	1,316	588	
In kind	1.975	88	1,887	1,975	88	1,887	-	- ,	-1			-	-	-		3,
Both	1,065	1,065		101	101	- 1001	50	50	_	599	599	_				
Not stated	.,000	.,000		.51	.51		30	30		577	377					

Table 5.22 Cont'd

Type of Payment	Total Paid Employees			Employee-Government			Employee-Government Parasitatal			Employee-Private organization			Employee-NGO (including International organization)			E
	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	To
BENISHANGUL GUM	IUZ REGION															
Total	24,970	16,498	8,473	12,448	8,584	3,864	3,420	2,729	691	6,547	4,342	2,205	688	422	266	1,4
In cash	24,443	16,114	8,329	12,289	8,448	3,841	3,420	2,729	691	6,261	4,095	2,165	688	422	266	1,3
In kind	87	87	-	54	54	-	-	-	-	33	33	-	-	-	-	
Both	303	222	81	52	52	-	-	-	-	171	171	-	-	-	-	
Not stated	137	74	63	54	31	23	-	-	-	82	43	40	-	-	-	
SNNP REGION																
Total	381,966	230,899	151,067	169,204	107,633	61,571	34,090	23,474	10,615	133,061	87,666	45,395	7,919	4,817	3,102	32,5
In cash	360,777	221,475	139,302	168,489	107,046	61,444	33,317	22,835	10,482	120,858	80,879	39,979	7,844	4,742	3,102	25,4
In kind	5,138	1,948	3,190	613	486	128	421	421	-	645	330	315	-	-	-	3,0
Both	15,771	7,195	8,575	101	101	-	352	218	133	11,278	6,177	5,101	75	75	-	3,9
Not stated	280	280	-	-	-	-	-	-	-	280	280	-	-	-	-	
GAMBELLA REGION																
Total	16,728	10,877	5,851	10,631	7,100	3,531	828	550	277	3,862	2,454	1,407	498	440	58	4
In cash	16,491	10,799	5,692	10,526	7,063	3,463	828	550	277	3,793	2,413	1,380	498	440	58	4
In kind	152	78	74	74	37	37	-	-	-	41	41	-	-	-	-	
Both	85	-	85	31	-	31	-	-	-	27	-	27	-	-	-	
Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
HARARI REGION																
Total	26,144	15,197	10,947	10,828	5,950	4,878	1,805	1,406	399	11,060	7,188	3,873	634	449	186	1,6
In cash	25,398	14,936	10,461	10,632	5,901	4,731	1,805	1,406	399	10,698	6,976	3,722	634	449	186	1,4
In kind	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
Both	747	261	486	196	49	147	-	-	-	362	211	151	-	-	-	1
Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
ADDIS ABABA CITY	ADMINISTRA	TION														
Total	878,145	463,149	414,996	156,428	80,817	75,611	107,681	70,542	37,139	451,525	275,556	175,969	21,298	12,361	8,937	123,0
In cash	855,264	452,802	402,461	154,339	79,073	75,265	105,021	69,010	36,011	443,165	270,157	173,008	20,952	12,361	8,591	113,9
In kind	2,651	1,414	1,238	346	346	-	-	-	-	722	722	-	-	-	-	1,5
Both	18,956	8,277	10,678	1,744	1,398	346	2,660	1,532	1,127	7,329	4,368	2,961	346	-	346	6,5
Not stated	1,275	656	619	-	-	-	-	-	-	310	310	-		-	-	9
DIRE DAWA ADMINIS	STRATION															
Total	49,847	29,622	20,225	11,946	8,026	3,920	2,855	2,310	545	27,132	17,084	10,048	1,501	1,054	447	5,5
In cash	48,945	29,229	19,716	11,309	7,632	3,677	2,855	2,310	545	27,041	17,084	9,957	1,501	1,054	447	5,3
In kind	178		178	-	- 100-	-	-	-	-	92	-	92	-	-	-	- 1-
Both	724	394	330	637	394	243	_	_	_		_	-			_	
Not stated	727	-	-	-	-	240	_	_	_	_	_	_	_	_	_	

Table 5.23 Paid Employees of Urban Population Aged Ten Years and Above by Sex and Period of Payment - COUNTRY TOTAL: 2012

			Pa	aid Employees	 }		
Period of Payment and Sex	Total Paid Employees	Employee- Government	Employee- Government Parasitatal	Employee- Private organization	Employee- NGO (including International organization)	Employee Domestic	Other Paid Employees
Total Paid Emp	oloyees				,		<u> </u>
Total	2,867,785	876,306	297,994	1,297,136	64,029	267,461	64,860
Male	1,682,365	541,563	206,729	822,436	38,077	19,940	53,620
Female	1,185,420	334,742	91,265	474,700	25,952	247,520	11,240
Hourly							
Total	20,972	2,357	1,570	16,075	-	356	614
Male	14,847	1,969	89	12,229	-	269	290
Female Daily	6,125	388	1,480	3,846	-	87	324
Total	124,396	1,946	2,015	108,770	317	2,270	9,077
Male	95,866	1,581	1,458	84,465	227	48	8,087
Female Weekly	28,530	365	558	24,305	90	2,222	990
Total	119,978	709	2,132	110,928	88	1,539	4,582
Male	91,241	415	2,015	84,200	88	346	4,176
Female Half Month	28,737	294	117	26,728	-	1,193	406
Total	89,676	3,081	6,536	75,256	1,073	885	2,844
Male	55,420	1,919	4,073	46,932	451	210	1,835
Female Monthly	34,255	1,161	2,463	28,323	623	675	1,009
Total	2,459,248	866,113	283,243	952,899	61,508	252,993	42,493
Male	1,395,219	534,886	196,901	573,800	36,907	18,009	34,716
Female Yearly	1,064,029	331,226	86,342	379,099	24,601	234,984	7,777
Total	12,056	1,254	_	1,901	403	5,475	3,022
Male	5,656	643	_	1,219	403	605	2,785
Female	6,400	611	-	682	-	4,870	237
Others							
Total	37,495	791	1,998	29,314	223	3,287	1,882
Male	20,940	118	1,694	17,636	-	107	1,385
Female Not Stated	16,555	673	305	11,678	223	3,179	497
Total	3,964	54	499	1,993	416	656	346
Male Female	3,176 788	31 23	499	1,953 40	- 416	346 310	346

Table 5.24 Paid Employees of Urban Population Aged Ten Years and Over by Sex, Period of Payment and Amount of Total Payment - COUNTRY TOTAL: 2012

Daried of	of Paymen	it and r			•			2012	
Period of Payment and		l -	Amou	int of Total	Payment P	er Month (li	n pilt)		Not
Sex	Total	<50	50-99	100-199	200-399	400-699	700-999	1000+	Stated
Total Paid Em	ployees								
Total	2,867,785	14,939	29,218	108,926	388,948	548,049	385,789	1,368,088	23,827
Male	1,682,365	1,658	5,576	24,032	118,389	283,880	250,551	984,180	14,099
Female	1,185,420	13,280	23,642	84,894	270,559	264,169	135,239	383,908	9,728
Hourly									
Total	20,972	1,043	173	1,290	2,901	3,777	2,642	8,496	650
Male	14,847	490	-	385	1,715	2,760	2,541	6,652	304
Female	6,125	553	173	904	1,186	1,017	101	1,844	346
Daily									
Total	124,396	2,000	3,322	7,271	15,658	27,827	18,203	48,694	1,419
Male	95,866	59	1,159	3,639	8,907	20,838	16,400	43,890	973
Female	28,530	1,941	2,163	3,632	6,752	6,988	1,803	4,804	447
Weekly									
Total	119,978	1,125	1,217	6,682	14,542	32,214	25,680	38,237	281
Male	91,241	209	642	3,887	7,491	20,783	21,180	36,769	281
Female	28,737	916	575	2,795	7,051	11,431	4,500	1,468	-
Half Month									
Total	89,676	552	1,187	1,043	11,084	33,463	16,824	25,319	204
Male	55,420	346	-	316	3,131	13,650	14,082	23,815	80
Female	34,255	206	1,187	727	7,953	19,813	2,742	1,504	124
Monthly									
Total	2,459,248	9,379	22,089	88,594	335,315	436,670	316,180	1,234,931	16,090
Male	1,395,219	554	3,439	14,804	92,971	220,076	191,074	863,133	9,167
Female	1,064,029	8,825	18,651	73,790	242,344	216,594	125,106	371,798	6,923
Yearly									
Total	12,056	633	802	1,403	2,062	2,283	654	4,219	-
Male	5,656	-	336	813	1,058	403	654	2,392	-
Female	6,400	633	465	590	1,004	1,880	-	1,827	-
Other									
Total	37,495	206	427	2,643	7,386	11,784	5,606	7,866	1,577
Male	20,940	-	-	187	3,117	5,338	4,619	7,225	453
Female	16,555	206	427	2,456	4,270	6,446	987	640	1,124
Not Stated									
Total	3,964	-	-	-	-	31	-	326	3,607
Male	3,176	-	-	-	-	31	-	303	2,841
Female	788	_				-	-	23	765

Table 5.25 Paid Employees of Urban Population Aged Ten Years and Above by Major Industrial Divisions, Sex and Amount of Payment Per Month - COUNTRY TOTAL: 2012

Amount of Total Payment Per Month (In Birr)

Major Industrial			Amour	nt of Total	Payment	Per Month	(In Birr)			Mean
Divisions and										Amount
Sex									Not	of Total
•••	Total	<50	50-99	100-199	200-399	400-699	700-999	1000+	stated	Payment
Total Employed F	•									
Total	2,867,785	14,939	29,218	108,926	388,948	548,049	385,789	1,368,088	23,827	1,255.9
Male	1,682,365	1,658	5,576	24,032	118,389	283,880	250,551	984,180	14,099	1,519.3
Female	1,185,420	13,280	23,642	84,894	270,559	264,169	135,239	383,908	9,728	882.1
Agriculture, Hunti	ng Forestry	& Fishin	g							
Total	68,394	731	1,241	4,290	14,366	22,127	8,502	15,914	1,224	821.3
Male	46,812	-	368	2,849	8,100	14,516	5,919	13,836	1,224	962.5
Female	21,583	731	873	1,440	6,267	7,611	2,583	2,078	-	522.9
Mining &										
Quarrying										
Total	9,225	-	-	-	1,057	1,831	344	5,993	-	1,787.2
Male	6,031	-	-	-	641	740	344	4,307	-	2,198.5
Female	3,194	-	-	-	416	1,091	-	1,686	-	1,010.5
Manufacturing										
Total	323,139	3,300	2,425	8,309	36,409	90,882	65,273	111,464	5,076	1,017.6
Male	211,812	645	783	3,840	17,891	53,638	47,312	84,894	2,810	1,105.6
Female	111,327	2,656	1,642	4,469	18,519	37,245	17,962	26,570	2,266	848.9
Electricity, Gas a	nd Water Su	ıpply								
Total	43,245	127	-	-	2,407	7,670	9,091	23,440	511	1,402.8
Male	33,986	-	-	-	1,472	7,104	7,013	17,886	511	1,456.5
Female	9,259	127	-	-	935	566	2,077	5,554	-	1,208.5
Construction										
Total	256,948	301	593	3,261	20,479	52,070	41,496	136,952	1,797	1,349.1
Male	202,994	59	320	2,020	12,043	32,101	33,579	121,075	1,797	1,477.8
Female	53,954	241	273	1,241	8,436	19,970	7,917	15,877	· -	869.3
Whole Sale and F	Retail Trade			•	,			,		
Total	190,212	2,185	2,917	9,743	30,771	56,446	34,822	51,499	1,831	874.1
Male	115,333	· -	479	5,367	12,962	28,881	24,666	41,915	1,064	1,032.2
Female	74,879	2,185	2,438	4,376	17,809	27,565	10,156	9,584	767	630.2
Hotels and	1,010	_,	_,	.,	,	_:,	,	-,		
Restaurants										
Total	167,346	565	1,103	9,588	61,176	50,132	20,150	23,188	1,444	741.8
Male	64,485	-	432	2,101	17,566	20,815	8,604	14,226	740	1,122.5
Female	102,861	565	671	7,487	43,610	29,317	11,546	8,962	703	504.2
Transport, Storag	je and Comi	municatio	ons							
Total	179,582	376	100	839	8,902	28,189	30,213	109,788	1,176	1,513.9
Male	167,601	376	100	839	8,327			102,954	1,176	1,486.2
Female	11,981	-	-	_	575	3,053		6,834	· -	1,898.0
Financial	,					•	,	,		•
Intermediation										
Total	114,654	-	-	825	1,653	9,458	11,410	90,804	503	2,030.3
Male	69,453	-	-	479	418	4,716	6,216	57,467	157	2,170.8
Female	45,201	-	-	346	1,235	4,743	5,194	33,337	346	1,813.1
Real Estate, Ren		siness Ad	ctivities		•	•	•	•		
Total	76,377	-	-	-	5,788	8,112	9,979	52,097	400	1,802.0
Male	40,051	-	-	-	1,276	3,931	4,613	30,177	54	2,149.9
Female	36,326	-	-	-	4,513	4,182	5,366	21,920	346	1,415.4
	-,-				,	,	,	,		, = -

Tab	le 5.25	Cont'd

Table 5.25	Cont a												7
Major		1	Am	nount of	Total P	aymer	nt Per	Month	(In Birr)		,		Mean
Industrial													Amount
Divisions						200-		400-	700-			lot	of Total
and Sex	Total	<50	50-99	9	199	399		699	999	1000+	sta	ited	Payment
PUBLIC AD	_	_		_									
Total		81,697	374	689	2,03		8,682	47,8	,			1,341	,
Male	2	51,433	-	640	26		4,822	22,3	40 33,0	62 189,	144	864	,
Female	1	30,263	374	50	1,77	'1 ;	3,860	25,5	55 18,9	10 79,2	266	477	1,293.4
Education													
Total	4	00,605	-	380	63	6 14	4,196	51,0	33 42,2	94 290,	193	1,872	1,655.1
Male	2	37,406	-	380	35	4	4,817	19,2	00 18,8	35 192,	724	1,097	1,932.1
Female	1	63,199	-	-	28	3 9	9,379	31,8	33 23,4	59 97,4	169	776	1,252.1
Health and													
Social Work	(
Total	1	64,914	-	-	1,03	34	4,883	26,9	45 24,0	28 106,6	697	1,327	1,584.4
Male		79,508	-	-	61	8	1,763	11,19	96 11,6	86 53,7	753	491	1,850.0
Female		85,406	-	-	41	5 3	3,120	15,7	48 12,3	42 52,9	944	836	1,336.3
Other Comr	nunity, So	cial and P	ersona	I Service	Activit	ies							
Total	1	88,710	2,774	3,275	10,13	9 34	4,546	55,5	71 27,9	82 52,5	503	1,919	863.7
Male	1	12,927	400	696	2,29	1 10	6,368	32,7	22 17,1	55 41,7	742	1,553	1,015.2
Female		75,783	2,374	2,579	7,84	9 18	8,178	22,8	49 10,8	28 10,7	761	366	640.1
Private Hou	seholds w	ith Emplo	yed Pei	rsons									
Total	2	67,461	4,121	16,309	57,48	8 14	1,954	34,8	5,0	70 4,2	253	3,406	267.4
Male		19,940	92	1,379	3,01	3 8	8,880	4,4			394	561	
Female		•	4,029	14,929	54,47		3,075	30,3			359	2,845	
Extra-Territo		•			- ,		,	/-	2,2	-,-		,	
Total		33,913	87	187	74	3	1,389	3,9	32 2,9	82 24,5	594		- 2,574.0
Male		21,696	87	-			1,043	1,4			386		3,033.8
Female		12,217	-	187	74		346	2,4			208		1,757.4
Not Stated		,— · ·		1	•	-		_,.	,-				-,
Total		1,363	-	-		-	287	8	96 1	79	-		- 566.2
Male		896	_	_		_	_	-	96	-	_		600.0
Female		466	_	_		_	287			79	_		501.2
									•				

Table 5.26 Total Employed Population Not at Work Aged Ten Years and Above by Region, Sex, and Reason for Absence: 2012

Region, Sex, and Reason for Absence: 2012											
				Reason for	or Absen	ce					
Region and Sex		Sick	Holiday/	Training/	Off	No work		Not			
	Total		Vacation	Education			Others	Stated			
COUNTRY TOTAL	i otai	ii ijai y	vacation	Ladodion	Codoon	tomporamy	Outoio	Otatoa			
Total	284,922	76 465	43,432	8,015	40,584	51,155	63,065	2,207			
Male	148,500	38,120	22,913	5,030	22,736	33,809		937			
Female	136,422		20,519	2,986	17,847	17,346	38,110	1,270			
TIGRAY REGION	100,422	50,545	20,010	2,500	17,047	17,040	30,110	1,270			
Total	24,003	4,915	4,251	1,385	2,235	5,036	5,901	281			
Male	13,895	2,735	2,465	750	1,957	3,513		281			
Female	10,108	2,180	1,786	635	277	1,524					
AFFAR REGION	10,100	2,100	1,700	000	211	1,021	0,707				
Total	5,641	1,731	1,838	168	323	1,410	170	_			
Male	3,975	1,111	1,541	156	323	747		_			
Female	1,666	620	297	12	-	664	73	_			
AMHARA REGION	.,000	0_0									
Total	71,532	17,658	8,780	1,600	19,840	11,080	12,416	158			
Male	27,801	8,338	4,279	1,166	6,202	4,655		79			
Female	43,731	9,320	4,501	434	13,637	6,425		79			
OROMIA REGION	.0,.0.	0,020	.,		. 0,00.	0,0	0,000				
Total	78,941	23,624	10,450	1,319	10,233	10,685	22,383	247			
Male	46,861	11,656	5,858	1,201	7,392	9,017		178			
Female	32,080	11,968	4,592	118	2,841	1,668	10,825	69			
SOMALI REGION	,	,	.,		_,	,,,,,,	,				
Total	7,162	2,008	243	_	1,797	1,487	1,627	_			
Male	5,117	1,228	140	_	1,736	1,325	688	_			
Female	2,045	780	102	_	61	162		_			
BENISHANGUL - GUMUZ											
Total	4,04	6 1,43°	1 41	8 54	653	485	1,006	_			
Male	2,00				235	213	412	-			
Female	2,04		9 17	9 -	418	272	594	-			
SNNP REGION											
Total	40,53	5 12,239	9 3,53	9 1,304	4,006	11,691	7,755	-			
Male	19,83	2 4,960	3 1,52	3 215	3,430	7,856	1,843	-			
Female	20,70	3 7,27	3 2,01	6 1,090	576	3,836	5,912	-			
GAMBELLA REGION											
Total	81	7 392	2 11	0 -	148	41	126	-			
Male	31	5 134	4		111	-	70	-			
Female	50	2 25	3 11	0 -	37	41	56	-			
HARARI REGION											
Total	3,35	7 982	2 88	5 92	49	483	865	-			
Male	1,44	5 190	3 43	9 92	49	336	332	-			
Female	1,91	2 780	3 44	6 -	-	147	533	-			
ADDIS ABABA CITY ADM	MINISTRAT	ION									
Total	44,92	9 10,73	7 11,65	8 2,093	1,038	7,964	9,917	1,522			
Male	25,16	9 6,73	1 5,75		1,038	5,530	4,322	400			
Female	19,76	0 4,000	5,90	5 697	-	2,434	5,595	1,122			
DIRE DAWA ADMINISTR	ATION										
Total	3,96				263	791	899	-			
Male	2,08				263	617	358	-			
Female	1,87	2 57:	3 58	4 -	-	173	541				

Table 5.27 Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex, Whether Available and Ready to Work Additional Hours and Percent Available and Ready to Work Additional Hours- Country Total: 2012

	Whether Availab	le and Ready to Work	Additional Hours	Percent
Age Group		-	Not Seeking or	Available and
		Seeking or Available	Available to	Ready to work
and Sex	Total Employed	to Work Additional	Work Additional	Additional Hours
	Persons	Hours	Hours	
All Ages				
Total	5,726,116	2,338,235	3,387,881	59.2
Male	3,202,042	1,366,671	1,835,371	57.3
Female	2,524,074	971,564	1,552,510	61.5
10-14	, ,	•	, ,	
Total	195,567	26,769	168,798	86.3
Male	94,187	14,804	79,383	84.3
Female	101,380	11,965	89,415	88.2
15-19				
Total	492,071	168,851	323,220	65.7
Male	205,515	78,707	126,808	61.7
Female	286,556	90,144	196,412	68.5
20-24				
Total	902,304	415,775	486,529	53.9
Male	457,454	227,097	230,357	50.4
Female	444,850	188,678	256,171	57.6
25-29 T-1-1	4 400 407	500.440	005.040	55.4
Total	1,128,487	503,146	625,340	55.4
Male	653,525	298,028	355,496	54.4
Female 30-34	474,962	205,118	269,844	56.8
Total	767.262	220 242	429.040	55.9
Male	767,262 448,530	338,313 202,122	428,949 246,409	54.9
Female	318,731	136,191	182,540	57.3
35-39	310,731	130,191	162,540	57.3
Total	742,001	330,359	411,643	55.5
Male	423,955	192,948	231,007	54.5
Female	318,046	137,411	180,635	56.8
40-44	0.0,0.0	,	.00,000	00.0
Total	445,655	181,389	264,266	59.3
Male	273,476	110,582	162,895	59.6
Female	172,179	70,807	101,372	58.9
45-49				
Total	361,687	143,046	218,640	60.5
Male	211,592	87,543	124,049	58.6
Female	150,094	55,504	94,591	63.0
50-54				
Total	266,025	98,134	167,891	63.1
Male	153,598	59,975	93,622	61.0
Female	112,428	38,159	74,269	66.1
55-59				
Total	152,584	53,183	99,401	65.1
Male	96,469	34,433	62,035	64.3
Female	56,115	18,750	37,366	66.6
60-64	440.645	20.000	75.040	66.0
Total	113,615	38,602	75,013	66.0
Male	75,221	27,033	48,188	64.1 69.9
Female 65+	38,393	11,569	26,825	09.9
Total	158,859	40,668	118,191	74.4
Male	108,519	33,399	75,120	69.2
Female	50,340	7,270	43,070	85.6
1 CITIAIC	50,540	1,210	43,070	03.0

Table 5.27 Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex, Whether Available and Ready to Work Additional Hours and Percent Available and Ready to Work Additional Hours- Country Total: 2012

	Whether Availab	le and Ready to Work	Additional Hours	Percent
Age Group		-	Not Seeking or	Available and
		Seeking or Available	Available to	Ready to work
and Sex	Total Employed	to Work Additional	Work Additional	Additional Hours
	Persons	Hours	Hours	
All Ages				
Total	5,726,116	2,338,235	3,387,881	59.2
Male	3,202,042	1,366,671	1,835,371	57.3
Female	2,524,074	971,564	1,552,510	61.5
10-14	, ,	•	, ,	
Total	195,567	26,769	168,798	86.3
Male	94,187	14,804	79,383	84.3
Female	101,380	11,965	89,415	88.2
15-19				
Total	492,071	168,851	323,220	65.7
Male	205,515	78,707	126,808	61.7
Female	286,556	90,144	196,412	68.5
20-24				
Total	902,304	415,775	486,529	53.9
Male	457,454	227,097	230,357	50.4
Female	444,850	188,678	256,171	57.6
25-29 T-1-1	4 400 407	500.440	005.040	55.4
Total	1,128,487	503,146	625,340	55.4
Male	653,525	298,028	355,496	54.4
Female 30-34	474,962	205,118	269,844	56.8
Total	767.262	220 242	429.040	55.9
Male	767,262 448,530	338,313 202,122	428,949 246,409	54.9
Female	318,731	136,191	182,540	57.3
35-39	310,731	130,191	162,540	57.3
Total	742,001	330,359	411,643	55.5
Male	423,955	192,948	231,007	54.5
Female	318,046	137,411	180,635	56.8
40-44	0.0,0.0	,	.00,000	00.0
Total	445,655	181,389	264,266	59.3
Male	273,476	110,582	162,895	59.6
Female	172,179	70,807	101,372	58.9
45-49				
Total	361,687	143,046	218,640	60.5
Male	211,592	87,543	124,049	58.6
Female	150,094	55,504	94,591	63.0
50-54				
Total	266,025	98,134	167,891	63.1
Male	153,598	59,975	93,622	61.0
Female	112,428	38,159	74,269	66.1
55-59				
Total	152,584	53,183	99,401	65.1
Male	96,469	34,433	62,035	64.3
Female	56,115	18,750	37,366	66.6
60-64	440.645	20.000	75.040	66.0
Total	113,615	38,602	75,013	66.0
Male	75,221	27,033	48,188	64.1 69.9
Female 65+	38,393	11,569	26,825	09.9
Total	158,859	40,668	118,191	74.4
Male	108,519	33,399	75,120	69.2
Female	50,340	7,270	43,070	85.6
1 CITIAIC	50,540	1,210	43,070	03.0

Table 5.28 Currently Employed Population of Urban Areas Aged Ten Years and Above Who Sought Additional Hours and Number of Hours Worked Per Week by Age Group and Sex - Cour

	itional nou				nber of Hou					
Age Group										Ī
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	
All Ages				1		,		,	<u>, </u>	-
Total	2,338,235	110,589	4,414	138,815	200,457	237,326	274,522	524,506	341,461	3
Male	1,366,671	63,558	1,743	59,552	96,283	130,215	145,207	323,310	225,019	
Female	971,564	47,031	2,671	79,263	104,174	107,111	129,315	201,196	116,441	1
10-14	•	•	,	, -	,	,	,	,	•	
Total	26,769	425	-	6,266	3,327	5,111	3,350	1,247	2,993	
Male	14,804	425	-	3,579	1,751	3,117	1,593	1,050	1,428	
Female	11,965	-	-	2,687	1,576	1,995	1,756	196	1,565	
15-19	•			•			•		•	
Total	168,851	6,085	190	12,769	16,181	17,748	13,150	29,433	26,574	2
Male	78,707	2,572	-	5,324	8,260	8,140	5,704	16,231	10,183	•
Female	90,144	3,513	190	7,445	7,921	9,608	7,446	13,202	16,392	•
20-24										
Total	415,775	14,813	724	23,334	30,489	44,483	53,709	94,828	56,954	6
Male	227,097	7,448	-	9,003	15,426	23,863	25,789	51,156	34,298	4
Female	188,678	7,365	724	14,331	15,064	20,620	27,920	43,672	22,656	2
25-29										
Total	503,146	23,691	1,730	26,258	36,747	53,120	55,327	129,182	73,721	6
Male	298,028	12,636	948	11,260	17,616	31,421	30,420	80,113	49,966	4
Female	205,118	11,055	782	14,997	19,130	21,699	24,907	49,069	23,755	2
30-34										
Total	338,313	17,693	457	19,046	26,890	34,229	42,509	76,184	54,865	4
Male	202,122	11,415	-	6,866	10,324	18,053	23,789	48,123	39,341	3
Female	136,191	6,278	457	12,179	16,566	16,177	18,720	28,061	15,523	•
35-39										
Total	330,359	14,868	517	18,487	29,954	27,153	36,248	78,240	47,764	4
Male	192,948	7,623	-	8,221	11,586	15,131	18,153	48,383	32,294	3
Female	137,411	7,245	517	10,266	18,369	12,022	18,095	29,858	15,469	•
10-44										
Total	181,389	9,053	-	12,193	16,827	15,271	21,653	44,855	25,398	2
Male	110,582	5,060	-	6,484	9,018	7,779	10,234	28,830	17,987	•
Female	70,807	3,993	-	5,708	7,809	7,492	11,419	16,025	7,411	

Table 5.28 Cont'd

A O				Num	ber of Hour	s Worked	Per Week		
Age Group and Sex									
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57
45-49	1		l.						
Total	143,046	5,924	-	7,616	13,747	14,201	20,717	26,994	24,369
Male	87,543	2,951	-	3,841	6,530	6,999	13,229	16,523	18,983
Female	55,504	2,973	-	3,775	7,216	7,202	7,488	10,470	5,385
50-54									
Total	98,134	7,255	367	4,049	11,741	11,845	12,600	20,683	10,331
Male	59,975	3,850	367	459	7,701	5,929	6,862	14,196	7,032
Female	38,159	3,405	-	3,590	4,040	5,916	5,737	6,487	3,300
55-59									
Total	53,183	2,968	-	2,849	4,830	6,633	8,263	11,191	5,731
Male	34,433	2,011	-	1,835	1,654	4,147	4,626	8,520	4,263
Female	18,750	957	-	1,015	3,176	2,486	3,637	2,671	1,468
60-64									
Total	38,602	2,469	428	3,222	4,818	2,648	3,442	5,489	6,553
Male	27,033	2,322	428	1,156	2,950	1,767	2,008	4,857	4,532
Female	11,569	147	-	2,066	1,868	881	1,434	632	2,021
65+									
Total	40,668	5,345	-	2,727	4,907	4,884	3,555	6,180	6,209
Male	33,399	5,245	-	1,523	3,468	3,869	2,801	5,327	4,712
Female	7,270	100	-	1,204	1,438	1,015	754	853	1,497

Currently Employed Population of Urban Areas Aged Ten Years and Above Who were Willing and Available to Work Additional Hours and Type of Work Sought by Region and Sex: 2012 **Table 5.29**

	gion and Sex.		ork Sought or Av	ailable for	
		l ypo or w	l circ ocagint of 7th	Other job with	
Region and Sex			Other job in	more hours to	
3		At the present	addition to the	replace the	
	Total	main job	present	present job	Not Stated
COUNTRY TOTAL	L).
Total	2,338,235	701,121	1,111,159	524,323	1,632
Male	1,366,671	443,822	643,326	278,268	1,254
Female	971,564	257,299	467,832	246,055	378
TIGRAY REGION	,		,	_ :=,===	
Total	191,629	70,990	93,392	27,248	_
Male	95,777	40,515	44,910	10,351	_
Female	95,853	30,475	48,481	16,897	_
AFFAR REGION	00,000	00,110	10,101	10,001	
Total	41,130	10,549	20,088	10,493	_
Male	23,430	6,041	10,805	6,584	_
Female	17,700	4,508	9,283	3,909	_
AMHARA REGION	17,700	1,000	0,200	0,000	
Total	427,523	165,880	160,584	101,059	_
Male	244,888	99,232	92,316	53,340	_
Female	182,634	66,647	68,268	47,719	_
OROMIA REGION	102,004	00,047	00,200	47,713	
Total	793,115	187,537	395,035	210,044	499
Male	481,801	125,260	237,209	118,833	499
Female	311,314	62,277	157,826	91,211	
SOMALI REGION	311,314	02,211	107,020	31,211	_
Total	72,980	22,446	38,844	11,690	_
Male	46,922	13,877	25,834	7,210	_
Female	26,058	8,569	13,010	4,480	_
BENISHANGUL GUMUZ		0,509	13,010	4,400	_
Total	27,104	8,428	12,905	5,771	_
Male	14,782	4,748	7,340	2,694	_
Female	12,322	3,680	5,565	3,077	
SNNP REGION	12,322	3,000	3,303	3,077	_
Total	319,331	94,092	166,349	58,890	_
Male	188,733	65,937	94,555	28,241	_
Female	130,598	28,154	71,794	30,650	
GAMBELLA REGION	130,330	20,134	71,734	30,030	_
Total	9,932	3,138	6,077	717	_
Male	5,762	2,094	3,325	342	_
Female	4,170	1,043	2,752	375	_
HARARI REGION	4,170	1,040	2,732	070	
Total	25,370	7,573	13,495	4,302	_
Male	14,160	5,415	6,709	2,037	_
Female	11,210	2,158	6,786	2,265	_
ADDIS ABABA CITY	11,210	2,130	0,700	2,203	_
ADMINISTRATION					
Total	383,454	116,563	187,341	78,417	1,133
Male	224,745	71,635	110,047	42,308	755
Female	158,709	44,928	77,294	36,109	378
DIRE DAWA ADMINISTI	·	,	,	,	2.0
Total	46,666	13,927	17,049	15,690	-
Male	25,671	9,067	10,277	6,327	-
Female	20,995	4,859	6,772	9,363	_
- I OITIGIO	20,000	7,000	0,172	3,500	

Table 5.30 Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex, Employment Sector and Percentage Share of Informal-COUNTRY TOTAL: 2012

1 0		S	ector of Econo	my		Б ,
Age Group and Sex	Total		Informal Job	Not Identified	Not Stated	Percent Informal
A II . A						
All Ages	= 4=0 000	0.440.004		- 0.044	0.400	
Total	5,152,260	3,443,224	1,634,069	72,841	2,126	31.7
Male	2,989,199	2,216,197	722,316	49,482	1,204	24.2
Female 10-14	2,163,061	1,227,027	911,753	23,359	921	42.2
Total	153,137	48,342	103,954	840	_	67.9
Male	82,234	28,830	52,905	499	_	64.3
Female	70,903	19,512	51,050	341	-	72.0
15-19	,	,	,			
Total	356,108	187,917	161,426	6,765	-	45.3
Male	183,587	100,197	80,000	3,391	-	43.6
Female	172,521	87,720	81,427	3,374	-	47.2
20-24						
Total	791,899	560,730	215,126	15,355	688	27.2
Male .	429,898	313,147	107,713	8,639	399	25.1
Female	362,001	247,583	107,413	6,715	289	29.7
25-29 Total	1,046,391	782,483	248,911	14,464	533	23.8
Male	619,748	493,768	114,933	10,616	431	23.6 18.5
Female	426,643	288,715	133,978	3,848	102	31.4
30-34	720,043	200,713	133,370	3,040	102	51.4
Total	722,381	495,781	213,177	12,992	431	29.5
Male	425,379	320,764	95,039	9,425	151	22.3
Female	297,002	175,017	118,138	3,567	280	39.8
35-39						
Total	697,335	475,956	210,559	10,613	207	30.2
Male	401,353	311,075	83,386	6,891	-	20.8
Female	295,982	164,881	127,172	3,722	207	43.0
40-44	445.040	204 500	440.000	4.040	040	20.4
Total Male	415,642 253,857	294,506 204,098	116,602 45,537	4,316 4,046	218 176	28.1 17.9
Female	161,785	90,407	71,064	270	42	43.9
45-49	101,703	30,407	7 1,004	210	72	+0.0
Total	339,675	229,619	106,688	3,321	47	31.4
Male	201,516	160,333	38,730	2,405	47	19.2
Female	138,160	69,286	67,958	916	_	49.2
50-54						
Total	245,259	153,661	90,492	1,107	-	36.9
Male	142,201	110,191	30,950	1,060	-	21.8
Female	103,059	43,470	59,542	46	-	57.8
55-59						
Total	146,012	93,825	50,803	1,383	-	34.8
Male	92,094	71,227	19,830	1,037	-	21.5
Female 60-64	53,918	22,599	30,973	346	-	57.4
Total	101,105	57,234	42,729	1,143	_	42.3
Male	65,586	47,689	16,967	930	_	25.9
Female	35,519	9,544	25,762	213	_	72.5
65+	,0.0	-,	,· ~-			0
Total	137,315	63,171	73,602	542	-	53.6
Male	91,746	54,878	36,326	542	-	39.6
Female	45,569	8,293	37,276	-	-	81.8

Table 5.31 Currently Employed Population of Urban Areas Aged Ten Years and above by Region, Sex, Employment Sector of Economy and Percentage Share of Informal Sector: 2012

IIIIOI	mai Sector:		tor of Free	~	1	
Region and Sex		Sed	ctor of Econor			Percent
	Total	Formal Job	Informal Job	Not Identified	Not Stated	Informal
COLINTRY TOTAL	Total	FUIIIai Jub	JUD	identined	Not Stated	
COUNTRY TOTAL	F 450 000	2 442 224	4 604 000	70.044	0.400	24.7
Total	5,152,260	3,443,224	1,634,069	72,841	2,126	31.7
Male	2,989,199	2,216,197	722,316	49,482	1,204	24.2
Female	2,163,061	1,227,027	911,753	23,359	921	42.2
TIGRAY REGION	004 704	040 404	407.547	4.540	450	00.7
Total	361,724	249,484	107,547	4,540	153	29.7
Male	191,545	144,778	44,190	2,577	-	23.1
Female	170,180	104,706	63,356	1,964	153	37.2
AFFAR REGION						
Total	61,981	30,291	31,666	12	13	51.1
Male	34,262	20,398	13,839	12	13	40.4
Female	27,719	9,893	17,826	-	-	64.3
AMHARA REGION						
Total	954,833	601,375	341,762	11,528	169	35.8
Male	519,392	376,484	134,945	7,872	91	26.0
Female	435,441	224,890	206,818	3,655	78	47.5
OROMIA REGION						
Total	1,503,530	959,416	529,107	14,479	528	35.2
Male	899,247	648,305	242,184	8,582	176	26.9
Female	604,283	311,111	286,923	5,898	352	47.5
SOMALI REGION						
Total	175,661	78,989	95,558	997	116	54.4
Male	94,152	53,772	39,733	590	58	42.2
Female	81,509	25,218	55,826	408	58	68.5
BENISHANGUL -GUMI	UZ REGION					
Total	64,363	40,204	23,874	285	-	37.1
Male	36,334	27,500	8,705	129	-	24.0
Female	28,029	12,705	15,169	155	-	54.1
SNNP REGION						
Total	759,119	449,476	299,835	9,528	280	39.5
Male	435,676	302,415	126,350	6,912	-	29.0
Female	323,443	147,061	173,485	2,616	280	53.6
GAMBELLA REGION						
Total	32,543	19,368	12,610	443	122	38.7
Male	16,695	12,604	3,776	194	122	22.6
Female	15,848	6,764	8,835	249	_	55.7
HARARI REGION	,	,	•			
Total	47,609	30,502	16,825	283	-	35.3
Male	26,202	18,814	7,149	238	_	27.3
Female	21,407	11,687	9,676	44	_	45.2
ADDIS ABABA CITY	,	,	3,3. 3			
ADMINISTRATION						
Total	1,096,196	925,658	139,834	29,960	745	12.8
Male	681,993	573,096	86,213	21,940	745	12.6
Female	414,203	352,562	53,621	8,020	-	12.9
DIRE DAWA ADMINIST		,	•	,		
Total	94,700	58,462	35,451	786	-	37.4
Male	53,700	38,032	15,232	437	-	28.4
Female	40,999	20,430	20,219	350	-	49.3
	.0,000	_0,.00	,			



Table 5.32 Currently Employed Populations of Major Towns Aged Ten Years and above by Sex, Employment Sector and Percentage share of Informal: 2012

by Sex, L	Sector of Economy											
		060	OI OI LOUIR	JIII y		Percent						
Major Towns and Sex		Formal	Informal	Not	Not	Informal						
jo olo ana ook	Total	Job	Job	Identified	Stated							
MAJOR TOWNS TOTAL												
Total	1,972,859	1,538,311	391,352	41,711	1,485	19.8						
Male	1,192,225	958,999	201,601	30,421	1,204	16.9						
Female	780,634	579,312	189,751	11,289	281	24.3						
MEKELE TOWN	,	,-	,	,								
Total	109,035	79,464	28,254	1,317	-	25.9						
Male	62,615	48,967	12,662	985	-	20.2						
Female	46,420	30,497	15,592	332	-	33.6						
ASAYITA TOWN												
Total	7,240	4,313	2,902	12	13	40.1						
Male	3,940	2,842	1,074	12	13	27.3						
Female	3,300	1,471	1,828	-	-	55.4						
GONDER TOWN												
Total	90,957	63,400	26,135	1,254	169	28.7						
Male	50,101	38,212	10,620	1,178	91	21.2						
Female	40,856	25,187	15,515	75	78	38.0						
DESSIE TOWN												
Total	43,171	31,579	10,649	943	-	24.7						
Male	25,671	19,227	5,870	574	-	22.9						
Female	17,499	12,352	4,779	369	-	27.3						
BAHIR DAR TOWN												
Total	86,364	62,834	22,045	1,484	-	25.5						
Male	45,856	36,402	8,830	624	-	19.3						
Female	40,508	26,432	13,215	861	-	32.6						
BISHOFTU TOWN	00.404	00.705	0.440	0.40		05.0						
Total	36,491	26,795	9,446	249	-	25.9						
Male	21,973	17,008	4,755	210	-	21.6						
Female	14,518	9,787	4,691	40	-	32.3						
ADAMA TOWN	05 704	62.205	10.161	2 107	220	22.2						
Total	85,784	63,205	19,161	3,197	220	22.3						
Male Female	53,409	39,842	10,973	2,476 722	118 102	20.5 25.3						
JIMMA TOWN	32,375	23,364	8,188	122	102	25.5						
Total	52,997	38,683	14,167	47	101	26.7						
Male	32,103	24,462	7,536	47 47	58	23.5						
Female	20,895	14,221	6,631		42	31.7						
SHASHEMENE TOWN	20,000	17,221	0,001		72	31.7						
Total	38,776	25,607	12,818	351	_	33.1						
Male	25,016	18,858	5,895	262	_	23.6						
Female	13,760	6,748	6,923	88	_	50.3						
JIJIGA TOWN		0,1.10	0,020			00.0						
Total	37,557	20,958	16,209	273	116	43.2						
Male	22,646	15,152	7,212	223	58	31.8						
Female	14,911	5,806	8,996	50	58	60.3						
ASOSA TOWN	,-	,	7									
Total	16,816	10,987	5,829	-	-	34.7						
Male	9,802	7,501	2,301	-	-	23.5						
Female	7,014	3,487	3,528	-	-	50.3						
	•	•	•									

Table 5.32 Cont'd

	Sector of Economy								
Major Towns and Sex		Formal	Informal	Not	Not	Informal			
	Total	Job	Job	Identified	Stated				
HAWASSA TOWN									
Total	112,676	85,194	26,039	1,443	-	23.1			
Male	68,258	53,645	13,508	1,105	-	19.8			
Female	44,419	31,549	12,531	339	-	28.2			
GAMBELLA TOWN									
Total	16,491	10,670	5,588	111	122	33.9			
Male	8,941	6,938	1,770	111	122	19.8			
Female	7,550	3,732	3,818	-	-	50.6			
HARAR TOWN									
Total	47,609	30,502	16,825	283	-	35.3			
Male	26,202	18,814	7,149	238	-	27.3			
Female	21,407	11,687	9,676	44	-	45.2			
ADDIS ABABA CITY									
Total	1,096,196	925,658	139,834	29,960	745	12.8			
Male	681,993	573,096	86,213	21,940	745	12.6			
Female	414,203	352,562	53,621	8,020	-	12.9			
DIRE DAWA TOWN									
Total	94,700	58,462	35,451	786	-	37.4			
Male	53,700	38,032	15,232	437	-	28.4			
Female	40,999	20,430	20,219	350	-	49.3			

Table 5.33 Distribution of Urban Population Aged Ten years and above Working in the Informal Sector Group and Regions: 2012

		Major Occupational Group											
					<u> </u>	Service,	ľ						
Region and Sex		Legislator		Technical		Shop &	Skilled	Cra					
Region and Sex	Total	Senior		and		Market	agricultural	ar					
	Employed	Officials &		Associate		sales	and	rela					
	Persons	Managers	Professional	Professionals	Clerks	Workers	Fishery	trac					
COUNTRY TOTAL													
Total	1,634,069	1,455	2,533	13,310	9,114	- ,	100,118	515					
Male	722,316	1,316	2,325	11,226	5,029		62,199	216					
Female	911,753	140	207	2,084	4,085	432,553	37,919	299					
TIGRAY REGION													
Total	107,547	99	498	218	109	30,298	7,602	46,2					
Male	44,190	99	498	218	-	4,499	4,842	17,					
Female	63,356	-	-	-	109	25,799	2,760	29,0					
AFFAR REGION													
Total	31,666	12	-	156	180	15,721	6,266	5,0					
Male	13,839	-	-	156	180	3,252	5,622	2,1					
Female	17,826	12	-	-	-	12,468	644	2,8					
AMHARA REGION													
Total	341,762	45	617	2,441	3,961	104,599	16,422	128					
Male	134,945	45	522	2,441	1,188	24,695	12,531	49,9					
Female	206,818	-	95	-	2,772	79,905	3,891	78,					
OROMIA REGION													
Total	529,107	126	502	2,010	2,002	196,377	30,229	169					
Male	242,184	126	466	1,268	2,002	53,101	15,128	75,3					
Female	286,923	-	36	743	-	143,276	15,100	93,					
SOMALI REGION													
Total	95,558	-	-	1,959	-	67,441	8,341	5,6					
Male	39,733	-	-	1,520	-	20,050	6,660	2,8					
Female	55,826	-	-	439	-	47,391	1,681	2,7					
BENISHANGUL GUMUZ REGIO	NC												
Total	23,874	-	-	360	989	5,759	1,143	8,4					
Male	8,705	-	-	240	153	1,732	887	2,0					
Female	15,169	-	-	119	836	4,028	256	6,4					
SNNP REGION													
Total	299,835	338	245	2,974	949	112,405	26,383	87,9					
Male	126,350	210	168	2,586	631	36,628	14,472	30,3					
Female	173,485	128	77	388	319	75,777	11,911	57,6					

Table 5.33 Cont'd

Table 3.33 Colli u								
				Major (Occupat	ional Grou	ıp	
Region and Sex	T	Legislator		Technical		Service, Shop &	Skilled	Cra
	Total	Senior		and		Market	agricultural	ar
	Employed	Officials &	D	Associate	01	sales	and	rela
	Persons	Managers	Professional	Professionals	Clerks	Workers	Fishery	trac
GAMBELLA REGION								
Total	12,610	-	-	-	33	4,502	1,037	5,8
Male	3,776	-	-	-	33	1,399	754	92
Female	8,835	-	-	-	-	3,103	284	4,9
HARARI REGION								
Total	16,825	-	-	49	195	9,414	-	4,0
Male	7,149	-	-	-	147	1,939	-	2,8
Female	9,676	-	-	49	49	7,474	-	1,1
ADDIS ABABA CITY ADMINISTRATION								
Total	139,834	746	670	2,710	346	48,993	1,039	47,2
Male	86,213	746	670	2,364	346	27,053	692	28,8
Female DIRE DAWA ADMINISTRATION	53,621	-	-	346	-	21,941	346	18,
Total	35,451	90	-	434	350	16,471	1,656	7,2
Male	15,232	90	-	434	350	5,079	610	3,5
Female	20,219	-	-	-	-	11,391	1,047	3,6

Table 5.34: Distribution of Urban Population Aged Ten Years and Above Working in the Informal Sector by Region, Sex and Major Branches of Industry: 2012

		Maior S	Sector of Econ	omv	
		Manufacturing,	COLOT OF LCOTE	Office	Agriculture,
Region and Sex		Construction,	Whole Sale	Other	Hunting,
rtogion and cox		Mining &	and Retail	Service	Forestry &
	Total	Quarrying	Trade	Sectors	Fishing
COUNTRY TOTAL	rotar	Qualitying	Trade	0001013	i ioning
Total	1,634,069	429,067	544,201	531,414	129,387
Male	722,316	220,541	202,458	221,462	77,855
Female			341,742		
	911,753	208,526	341,742	309,952	51,532
TIGRAY REGION	407.547	04.007	40.007	40.404	40.000
Total	107,547	34,867	19,327	43,121	10,232
Male	44,190	18,244	5,540	13,074	7,334
Female	63,356	16,623	13,787	30,047	2,898
AFFAR REGION					
Total	31,666	4,620	9,413	9,358	8,275
Male	13,839	2,248	2,913	2,171	6,507
Female	17,826	2,372	6,500	7,187	1,768
AMHARA REGION					
Total	341,762	110,422	89,833	112,560	28,947
Male	134,945	53,287	25,971	38,123	17,563
Female	206,818	57,135	63,862	74,437	11,383
OROMIA REGION					
Total	529,107	141,029	190,858	164,304	32,916
Male	242,184	77,157	70,617	77,282	17,128
Female	286,923	63,872	120,240	87,022	15,788
SOMALI REGION	,-	/-	-,	- ,-	,
Total	95,558	4,666	54,107	27,428	9,357
Male	39,733	2,509	15,882	14,002	7,340
Female	55,826	2,157	38,225	13,426	2,017
BENISHANGUL GU			00,220	10,120	2,011
Total	23,874	6,989	3,947	9,497	3,441
Male	8,705	2,454	1,548	2,664	2,038
Female	15,169	4,535	2,398	6,833	1,403
SNNP REGION	15,105	+,555	2,000	0,033	1,403
Total	299,835	71,119	87,393	109,948	31,375
Male	126,350	29,077	37,218	42,255	17,801
Female	173,485	42,042	50,175	67,693	13,574
GAMBELLA REGIC		4 000	0.000	0.050	4.070
Total	12,610	1,923	2,660	6,052	1,976
Male	3,776	971	1,229	823	754
Female	8,835	952	1,432	5,229	1,222
HARARI REGION					
Total	16,825	4,055	10,127	2,643	-
Male	7,149	2,645	2,344	2,161	-
Female	9,676	1,410	7,784	482	-
ADDIS ABABA CIT					
Total	139,834	43,301	58,665	36,830	1,039
Male	86,213	28,304	33,347	23,870	692
Female	53,621	14,997	25,318	12,960	346
DIRE DAWA ADMI	NISTRATION				
Total	35,451	6,076	17,871	9,674	1,830
Male	15,232	3,646	5,850	5,040	697
Female	20,219	2,431	12,021	4,634	1,133



Distribution of Urban Population Aged ten years and Above Working in the Informal Sector by Liter Educational Level, Sex and Regions: 2012 **Table 5.35**

Edu	icational Le	evel, Sex a	na Keg	10/15: 20		rada Campla	tod for Literat			
					rignest G	raue Comple	ted for Literat	le I		
Region and Sex					Secondary					Deg
J 2 2 1 1 2 2 2 1 1			Non	Grades	not	Secondary		Certificate	Diploma	and a
	Total	Pre-school		1-8	completed		Preparatory			
COUNTRY TOTAL					·			·	·	
Total	1,634,069	13,754	33,145	744,128	89,605	135,897	17,731	13,678	21,899	
Male	722,316			392,456	50,073	77,853	9,959	6,867	14,432	
Female	911,753	7,377	17,793	351,672	39,532	58,044	7,771	6,811	7,467	
TIGRAY REGION										
Total	107,547	255	2,496	46,173	4,465	11,519	1,115	534		
Male	44,190	-	1,351	24,780	1,728	5,771	213	425		
Female	63,356	255	1,145	21,392	2,737	5,748	901	109	218	
AFFAR REGION										
Total	31,666	-	1,065	14,113	1,669	464	156	25	_	
Male	13,839	-	682	7,064	1,345	369	156	-	24	
Female	17,826	-	384	7,048	324	96	-	25	168	
AMHARA REGION										
Total	341,762		10,880	136,340	11,412	26,585	2,206	2,761	3,686	
Male	134,945	2,005	5,571	69,074	6,023	13,152	1,593	1,121	3,084	
Female	206,818	2,284	5,310	67,266	5,389	13,433	614	1,640	602	
OROMIA REGION										
Total	529,107	6,222	9,061	264,614	31,873	42,951	4,095	4,759		
Male	242,184	2,366	4,430	144,928	16,583	22,136	1,793	2,897		
Female	286,923	3,857	4,632	119,686	15,290	20,815	2,302	1,862	5,259	
SOMALI REGION										
Total	95,558	314	1,331	29,703	2,686	4,371	152	-	-	
Male	39,733	314	492	15,135	2,026	3,540	101	-	-	
Female	55,826	-	840	14,568	661	831	50	-	-	
BENISHANGUL										
GUMUZ REGION	00.07:	4=6	 -	0.00=	4 000	4 04-	40-	-		
Total	23,874	150	772	9,895	1,639	1,617	498	37		
Male	8,705	113	119	4,405	725	766	253	-		
Female	15,169	37	653	5,490	914	852	245	37	143	
SNNP REGION			0.5				<u>.</u>	<u>.</u>	<u>.</u> :=	
Total	299,835	1,215	2,628	159,202	16,641	27,269	3,120	3,396		
Male .	126,350	444	754	76,407	9,803	15,492	1,239	1,237	•	
Female	173,485	771	1,874	82,796	6,838	11,778	1,881	2,159	498	

Table 5.35 Cont'd

Table 3.33 Cont a										
					Highest G	rade Comple	ted for Literat	e	<u>, </u>	
Region and Sex	Total	Pre-school	Non Formal	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate Completed	Diploma Completed	De and Com
GAMBELLA REGION	Total	1 16-3011001	Tomai		compiciou	Completed	rieparatory	Completed	Completed	Com
Total	12,610	_	_	5,459	926	614	221	_	_	
Male	3,776	-	-	2,032	361	218	_	-	_	
Female	8,835	-	-	3,427	565	395	221	-	-	
HARARI REGION										
Total	16,825	-	590	9,186	1,132	1,215	142	49	245	
Male	7,149	-	49	4,269	595	728	49	49	98	
Female ADDIS ABABA CITY ADMINISTRATION	9,676	-	541	4,917	538	487	93	-	147	
Total	139,834	1,038	3,107	53,142	14,797	16,956	5,503	2,031	3,128	
Male	86,213	1,038	1,038	35,614	9,884	13,828	4,130	1,052	2,782	
Female DIRE DAWA ADMINISTRATION	53,621	-	2,069	17,528	4,913	3,129	1,373	979	346	
Total	35,451	270	1,214	16,300	2,363	2,335	523	87	437	
Male	15,232	97	867	8,748	1,000	1,855	434	87	350	
Female	20,219	173	347	7,552	1,363	480	90	-	87	

Table 5.36 Distribution of Urban Population Aged Ten Years and Above Working in the Informal Sector by Regions, Sex and Size of Workers: 2012

	111101111	al Sector	by iteg	10113, 0				3. 2012	•			
Region					S	ize of Wor	rkers					
and Sex											_	Not
	Total	<10	11-19	20-29	30-39	40-49	50-59	60-69	70-79	80-89	90+	Stated
COUNTRY TO												
Total	1,634,069	1,597,368		4,053	2,913	763	552	56	-	-	3,247	8,339
Male	722,316	697,450	14,164	2,636	1,435	453	484	-	-	-	1,738	3,956
Female	911,753	899,918	2,615	1,417	1,478	310	67	56	-	-	1,510	4,383
TIGRAY REG												
Total	107,547	104,951	1,506	226	109	-	-	-	-	-	109	646
Male	44,190	42,144	1,279	226	109	-	-	-	-	-	109	323
Female	63,356	62,807	226	-	-	-	-	-	-	-	-	323
AFFAR REGIO												
Total	31,666	31,378	-	-	-	144	-	-	-	-	-	144
Male	13,839	13,695	-	-	-	144	-	-	-	-	-	-
Female	17,826	17,683	-	-	-	-	-	-	-	-	-	144
AMHARA REC												
Total	341,762	334,959	3,737	1,570	438	-	-	-	-	-	573	486
Male	134,945	130,493	2,553	735	373	-	-	-	-	-	573	217
Female	206,818	204,466	1,184	835	65	-	-	-	-	-	-	269
Total	529,107	520,270	6,091	700	750	-	552	-	-	-	456	288
Male	242,184	234,900	5,561	583	220	-	484	-	-	-	357	80
Female	286,923	285,370	531	117	531	-	67	-	-	-	99	207
SOMALI REG	ION											
Total	95,558	95,346	213	-	-	-	-	-	-	-	-	-
Male	39,733	39,571	162	-	-	-	-	-	-	-	-	-
Female	55,826	55,775	50	-	-	-	-	-	-	-	-	-
BENISHANGU	JL GUMUZ	REGION										
Total	23,874	23,846	28	-	-	-	-	-	-	-	-	-
Male	8,705	8,677	28	-	-	-	-	-	-	-	-	-
Female	15,169	15,169	-	-	-	-	-	-	-	-	-	-
SNNP REGIO	Ν											
Total	299,835	294,558	1,534	633	961	-	-	-	-	-	327	1,822
Male	126,350	123,523	1,307	245	388	-	-	-	-	-	327	561
Female	173,485	171,035	226	388	573	-	-	-	-	-	-	1,261
Total	12,610	12,132	295	127	-	-	-	56	-	-	-	-
Male	3,776	3,469	207	99	-	-	-	-	-	-	-	-
Female	8,835	8,663	88	28	-	-	-	56	-	-	-	-
HARARI REG	ION											
Total	16,825	16,583	101	141	-	-	-	-	-	-	-	-
Male	7,149	6,956	101	92	-	-	-	-	-	-	-	-
Female	9,676	9,627	-	49	-	-	-	-	-	-	-	-
ADDIS ABABA	A CITY ADN	MINISTRAT	ON									
Total	139,834	128,153	3,101	656	656	619	-	-	-	-	1,783	4,866
Male	86,213	79,050	2,792	656	346	310	-	-	-	-	372	2,688
Female	53,621	49,103	310	-	310	310	-	-	-	-	1,410	2,179
DIRE DAWA	ADMINISTR	ATION										
Total	35,451	35,191	173	-	-	-	-	-	-	-	-	87
Male	15,232	14,972	173	-	-	-	-	-	-	-	-	87
Female	20,219	20,219	-	-	-	-	-	-	-	-	-	-

Table 5.37 Distribution of Urban Population aged ten years and Above Working in the Informal Sector by Re Working Places: 2012

							Тур	e of Work	ing Place		
						Gulit or	Farm			Any	
Region and Sex		Business		At	On	Open	Area		Quarrying/	Where as	
	Total	House	Office	Home	Street	Market	/Field	Factory	mining	Found	Area
COUNTRY TOTAL											
Total	1,634,069	151,500	11,810	499,431	136,996	302,073	85,059	4,645	3,502	263,256	51,0
Male	722,316	70,258	6,526	114,125	75,085	86,727	55,817	3,380	3,274	202,648	42,6
Female	911,753	81,242	5,284	385,306	61,912	215,346	29,242	1,265	228	60,608	8,3
TIGRAY REGION											
Total	107,547	11,428	850	49,752	5,047	9,443	4,406	-	1,462	16,400	2,6
Male	44,190	2,543	386	10,227	2,605	2,678	4,065	-	1,462	14,382	2,5
Female	63,356	8,886	464	39,525	2,443	6,766	341	-	-	2,018	
AFFAR REGION											
Total	31,666	2,683	72	11,426	3,779	1,530	3,167	12	464	4,017	
Male	13,839	983	-	1,930	1,232	427	2,461	12	464	2,077	
Female	17,826	1,700	72	9,497	2,547	1,102	706	-	-	1,940	
AMHARA REGION											
Total	341,762	27,220	2,467	121,852	18,748	56,718	17,716	368	207	57,194	11,7
Male	134,945	11,507	2,009	26,039	11,197	14,058	13,417	368	207	38,579	8,2
Female	206,818	15,713	458	95,813	7,551	42,659	4,298	-	-	18,615	3,4
OROMIA REGION											
Total	529,107	49,023	2,578	154,473	47,779	117,410	17,659	2,045	314	86,874	17,8
Male	242,184	24,141	976	36,913	28,420	29,872	10,193	1,531	314	72,792	15,9
Female	286,923	24,882	1,602	117,559	19,359	87,538	7,466	515	-	14,081	1,9
SOMALI REGION											
Total	95,558	19,447	1,375	6,669	7,650	20,137	16,869	-	314	12,603	
Male	39,733	6,952	1,010	1,913	2,039	5,426	7,803	-	314	10,203	
Female	55,826	12,495	365	4,756	5,610	14,711	9,066	-	-	2,400	
BENISHANGUL GU	JMUZ REGI	ON									
Total	23,874	1,595	139	8,444	980	2,264	3,689	-	-	2,184	
Male	8,705	540	23	802	375	723	2,221	-	-	1,868	
Female	15,169	1,055	115	7,642	606	1,541	1,469	-	-	316	
SNNP REGION											
Total	299,835	16,917	1,462	95,180	19,009	62,661	18,370	1,583	594	48,856	7,0
Male	126,350	9,247	378	18,292	10,100	20,212	13,687	967	415	34,581	6,0
Female	173,485	7,670	1,084	76,888	8,909	42,449	4,683	617	179	14,275	
	•	•	•	•	•	•				•	

Table 5.37 Cont'd

							Тур	e of Work	ing Place		
						Gulit or	Farm			Any	
Region and Sex		Business		At	On	Open	Area		Quarrying/	Where as	Construc
rtogion and cox	Total	House	Office	Home	Street	Market	/Field	Factory	mining	Found	Area
GAMBELLA REGIO	N										
Total	12,610	1,161	276	4,514	486	3,154	410	-	-	1,187	
Male	3,776	482	71	372	127	764	410	-	-	856	
Female	8,835	679	204	4,142	358	2,390	-	-	-	331	
HARARI REGION											
Total	16,825	847	191	5,154	2,728	3,574	-	150	148	2,164	1,
Male	7,149	394	49	1,042	949	905	-	106	98	2,074	1,
Female	9,676	452	142	4,112	1,780	2,669	-	44	49	89	
ADDIS ABABA CITY	ADMINIST	RATION									
Total	139,834	18,594	2,137	34,549	23,320	19,476	1,039	310	-	26,369	6,
Male	86,213	12,226	1,450	15,012	15,874	8,471	692	310	-	21,929	5,
Female	53,621	6,369	687	19,537	7,446	11,005	346	-	-	4,440	1,
DIRE DAWA ADMIN	IISTRATIOI	V									
Total	35,451	2,585	263	7,418	7,469	5,706	1,734	176	-	5,409	2,
Male	15,232	1,243	173	1,584	2,167	3,190	867	87	-	3,306	1,
Female	20,219	1,342	90	5,834	5,303	2,516	867	90	-	2,102	

Distribution of Urban Population Aged Ten Years and Above Working in the Informal Sector by Re Types of Difficulties faced: 2012 **Table 5.38**

Types of Difficultie	es laceu:	2012							
							Region)	
Types of Difficulties					_		Benishangul		_
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambella
First Difficulties									
Total	1,634,069	107,547	31,666	341,762	529,107	95,558	23,874	299,835	12,610
I do not have Difficulties	555,163	38,734	15,465	95,212	159,946	52,235	11,635	131,905	7,644
Lack of Premises	142,796	5,793	1,685	33,788	42,048	4,641	1,510	16,769	705
Lack of Working Capital	305,377	24,068	5,014	74,349	109,772	8,847	4,003	54,943	1,455
Limited Market	252,664	17,249	5,138	57,109	87,937	10,420	2,307	35,574	1,452
No Training/Inadequate Skill/	11,789	1,060	85	2,365	2,026	746	28	2,252	170
Difficulty in Obtaining License	11,356	396	96	1,325	5,917	421	381	1,471	-
Lack of credit	19,351	1,673	108	1,069	6,881	4,138	37	2,417	28
Access to raw material	74,913	4,581	1,161	15,257	24,517	2,838	523	12,979	379
Lack of Working Material	10,657	1,388	188	3,089	2,587	-	222	1,981	28
Lack of Enough Information	1,542	675	-	-	469	52	-	-	-
Pressure from Government Regulation	18,264	514	50	3,402	5,076	574	138	4,822	41
Inadequate salary	41,377	3,262	376	11,028	13,705	450	318	3,956	175
Inconvenient work place	79,730	2,394	1,640	20,115	26,467	4,593	489	14,810	252
Others	107,248	5,761	516	23,117	41,677	5,605	2,282	15,676	281
Not Stated	1,842	-	144	538	80	-	-	280	-
Second Difficulties									
Total	1,634,069	107,547	31,666	341,762	529,107	95,558	23,874	299,835	12,610
I do not have Difficulties	933,302	66,675	22,644	177,007	287,956	78,443	16,075	194,382	9,117
Lack of Premises	81,102	4,569	1,466	22,691	25,836	962	602	10,110	641
Lack of Working Capital	139,834	8,481	1,545	33,973	43,949	4,036	1,916	24,409	1,087
Limited Market	155,604	4,815	2,901	41,091	53,674	3,933	1,825	21,559	486
No Training/Inadequate Skill/	15,966	1,643	48	4,439	4,740	999	63	1,244	83
Difficulty in Obtaining License	17,744	457	48	2,888	7,728	632	129	2,267	83
Lack of credit	32,748	2,350	396	2,684	14,855	2,085	139	6,692	104
Access to raw material	73,994	5,485	158	18,385	23,497	1,454	910	10,733	214
Lack of Working Material	11,303	213	60	4,824	2,447	-	308	2,024	165
Lack of Enough Information	6,662	1,816	-	1,867	1,153	-	-	1,257	-
Pressure from Government Regulation	12,784	638	60	1,931	2,428	358	267	3,485	-
Pressure from Government Regulation	12,784	638	60	1,931	2,428	358	267	3,485	-
Inadequate salary	21,007	2,005	176	5,529	10,340	151	-	1,813	31
Inconvenient work place	80,746	5,713	1,720	13,550	27,673	2,343	972	13,038	490
Others	49,326	2,687	299	10,258	22,751	162	671	6,541	109
Not Stated	1,948	-	144	645	80	-	-	280	-

Table 5.38 Cont'd

						Regior	า		_
Types of Difficulties	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNN PR	G
Third Difficulties									
Total	1,634,069	107,547	31,666	341,762	529,107	95,558	23,874	299,8 35 255,2	
I do not have Difficulties	1,317,018	89,816	26,896	264,446	438,112	89,038	19,690	48	
Lack of Premises	28,289	2,051	567	6,451	9,136	207	175	3,909	
Lack of Working Capital	30,208	1,632	911	7,716	9,060	310	465	2,864	
Limited Market	52,604	3,050	858	15,845	14,139	1,656	1,144	5,794	
No Training/Inadequate Skill/	15,634	1,852	236	4,452	2,309	679	52	1,817	ŀ
Difficulty in Obtaining License	9,322	-	496	1,117	1,971	365	28	1,504	
Lack of credit	23,066	1,210	242	2,834	7,045	917	157	3,307	
Access to raw material	39,838	1,801	110	14,640	11,844	417	841	5,058	
Lack of Working Material	10,035	255	-	3,797	3,981	-	22	703	ŀ
Lack of Enough Information	2,988	842	12	96	245	52	-	-	l
Pressure from Government Regulation	13,548	-	188	4,026	1,953	70	363	2,731	
Pressure from Government Regulation	13,548	-	188	4,026	1,953	70	363	2,731	
Inadequate salary	5,057	508	353	1,477	1,061	52	-	297 12,30	
Inconvenient work place	57,404	3,292	510	8,312	18,486	1,380	602	3	
Others	27,110	1,238	144	5,909	9,683	415	334	4,020	
Not Stated	1,948	-	144	645	80	-	-	280	

Table 5.39: Currently Employed population of Urban `areas Aged ten years and above by Regions, Sex, Mean Number of Hours Worked Per Week in Major Branches of Industry: 2012

Region and Sex	iliuusti						
Region and Sex		IV		lours worked	by Major Brand		/
Total 43.8 42.9 44.6 46.1 30.6 24.0 Male 46.3 46.0 48.7 48.4 32.9 32.9 22.1 TIGRAY REGION Total 46.9 44.8 51.3 47.7 35.2 2.1 Male 48.9 46.6 56.8 49.8 37.1 - Female 48.8 42.4 41.1 48.9 43.2 38.3 - Male 43.7 40.5 51.5 46.1 40.6 56.7 40.5 32.1 - Female 49.8 42.2 47.2 40.5 32.1 - Female 49.8 42.2 47.2 40.5 32.1 - Female 49.8 42.2 47.2 40.5 32.1 - Female 49.8 42.1 43.6 46.1 28.7 - Female 36.2 33.1 35.3 41.7 15.1 - Female 36.8 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 42.3 42.1 42.3 44.5 47.2 31.8 47.7 Female 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Female 37.8 47.5 40.4 33.7 37.1 - Female 37.8 47.5 40.4 33.7 37.1 - Female 37.8 47.5 40.4 33.7 37.1 - Female 38.6 39.6 33.8 41.6 34.1 - Female 38.6 39.6 33.8 41.6 34.1 - Female 38.6 39.6 33.8 41.6 34.1 - Female 32.1 34.7 31.5 33.0 24.8 - Female 32.1 34.7 31.9 36.9 40.0 25.2 56.0 46.9 32.6 21.0 46.9 32.6 21.0 46.9 32.6 21.0 46.9 32.6 21.0 46.9 32.6 21.0 46.9 32.6 21.0 46.9 32.6 21.0 46.9 32.6 21.0 46.9 32.5 56.0 46.9 32.6 21.0 46.9 32.5 56.0 46.9 32.0 56.0 46.9 32.0 56.0 56.0 56.0 56.0 56.0 56.0 56.0 5	Region and Sex	Total	Construction, Mining &	and Retail	Service	Hunting, Forestry &	
Male 46.3 46.0 37.1 40.4 43.5 25.9 25.0 Female 40.6 37.1 40.4 43.5 25.9 22.1 TIGRAY REGION Total 46.9 44.8 51.3 47.7 35.2 - Male 48.9 46.6 56.8 49.8 37.1 - Female 44.6 41.9 46.7 45.8 31.2 - AFFAR REGION Total 42.4 41.1 48.9 43.2 38.3 - Female 40.8 42.2 47.2 40.5 32.1 - AMHARA REGION Total 39.2 38.5 38.9 43.9 24.3 - Male 41.8 42.1 43.6 46.1 28.7 - Female 36.2 33.1 35.3 41.7 15.1 - OROMIA REGION Total 42.3 42.1 42.3 44.5 43.2 31.8 47.7 </td <td>COUNTRY TOTAL</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	COUNTRY TOTAL						
Male 46.3 46.0 37.1 40.4 43.5 25.9 25.0 Female 40.6 37.1 40.4 43.5 25.9 22.1 TIGRAY REGION Total 46.9 44.8 51.3 47.7 35.2 - Male 48.9 46.6 56.8 49.8 37.1 - Female 44.6 41.9 46.7 45.8 31.2 - AFFAR REGION Total 42.4 41.1 48.9 43.2 38.3 - Female 40.8 42.2 47.2 40.5 32.1 - AMHARA REGION Total 39.2 38.5 38.9 43.9 24.3 - Male 41.8 42.1 43.6 46.1 28.7 - Female 36.2 33.1 35.3 41.7 15.1 - OROMIA REGION Total 42.3 42.1 42.3 44.5 43.2 31.8 47.7 </td <td>Total</td> <td>43.8</td> <td>42.9</td> <td>44.6</td> <td>46.1</td> <td>30.6</td> <td>24.0</td>	Total	43.8	42.9	44.6	46.1	30.6	24.0
Female	Male						
TIGRAY REGION Total	Female						
Total 46.9 44.8 51.3 47.7 35.2 - Male 48.9 46.6 56.8 49.8 37.1 - Female 44.6 41.9 46.7 45.8 31.2 - AFFAR REGION Total 42.4 41.1 48.9 43.2 38.3 - Male 43.7 40.5 51.5 46.1 40.6 - Female 40.8 42.2 47.2 40.5 32.1 - AMHARA REGION Total 39.2 38.5 38.9 43.9 24.3 - Male 41.8 42.1 43.6 46.1 28.7 - Emale 36.2 33.1 35.3 41.7 15.1 - OROMIA REGION Total 42.3 42.1 42.3 44.5 31.5 37.3 Male 45.0 44.9 46.5 47.2 31.8 47.7 Female 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Female 37.8 47.5 40.4 33.7 37.1 - Emale 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Female 37.8 47.5 40.4 33.7 37.1 - Emale 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Total 35.7 37.5 32.8 37.4 30.7 37.1 - BENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 - Male 38.6 39.6 33.8 41.6 34.1 - Female 32.1 34.7 31.5 33.0 24.8 - SONNP REGION Total 42.2 41.3 44.2 43.7 39.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 30.9 32.6 21.0 Female 30.1 34.7 39.8 43.8 41.0 42.2 - Female 30.5 33.3 40.9 39.9 - ADDIS ABABA CITY ADMINISTRATION Total 41.7 39.8 43.8 44.0 39.9 39.9 - ADDIS ABABA CITY ADMINISTRATION Total 46.2 46.2 46.5 55.5 55.5 37.5 - Female DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.5 46.7 33.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.5 46.7 33.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.5 46.7 33.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.5 46.7 33.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.5 46.7 33.0 DIRE DAWA 40.4 47.7 46.7 46.5 46.5 46.7 33.0 DIRE DAWA 40.4 47.7 46.7 46.2 49.6 53.5 5.5 Male 47.7 46.7 46.2 49.6 49.6 53.5 5.5 Total 46.2 46.2 46.2 46.5 46.5 46.7 Male 47.7 46.7 46.7 46.2 49.6 53.5 5.5 Total 46.2 46.2 46.2 46.5 46.5 46.7 Male 47.7 46.7 46.7 46.2 49.6 53.5 Total 46.2 46.2 46.2 46.5 46.5 46.7							
Male 48,9 46.6 56.8 49.8 37.1	Total	46.9	44.8	51.3	47.7	35.2	-
Female 44.6 41.9 46.7 45.8 31.2 AFARR REGION Total 42.4 41.1 48.9 43.2 38.3 - Male 43.7 40.5 51.5 46.1 40.6 - Female 40.8 42.2 47.2 40.5 32.1 - AMHARA REGION Total 39.2 38.5 38.9 43.9 24.3 - Male 41.8 42.1 43.6 46.1 28.7 - Female 36.2 33.1 35.3 41.7 15.1 - OROMIA REGION Total 42.3 42.1 42.3 44.5 31.5 37.3 Male 45.0 44.9 46.5 47.2 31.8 47.7 Female 30.7 7.1 15.1 - SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - - - - - - -							-
AFFAR REGION Total 42.4 41.1 48.9 43.2 38.3 - Male 43.7 40.5 51.5 46.1 40.6 - Female 40.8 42.2 47.2 40.5 32.1 - AMHARA REGION Total 39.2 38.5 38.9 43.9 24.3 - Male 41.8 42.1 43.6 46.1 28.7 - Female 36.2 33.1 35.3 41.7 15.1 0- CROMIA REGION Total 42.3 42.1 42.3 44.5 31.5 37.3 Male 45.0 44.9 46.5 47.2 31.8 47.7 - Female 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Female 37.8 47.5 40.4 33.7 37.1 4.1 - Female 37.8 47.5 40.4 33.7 37.1 37.1 - Female 37.1 31.9 36.9 38.0 32.4 8 - SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 - Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Female 40.4 36.1 44.5 39.7 41.6 40.1 - Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Female 40.4 36.1 44.5 39.7 41.6 - Female 39.5 33.3 40.9 39.9 - Total 43.1 44.2 49.7 41.9 40.7 - Female 39.5 33.3 40.9 39.9 - Total 43.1 44.2 49.7 41.9 40.7 - Female 39.5 33.3 40.9 39.9 - Female 39.5 33.3 30.9 39.9 30.5 - Female 49.0 43.1 49.2 50.3 35.5 - Female 49.0 43.1 49.2 50.							_
Total 42.4 41.1 48.9 43.2 38.3 - Male 43.7 40.5 51.5 46.1 40.6 - Female 40.8 42.2 47.2 40.5 32.1 - Female 40.8 41.8 42.1 43.6 46.1 28.7 - Female 36.2 33.1 35.3 41.7 15.1 - Female 36.2 33.1 35.3 41.7 15.1 - Female 36.2 33.1 35.3 41.7 15.1 - Female 45.0 44.9 46.5 47.2 31.8 47.7 Female 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Female 37.8 47.5 40.4 33.7 37.1 42.1 - Female 37.8 47.5 40.4 33.7 37.1 42.1 - Female 37.8 47.5 40.4 33.7 37.1 5.1 5.2 Female 38.6 39.6 33.8 41.6 34.1 - Female 37.8 47.5 40.4 33.7 37.1 5.1 5.2 SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Female 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 Female 40.4 36.1 44.5 39.7 41.6 41.7 42.1 Female 40.4 36.1 44.5 39.7 41.6 41.7 42.1 Female 40.4 36.1 44.5 39.7 41.6 Female 40.4 43.5 42.3 48.4 41.9 42.2 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 Female 40.4 36.1 44.5 39.7 41.6 Female 40.4 43.5 42.3 48.4 41.9 42.2 5.5 60.0 GAMBELLA REGION Total 43.7 39.8 43.8 41.0 42.2 5.5 60.0 GAMBELLA REGION Total 43.7 39.8 43.8 41.0 42.2 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 5.5 60.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 5.5 60.0							
Male 43.7 40.5 51.5 46.1 40.6		42.4	41.1	48.9	43.2	38.3	_
Female							-
AMHARA REGION							_
Total 39.2 38.5 38.9 43.9 24.3 — Male 41.8 42.1 43.6 46.1 28.7 — Female 36.2 33.1 35.3 41.7 15.1 — OROMIA REGION Total 42.3 42.1 42.3 44.5 31.5 37.3 Male 45.0 44.9 46.5 47.2 31.8 47.7 Female 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 — Male 43.1 41.6 47.6 41.7 42.1 — Female 37.8 47.5 40.4 33.7 37.1 — BENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 37.1 SENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 — Male 38.6 39.6 33.8 41.6 34.1 — Female 32.1 34.7 31.5 33.0 24.8 — SNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 — Male 45.5 49.4 53.7 44.1 40.1 — Female 40.4 36.1 44.5 39.7 41.6 — Female 40.4 36.1 44.5 39.7 41.6 — Female 40.4 36.1 44.5 39.7 41.6 — Female 39.5 33.3 40.9 39.9 — Pemale ADDIS ABABA CITY ADMINISTRATION Total 52.2 50.2 52.5 53.5 37.5 — Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 48.2 46.2 46.2 46.5 46.7 33.0 — Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 48.2 46.2 46.5 46.5 53.5 37.5 — Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.5 46.7 33.0 — Total 46.2 46.2 46.5 46.5 32.5 —						0=	
Male 41.8 42.1 43.6 46.1 28.7		39.2	38.5	38.9	43.9	24.3	_
Female							_
OROMIA REGION Total 42.3 42.1 42.3 44.5 31.5 37.3 Male 45.0 44.9 46.5 47.2 31.8 47.7 Female 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Male 43.1 41.6 47.6 41.7 42.1 - Female 37.8 47.5 40.4 33.7 37.1 - BENISHANGUL GUMUZ REGION 35.7 37.5 32.8 37.4 30.7 - Male 38.6 39.6 33.8 41.6 34.1 - Female 32.1 34.7 31.5 33.0 24.8 - SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.							_
Total 42.3 42.1 42.3 44.5 31.5 37.3 Male 45.0 44.9 46.5 47.2 31.8 47.7 Female 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Male 43.1 41.6 47.6 41.7 42.1 - Female 37.8 47.5 40.4 33.7 37.1 - BENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 - Male 38.6 39.6 33.8 41.6 34.1 - Female 37.1 31.5 33.0 24.8 - Female 37.1 31.9 36.9 40.0 25.2 56.0 25.2 56.0 GAMBELLA REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 45.5 49.4 53.7 55.7 55.0 55.0 55.0 55.0 55.0 55.0 55		00.2	00.1	00.0		10.1	
Male 45.0 44.9 46.5 47.2 31.8 47.7 Female 38.6 36.3 38.1 40.9 30.7 7.1 SOMALI REGION Total 40.7 43.0 43.0 38.5 40.6 - Male 43.1 41.6 47.6 41.7 42.1 - Female 37.8 47.5 40.4 33.7 37.1 - BENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 - Male 38.6 39.6 33.8 41.6 34.1 - - Female 32.1 34.7 31.5 33.0 24.8 - SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 - Female 37.1 31.9 36.9 40.0 25.2 56.0		423	42.1	423	44.5	31.5	37 3
Female 38.6 36.3 38.1 40.9 30.7 7.1							
SOMALI REGION Total							
Total 40.7 43.0 43.0 38.5 40.6 — Male 43.1 41.6 47.6 41.7 42.1 — Female 37.8 47.5 40.4 33.7 37.1 — BENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 — Male 38.6 39.6 33.8 41.6 34.1 — Female 32.1 34.7 31.5 33.0 24.8 — SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 — Male 45.5 49.4 53.7 44.1 40.1 — Female 40.4 36.1 44.5 39.7 41.6 — Female 40.4 36.1 44.5 39.7 41.6 — Female 40.4 36.1 44.5 39.7 41.6 — Female 43.5 42.3 48.4 41.9 42.2 — Male 43.5 42.3 48.4 41.9 42.2 — Female 39.5 33.3 40.9 39.9 — 2 — Cappella ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 — Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.5 46.7 33.0 — Male 46.2 46.5 46.7 33.0 — Cappella Additional Addit		30.0	30.3	30.1	40.9	30.7	7.1
Male 43.1 41.6 47.6 41.7 42.1 - Female 37.8 47.5 40.4 33.7 37.1 - BENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 - Male 38.6 39.6 33.8 41.6 34.1 - Female 32.1 34.7 31.5 33.0 24.8 - SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 -		40.7	12.0	42.0	20.5	40.6	
Female 37.8 47.5 40.4 33.7 37.1 - BENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 - Colspan="8">-							_
BENISHANGUL GUMUZ REGION Total 35.7 37.5 32.8 37.4 30.7 - Male 38.6 39.6 33.8 41.6 34.1 - Female 32.1 34.7 31.5 33.0 24.8 - SINP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Male 43.5 33.3 40.9 39.9 - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -							_
REGION Total 35.7 37.5 32.8 37.4 30.7 - Male 38.6 39.6 33.8 41.6 34.1 - Female 32.1 34.7 31.5 33.0 24.8 - SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Male 39.5 33.3 40.9 39.9 - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.5 46.5 46.7 33.0 -		37.0	47.5	40.4	33.1	37.1	-
Total 35.7 37.5 32.8 37.4 30.7 - Male 38.6 39.6 33.8 41.6 34.1 - Female 32.1 34.7 31.5 33.0 24.8 - SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Male 43.5 53.3 40.9 39.9 - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -							
Male 38.6 39.6 33.8 41.6 34.1		35.7	37.5	32.8	37.4	30.7	_
Female 32.1 34.7 31.5 33.0 24.8 - SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>_</td>							_
SNNP REGION Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - - - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -							_
Total 42.2 41.3 44.2 43.7 29.9 38.5 Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - Female 40.4 36.1 44.5 39.7 41.6 - Female 43.5 42.3 48.4 41.9 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - CADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -		02	0	01.0	00.0	21.0	
Male 46.2 47.6 50.0 46.9 32.6 21.0 Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.5 46.5 46.		42.2	41.3	44 2	43.7	29.9	38.5
Female 37.1 31.9 36.9 40.0 25.2 56.0 GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION - <							
GAMBELLA REGION Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION -							
Total 43.1 44.2 49.7 41.9 40.7 - Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -		07	01.0	00.0	10.0	20.2	00.0
Male 45.5 49.4 53.7 44.1 40.1 - Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION -		43.1	44.2	49.7	41.9	40.7	_
Female 40.4 36.1 44.5 39.7 41.6 - HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -							_
HARARI REGION Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -							_
Total 41.7 39.8 43.8 41.0 42.2 - Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - - - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -		10.1	00.1	11.0	00.1	11.0	
Male 43.5 42.3 48.4 41.9 42.2 - Female 39.5 33.3 40.9 39.9 - - - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -		<i>4</i> 1 7	30.8	43.8	41 0	42.2	_
Female 39.5 33.3 40.9 39.9 - - - ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -							_
ADDIS ABABA CITY ADMINISTRATION Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -						72.2	_
Total 50.8 48.1 51.2 51.9 41.0 10.9 Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -				70.3	00.9	_	_
Male 52.2 50.2 52.5 53.5 37.5 - Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -				51.2	51 Q	<i>/</i> 11 O	10.0
Female 49.0 43.1 49.2 50.3 52.8 39.0 DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -							-
DIRE DAWA ADMINISTRATION Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -							30 U
Total 46.2 46.2 46.5 46.7 33.0 - Male 47.7 46.7 46.2 49.6 32.5 -			43.1	+3.∠	50.5	52.0	39.0
Male 47.7 46.7 46.2 49.6 32.5 -			16.2	16 5	167	33 0	
							-
	Female	44.4	45.5	46.8	42.8	33.4	_



CHAPTER VI

UNEMPLOYEMENT IN URBAN AREAS

6.1. Introduction

Unemployment is one of the indicators that have been used to measure the extent of unutilized human resource in the economy at a given area or region in a certain period of time. It indicates the quantity and quality of the human labour a nation or a region would have, and should be used to promote its economic development, but not actually used. The unutilized human resource as a result of unemployment has its own social and economic implication. Therefore, the availability of reliable and timely statistical data on unemployment is vital to formulate and design new strategies as well as to monitor the existing employment policies.

A wide range of information regarding unemployment and the characteristics of unemployed population were collected in the 2012 Urban Employment and Unemployment Survey. This information is useful to estimate the number of unemployed population and rate of unemployment at various reporting levels.

This chapter presents the size of unemployed population and rate of unemployment at national, regional and major urban levels. In addition, the characteristics of unemployed population such as age, sex, education, training status, marital status, previous work experience, duration of unemployment, type of jobs looking for, problems in establishing own business or enterprise, actions taken to seek work ... etc., are presented in detail. These information were collected from persons aged ten years and above by using the two known measurement approaches, namely current and usual approaches. But, the results presented in this chapter refer only the current activity status approach.

During the data collection period, respondents in the sampled household were asked detailed labour force questions using structured questionnaires. These includes:-

- a. whether they were seeking for employment, if not reason for not seeking.
- b. whether available for work during the coming one month, if not reason for not being available.
- c. type of job looking for /i.e. self employment or paid employment or other/
- d. type of problems faced to establish own business.
- e. previous work experience and duration of unemployment.

Based on the responses obtained from the respondents aged ten years and above, the measurement of unemployment was done using various methods.

6.2 Measurement of Unemployment

The measurement of unemployment is based on the following three criteria that must be satisfied simultaneously: "without work", "currently available for work" 'and "seeking work" (ILO, 1983). The standard definition of unemployment is based on the "seeking work" criterion that can be interpreted as activities or efforts that non-working persons performed to look for jobs during a specified reference period or prior to it (i.e., paid or self employment). The specific steps include registration at a public or private exchange; applying to employers; checking at worksites, farms, factory gates, market or other assembly places; placing or answering news paper advertisements; seeking assistance of friends or relatives; visit vacancy boards; looking for land, building, machinery or equipment to establish own enterprise, and arranging financial resources; applying for work permits and licenses...etc.

However, the standard definition of unemployment with its emphasis on seeking work criterion might be restrictive and might not fully capture the prevailing employment situations in many developing countries including Ethiopia, where the labour market is largely unorganized or of limited scope, where labour absorption is inadequate or where the labour force is largely self-employed.

Hence, the International standards introduced provisions, which allows for the relaxation of the seeking work criterion in certain situations. The provisions are two types, namely, partial relaxation and complete relaxation. Following the recommendations of the International standard and considering the prevailing national situations, the employment and unemployment and related labour force surveys used the above provisions to measure the different forms of unemployment. The definition of the two provisions is described as follows.

Under partial relaxation, the definition of unemployment includes discouraged job seekers, future start and layoffs in addition to persons satisfying the standard definition. Discouraged job seekers are persons who want a job but did not take any active step to search for work because they believe that they cannot find one. Future starts are those persons without work who have made arrangements to take up paid employment or to undertake self-employment activity at a date subsequent to the reference period. Layoff is a person whose contract of employment or activity has been suspended by the employer for a specified or unspecified period at the end of which the person concerned has a recognized right or recognized expectation to cover employment with the employer (ILO, 1990).

The completely relaxed definition of unemployment includes persons without work and those who are available for work, including those who were or were not seeking work. That is, the seeking work criterion is completely relaxed and unemployment is based on the "without work" and "availability" criterion only. The availability in this situation is tested by asking the willingness to take up work for wage or salary in locally prevailing terms, or readiness to undertake self-employment activity; given the necessary resources and facilities. It should be noted that fulltime students are considered as available if and only if they are ready to withdraw from their studies in order to accept a job.

This survey collected unemployment data in the standard, partially relaxed and completely relaxed options of measurements. After thorough evaluation and assessment of the results obtained using the three alternative and complementary

measures; the rates obtained using the completely relaxed definition was found most plausible and hence selected for reporting. In this survey, those persons aged ten years and above who had no work or did not work at least four hours or did not have job to return to, were asked to respond whether they were available or willing to work if job was found during the coming one month. Those who responded "Yes" answer to this question were further tested whether they were ready to take a job under prevailing conditions. By doing so, those persons who had no job but available to work are considered as unemployed at the time of the survey.

6.3. Size and Rate of Urban Unemployment

The proportion of unemployed population aged ten years and above is described using unemployment rate. Unemployment rate is computed as percentage of the unemployed population over the total number of economically active population. The rate indicates that the number of persons who are ready to participate or engaged in the production of goods and services. A higher rate of unemployment implies the loss of individual contributions to the growth of Growth Domestic Product (GDP) as well as the national economy, and hence a waste of human resource.

The distribution of currently unemployed population aged ten years and above and unemployment rate of urban areas of the country by region, age group and sex is presented in Table 6.1. The survey result reveals that the unemployment rate at national urban level was 17.5 percent. The corresponding rates for males and females were 11.4 percent and 24.2 percent, respectively (Table 6.1). Unemployment rate in the regions illustrates that the highest rate was recorded in Addis Ababa 23.0 percent and Dire Dawa Administration 22.7 percent, while the lowest rate as compared to the highest ones was found in Benishangul-Gumuz (7.1 percent) and Gambella (7.8 percent) regions. But in the other regions of the country, the rate was found in the range 11 - 19 percent (Table 6.1).

Table 6.2 shows the distribution of currently unemployed population of total major towns and sixteen selected major towns classified by age group and sex. Among the major towns, the highest unemployment rate above 20.0 percent was recorded in Addis Ababa, Dire Dawa, Dessie, Bishoftu, Adama and Shashemene towns. The lowest unemployment rate as compared to the highest ones was found in Gambella town (7.9 percent).

With regard to literacy status and educational attainment of the currently unemployed population of each region is shown in Table 6.3 and 6.4, while the training status and types of training of the unemployed illustrated in Table 6.5. The distribution of unemployed persons disaggregated by marital status, age group and sex is presented in Table 6.6. Table 6.7 shows similar information for each region classified by sex.

Those persons aged ten years and above who were looking for work during the three months prior to date of interview were asked to report what steps they have taken to seek work and type of work they look for during this period. Accordingly, Table 6.8 presents the distribution of unemployed persons who were and not looking for work during the three months prior to the survey date by sex and step taken to seek work for urban areas of each region. Table 6.9 also provides data on type of job they were looking for by sex and region.

The size of unemployed persons of urban areas of the country categorized by age group sex, previous work experience and duration of unemployment is presented in Table 6.10. The corresponding figure by sex and region is presented in Table 6.11. The distribution of unemployed persons according to the type of problems they faced to start own business by region and sex is shown in Table 6.12.

Table 6.1 Currently Unemployed Population of Urban Areas Aged Ten Years and Above by Region, Age Group, Sex, and Unemployment Rate: 2012

		nomically Ac			nemployed		Unemployment Rate			
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female	
COUNTRY TO	OTAL									
All Ages	6,953,202	3,614,936		1,218,574	410,381	808,193	17.5	11.4	24.2	
10-14	223,235	107,623	115,613	24,618	11,787	12,831	11.0	11.0	11.1	
15-19	630,232	257,775	372,457	136,069	51,396	84,673	21.6	19.9	22.7	
20-24	1,265,279	584,861	680,418	361,492	127,407	234,085	28.6	21.8	34.4	
25-29	1,397,738	733,144	664,594	268,787	79,619	189,168	19.2	10.9	28.5	
30-34	896,084	488,015	408,069	128,378	39,484	88,893	14.3	8.1	21.8	
35-39	841,095	450,158	390,937	98,500	26,203	72,297	11.7	5.8	18.5	
40-44	503,817	290,065	213,752	58,078	16,589	41,489	11.5	5.7	19.4	
45-49	405,044	224,308	180,736	43,186	12,716	30,470	10.7	5.7	16.9	
50-54	302,251	166,568	135,682	36,225	12,971	23,254	12.0	7.8	17.1	
55-59	177,055	106,145	70,910	24,471	9,677	14,794	13.8	9.1	20.9	
60-64	132,266	83,244	49,022	18,566	8,023	10,544	14.0	9.6	21.5	
65+	179,105	123,030	56,076	20,205	14,510	5,694	11.3	11.8	10.2	
TIGRAY REG	ION									
All Ages	487,785	231,098	256,688	94,756	25,996	68,761	19.4	11.2	26.8	
10-14	8,130	3,701	4,430	603	281	323	7.4	7.6	7.3	
15-19	48,786	18,557	30,229	12,014	4,398	7,615	24.6	23.7	25.2	
20-24	96,134	37,694	58,441	30,545	9,254	21,291	31.8	24.6	36.4	
25-29	102,681	46,399	56,283	23,393	5,335	18,059	22.8	11.5	32.1	
30-34	62,243	33,157	29,086	8,462	1,365	7,097	13.6	4.1	24.4	
35-39	53,728	27,800	25,928	6,513	1,480	5,033	12.1	5.3	19.4	
40-44	31,431	17,916	13,515	4,051	1,298	2,753	12.9	7.2	20.4	
45-49	24,888	11,624	13,264	3,210	630	2,580	12.9	5.4	19.5	
50-54	20,545	10,689	9,856	1,941	536	1,406	9.4	5.0	14.3	
55-59	12,404	6,352	6,052	1,290	254	1,036	10.4	4.0	17.1	
60-64	12,696	7,443	5,254	1,506	434	1,072	11.9	5.8	20.4	
65+	14,119	9,768	4,351	1,227	732	495	8.7	7.5	11.4	
AFFAR REGI	ON									
All Ages	86,914	46,082	40,832	9,846	2,425	7,421	11.3	5.3	18.2	
10-14	4,358	1,607	2,751	227	-	227	5.2	-	8.3	
15-19	6,049	3,060	2,989	684	311	372	11.3	10.2	12.4	
20-24	11,302	5,811	5,491	2,455	506	1,949	21.7	8.7	35.5	
25-29	17,134	8,613	8,521	2,266	559	1,707	13.2	6.5	20.0	
30-34	11,896	6,277	5,619	1,173	409	763	9.9	6.5	13.6	
35-39	11,420	6,138	5,282	1,203	48	1,156	10.5	0.8	21.9	
40-44	9,520	5,096	4,424	856	343	513	9.0	6.7	11.6	
45-49	6,205	3,561	2,645	541	12	529	8.7	0.3	20.0	
50-54	4,042	2,578	1,464	406	201	204	10.0	7.8	13.9	
55-59	1,850	996	855	12	12	-	0.6	1.2	-	
60-64	1,988	1,268	720	12	12	-	0.6	0.9	-	
65+	1,149	1,077	72	12	12	-	1.0	1.1	-	

Table 6.1 Cont'd

Table 6.1	Cont'd			Ī		Т			
A C	Ecor	nomically A	ctive	U	nemploye	d	Unemployment Rate		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Femal e
AMHARA RE	GION								
All Ages	1,273,491	631,044	642,448	209,533	68,229	141,304	16.5	10.8	22.0
10-14	51,398	24,611	26,787	4,505	3,900	605	8.8	15.8	2.3
15-19	133,872	59,806	74,067	27,257	9,267	17,990	20.4	15.5	24.3
20-24	246,832	105,461	141,371	75,607	25,276	50,331	30.6	24.0	35.6
25-29	243,244	124,047	119,197	42,132	12,342	29,790	17.3	9.9	25.0
30-34	138,453	72,425	66,028	14,303	4,394	9,909	10.3	6.1	15.0
35-39	138,088	68,859	69,229	15,404	3,790	11,614	11.2	5.5	16.8
40-44	76,018	42,501	33,517	8,239	2,165	6,074	10.8	5.1	18.1
45-49	80,393	41,401	38,992	8,004	2,388	5,617	10.0	5.8	14.4
50-54	58,155	27,663	30,492	4,470	849	3,621	7.7	3.1	11.9
55-59	36,802	19,213	17,589	4,782	1,438	3,344	13.0	7.5	19.0
60-64	26,064	17,403	8,661	1,192	520	672	4.6	3.0	7.8
65+	44,171	27,654	16,516	3,639	1,901	1,738	8.2	6.9	10.5
05+	44,171	27,034	10,510	3,039	1,301	1,730	0.2	0.5	10.5
OROMIA RE	GION								
All Ages	1,979,212	1,080,818	898,394	330,723	115,759	214,964	16.7	10.7	23.9
10-14	75,862	41,501	34,362	11,148	4,801	6,347	14.7	11.6	18.5
15-19	183,487	87,633	95,854	48,172	18,526	29,646	26.3	21.1	30.9
20-24	388,258	196,837	191,421	104,073	37,409	66,664	26.8	19.0	34.8
25-29	388,165	217,709	170,456	71,304	20,499	50,804	18.4	9.4	29.8
30-34	245,089	130,361	114,728	27,749	7,981	19,768	11.3	6.1	17.2
35-39	252,472	139,458	113,014	25,353	7,634	17,720	10.0	5.5	15.7
40-44	135,998	76,525	59,474	12,715	4,353	8,362	9.3	5.7	14.1
45-49	109,972	67,925	42,047	6,515	2,268	4,246	5.9	3.3	10.1
50-54	73,360	40,164	33,196	8,771	3,292	5,479	12.0	8.2	16.5
55-59	49,283	30,283	19,000	5,573	1,953	3,620	11.3	6.4	19.1
60-64	31,710	22,581	9,129	4,515	3,166	1,348	14.2	14.0	14.8
65+	45,555	29,842	15,713	4,835	3,876	960	10.6	13.0	6.1
SOMALI REC	GION								
All Ages	236,974	123,994	112,980	34,427	12,124	22,302	14.5	9.8	19.7
10-14	14,132	7,652	6,481	1,683	680	1,003	11.9	8.9	15.5
15-19	22,217	9,216	13,000	4,557	2,056	2,502	20.5	22.3	19.2
20-24	31,341	14,533	16,808	7,604	1,741	5,863	24.3	12.0	34.9
25-29	34,499	17,684	16,816	6,470	1,775	4,695	18.8	10.0	27.9
30-34	32,334	15,686	16,648	3,417	875	2,542	10.6	5.6	15.3
35-39	28,827	15,672	13,155	3,641	1,369	2,272	12.6	8.7	17.3
40-44	26,791	14,493	12,298	2,510	1,153	1,357	9.4	8.0	11.0
45-49	13,634	8,512	5,122	1,075	372	704	7.9	4.4	13.7
50-54	12,967	7,457	5,509	2,008	1,203	805	15.5	16.1	14.6
55-59	5,019	3,278	1,740	543	229	314	10.8	7.0	18.0
60-64	7,400	4,712	2,688	830	584	246	11.2	12.4	9.2
65+	7,400	5,097	2,715	88	88		1.1	1.7	-
001	7,012	0,001	2,110	00	00	_	1.1	1.7	_

Table 6.1 Cont'd

Age Group	Econ	omically Ac	tive	U	nemploye	ed	Unemployment Rate		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
BENISHANG	UL GUMUZ RE	EGION							
All Ages	78,024	41,751	36,273	5,505	1,897	3,609	7.1	4.5	9.9
10-14	4,905	2,252	2,653	-	-	-	-	-	-
15-19	9,941	4,507	5,435	901	562	339	9.1	12.5	6.2
20-24	12,772	5,842	6,930	1,846	568	1,278	14.5	9.7	18.4
25-29	14,093	8,249	5,843	891	340	551	6.3	4.1	9.4
30-34	8,753	4,197	4,555	442	147	295	5.0	3.5	6.5
35-39	8,937	4,686	4,250	614	66	548	6.9	1.4	12.9
40-44	6,067	4,153	1,914	459	103	356	7.6	2.5	18.6
45-49	4,283	2,754	1,530	63	24	40	1.5	0.9	2.6
50-54	2,649	1,269	1,380	107	-	107	4.0	-	7.8
55-59	1,822	1,230	592	33	33	-	1.8	2.7	-
60-64	1,401	811	590	95	-	95	6.8	-	16.1
65+	2,400	1,799	600	53	53	-	2.2	2.9	-
SNNP REGIO	N			110 17					
All Ages	957,813	507,014	450,799	119,17 6	40,404	78,771	12.4	8.0	17.5
10-14	47,049	20,093	26,956	3,566	628	2,938	7.6	3.1	10.9
15-19	93,698	38,973	54,724	12,899	5,855	7,044	13.8	15.0	12.9
20-24	168,084	84,187	83,897	36,312	13,773	22,540	21.6	16.4	26.9
25-29	187,389	103,942	83,447	27,120	7,899	19,221	14.5	7.6	23.0
30-34	121,863	70,160	51,703	13,033	4,344	8,688	10.7	6.2	16.8
35-39	113,550	56,327	57,224	8,881	1,698	7,182	7.8	3.0	12.6
40-44	74,895	45,409	29,486	4,519	1,232	3,287	6.0	2.7	11.1
45-49	52,509	28,672	23,838	3,177	1,007	2,170	6.1	3.5	9.1
50-54	39,803	25,212	14,591	3,479	1,491	1,988	8.7	5.9	13.6
55-59	19,868	11,612	8,256	1,916	616	1,300	9.6	5.3	15.7
60-64	15,789	7,229	8,560	1,815	601	1,215	11.5	8.3	14.2
65+	23,317	15,200	8,117	2,460	1,262	1,198	10.6	8.3	14.8
GAMBELLA I	REGION								
All Ages	38,759	19,281	19,478	3,020	484	2,536	7.8	2.5	13.0
10-14	429	190	239	115	27	88	26.8	14.2	36.8
15-19	2,454	769	1,686	417	27	390	17.0	3.5	23.1
20-24	7,115	2,644	4,471	814	55	759	11.4	2.1	17.0
25-29	9,710	4,972	4,738	858	143	715	8.8	2.9	15.1
30-34	5,658	3,266	2,392	406	152	254	7.2	4.7	10.6
35-39	5,663	3,157	2,506	221	-	221	3.9	-	8.8
40-44	2,989	1,631	1,358	188	80	109	6.3	4.9	8.0
45-49	2,306	1,036	1,270	-	-	-	-	-	-
50-54	1,537	1,025	511	_	_	_	_	_	_
55-59	382	274	108	_	_	_	_	_	_
60-64	266	129	137	_	_	_	_	_	_
65+	250	188	62	_	_	_	_	_	_
551	200	100	02						

Table 6.1 Cont'd

Age Group	Econ	omically Ac	tive	U	nemploye	d	Unemployment Rate		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
HARARI REG	SION								
All Ages	56,942	28,987	27,955	6,973	2,086	4,887	12.2	7.2	17.5
10-14	557	325	231	-	-	-	-	-	
15-19	3,418	1,209	2,210	548	151	398	16.0	12.5	18.0
20-24	8,080	3,746	4,334	1,591	692	899	19.7	18.5	20.7
25-29	10,184	5,375	4,809	1,218	248	970	12.0	4.6	20.2
30-34	8,567	4,151	4,417	1,152	245	907	13.4	5.9	20.5
35-39	7,798	4,298	3,500	948	205	743	12.2	4.8	21.2
40-44	4,617	2,367	2,250	240	44	196	5.2	1.9	8.7
45-49	4,216	2,265	1,951	392	49	343	9.3	2.2	17.6
50-54	3,325	1,709	1,616	196	49	147	5.9	2.9	9.1
55-59	3,040	1,604	1,436	369	222	147	12.1	13.8	10.2
60-64	1,153	723	430	49		49	4.2	-	11.4
65 +	1,987	1,215	772	271	182	89	13.6	15.0	11.5
ADDIS ABAE	BA CITY ADMIN	NISTRATIO	N						
All Ages	1,625,468	839,892	785,576	374,626	130,686	243,940	23.0	15.6	31.1
10-14	12,300	3,582	8,718	2,423	1,384	1,038	19.7	38.6	11.9
15-19	114,076	29,832	84,244	25,488	8,923	16,564	22.3	29.9	19.7
20-24	273,481	117,270	156,211	93,482	35,314	58,168	34.2	30.1	37.2
25-29	365,840	183,509	182,330	87,535	28,830	58,705	23.9	15.7	32.2
30-34	244,892	139,844	105,048	54,507	18,026	36,481	22.3	12.9	34.7
35-39	204,612	115,698	88,914	33,520	9,473	24,046	16.4	8.2	27.0
40-44	126,925	75,899	51,025	22,412	5,192	17,220	17.7	6.8	33.7
45-49	95,390	51,268	44,122	17,560	5,445	12,115	18.4	10.6	27.5
50-54	78,907	45,315	33,592	13,738	5,176	8,562	17.4	11.4	25.5
55-59	41,798	28,407	13,392	8,835	4,331	4,504	21.1	15.2	33.6
60-64				7,770					
65+	31,524 35,723	19,721 29,547	11,804 6,176	7,770	2,273 6,318	5,497 1,038	24.6 20.6	11.5 21.4	46.6 16.8
	ADMINISTRAT		3,	.,,,,,	3,0.0	.,000			
All Ages	131,820	64,975	66,844	29,988	10,291	19,697	22.7	15.8	29.5
71171g00 10-14	4,113	2,108	2,005	348	87	262	8.5	4.1	13.1
15-1 4 15-19	12,235	4,213	8,021	3,132	1,320	1,812	25.6	31.3	22.6
20-24	21,880	10,836	11,044	7,164	2,820	4,345	32.7	26.0	39.3
20-2 4 25-29	24,798	12,645	12,153	5,599	1,649	3,950	22.6	13.0	32.5
20-29 30-34	16,338	8,493	7,845	3,735	1,545	2,189	22.9	18.2	27.9
35-3 4 35-39	15,999	8,065	7,845	2,201	440	1,762	13.8	5.5	22.2
40-44 45-40	8,566	4,075	4,491	1,888	626	1,262	22.0	15.4	28.1
45-49 50.54	11,248	5,292	5,956	2,648	523	2,125	23.5	9.9	35.7
50-54 55-50	6,962	3,487	3,475	1,109	173	935	15.9	5.0	26.9
55-59 60-64	4,785	2,895	1,890	1,118	588	529	23.4	20.3	28.0
60-64	2,274	1,224	1,049	783	434	350	34.4	35.5	33.4
65+	2,622	1,641	981	263	87	176	10.0	5.3	17.9

Table 6.2 Currently Unemployed Populations of Major Towns Aged Ten Years and Above by Sex, Age Group and Unemployment Rate: 2012

Age **Economically Active** Unemployed **Unemployment Rate** Group Male **Female** Total Male Female Total Male **Female** Total **COUNTRY TOTAL** All Ages 2,817,517 1,438,991 1,378,526 602,000 197,663 404,337 21.4 13.7 29.3 10-14 41,468 15,749 25,719 5,211 2,694 2,517 12.6 17.1 9.8 15-19 227,469 70,547 156,922 51,976 18,833 33,143 22.8 26.7 21.1 20-24 499,149 215,737 283,412 158,256 54,604 103,651 31.7 25.3 36.6 25-29 617,504 308,561 308,943 140,145 41,907 98,238 22.7 13.6 31.8 30-34 392,860 221,442 171,419 78,496 24,650 53,845 20.0 11.1 31.4 35-39 347,896 193,375 154,522 51,975 13,769 38,206 14.9 7.1 24.7 40-44 214,079 126,395 87,684 33,503 8,522 24,981 15.6 6.7 28.5 45-49 166,298 91,124 75,174 27,668 7,893 19,775 16.6 8.7 26.3 50-54 126,914 70,804 56,110 20,300 6,496 13,804 16.0 9.2 24.6 55-59 69,229 45,173 24,056 12,865 6,074 6,791 18.6 13.4 28.2 60-64 52,784 33,314 19,470 11,251 4,000 7,251 21.3 12.0 37.2 65+ 61,867 46,770 15,097 10,355 8,220 2,135 16.7 17.6 14.1 **MEKELE TOWN** 73,787 77,702 All Ages 151,489 29,944 7,951 21,993 19.8 10.8 28.3 10-14 2,150 436 1,714 9,074 2,908 9.5 15-19 14,423 5,349 2,042 866 20.2 38.2 20-24 28,167 10,314 17,853 8,808 2,162 6,646 31.3 21.0 37.2 14,374 25-29 35,278 20,904 10,268 2,142 8,127 29.1 14.9 38.9 30-34 18,804 10,775 8,029 2,914 2,486 428 15.5 4.0 31.0 35-39 20,178 12,425 7,752 2,257 634 1,622 11.2 5.1 20.9 40-44 12,710 8,595 4,115 543 322 221 4.3 3.7 5.4 45-49 6,556 3,312 3,244 1,057 1,057 16.1 32.6 2,041 338 3.5 50-54 5,214 3,173 111 227 6.5 11.1 1,326 1,163 307 307 55-59 2,489 12.3 26.4 60-64 1,505 2,253 748 215 215 9.5 28.7 65+ 3,268 2,202 1,066 327 109 218 10.0 5.0 20.5 **ASAYITA TOWN** 9,218 4,910 4,309 849 286 564 9.2 5.8 13.1 All Ages 10-14 251 108 143 15-19 756 289 467 108 24 84 14.3 8.3 18.0 20-24 1,511 695 816 288 72 215 19.1 26.3 10.4 2,064 107 25-29 1,171 892 59 48 5.2 5.0 5.4 96 30-34 1,136 551 12 2.0 15.2 586 84 8.5 35-39 1,081 491 107 48 60 9.9 12.2 589 8.1 40-44 634 370 264 47 23 24 7.4 6.2 9.1 45-49 599 372 227 36 12 24 6.0 3.2 10.6 50-54 421 168 254 25 25 5.9 9.8 252 156 96 12 12 4.8 7.7 55-59 60-64 228 156 72 12 12 5.3 7.7 65+ 286 214 72 12 12 4.2 5.6

Table 6.2 Cont'd

	Econo	mically Act	ive	U	nemploye	d	Unemployment Rate		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
GONDER TOV	VN								
All Ages	116,260	57,528	58,732	18,392	5,780	12,612	15.8	10.0	21.5
10-14	8,532	4,421	4,111	608	512	96	7.1	11.6	2.3
15-19	12,218	4,987	7,232	2,381	912	1,469	19.5	18.3	20.3
20-24	20,895	7,335	13,560	5,419	633	4,785	25.9	8.6	35.3
25-29	20,998	11,917	9,081	3,498	1,646	1,853	16.7	13.8	20.4
30-34	12,277	6,266	6,011	2,018	910	1,108	16.4	14.5	18.4
35-39	11,429	5,836	5,594	1,119	201	919	9.8	3.4	16.4
40-44	8,285	4,442	3,843	1,556	293	1,263	18.8	6.6	32.9
45-49	7,087	3,832	3,255	720	475	245	10.2	12.4	7.5
50-54	4,244	2,276	1,969	505	123	382	11.9	5.4	19.4
55-59	2,994	1,553	1,440	247	-	247	8.2	-	17.2
60-64	3,415	1,913	1,502	321	75	245	9.4	3.9	16.3
65+	3,885	2,750	1,135	-	-	-	-	-	-
DESSIE TOW	N								
All Ages	61,298	30,546	30,752	13,755	4,136	9,618	22.4	13.5	31.3
10-14	796	54	742	92	-	92	11.6	-	12.4
15-19	6,453	2,471	3,981	2,218	582	1,636	34.4	23.6	41.1
20-24	12,466	6,152	6,314	4,268	1,494	2,774	34.2	24.3	43.9
25-29	11,229	5,670	5,559	2,480	630	1,850	22.1	11.1	33.3
30-34	6,480	3,821	2,659	842	421	421	13.0	11.0	15.8
35-39	7,891	3,767	4,124	1,398	309	1,089	17.7	8.2	26.4
40-44	3,986	2,321	1,665	616	155	461	15.5	6.7	27.7
45-49	3,367	1,748	1,619	640	122	518	19.0	7.0	32.0
50-54	3,205	1,124	2,081	586	54	532	18.3	4.8	25.6
55-59	1,662	964	698	353	154	198	21.2	16.0	28.4
60-64	1,830	1,379	451	217	170	47	11.9	12.3	10.4
65+	1,934	1,076	858	46	46	-	2.4	4.3	-
BAHIR DAR T	OWN								
All Ages	115,388	52,479	62,909	21,606	5,672	15,934	18.7	10.8	25.3
10-14	1,961	901	1,060	348	146	202	17.7	16.2	19.1
15-19	12,624	2,856	9,768	2,436	547	1,889	19.3	19.2	19.3
20-24	28,442	9,277	19,165	8,633	2,436	6,197	30.4	26.3	32.3
25-29	26,019	13,804	12,216	4,795	792	4,003	18.4	5.7	32.8
30-34	12,720	6,178	6,541	1,902	214	1,688	15.0	3.5	25.8
35-39	12,033	6,151	5,882	983	339	644	8.2	5.5	10.9
40-44	6,652	4,313	2,339	586	405	181	8.8	9.4	7.7
45-49	5,493	3,050	2,444	632	150	482	11.5	4.9	19.7
50-54	4,092	2,193	1,899	587	103	484	14.3	4.7	25.5
55-59	2,057	1,362	695	-	-	-	-	-	-
60-64	1,710	1,177	533	439	275	164	25.7	23.4	30.8
65+	1,584	1,217	367	266	266	-	16.8	21.9	-

Table 6.2 Cont'd

Age Group	Econor	nically Acti	ve	Uı	nemploye	d	Uner	Unemployment Rate		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female	
BISHOFTU TO	OWN									
All Ages	56,377	28,582	27,795	15,052	4,901	10,151	26.7	17.1	36.5	
10-14	499	286	213	86	46	40	17.2	16.1	18.8	
15-19	4,089	1,443	2,646	1,635	503	1,131	40.0	34.9	42.7	
20-24	11,209	5,662	5,547	4,047	1,606	2,440	36.1	28.4	44.0	
25-29	12,513	6,390	6,124	3,573	1,092	2,481	28.6	17.1	40.5	
30-34	7,561	4,653	2,907	1,364	412	952	18.0	8.9	32.7	
35-39	7,475	3,551	3,924	1,620	386	1,233	21.7	10.9	31.4	
40-44	3,540	1,655	1,885	649	142	507	18.3	8.6	26.9	
45-49	2,992	1,441	1,550	563	114	449	18.8	7.9	29.0	
50-54	2,384	992	1,392	583	164	419	24.5	16.5	30.1	
55-59	1,425	805	620	495	139	356	34.7	17.3	57.4	
60-64	1,052	595	457	192	96	96	18.3	16.1	21.0	
65+	1,639	1,109	530	247	201	46	15.1	18.1	8.7	
ADAMA TOW	N									
All Ages	123,171	63,334	59,837	29,899	8,097	21,802	24.3	12.8	36.4	
10-14	1,152	477	675	427	300	128	37.1	62.9	19.0	
15-19	13,169	5,844	7,325	4,386	1,885	2,502	33.3	32.3	34.2	
20-24	22,766	10,509	12,257	7,106	1,701	5,405	31.2	16.2	44.1	
25-29	29,002	12,947	16,055	7,931	1,341	6,590	27.3	10.4	41.0	
30-34	16,463	8,982	7,480	3,256	663	2,592	19.8	7.4	34.7	
35-39	14,967	9,062	5,905	2,447	764	1,683	16.3	8.4	28.5	
40-44	8,341	4,858	3,482	1,406	498	908	16.9	10.3	26.1	
45-49	7,168	4,741	2,427	872	183	689	12.2	3.9	28.4	
50-54	3,957	2,075	1,882	776	186	590	19.6	9.0	31.3	
55-59	1,990	1,267	723	461	267	195	23.2	21.1	27.0	
60-64	1,698	842	857	225	-	225	13.3	-	26.3	
65+	2,499	1,729	770	605	309	295	24.2	17.9	38.3	
JIMMA TOWN	I									
All Ages	70,819	37,099	33,720	11,711	3,449	8,263	16.5	9.3	24.5	
10-14	2,362	852	1,510	90	44	46	3.8	5.2	3.0	
15-19	7,114	2,535	4,579	1,870	519	1,351	26.3	20.5	29.5	
20-24	14,210	7,079	7,131	3,450	1,332	2,118	24.3	18.8	29.7	
25-29	14,050	7,832	6,218	2,190	739	1,451	15.6	9.4	23.3	
30-34	8,492	4,688	3,804	1,246	170	1,075	14.7	3.6	28.3	
35-39	8,671	4,662	4,008	846	-	846	9.8	-	21.1	
40-44	4,729	2,660	2,068	589	175	415	12.5	6.6	20.1	
45-49	3,903	2,413	1,490	362	50	311	9.3	2.1	20.9	
50-54	2,838	1,685	1,152	349	53	295	12.3	3.1	25.6	
55-59	1,845	1,226	619	281	120	161	15.2	9.8	26.0	
60-64	1,090	754	336	155	114	41	14.2	15.1	12.2	
65+	1,516	712	804	284	131	153	18.7	18.4	19.0	

Table 6.2	Cont'd

A O	Econo	Uı	d	Unemployment Rate					
Age Group	Total	Male	Female	Total	Male	Female	Total Male		Female
SHASHEMENE TOWN									
All Ages	60,565	32,633	27,932	14,606	4,098	10,508	24.1	12.6	37.6
10-14	1,182	526	655	234	97	137	19.8	18.4	20.9
15-19	5,612	2,558	3,055	1,793	494	1,300	31.9	19.3	42.6
20-24	10,845	5,481	5,364	3,464	940	2,524	31.9	17.2	47.1
25-29	14,343	7,111	7,232	4,017	997	3,020	28.0	14.0	41.8
30-34	8,968	5,018	3,949	1,660	378	1,282	18.5	7.5	32.5
35-39	7,193	4,217	2,977	1,404	509	895	19.5	12.1	30.1
40-44	4,912	2,904	2,007	745	99	646	15.2	3.4	32.2
45-49	2,873	1,750	1,123	387	128	259	13.5	7.3	23.1
50-54	1,972	1,079	893	477	126	352	24.2	11.7	39.4
55-59	803	630	174	74	39	35	9.2	6.2	20.1
60-64	1,076	786	290	299	239	60	27.8	30.4	20.7
65+	785	573	212	52	52	-	6.6	9.1	-
JIJIGA TOWN									
All Ages	50,418	26,548	23,870	9,476	2,558	6,918	18.8	9.6	29.0
10-14	1,355	420	935	225	51	174	16.6	12.1	18.6
15-19	3,631	1,411	2,220	713	298	415	19.6	21.1	18.7
20-24	7,405	2,860	4,546	2,617	462	2,155	35.3	16.2	47.4
25-29	9,792	4,852	4,940	1,973	496	1,477	20.1	10.2	29.9
30-34	6,865	3,820	3,045	957	243	714	13.9	6.4	23.4
35-39	7,080	4,074	3,006	1,377	394	983	19.4	9.7	32.7
40-44	4,990	3,145	1,844	602	52	550	12.1	1.7	29.8
45-49	3,934	2,775	1,158	422	211	211	10.7	7.6	18.2
50-54	2,202	1,144	1,058	253	102	151	11.5	8.9	14.3
55-59	1,326	895	430	50	50	-	3.8	5.6	-
60-64	892	562	330	197	109	89	22.1	19.4	27.0
65+	947	590	357	88	88	-	9.3	14.9	-
ASOSA TOWN	I								
All Ages	20,119	10,962	9,157	2,189	803	1,385	10.9	7.3	15.1
10-14	744	345	398	-	-	-	-	-	-
15-19	2,416	731	1,685	570	231	339	23.6	31.6	20.1
20-24	3,698	1,782	1,916	609	185	425	16.5	10.4	22.2
25-29	4,195	2,339	1,855	355	165	190	8.5	7.1	10.2
30-34	2,864	1,597	1,267	256	23	233	8.9	1.4	18.4
35-39	2,679	1,767	912	147	66	80	5.5	3.7	8.8
40-44	1,417	1,029	387	70	23	47	4.9	2.2	12.1
45-49	779	640	139	63	24	40	8.1	3.8	28.8
50-54	506	274	232	31	-	31	6.1	-	13.4
55-59	399	199	200	33	33	-	8.3	16.6	-
60-64	195	115	80	-	-	-	-	-	-
65+	229	142	87	53	53	-	23.1	37.3	-

Table 6.2 Cont'd

Age Group	Economically Active			U	d	Unemployment Rate			
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
HAWASSA TOWN									
All Ages	149,173	77,073	72,100	21,428	6,539	14,889	14.4	8.5	20.7
10-14	3,294	762	2,532	302	-	302	9.2	-	11.9
15-19	14,239	4,570	9,669	1,648	376	1,272	11.6	8.2	13.2
20-24	30,677	15,675	15,002	6,862	2,701	4,161	22.4	17.2	27.7
25-29	32,253	15,950	16,303	4,224	970	3,253	13.1	6.1	20.0
30-34	17,768	10,983	6,785	2,349	848	1,501	13.2	7.7	22.1
35-39	15,939	7,559	8,379	1,414	-	1,414	8.9	-	16.9
40-44	12,292	7,007	5,285	1,476	471	1,005	12.0	6.7	19.0
45-49	9,549	5,560	3,990	1,315	408	907	13.8	7.3	22.7
50-54	6,122	3,695	2,427	746	77	670	12.2	2.1	27.6
55-59	2,051	1,679	373	229	117	112	11.2	7.0	30.0
60-64	2,276	1,770	506	376	204	172	16.5	11.5	34.0
65+	2,713	1,865	849	487	368	119	18.0	19.7	14.0
GAMBELLA T	OWN								
All Ages	18,992	9,656	9,336	1,506	330	1,176	7.9	3.4	12.6
10-14	221	110	111	27	27	-	12.2	24.5	-
15-19	995	250	745	143	27	116	14.4	10.8	15.6
20-24	3,417	1,065	2,352	448	55	393	13.1	5.2	16.7
25-29	4,946	2,675	2,271	383	110	273	7.7	4.1	12.0
30-34	2,666	1,585	1,081	241	111	131	9.0	7.0	12.1
35-39	2,871	1,653	1,218	188	-	188	6.5	-	15.4
40-44	1,485	753	731	75	-	75	5.1	-	10.3
45-49	1,145	665	480	-	-	-	-	-	-
50-54	563	414	149	-	-	-	-	-	-
55-59	314	206	108	-	-	-	-	-	-
60-64	119	92	27	-	-	-	-	-	-
65+	250	188	62	-	-	-	-	-	-
HARAR TOW	N								
All Ages	56,942	28,987	27,955	6,973	2,086	4,887	12.2	7.2	17.5
10-14	557	325	231	-	-	-	-	-	-
15-19	3,418	1,209	2,210	548	151	398	16.0	12.5	18.0
20-24	8,080	3,746	4,334	1,591	692	899	19.7	18.5	20.7
25-29	10,184	5,375	4,809	1,218	248	970	12.0	4.6	20.2
30-34	8,567	4,151	4,417	1,152	245	907	13.4	5.9	20.5
35-39	7,798	4,298	3,500	948	205	743	12.2	4.8	21.2
40-44	4,617	2,367	2,250	240	44	196	5.2	1.9	8.7
45-49	4,216	2,265	1,951	392	49	343	9.3	2.2	17.6
50-54	3,325	1,709	1,616	196	49	147	5.9	2.9	9.1
55-59	3,040	1,604	1,436	369	222	147	12.1	13.8	10.2
60-64	1,153	723	430	49	-	49	4.2	-	11.4
65+	1,987	1,215	772	271	182	89	13.6	15.0	11.5

Table 6.2 Cont'd

Table 6.2	Cont'd								
Age Group	Economically Active			U	Unemployment Rate				
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
ADDIS ABAB ADMINISTRA	-								
All Ages	1,625,468	839,892	785,576	374,626	130,686	243,940	23.0	15.6	31.1
10-14	12,300	3,582	8,718	2,423	1,384	1,038	19.7	38.6	11.9
15-19	114,076	29,832	84,244	25,488	8,923	16,564	22.3	29.9	19.7
20-24	273,481	117,270	156,211	93,482	35,314	58,168	34.2	30.1	37.2
25-29	365,840	183,509	182,330	87,535	28,830	58,705	23.9	15.7	32.2
30-34	244,892	139,844	105,048	54,507	18,026	36,481	22.3	12.9	34.7
35-39	204,612	115,698	88,914	33,520	9,473	24,046	16.4	8.2	27.0
40-44	126,925	75,899	51,025	22,412	5,192	17,220	17.7	6.8	33.7
45-49	95,390	51,268	44,122	17,560	5,445	12,115	18.4	10.6	27.5
50-54	78,907	45,315	33,592	13,738	5,176	8,562	17.4	11.4	25.5
55-59	41,798	28,407	13,392	8,835	4,331	4,504	21.1	15.2	33.6
60-64	31,524	19,721	11,804	7,770	2,273	5,497	24.6	11.5	46.6
65+	35,723	29,547	6,176	7,356	6,318	1,038	20.6	21.4	16.8
DIRE DAWA 1	OWN								
All Ages	131,820	64,975	66,844	29,988	10,291	19,697	22.7	15.8	29.5
10-14	4,113	2,108	2,005	348	87	262	8.5	4.1	13.1
15-19	12,235	4,213	8,021	3,132	1,320	1,812	25.6	31.3	22.6
20-24	21,880	10,836	11,044	7,164	2,820	4,345	32.7	26.0	39.3
25-29	24,798	12,645	12,153	5,599	1,649	3,950	22.6	13.0	32.5
30-34	16,338	8,493	7,845	3,735	1,545	2,189	22.9	18.2	27.9
35-39	15,999	8,065	7,935	2,201	440	1,762	13.8	5.5	22.2
40-44	8,566	4,075	4,491	1,888	626	1,262	22.0	15.4	28.1
45-49	11,248	5,292	5,956	2,648	523	2,125	23.5	9.9	35.7
50-54	6,962	3,487	3,475	1,109	173	935	15.9	5.0	26.9
55-59	4,785	2,895	1,890	1,118	588	529	23.4	20.3	28.0
60-64	2,274	1,224	1,049	783	434	350	34.4	35.5	33.4
65+	2,622	1,641	981	263	87	176	10.0	5.3	17.9

Table 6.3 Currently Unemployed Population of Urban Areas Aged Ten Years and Above by Age Group, S and Educational Attainment : 2012

and Educational Attainment : 2012										
	Total Unemployed Persons	Literate								
Age Group and Sex		Non Formal	Pre-school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate Completed	Diplom Comple	
COUNTRY TOTAL										
Total	1,218,574	14,845	5,024	408,959	104,955	279,438	42,464	35,314	105,84	
Male	410,381	5,720	355	126,308	46.542	108.597	19,296	9,608	45,50	
Female 10-14	808,193	9,125	4,669	282,650	58,412	170,841	23,168	25,706	60,34	
Total	24,618	_	1,345	19,127	210	25	_	_	_	
Male	11,787	_	314	10,877	-	-	_	_	_	
Female 15-19	12,831	-	1,031	8,250	210	25	-	-	-	
Total	136,069	210	346	57,147	12,045	41,911	3,672	6,878	6,488	
Male	51,396	210	-	17,545	5,948	19,040	1,636	2,137	3,377	
Female 20-24	84,673	-	346	39,602	6,098	22,872	2,036	4,741	3,112	
Total	361,492	12	97	101,443	25,342	92,187	21,445	15,014	57,90	
Male	127,407	12	-	24,955	12,297	39,250	10,173	2,944	24,67	
Female 25-29	234,085	-	97	76,488	13,046	52,937	11,272	12,070	33,22	
Total	268,787	908	2,551	89,877	25,762	61,321	11,758	9,972	20,86	
Male	79,619	319	-	24,935	8,298	19,069	4,964	2,317	9,120	
Female 30-34	189,168	588	2,551	64,942	17,464	42,253	6,794	7,655	11,74	
Total	128,378	717	77	42,659	15,377	31,774	3,738	1,737	8,441	
Male	39,484	454	-	12,645	6,739	9,571	1,748	1,223	2,841	
Female 35-39	88,893	263	77	30,014	8,637	22,202	1,991	513	5,600	
Total	98,500	2,095	102	30,796	11,202	23,211	727	1,161	4,714	
Male	26,203	509	-	8,234	3,912	7,222	204	648	1,396	
Female	72,297	1,587	102	22,561	7,290	15,989	522	513	3,318	

Table 6.3 Cont'd

Table 0.5 Com												
	Total	Literate										
Age Group and Sex	Unemployed Persons	Non Formal	Pre-school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate Completed	Diplom Complet			
40-44									<u>, </u>			
-	50.070	4 004		40.000	F 404	45.000	604		4 0 40			
Total	58,078	1,901	-	18,688	5,464	15,036	624	-	1,349			
Male	16,589	934	-	4,510	2,364	4,977	72	-	109			
Female 45-49	41,489	967	-	14,179	3,100	10,059	552	-	1,239			
Total	43,186	1,591	-	15,718	3,074	5,598	-	213	1,498			
Male	12,716	-	-	5,173	1,386	3,255	-	-	895			
Female	30,470	1,591	-	10,545	1,687	2,343	-	213	603			
50-54												
Total	36,225	2,799	-	11,440	3,407	3,454	-	287	1,586			
Male	12,971	401	-	4,270	2,574	2,181	-	287	432			
Female	23,254	2,397	-	7,170	833	1,272	-	-	1,155			
55-59												
Total	24,471	1,527	41	8,962	761	2,657	-	50	465			
Male	9,677	805	41	4,383	713	1,770	-	50	119			
Female	14,794	722	-	4,578	47	887	-	-	346			
60-64												
Total	18,566	1,551	-	7,236	449	667	153	-	1,667			
Male	8,023	865	-	3,348	449	667	153	-	1,667			
Female	10,544	687	-	3,888	-	-	-	-	-			
65+												
Total	20,205	1,534	465	5,866	1,863	1,597	346	-	868			
Male	14,510	1,211	-	5,433	1,863	1,597	346	-	868			
Female	5,694	323	465	433	-	-	-	-	-			

Table 6.4 Currently Unemployed Population of Urban Areas Aged Ten Years and Above by Region, Sex a Educational Attainment: 2012

Laucane	mai Attailine	All Literate								
	Total				C	All	Literate			
Region and Sex	unemployed	Non	Pre-	Grades	Secondary not	Secondary		Certificate	Dipl	
	persons	Formal	school	1-8	completed		Preparatory	Completed		
	ролосия	1 Office	SCHOOL	1-0	completed	Completed	rieparatory	Completed	Comp	
COUNTRY TOTAL										
Total	1,218,574	14,845	5,024	408,959	104,955	279,438	42,464	35,314	105	
Male	410,381	5,720	355	126,308	46,542	108,597	19,296	9,608	45,	
Female	808,193	9,125	4,669	282,650	58,412	170,841	23,168	25,706	60,	
TIGRAY REGION										
Total	94,756	1,190	-	33,725	5,462	25,551	4,707	1,655	4,7	
Male	25,996	412	-	8,233	1,521	8,042	3,016	218	1,9	
Female	68,761	778	-	25,492	3,941	17,509	1,691	1,438	2,8	
AFFAR REGION										
Total	9,846	526	-	3,676	406	1,192	24	85	51	
Male	2,425	36	-	1,434	49	156	12	61	22	
Female	7,421	491	-	2,242	357	1,037	13	24	28	
AMHARA REGION										
Total	209,533	4,110	77	65,627	13,023	46,529	6,167	9,225	18,	
Male	68,229	1,865	-	20,697	6,201	15,404	2,890	1,695	6,4	
Female	141,304	2,245	77	44,931	6,822	31,124	3,277	7,531	12,	
OROMIA REGION										
Total	330,723	2,340	2,755	119,082	23,941	70,570	9,790	8,855	37,	
Male	115,759	1,342	-	38,174	11,734	32,182	3,395	2,407	16,	
Female	214,964	998	2,755	80,907	12,207	38,388	6,395	6,448	21,	
SOMALI REGION										
Total	34,427	681	629	11,592	1,939	1,718	357	50	25	
Male	12,124	367	314	5,276	980	814	50	50		
Female	22,302	314	314	6,315	959	904	306	-	25	
BENISHANGUL										
GUMUZ REGION										
Total	5,505	-	-	2,175	500	1,018	273	80	37	
Male	1,897	-	-	680	204	574	137	-	15	
Female	3,608	-	-	1,495	297	444	136	80	22	

Table 6.4 Cont'd

		All Literate											
Region and Sex	Total unemployed persons	Non Formal	Pre- school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate Completed	Dip Com				
SNNP REGION													
Total	119,176	786	-	46,110	6,621	26,141	3,151	3,786	12,				
Male	40,404	210	-	13,971	2,733	11,547	1,051	1,262	5,6				
Female GAMBELLA REGION	78,771	576	-	32,139	3,888	14,594	2,100	2,524	7,1				
Total	3,020	_	-	1,470	276	544	_	29	5				
Male	484	-	-	235	47	81	-	-	•				
Female HARARI REGION	2,536	-	-	1,236	228	463	-	29	5				
Total	6,973	147	41	2,206	726	1,086	288	304	9				
Male	2,086	-	41	577	331	378	101	157	5				
Female ADDIS ABABA CITY ADMINISTRATION	4,887	147	-	1,628	395	707	187	147	4				
Total	374,626	4,098	1,425	112,492	50,051	99,931	16,996	10,639	27,				
Male	130,686	1,053	-	33,727	21,974	37,148	8,380	3,579	13,				
Female DIRE DAWA ADMINISTRATION	243,940	3,045	1,425	78,765	28,077	62,784	8,616	7,060	13,				
Total	29,988	967	97	10,803	2,009	5,157	710	604	2,0				
Male	10,291	435	-	3,304	768	2,271	264	179	9				
Female	19,697	532	97	7,499	1,241	2,886	446	425	1,0				

Table 6.5 Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Region, Second Training: 2012

Training status and Types of Training Total Tigray Afar Amhara Oromia Somalie Benishangul Gumuz SNNPR GA	Training status and						Regi	on		
All Persons Total 1,218,574 94,756 9,846 209,533 330,723 34,427 5,505 119,176 Male 410,381 25,996 2,425 68,229 115,759 12,124 1,897 40,404 Female 808,193 68,761 7,421 141,304 214,964 22,302 3,609 78,771 Not Trained 710tal 922,871 71,907 9,151 159,426 246,962 32,012 4,846 94,733 Male 278,468 17,309 2,078 47,810 76,083 10,842 1,612 29,790 Female 644,403 54,598 7,073 111,615 170,879 21,170 3,234 64,943 Trained 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated 710,879 7								Benishangul-		
Total Male 1,218,574 94,756 9,846 209,533 330,723 34,427 5,505 119,176 Male 410,381 25,996 2,425 68,229 115,759 12,124 1,897 40,404	Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	G
Male Female 410,381 25,996 2,425 68,229 115,759 12,124 1,897 40,404 Female 808,193 68,761 7,421 141,304 214,964 22,302 3,609 78,771 Not Trained Total 922,871 71,907 9,151 159,426 246,962 32,012 4,846 94,733 Male 278,468 17,309 2,078 47,810 76,083 10,842 1,612 29,790 Female 644,403 54,598 7,073 111,615 170,879 21,170 3,234 64,943 Trained 10tal 295,703 22,849 695 50,107 83,761 2,414 659 24,443 Male 131,913 8,686 347 20,419 39,676 1,282 285 10,614 Female 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated 7 - - - - </td <td>All Persons</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	All Persons									
Female 808,193 68,761 7,421 141,304 214,964 22,302 3,609 78,771 Not Trained	Total	1,218,574	94,756	9,846	209,533	330,723	34,427	5,505	119,176	
Not Trained	Male	410,381	25,996	2,425	68,229	115,759	12,124	1,897	40,404	
Total 922,871 71,907 9,151 159,426 246,962 32,012 4,846 94,733 Male 278,468 17,309 2,078 47,810 76,083 10,842 1,612 29,790 Female 644,403 54,598 7,073 111,615 170,879 21,170 3,234 64,943 Trained 295,703 22,849 695 50,107 83,761 2,414 659 24,443 Male 131,913 8,686 347 20,419 39,676 1,282 285 10,614 Female 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated 7 - <t< td=""><td>Female</td><td>808,193</td><td>68,761</td><td>7,421</td><td>141,304</td><td>214,964</td><td>22,302</td><td>3,609</td><td>78,771</td><td></td></t<>	Female	808,193	68,761	7,421	141,304	214,964	22,302	3,609	78,771	
Male Female 278,468 or 17,309 or 17,309 or 17,309 or 111,615 or 170,879 or 17,309 or	Not Trained									
Female Trained 644,403 54,598 7,073 111,615 170,879 21,170 3,234 64,943 Trained 295,703 22,849 695 50,107 83,761 2,414 659 24,443 Male 131,913 8,686 347 20,419 39,676 1,282 285 10,614 Female 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated -	Total	922,871	71,907	9,151	159,426	246,962	32,012	4,846	94,733	
Trained Total 295,703 22,849 695 50,107 83,761 2,414 659 24,443 Male 131,913 8,686 347 20,419 39,676 1,282 285 10,614 Female 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated Total	Male	278,468	17,309	2,078	47,810	76,083	10,842	1,612	29,790	
Total 295,703 22,849 695 50,107 83,761 2,414 659 24,443 Male 131,913 8,686 347 20,419 39,676 1,282 285 10,614 Female 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated Total - </td <td>Female</td> <td>644,403</td> <td>54,598</td> <td>7,073</td> <td>111,615</td> <td>170,879</td> <td>21,170</td> <td>3,234</td> <td>64,943</td> <td></td>	Female	644,403	54,598	7,073	111,615	170,879	21,170	3,234	64,943	
Male Female 131,913 8,686 347 20,419 39,676 1,282 285 10,614 Female Female 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated 7 color 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated 7 color 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated 7 color 14 2	Trained									
Female Not Stated 163,790 14,163 348 29,689 44,085 1,132 374 13,829 Not Stated Total -<	Total	295,703	22,849	695	50,107	83,761	2,414	659	24,443	
Not Stated Total	Male	131,913	8,686	347	20,419	39,676	1,282	285	10,614	
Total - <td></td> <td>163,790</td> <td>14,163</td> <td>348</td> <td>29,689</td> <td>44,085</td> <td>1,132</td> <td>374</td> <td>13,829</td> <td></td>		163,790	14,163	348	29,689	44,085	1,132	374	13,829	
Male -	Not Stated									
Female 1 2 2 2 2 2 2 2 2 7 101 8 8 9 102 101 8 9 101 101 8 9 101	Total	-	-	-	-	-	-	-	-	
Types of Training Agriculture Total 3,002 529 176 704 529 50 - 101 Male 2,112 206 176 654 489 50 - 101 Female 890 323 - 50 40 - - - Water and Soil Conservation Total 109 - - 34 - - - Male - - - 34 - - - - Female 109 - - - 34 - - - - Female 109 - - - 34 - - - - Forestry Science - - - 34 -	Male	-	-	-	-	-	-	-	-	
Agriculture Total 3,002 529 176 704 529 50 - 101 Male 2,112 206 176 654 489 50 - 101 Female 890 323 - 50 40 - - - Water and Soil Conservation - - 50 40 - - - - Total 109 - - - 34 - - - - Female 109 - - - 34 - - - - Forestry Science - - - - 34 -<		-	-	-	-	-	-	-	-	
Total 3,002 529 176 704 529 50 - 101 Male 2,112 206 176 654 489 50 - 101 Female 890 323 - 50 40 - - - Water and Soil Conservation Total 109 - - - 34 - - - 75 Male - - - - - 34 - <td< td=""><td>Types of Training</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	Types of Training									
Male 2,112 206 176 654 489 50 - 101 Female 890 323 - 50 40 - - - Water and Soil Conservation Total 109 - - - 34 - - 75 Male - - - - 34 - - 75 Forestry Science - - - 34 - - 75 Forestry Science - - - 34 - - - 75 Male -	Agriculture									
Female Water and Soil Conservation 890 323 - 50 40 - - - Total Total 109 - - - 34 - - 75 Male	Total	3,002	529	176	704	529	50	-	101	
Water and Soil Conservation Total 109 - - - 34 - - 75 Male -	Male	2,112	206	176	654	489	50	-	101	
Total 109 - - - 34 - - 75 Male -			323	-	50	40	-	-	-	
Male -	Water and Soil Conserva	ation								
Female 109 - - - 34 - - 75 Forestry Science Total -	Total	109	-	-	-	34	-	-	75	
Forestry Science Total - <td>Male</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td></td>	Male	-	-	-	-	-	-	-	-	
Total - <td>Female</td> <td>109</td> <td>-</td> <td>-</td> <td>-</td> <td>34</td> <td>-</td> <td>-</td> <td>75</td> <td></td>	Female	109	-	-	-	34	-	-	75	
Male -	Forestry Science									
Female	Total	-	-	-	-	-	-	-	-	
General medicine, Ophthalmology Pathology, Internal medicine, Gynecology, Surgery, Anesthesiology Total 318 276 42 Male 318 276 42	Male	-	-	-	-	-	-	-	-	
Total 318 276 42 Male 318 276 42	Female	-	-	-	-	-	-	-	-	
Male 318 276 42	General medicine, Ophtha	almology Path	nology, Interna	al medicin	e, Gynecol	ogy, Surge	ry, Anesth	esiology		
	Total	318	-	-	276	42	-	-	-	
Female	Male	318	-	-	276	42	-	-	-	
	Female	-	-	-	-	-	-	-	-	

Table 6.5 Cont'd

Table 6.5 Cont d									
Training status and	,					Regi			
Types of Training					_		Benishangul-		
	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	G
Medical Laboratory / Radio	••								
Total	1,257	-	-	539	42	-	-	-	
Male	1,147	-	-	471	-	-	-	-	
Female	110	-	-	67	42	-	-	-	
Pharmacy/Pharmacy Te	chnician								
Total	842	-	-	171	50	-	-	179	
Male	261	-	-	171	-	-	-	90	
Female	581	-	-	-	50	-	-	90	
Sanitary Science									
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Clinical Science/Nursing									
Total	1,849	-	-	365	226	52	-	597	
Male	474	-	-	112	-	-	-	275	
Female	1,376	-	-	253	226	52	-	322	
Public health Nurse, Health	officer / Sani	tarian/Post He	alth Office	er					
Total	1,523	281	-	-	190	-	-	-	
Male	717	281	-	-	77	-	-	-	
Female	806	-	-	-	113	-	-	-	
Personal Business Manage	ement								
Total	3,559	361	-	107	147	-	-	304	
Male	1,232	-	-	-	53	-	-	210	
Female	2,327	361	-	107	94	-	-	94	
Economics	,								
Total	1,874	-	-	758	181	51	-	94	
Male	945	-	-	758	50	51	-	-	
Female	929	-	-	-	131	-	-	94	
Book Keeping / Accounting	9								
Total	6,096	310	12	572	1,099	153	-	1,331	
Male	3,280	-	-	295	671	102	-	670	
Female	2,816	310	12	278	427	50	-	661	
Secretarial Science	,			-					
Total	2,081	-	_	56	304	_	-	_	
Male	326	-	-	-	39	-	-	-	
Female	1,755	-	_	56	265	_	-	_	
. 3	. ,								

Table 6.5 Cont'd

Training status and						Regio	on		_
Types of Training						_	Benishangul-		\Box
J. 0	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	G
Banking and Finance									
Total	514	-	-	-	-	-	-	168	
Male	168	-	-	-	-	-	-	168	
Female	346	-	-	-	-	-	-	-	
Law									
Total	1,649	281	-	78	643	-	-	-	
Male	923	281	-	-	643	-	-	-	
Female	726	-	-	78	-	-	-	-	
Language and Literature		nd Foreign)							
Total	2,280	-	-	889	611	-	-	77	
Male	1,485	-	-	797	611	-	-	77	
Female	796	-	-	92	-	-	-	-	
Theatrical art									
Total	346	-	-	-	-	-	-	-	
Male	346	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Educational Pedagog	gy, Teaching o	curriculum instr	ruction Ed	lucational a	dministrat	ion, Adults	education		
Total	2,738	-	-	328	571	50	-	254	
Male	1,864	-	-	328	-	50	-	-	
Female	874	-	-	-	571	-	-	254	
Geography									
Total	673	102	38	533	-	-	-	-	
Male	571	102	38	432	-	-	-	-	
Female	102	-	-	102	-	-	-	-	
History									
Total	82	-	-	82	-	-	-	-	
Male	82	-	-	82	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Political science									
Total	346	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	346	-	-	-	-	-	-	-	
Sociology and Anthropolog	gy								
Total	1,835	-	-	-	888	-	-	201	
Male	511	-	-	-	-	-	-	201	
Female	1,324	-	-	-	888	-	-	-	

Table 6.5 Cont'd

Table 0.5 Cont u	1								_
Training status and		1			1	Regio		T	_
Types of Training	+					0 "	Benishangul-	ONINIDD	
	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	G
Psychology									
Total	385	281	-	105	-	-	-	-	
Male	385	281	-	105	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Civil Engineering/Building									
Total	626	-	-	-	-	-	-	280	
Male	-	-	-	-	-	-	-	-	
Female	626	-	-	-	-	-	-	280	
Mechanical Engineering									
Total	287	-	-	-	-	-	-	-	
Male	287	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Electrical Engineering									
Total	391	-	-	45	-	-	-	-	
Male	391	-	-	45	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Industrial Engineering									
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Municipality Engineering / A	Architecture								
Total	44	-	-	-	44	-	-	-	
Male	44	-	-	-	44	-	-	-	
Female	-	-	-	-	-	-	-	-	
Other types of Engineering	Dry Land, B	io-diversity, Hy	draulic Er	gineering I	rrigation E	ngineering			
Total	394	-	-	-	306	-	-	-	
Male	394	-	-	-	306	-	-	-	
Female	-	-	-	-	-	-	-	-	
Automotive Technology									
Total	122	-	-	-	122	-	-	-	
Male	-	-	-	-	-	-	_	-	
Female	122	-	-	-	122	-	_	-	
Construction Technology									
Total	1,822	109	_	269	-	-	_	-	
Male	961	-	-	269	-	-	_	-	
Female	861	109	-	-	-	-	_	-	

Table 6.5 Cont'd

Table 0.5 Cont u						Regio	n e		—
Training status and						Negic	Benishangul-		_
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	G
Drafting technology		J - 7							
Total	269	-	-	269	-	-	_	-	
Male	269	-	-	269	-	-	_	-	
Female	-	-	-	-	-	-	-	-	
Manufacturing Technology									
Total	_	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Metal Technology									
Total	961	-	-	45	360	-	-	210	
Male	615	-	-	45	360	-	-	210	
Female	346	-	-	-	-	-	-	-	
Veterinary Science, Animal	Husbandry								
Total	455	-	-	109	-	-	-	-	
Male	455	-	-	109	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Communication									
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Power, Energy and of	thers like Instr	umentation, El	ectrical Po	ower Mech	anical and	Manufactu	ring Power		
Total	492	-	-	89	-	-	-	-	
Male	492	-	-	89	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Biology									
Total	790	-	-	82	451	-	-	257	
Male	-	-	-	-	-	-	-	-	
Female	790	-	-	82	451	-	-	257	
Chemistry									
Total	1,100	-	-	258	496	-	-	-	
Male	709	-	-	258	451	-	-	-	
Female	392	-	-	-	46	-	-	-	
Geology									
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	ŀ

Table 6.5 Cont'd

Table 0.5 Cont u						Regio	on		
Training status and Types of Training							Benishangul-		Γ
	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	G
Physics									
Total	187	-	-	-	-	-	-	187	
Male	187	-	-	-	-	-	-	187	
Female	-	-	-	-	-	-	-	-	
Mathematics									
Total	791	-	12	45	735	-	-	-	
Male	791	-	12	45	735	-	-	-	
Female	-	-	-	-	-	-	-	-	
Statistics									
Total	416	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	416	-	-	-	-	-	-	-	
Demography									
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Computer Science / Inform	nation Science	•							
Total	3,923	109	-	742	505	-	33	-	
Male	2,336	109	-	-	51	-	-	-	
Female	1,588	-	-	742	454	-	33	-	
Health and Physical Education	ation								
Total	1,433	-	-	-	1,433	-	-	-	
Male	1,433	-	-	-	1,433	-	-	-	
Female	-	-	-	-	-	-	-	-	
Meteorological Science									
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Chemical Engineering									
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Textile Engineering									
Total	692	-	-	-	-	-	-	-	
Male	346	-	-	-	-	-	-	-	
Female	346	-	-	-	-	-	-	-	

Table 6.5 Cont'd

Table 6.5 Cont d	1					Б.			_
Training status and				T		Regio			_
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	(
Purchasing Supply	rotar	. igiay	7 11 (41	7 11111111111	OTOTTIC	Comano	Gamaz	Oran IX	
Total	1,294	109	_	454	290	_	_	_	
Male	295	-	_	89	206	_	_	_	
Female	999	109	_	365	84	_	_	_	
Software Engineering	333	103		303	04	_	_		
Total	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	
Home Economics									
Total	4,284	552	-	1,460	503	-	-	-	
Male	1,526	-	-	1,168	-	-	-	-	
Female	2,759	552	-	292	503	-	-	-	
Leather Handicraft Techno	logy								
Total	1,202	541	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	1,202	541	-	-	-	-	-	-	
Decorative Work Training									
Total	12,500	1,311	-	1,519	1,807	-	20	501	
Male	787	-	-	-	-	-	-	-	
Female	11,713	1,311	-	1,519	1,807	-	20	501	
Video Photographic Trainir	ng								
Total	692	-	-	-	-	-	-	-	
Male	346	-	-	-	-	-	-	-	
Female	346	-	-	-	-	-	-	-	
Pottery									
Total	104	104	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	104	104	-	-	-	-	-	-	
Weavery									
Total	829	-	-	829	-	-	-	-	
Male	-	-	-	-	-	-	-	-	
Female	829	_	-	829	-	-	-	-	
Wood Work / Carpentry									
Total	4,402	255	-	1,313	681	-	-	258	
Male	3,492	255	-	1,055	543	-	-	90	
Female	910	-	-	258	137	-	-	168	

Table 6.5 Cont'd

Table 6.5 Cont d	1							
Training status and						Regio		
Types of Training		_ .					Benishangul-	0.11.155
	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR
Electrical Work								
Total	8,291	436	36	2,394	1,928	-	43	420
Male	5,029	212	36	291	1,815	-	-	420
Female	3,262	224	-	2,103	113	-	43	-
Metal Welding and Binding								
Total	1,826	667	-	750	225	-	-	-
Male	1,398	667	-	477	168	-	-	-
Female	428	-	-	274	57	-	-	-
Auto Mechanics Including N	-	rcraft Maintena	nce					
Total	4,214	937	-	380	645	-	-	275
Male	3,617	341	-	380	645	-	-	275
Female	596	596	-	-	-	-	-	-
Mason								
Total	3,384	904	12	-	195	-	-	280
Male	2,092	623	12	-	155	-	-	-
Female	1,293	281	-	-	40	-	-	280
Typing								
Total	5,288	650	-	430	2,172	-	27	-
Male	-	-	-	-	-	-	-	-
Female	5,288	650	-	430	2,172	-	27	-
Tailoring								
Total	5,838	1,389	-	1,165	1,025	-	-	234
Male	372	254	-	-	118	-	-	-
Female	5,466	1,135	-	1,165	907	-	-	234
Driver								
Total	22,026	963	-	2,752	9,031	387	94	1,095
Male	20,678	748	-	2,676	9,031	317	94	909
Female	1,348	215	-	75	-	70	-	187
Development Agent								
Total	692	-	-	-	-	-	-	-
Male	346	-	-	-	-	-	-	-
Female	346	-	-	-	-	-	-	-
Traditional Birth Attendant								
Total	102	-	-	-	102	-	_	-
Male	-	-	-	-	-	-	-	-
Female	102	-	-	-	102	-	-	-

Table 6.5 Cont'd

Table 6.5 Cont d						Regio	n	
Training status and						Regio	Benishangul-	
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR (
Computer Applications and		g,	7	7	0.0	•		
Total	10,260	270	37	848	1,304	758	_	787
Male	2,052	-	-	267	-	-	_	-
Female	8,208	270	37	582	1,304	758	_	787
Lower Scale Training of Ag					.,			
Total	42	-	_	_	42	_	_	_
Male	-	_	_	_	-	_	_	_
Female	42	_	_	_	42	_	_	_
Hotels and Tourism					'-			
Total	22,074	1,267	12	2,459	6,790	_	24	3,440
Male	5,317	577	-	223	1,412	_		1,262
Female	16,757	691	12	2,236	5,378	_	24	2,178
Police				_,	-,			_,
Total	1,701	-	_	-	453	_	_	421
Male	1,433	-	-	-	185	-	-	421
Female	268	-	_	-	268	_	_	-
Military								
Total	1,525	254	-	196	41	-	-	210
Male	921	254	-	148	41	-	-	-
Female	604	-	-	47	-	-	-	210
Cobblestone								
Total	3,569	1,098	-	931	904	50	-	-
Male	1,020	440	-	256	233	-	-	-
Female	2,548	658	-	675	672	50	-	-
Embroidery	,							
Total	2,553	431	-	319	990	-	-	107
Male	-	-	-	-	-	-	-	-
Female	2,553	431	-	319	990	-	-	107
Plumbery								
Total	1,666	-	-	-	599	-	-	375
Male	1,666	-	-	-	599	-	-	375
Female	-	-	-	-	-	-	-	-

Table 6.5 Cont'd

Training status and						Regio	on		
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	(
Other Senior and Junior Pr	rofessions / Sp	pecified							
Total	98,056	5,285	335	18,816	36,019	392	378	10,542	
Male	41,660	1,889	48	6,455	15,338	240	151	4,490	
Female	56,396	3,395	287	12,361	20,682	152	227	6,052	
Other Skilled and Technica	al Professions	/ Specified							
Total	27,954	2,957	25	5,511	7,401	472	40	1,182	
Male	10,834	1,058	25	1,258	3,134	472	40	187	
Female	17,121	1,899	-	4,253	4,267	-	-	996	
Not Stated for type of traini	ing								
Total	779	109	-	69	602	-	-	-	
Male	177	109	-	69	-	-	-	-	
Female	602	-	-	-	602	-	-	-	

Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and Marital Status, COUNTRY TOTAL: 2012 Table 6.6

Aye	e Group, Sex	and Man	iai Status,			2012		
Age Group and Sex		Never		Marital Sta	alus		Live	Not
Age Gloup and Gex	All Persons	Married	Married	Divorced	Widowed	Separated	together	Stated
All Ages	7.11 0100110	Mamou	mamou	Divoloca	macmou	Coparatoa	togotiro.	Otatoa
Total	1,218,574	540,337	536,678	55,232	52,852	22,331	10,964	179
Male	410,381	279,311	106,270	9,789	7,588	4,985	2,260	179
Female	808,193	261,026	430,408	45,444	45,264	17,347	8,704	-
10-14	000,193	201,020	430,400	45,444	75,207	17,577	0,704	
Total	24,618	24,618	_	_	_	_	_	_
Male	11,787	11,787	_	_	_	_	_	_
Female	12,831	12,831	_	_	_	_	_	_
15-19	12,031	12,031	-	-	-	-	-	-
Total	136,069	103,143	29,464	1,696	144	908	714	
Male	51,396	48,936	29,404	-	1 44 -	-	310	-
								-
Female 20-24	84,673	54,207	27,314	1,696	144	908	405	-
	204 402	007.004	440.000	0.005	050	0.054	F 200	
Total	361,492 127,407	227,064	116,036	8,685	853	3,654	5,200	-
Male	•	117,500	8,089	791	-	306	722	-
Female	234,085	109,564	107,947	7,894	853	3,348	4,479	-
25-29		4.4= =00	100 000	40.000			4 0 4=	
Total	268,787	117,700	129,269	12,039	4,044	3,887	1,847	-
Male .	79,619	62,988	13,717	1,213	652	779	269	-
Female	189,168	54,711	115,552	10,826	3,392	3,108	1,578	-
30-34								
Total	128,378	41,479	73,325	5,611	3,003	3,692	1,089	179
Male	39,484	23,760	12,793	1,570	210	685	287	179
Female	88,893	17,719	60,532	4,041	2,793	3,006	802	-
35-39								
Total	98,500	15,447	65,810	7,133	5,730	3,204	1,176	-
Male	26,203	7,219	16,102	1,195	485	1,203	-	-
Female	72,297	8,229	49,708	5,939	5,245	2,001	1,176	-
40-44								
Total	58,078	5,289	38,788	6,782	6,189	838	193	-
Male	16,589	2,538	10,879	1,561	1,317	189	104	-
Female	41,489	2,751	27,908	5,221	4,872	648	88	-
45-49								
Total	43,186	2,508	27,981	1,845	8,019	2,743	90	-
Male	12,716	1,782	9,555	604	612	163	-	-
Female	30,470	726	18,426	1,241	7,407	2,580	90	-
50-54								
Total	36,225	1,695	19,186	5,969	8,229	1,059	87	-
Male	12,971	1,408	9,325	1,214	398	626	-	-
Female	23,254	287	9,862	4,755	7,831	432	87	-
55-59								
Total	24,471	324	15,299	1,865	6,255	246	482	-
Male	9,677	324	7,108	430	1,206	127	482	-
Female	14,794	-	8,191	1,435	5,049	119	-	-
60-64								
Total	18,566	338	10,121	2,841	3,949	1,230	87	-
Male	8,023	338	6,567	722	274	35	87	-
Female	10,544	-	3,554	2,119	3,675	1,195	-	-
65+	,		•	, -	, -	, -		
Total	20,205	732	11,399	767	6,437	870	_	_
Male	14,510	732	9,985	490	2,434	870	-	_
Female	5,694	-	1,414	277	4,003	-	_	_
1 GITIAIG	U,UU T	*	1,717	411	7,000	-	-	·

Table 6.7 Currently Unemployed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Marital Status: 2012

Region and Sex	Region	Sex and Ma	rital Statu	ıs: 2012					
Married Married Divorced Widowed Separated together Stated				1	Marital S	Status			
COUNTRY TOTAL Total 1,218,574 540,337 536,678 55,232 52,852 22,331 10,964 179 Maile 410,381 279,311 106,270 9,789 7,588 4,985 2,260 179 Female 808,193 261,026 430,408 45,444 45,264 17,347 8,704 - TIGRAY REGION Total 94,756 34,291 47,652 7,737 4,180 688 208 - Female 68,761 17,489 40,291 7,159 3,395 324 103 - AFFAR REGION Total 9,846 1,914 5,767 945 1,125 24 72 - Female 7,421 670 4,801 766 1,25 24 72 - AMHAR REGION 701 40,996 79,619 11,214 8,905 1,157 311 - OROMIA REGION Total 330,723	Region and Sex				<u>.</u>		•		
Total Male		All Persons	Married	Married	Divorced	Widowed	Separated	together	Stated
Male Female 410,381 279,311 106,270 9,788 7,588 4,985 2,260 179 TIGRAY REGION Total 94,756 34,291 47,652 7,737 4,180 688 208 - Female 68,761 17,489 40,291 7,159 3,395 324 103 - Female 68,761 1,914 5,767 945 1,125 24 72 - AFFAR REGION 7,421 670 4,801 766 1,125 24 72 - Female 7,421 670 4,801 766 1,125 24 72 - AMHARA REGION 7 167 985.94 13,179 9,822 1,202 806 - Female 141,304 40,996 79,619 11,214 8,905 11,157 311 - OROMIA REGION 7 11,214 8,905 1,167 311 - - - - <									
Female 808,193 261,026 430,408 45,444 45,264 17,347 8,704 TIGRAY REGION Total 94,756 34,291 47,652 7,737 4,180 688 208 Male 25,996 16,803 7,361 578 785 364 104 Female 68,761 17,489 40,291 7,159 3,395 324 103 AFFAR REGION Total 9,846 1,914 5,767 945 1,125 24 72 Male 2,425 1,244 966 179 36 Female 7,421 670 4,801 766 1,125 24 36 AMARA REGION Total 209,533 85,831 98,694 13,179 9,822 1,202 806 Female 68,229 45,735 19,075 1,965 916 45 494 Female 141,304 40,096 79,619 11,214 8,905 1,157 311 OROMIA REGION Total 330,723 148,170 147,781 9,524 12,343 6,455 6,450 Total 330,723 148,170 147,781 9,524 12,343 6,455 6,450 Female 214,964 67,467 118,859 7,409 11,142 4,365 5,721 Female 214,964 67,467 118,859 7,409 11,142 4,365 5,721 Total 34,427 8,574 22,803 967 749 1,282 51 Female 22,302 3,515 16,088 857 661 1,181 Elemish-IANGUL GUMUZ REGION Total 3,609 6,303 2,690 137 115 62 Total 1,997 1,378 519 Male 1,897 1,378 519 .						•	•		
TIGRAY REGION									179
Total Male 25,966 16,803 7,361 578 785 364 104 - Male 25,966 16,803 7,361 578 785 364 104 - Female 68,761 17,489 40,291 7,159 3,395 324 103 - AFFAR REGION Total 9,846 1,914 5,767 945 1,125 24 72 - Male 2,425 1,244 966 179 - 0 36 - Female 7,421 670 4,801 766 1,125 24 36 - Female 7,421 670 4,801 766 1,125 24 36 - AMHARA REGION Total 209,533 85,831 98,694 13,179 9,822 1,202 806 - Female 141,304 40,096 79,619 11,214 8,905 1,157 311 - CROMIA REGION Total 330,723 148,170 147,781 9,524 12,343 6,455 6,450 - Female 115,759 80,703 28,922 2,115 1,201 2,091 728 - Female 214,964 67,467 118,859 7,409 11,142 4,365 5,721 - SOMALI REGION Total 34,427 8,574 22,803 967 749 1,282 51 - Female 22,302 3,515 16,088 857 661 1,181 57 - Female 22,302 3,515 16,088 857 661 1,181 57 - Female 3,609 603 2,690 137 115 62 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - SNNP REGION Total 1,997 1,378 519 - 0 - 0 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - 0 - SNNP REGION Total 1,997 4,378 519 - 0 - 0 - 0 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - 0 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - 0 - 0 - 0 - Female 3,609 603 2,690 137 115 62 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 -		808,193	261,026	430,408	45,444	45,264	17,347	8,704	-
Male Female 25,996 16,803 7,361 578 785 364 104									
Female									-
AFFAR REGION Total									-
Total Male 9,846 1,914 5,767 945 1,125 24 72 - 36 Male Female 7,421 670 4,801 766 1,79 - 2 36 - 36 AMHARA REGION Total 29,533 85,831 98,694 13,179 9,822 1,202 806 - 2 Male Male 68,229 45,735 19,075 1,965 916 45 494 - 2 Female Pemale 141,304 40,096 79,619 11,214 8,905 1,157 311 - 3 OROMIA REGION 101 330,723 148,170 147,781 9,524 12,343 6,455 6,450 - 2 Female 115,759 80,703 28,922 2,115 1,201 2,091 728 - 2 SOMALI REGION 115,759 80,703 28,922 2,115 1,201 2,091 728 - 2 SOMALI REGION 12,124 5,059 6,716 110 88 101 <		68,761	17,489	40,291	7,159	3,395	324	103	-
Male Female 2,425 1,244 966 179 - - 36 - Female Female 7,421 670 4,801 766 1,125 24 36 - AMHARA REGION Total 209,533 85,831 98,694 13,179 9,822 1,202 806 - Female 141,304 40,096 79,619 11,214 8,905 1,157 311 - OROMIA REGION Total 330,723 148,170 147,781 9,524 12,343 6,455 6,450 - Female 214,964 67,467 118,895 7,409 11,142 4,365 5,721 - SOMALI REGION Total 34,427 8,574 22,803 967 749 1,282 51 - Female 22,302 3,515 16,088 857 661 1,181 - - SENISHANGUL GUMUZ 7 1,378 519 - - - - - -									
Female AMHARA REGION 7,421 670 4,801 766 1,125 24 36			1,914			1,125	24		-
Male	Male		1,244	966	179		-	36	-
Total Male 209,533 85,831 98,694 13,179 9,822 1,202 806 - Male 68,229 45,735 19,075 1,965 916 45 494 - Female 141,304 40,096 79,619 11,214 8,905 1,157 311 - OROMIA REGION Total 330,723 148,170 147,781 9,524 12,343 6,455 6,450 - Female 214,964 67,467 118,859 7,409 11,142 4,365 5,721 - SOMALI REGION Total 34,427 8,574 22,803 967 749 1,282 51 - Male 12,124 5,059 6,716 110 88 101 51 - Female 22,302 3,515 16,088 857 661 1,181 - - BENISHANGUL GUMUZ REGION 1,378 519 - - - - - <td>Female</td> <td>7,421</td> <td>670</td> <td>4,801</td> <td>766</td> <td>1,125</td> <td>24</td> <td>36</td> <td>-</td>	Female	7,421	670	4,801	766	1,125	24	36	-
Male Female 68,229 45,735 79,075 1,965 916 45 494 79,076 79,619 11,214 8,905 1,157 311 70,0000 70,619 71,214 8,905 1,157 311 70,0000 70,0000 70,619 71,214 8,905 7,100 70,000	AMHARA REGION								
Female OROMIA REGION 141,304 40,096 79,619 11,214 8,905 1,157 311 Total Male Male 115,759 330,723 148,170 147,781 9,524 12,343 6,455 6,450 Female Female 214,964 67,467 118,859 7,409 11,142 4,365 5,721 SOMALI REGION Total Male 12,124 5,059 6,716 110 88 101 51 Male 12,124 5,059 6,716 110 88 101 51 Female 22,302 3,515 16,088 857 661 1,181 BENISHANGUL GUMUZ REGION Total 3,897 1,378 519	Total								-
OROMIA REGION Total 330,723 148,170 147,781 9,524 12,343 6,455 6,450 - Male 115,759 80,703 28,922 2,115 1,201 2,091 728 - Female 214,964 67,467 118,859 7,409 11,12 4,365 5,721 - SOMALI REGION Total 34,427 8,574 22,803 967 749 1,282 51 - Male 12,124 5,059 6,716 110 88 101 51 - Female 22,302 3,515 16,088 857 661 1,181 - - BENISHANGUL GUMUZ REGION 1 3,210 137 115 62 - - Total 5,505 1,981 3,210 137 115 62 - - SNNP REGION 1 119,176 48,297 59,002 4,027 6,229 1,116<	Male	68,229	45,735				45		-
Total 330,723 148,170 147,781 9,524 12,343 6,455 6,450 - Male 115,759 80,703 28,922 2,115 1,201 2,091 728 - Female 214,964 67,467 118,859 7,409 11,142 2,091 728 - SOMALI REGION Total 34,427 8,574 22,803 967 749 1,282 51 - Male 12,124 5,059 6,716 110 88 101 51 - Female 22,302 3,515 16,088 857 661 1,181 - - BENISHANGUL GUMUZ REGION Total 5,505 1,981 3,210 137 115 62 - - Male 1,897 1,378 519 - - - - - - - - - - - - - - - - - <td>Female</td> <td>141,304</td> <td>40,096</td> <td>79,619</td> <td>11,214</td> <td>8,905</td> <td>1,157</td> <td>311</td> <td>-</td>	Female	141,304	40,096	79,619	11,214	8,905	1,157	311	-
Male Female 115,759 Female 80,703 Female 214,964 Female 67,467 Female 118,859 Female 7,409 Female 11,142 Female 2,091 Female 728 Female 5,721 Female 5,721 Female 5,721 Female 5,741 Female 11,142 Female 11,142 Female 11,142 Female 5,721 Female 5,721 Female 5,721 Female 5,721 Female 11,142 Female 5,721 Female 5,721 Female 5,059 Female 6,716 Female 110 Female 88 Female 101 Female 5,505 Female 1,817 Fema	OROMIA REGION								
Female	Total	330,723	148,170	147,781	9,524	12,343	6,455	6,450	-
SOMALI REGION Total 34,427 8,574 22,803 967 749 1,282 51 51 51 51 51 51 51 5	Male	115,759	80,703	28,922	2,115	1,201	2,091	728	-
Total 34,427 8,574 22,803 967 749 1,282 51 - Male 12,124 5,059 6,716 110 88 101 51 - Female 22,302 3,515 16,088 857 661 1,181 - BENISHANGUL GUMUZ REGION Total 5,505 1,981 3,210 137 115 62 - Female 3,609 603 2,690 137 115 62 - Female 3609 603 2,690 137 115 62 - SINNP REGION Total 119,176 48,297 59,002 4,027 6,229 1,116 327 179 Male 40,404 26,272 11,475 1,375 300 639 163 179 Female 78,771 22,025 47,527 2,652 5,929 476 163 - GAMBELLA REGION Total 3,020 569 2,261 116 47 27 - Male 484 169 315 - Female 2,536 400 1,946 116 47 27 - Female 6,973 2,313 3,572 196 573 320 - Male 4,887 1,119 2,809 196 573 320 - Female 4,887 1,119 2,809 196 573 190 - ADDIS ABABA CITY ADMINISTRATION Total 374,626 194,616 135,114 17,106 15,670 10,545 1,576 - Female 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -	Female	214,964	67,467	118,859	7,409	11,142	4,365	5,721	-
Male Female 12,124 5,059 6,716 110 88 101 51 - BENISHANGUL GUMUZ REGION Total 5,505 1,981 3,210 137 115 62 - - Male 1,897 1,378 519 - - - - - Female 3,609 603 2,690 137 115 62 - - SNNP REGION - - - - - - - Male 40,404 26,272 11,475 1,375 300 639 163 179 Male 40,404 26,272 11,475 1,375 300 639 163 179 Female 78,771 22,025 47,527 2,652 5,929 476 163 - GAMBELLA REGION - - - - - - - - - - - - - -	SOMALI REGION								
Female BENISHANGUL GUMUZ REGION	Total	34,427	8,574	22,803	967	749	1,282	51	-
BENISHANGUL GUMUZ REGION	Male	12,124	5,059	6,716	110	88	101	51	-
Total		22,302	3,515	16,088	857	661	1,181	-	-
Male Female 1,897 1,378 519 -									
Female 3,609 603 2,690 137 115 62 - - SNNP REGION Total 119,176 48,297 59,002 4,027 6,229 1,116 327 179 Male 40,404 26,272 11,475 1,375 300 639 163 179 Female 78,771 22,025 47,527 2,652 5,929 476 163 - GAMBELLA REGION Total 3,020 569 2,261 116 47 27 - - Male 484 169 315 -	Total				137	115	62	-	-
SNNP REGION Total 119,176 48,297 59,002 4,027 6,229 1,116 327 179 Male 40,404 26,272 11,475 1,375 300 639 163 179 Female 78,771 22,025 47,527 2,652 5,929 476 163 - GAMBELLA REGION Total 3,020 569 2,261 116 47 27 - - Male 484 169 315 -		1,897	1,378	519	-	-	-	-	-
Total 119,176 48,297 59,002 4,027 6,229 1,116 327 179 Male 40,404 26,272 11,475 1,375 300 639 163 179 Female 78,771 22,025 47,527 2,652 5,929 476 163 - GAMBELLA REGION Total 3,020 569 2,261 116 47 27 - - - Male 484 169 315 - </td <td></td> <td>3,609</td> <td>603</td> <td>2,690</td> <td>137</td> <td>115</td> <td>62</td> <td>-</td> <td>-</td>		3,609	603	2,690	137	115	62	-	-
Male Female 40,404 26,272 11,475 1,375 300 639 163 179 Female 78,771 22,025 47,527 2,652 5,929 476 163 - GAMBELLA REGION Total 3,020 569 2,261 116 47 27 - - Male 484 169 315 -	SNNP REGION								
Female 78,771 22,025 47,527 2,652 5,929 476 163 - GAMBELLA REGION Total 3,020 569 2,261 116 47 27 - - Male 484 169 315 -	Total	119,176					1,116	327	179
GAMBELLA REGION Total 3,020 569 2,261 116 47 27 - - Male 484 169 315 - - - - - - Female 2,536 400 1,946 116 47 27 - - HARARI REGION Total 6,973 2,313 3,572 196 573 320 - - Male 2,086 1,194 762 - - 130 - - Female 4,887 1,119 2,809 196 573 190 - - ADDIS ABABA CITY ADMINISTRATION Total 374,626 194,616 135,114 17,106 15,670 10,545 1,576 - Male 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -	Male	40,404	26,272			300	639	163	179
Total 3,020 569 2,261 116 47 27 Male 484 169 315		78,771	22,025	47,527	2,652	5,929	476	163	-
Male 484 169 315 -	GAMBELLA REGION								
Female 2,536 400 1,946 116 47 27 - - HARARI REGION Total 6,973 2,313 3,572 196 573 320 - - - Male 2,086 1,194 762 - - 130 - - - Female 4,887 1,119 2,809 196 573 190 - - - ADDIS ABABA CITY ADMINISTRATION ADDIS ABABA CITY ADDIS ABABA CITY -	Total				116	47	27	-	-
HARARI REGION Total 6,973 2,313 3,572 196 573 320 Male 2,086 1,194 762 130 Female 4,887 1,119 2,809 196 573 190 ADDIS ABABA CITY ADMINISTRATION Total 374,626 194,616 135,114 17,106 15,670 10,545 1,576 - Male 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -					-			-	-
Total 6,973 2,313 3,572 196 573 320 Male 2,086 1,194 762 130 - 130 - 130 - 130 130 - 130 - 1		2,536	400	1,946	116	47	27	-	-
Male 2,086 1,194 762 - - 130 - - Female 4,887 1,119 2,809 196 573 190 - - ADDIS ABABA CITY ADMINISTRATION 374,626 194,616 135,114 17,106 15,670 10,545 1,576 - Male 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -									
Female 4,887 1,119 2,809 196 573 190 - - ADDIS ABABA CITY ADMINISTRATION 374,626 194,616 135,114 17,106 15,670 10,545 1,576 - Male 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -	Total				196	573		-	-
ADDIS ABABA CITY ADMINISTRATION Total 374,626 194,616 135,114 17,106 15,670 10,545 1,576 - Male 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -	Male	2,086	1,194		-	-		-	-
ADMINISTRATION Total 374,626 194,616 135,114 17,106 15,670 10,545 1,576 - Male 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -		4,887	1,119	2,809	196	573	190	-	-
Total 374,626 194,616 135,114 17,106 15,670 10,545 1,576 - Male 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -									
Male 130,686 94,181 27,027 3,380 4,059 1,441 597 - Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -		074000	101010	40= 444	4= 400	4 = 0 = 0		4 ==0	
Female 243,940 100,434 108,086 13,726 11,611 9,104 979 -									-
						•			-
		243,940	100,434	108,086	13,726	11,611	9,104	979	-
ADMINISTRATION									
Total 29,988 13,780 10,824 1,299 1,999 610 1,476 -		29,988	13,780	10.824	1,299	1.999	610	1.476	_
Male 10,291 6,573 3,133 87 239 173 87 -									_
Female 19,697 7,208 7,691 1,212 1,760 437 1,390 -									_



Currently Unemployed Populations of Urban Areas Aged Ten Years and Above Who Were Seby Region, Sex and Steps Taken to Seek Work: 2012 Table 6.8

, u				Step	s taken to	seek Wo	rk	
			Through	Seeking				
Region and Sex		Searching	news	assistance	<u> </u>	Trying to	Direct	
	Total	Vacancy	paper,	of friends,	Checking	establish	application	I have
	Unemployed Persons	Advertising boards	Radio and TV	relatives,	at work sites	own enterprise	to employee	unemploymer card
COUNTRY TOTAL	r ersoris	Doalus	and iv	etc	31103	enterprise	employee	Calu
Total	1,218,574	221,967	61,658	177,748	107,354	176,217	25,347	14,676
Male	410,381	98,571	24,718	73,513	50,827	55,807	12,326	5,015
Female	808,193	123,396	36,940	104,235	56,527	120,411	13,021	9,661
TIGRAY REGION	000,100	120,000	30,540	104,200	30,321	120,411	10,021	3,001
Total	94,756	17,960	1,509	8,847	7,843	13,811	3,074	1,200
Male	25,996	7,831	533	3,934	3,580	3,493	651	281
Female	68,761	10,129	976	4,914	4,263	10,318	2,423	919
AFFAR REGION	,	•		•	,	•	•	
Total	9,846	514	188	734	670	1,160	120	12
Male	2,425	237	12	111	369	382	12	12
Female	7,421	277	176	624	301	778	108	-
AMHARA REGION								
Total	209,533	50,010	3,147	21,696	14,785	37,376	1,977	2,345
Male	68,229	19,476	1,534	10,221	6,109	16,050	1,037	314
Female	141,304	30,534	1,613	11,474	8,676	21,326	940	2,031
OROMIA REGION								
Total	330,724	55,042	13,336	54,903	31,966	50,871	5,726	3,958
Male	115,759	23,941	5,981	20,578	16,139	17,540	4,305	1,749
Female	214,964	31,101	7,355	34,326	15,827	33,331	1,421	2,209
SOMALI REGION								
Total	34,427	469	-	4,445	4,352	4,518	255	-
Male	12,124	190	-	2,021	2,700	1,112	153	-
Female	22,302	279	-	2,425	1,652	3,406	102	-
BENISHANGUL GUMUZ RE								
Total	5,505	1,116	23	163	670	972	20	92
Male	1,897	577	23	63	223	295	20	23
Female	3,609	539	-	100	447	677	-	69
SNNP REGION								
Total	119,176	22,128	1,953	16,615	14,263	22,115	1,784	651
Male	40,404	9,739	641	6,608	7,136	4,922	701	550
Female	78,771	12,389	1,313	10,007	7,127	17,193	1,083	101

Table 6.8 Cont'd

Table 6.8 Cont'd								
		- 		Step	s taken to	seek Wo	rk	
			Through	Seeking				
Region and Sex		Searching	news	assistance		Trying to	Direct	
region and Sex	Total	Vacancy	paper,	of friends,	Checking	establish	application	I have
	Unemployed	_	Radio	relatives,	at work	own	to	unemployme
	Persons	boards	and TV	etc	sites	enterprise	employee	card
GAMBELLA REGION								
Total	3,020	349	-	91	245	482	146	88
Male	484	27	-	62	125	113	27	-
Female	2,536	322	-	29	120	369	119	88
HARARI REGION								
Total	6,973	1,265	93	689	929	1,384	147	-
Male	2,086	517	43	493	239	195	49	-
Female	4,887	748	49	196	690	1,189	98	-
ADDIS ABABA CITY								
ADMIISTRATION								
Total	374,626	67,153	41,233	65,387	27,051	40,057	11,423	6,331
Male	130,686	33,680	15,864	27,262	11,996	10,831	4,960	2,086
Female	243,940	33,474	25,369	38,126	15,054	29,226	6,463	4,245
DIRE DAWA								
ADMINISTRATION								
Total	29,988	5,961	176	4,176	4,581	3,472	675	-
Male	10,291	2,357	87	2,161	2,211	874	412	-
Female	19,697	3,604	90	2,015	2,370	2,598	263	

Currently Unemployed Population of Urban Areas Aged ten Years and Above by Region, Sex and Type of job looking for: 2012 Table 6.9

Region, Sex	and Type	oi job io		of job lookir	na for		
		1	Self	OI JOB IOOKII	ig ioi		
5			Employed/				
Region and Sex		Any	To Establish				
		Available	Own	Private	Government		Not
	Total	work	Business	Employee	Employee	Others	Stated
COUNTRY TOTAL							
Total	1,218,574	648,529	349,456	65,704	143,276	10,000	1,608
Male	410,381	237,873	86,163	29,601	52,023	4,265	455
Female	808,193	410,656	263,293	36,103	91,253	5,735	1,153
TIGRAY REGION							
Total	94,756	45,644	31,882	2,474	13,412	1,235	109
Male	25,996	14,449	7,151	176	3,277	833	109
Female	68,761	31,195	24,732	2,297	10,135	402	-
AFFAR REGION							
Total	9,846	5,556	2,284	643	1,352	12	-
Male	2,425	1,337	645	215	228	-	-
Female	7,421	4,219	1,639	427	1,124	12	-
AMHARA REGION	,	, -	,		,		
Total	209,533	97,490	72,458	8,035	29,669	1,420	461
Male	68,229	33,859	18,320	3,700	12,026	324	_
Female	141,304	63,631	54,137	4,335	17,643	1,095	461
OROMIA REGION	,	,	, -	,	,	,	-
Total	330,723	171,298	101,682	18,201	37,285	2,257	_
Male	115,759	67,589	26,216	8,355	12,426	1,173	_
Female	214,964	103,709	75,465	9,847	24,859	1,084	_
SOMALI REGION	,00 .	.00,.00	. 0, .00	0,0	,000	.,	
Total	34,427	20,833	8,724	806	2,698	1,366	_
Male	12,124	7,862	2,404	243	1,301	314	_
Female	22,302	12,971	6,320	563	1,397	1,052	_
BENISHANGUL GUMUZ REG		12,071	0,020	000	1,007	1,002	
Total	5,505	2,401	1,647	255	1,159	43	_
Male	1,897	1,122	410	23	300	43	_
Female	3,609	1,279	1,238	233	860	-	_
SNNP REGION	0,000	1,270	1,200	200	000		
Total	119,176	55,439	37,464	5,676	20,227	370	_
Male	40,404	20,014	8,021	2,919	9,081	370	_
Female	78,771	35,425	29,443	2,758	11,145	-	_
GAMBELLA REGION	70,771	00,120	20,110	2,700	11,110		
Total	3,020	1.617	818	_	585	_	_
Male	484	320	29	_	135	_	_
Female	2,536	1,297	790	_	450	_	_
HARARI REGION	2,000	1,201	750		-100		
Total	6,973	3,979	1,974	196	771	54	
Male	2,086	1,510	328	98	151	J 4	_
Female	4,887	2,469	1,646	98	620	54	_
ADDIS ABABA CITY ADMINIS		2,403	1,040	90	020	J 4	_
Total	374,626	229,360	80,945	27,187	32,938	3,157	1,038
Male	130,686	83,966	20,388	13,021	32,936 11,757	1,208	346
Female	243,940	145,395	60,557	14,166	21,180	1,208	692
DIRE DAWA ADMINISTRATION		140,080	00,007	14,100	21,100	ı,5 4 9	092
Total	29,988	14,913	9,577	2,231	3,180	87	
Male		5,847		2,231 852		07	-
	10,291		2,251		1,341	- 07	-
Female	19,697	9,066	7,326	1,380	1,839	87	



Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Grou Experience and Duration of Unemployment - Country Total: 2012 **Table 6.10**

	хроп.		ilation of or	.cp.c	yoc	ooa						— '
		Unemplo	yed					Durat	ion of Ur	nemploy	nent (In	M
Age Group and Sex	Total Unemployed Persons	Unemployed Without Work Experience	Unemployed With Work Experience	Not Stated	< 1	1-6	7-12	13-24	25-36	37-48	49-60	
All Ages												
Total	1,218,574	650,742	565,020	2,812	116,520	368,296	291,522	166,740	72,322	46,261	20,092	1
Male	410,381	235,185	174,695	501	31,729	144,060	107,715	50,953	22,286	12,384	7,096	
Female 10-14	808,193	415,558	390,325	2,311	84,790	224,236	183,808	115,788	50,035	33,877	12,997	1
Total	24,618	8,985	15,633	-	7,694	9,914	4,047	386	-	923	314	
Male	11,787	4,127	7,660	-	3,844	4,471	1,749	191	-	692	314	
Female 15-19	12,831	4,858	7,973	-	3,850	5,443	2,299	195	-	231	-	
Total	136,069	46,046	89,677	346	16,321	48,316	38,421	22,003	4,232	1,814	-	
Male	51,396	18,443	32,953	-	4,303	20,743	13,593	8,328	2,507	140	-	ļ
Female 20-24	84,673	27,603	56,724	346	12,018	27,573	24,828	13,675	1,725	1,674	-	
Total	361,492	159,672	201,428	393	29,790	115,878	98,108	55,553	24,147	14,139	3,744	
Male	127,407	55,054	72,353	-	8,522	42,825	36,666	18,801	7,126	4,440	2,556	ļ
Female 25-29	234,085	104,618	129,075	393	21,268	73,053	61,442	36,752	17,021	9,698	1,188	
Total	268,787	147,922	120,264	601	21,015	80,991	63,659	40,408	17,667	11,267	5,799	
Male	79,619	44,419	35,091	109	4,626	31,089	20,132	10,737	4,458	2,731	1,023	-
Female 30-34	189,168	103,503	85,173	492	16,388	49,902	43,528	29,671	13,210	8,537	4,776	
Total	128,378	83,150	44,516	712	11,756	36,761	24,838	17,590	7,962	6,221	2,126	ļ
Male	39,484	29,393	10,044	46	3,508	14,134	8,839	4,538	1,836	1,166	483	
Female	88,893	53,757	34,471	665	8,248	22,627	15,999	13,052	6,126	5,055	1,643	

35-39											
Total	98,500	67,666	30,488	346	8,925	28,546	20,859	10,206	5,253	4,250	2,488
Male Female 40-44	26,203 72,297	21,964 45,702	3,893 26,595	346 -	1,589 7,336	10,018 18,528	6,696 14,163	2,489 7,717	1,352 3,901	1,087 3,163	588 1,900
Total	58,078	35,989	22,089	-	6,094	12,959	13,314	7,636	3,964	2,778	1,276
Male	16,589	13,586	3,003	-	270	4,737	5,823	1,934	805	476	442

5,824

8,222

7,491

5,702

3,159

2,302

834

Female

41,489

22,403

19,086

Table 6.10) Cont'd										
		Unemplo	yed					Durat	ion of Ur	nemployr	ment (In M
Age Group and Sex	Total Unemployed Persons	Unemployed Without Work Experience	Unemployed With Work Experience	Not Stated	< 1	1-6	7-12	13-24	25-36	37-48	49-60
45-49		,									<u> </u>
Total	43,186	27,982	15,204	-	4,801	9,679	10,388	4,170	1,541	1,134	1,988
Male	12,716	10,097	2,619	-	1,162	3,630	4,323	1,124	87	620	287

Total												
Male	8,023	7,870	153	-	458	2,672	1,157	493	433	39	405	
Female 65+	10,544	6,924	3,273	346	1,994	1,888	691	1,225	130	875	359	
Total	20,205	16,331	3,874	-	2,317	5,554	3,807	1,127	1,209	904	616	
Male	14,510	12,799	1,711	-	1,872	3,580	2,834	83	1,209	904	572	
Female	5.694	3.532	2.163	_	445	1.973	973	1.045	_	_	44	

Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Region, Se and Duration of Unemployment (In Months): 2012 **Table 6.11**

a	liu Duiation	•			7.2012						
		Unemplo	yed	T		-	ı	Duratio	n of Un	employ	<u>/ment</u> (
Region and Sex	Total Unemployed Persons	Unemployed Without Work Experience	Unemployed With Work Experience	Not Stated	< 1	1-6	7-12	13-24	25-36	37-48	49-60
COUNTRY TOTAL			•	1	ı		ı				
Total	1,218,574	650,742	565,020	2,812	116,520	368,296	291,522	166,740	72,322	46,261	20,092
Male	410,381	235,185	174,695	501	31,729	144,060	107,715	50,953	22,286	12,384	7,096
Female	808,193	415,558	390,325	2,311	84,790	224,236	183,808	115,788	50,035	33,877	12,997
TIGRAY REGION	•	·	•		•	·	•	•			
Total	94,756	48,322	45,901	533	5,072	27,504	19,044	14,864	7,884	2,068	920
Male	25,996	15,635	10,252	109	1,048	9,933	5,482	4,048	1,566	814	705
Female	68,761	32,687	35,649	424	4,024	17,572	13,562	10,816	6,319	1,254	215
AFFAR REGION											
Total	9,846	5,905	3,941	-	835	2,379	3,327	1,292	631	156	181
Male	2,425	2,019	406	-	203	357	882	349	191	156	-
Female	7,421	3,886	3,535	-	631	2,022	2,445	944	440	-	181
AMHARA REGION											
Total	209,533	102,717	106,240	576	28,180	69,359	48,436	28,891	6,983	7,723	3,496
Male	68,229	38,804	29,378	46	6,638	25,833	18,520	6,999	2,131	2,906	1,751
Female	141,304	63,913	76,861	530	21,542	43,525	29,916	21,892	4,853	4,817	1,745
OROMIA REGION											
Total	330,723	185,199	145,524	-	36,973	117,651	76,984	44,048	15,420	9,560	5,302
Male	115,759	68,964	46,796	-	10,062	51,151	24,277	17,160	3,790	1,473	1,617
Female	214,964	116,235	98,729	-	26,912	66,500	52,707	26,888	11,630	8,088	3,685
SOMALI REGION											
Total	34,427	15,905	18,522	-	5,576	7,579	9,538	3,884	1,656	1,064	679
Male	12,124	6,369	5,755	-	1,294	2,638	3,827	1,548	435	173	365
Female	22,302	9,535	12,767	-	4,282	4,941	5,711	2,335	1,221	891	314
BENISHANGUL GU											
Total	5,505	3,224	2,281	-	1,069	2,019	989	653	239	284	20
Male	1,897	1,165	732	-	446	752	324	85	40	197	-
Female	3,609	2,059	1,550	-	623	1,266	665	568	200	87	20
SNNP REGION											
Total	119,176	61,259	57,598	319	9,043	36,470	33,874	13,160	8,860	6,796	979
Male	40,404	19,941	20,464	-	2,885	15,547	12,312	3,431	2,182	2,464	96
Female	78,771	41,318	37,134	319	6,158	20,923	21,563	9,728	6,678	4,333	884

Table 6.11 Cont'd

		Unemplo	yed					Duratio	n of Un	employ	ment (
Region and Sex	Total Unemployed Persons	Unemployed Without Work Experience	Unemployed With Work Experience	Not Stated	< 1	1-6	7-12	13-24	25-36	37-48	49-60
GAMBELLA REGIO	N										
Total	3,020	1,280	1,740	-	212	1,692	477	137	110	135	113
Male	484	279	205	-	27	260	41	54	47	27	28
Female	2,536	1,001	1,535	-	185	1,433	437	83	62	108	85
HARARI REGION											
Total	6,973	4,157	2,816	-	489	2,104	1,208	1,383	1,094	286	43
Male	2,086	1,443	644	-	98	631	287	602	347	41	
Female ADDIS ABABA CITY ADMINISTRTAION	4,887	2,715	2,172	-	392	1,473	920	782	747	245	43
Total	374,626	202,880	170,361	1,384	23,318	94,509	92,404	54,581	27,033	16,968	7,686
Male	130,686	73,861	56,479	346	7,798	34,309	39,303	15,302	10,945	3,612	2,360
Female DIRE DAWA ADMINISTRATION	243,940	129,019	113,883	1,038	15,520	60,200	53,101	39,279	16,088	13,356	5,326
Total	29,988	19,894	10,094	-	5,752	7,030	5,241	3,847	2,411	1,223	672
Male	10,291	6,706	3,585	-	1,230	2,649	2,460	1,374	613	523	173
Female	19,697	13,188	6,510	-	4,522	4,381	2,781	2,473	1,799	700	499

Table 6.12 Currently Unemployed Population of Urban Areas Aged Ten Years and Above by Region, Se Establish Own Business: 2012

	-0.00011011 0	D usi	20 1	_						
<u> </u>	Problem Faced to start Own Business									
						Lastini	Lack of			
Region and Sex		Ihave	Shortage		Problem	Lack of Finance	working Place	Lack /	Shortage /	
		no	of	Lack of	of work	and	and	Absence	Absence of	D
	Total	problem	Finance	Training	place	Training	Finance	of License	Equipment	Kr
COUNTRY TOTAL					F.5.5 5					
Total	350,718	44,007	188,694	5,068	28,886	9,399	36,773	1,373	710	2,
Male	86,618	17,648	42,364	811	4,716	4,041	8,585	460	-	7
Female	264,100	26,359	146,330	4,257	24,170	5,357	28,189	913	710	1,
TIGRAY REGION	•	,	,	•	,	,	,			,
Total	31,991	3,967	18,293	896	648	1,147	3,258	-	-	1
Male	7,260	1,023	5,335	251	102	105	109	-	-	
Female	24,732	2,944	12,958	645	547	1,043	3,149	-	-	1
AFFAR REGION										
Total	2,284	12	1,767	-	62	-	156	-	-	
Male	645	12	451	-	38	-	-	-	-	
Female	1,639	-	1,316	-	24	-	156	-	-	
AMHARA REGION	1									
Total	72,919	4,999	40,196	888	7,217	1,519	13,736	483	-	
Male	18,320	2,365	9,202	169	1,837	780	3,680	-	-	
Female	54,599	2,634	30,994	719	5,380	738	10,056	483	-	
OROMIA REGION										
Total	101,682	13,960	59,824	1,522	7,145	1,897	6,902	407	304	5
Male	26,216	6,019	15,198	51	905	1,121	834	-	-	
Female	75,465	7,941	44,626	1,471	6,239	775	6,068	407	304	5
SOMALI REGION										
Total	8,724	1,551	5,492	161	832	-	373	-	-	
Male	2,404	629	1,564	161	-	-	51	-	-	
Female	6,320	923	3,928	-	832	-	323	-	-	
BENISHANGUL G										
Total	1,647	281	451	31	95	270	191	23	-	
Male	410	95	62	-	-	119	54	-	-	
Female	1,238	185	390	31	95	151	138	23	-	
SNNP REGION										
Total	37,464	4,602	20,247	1,137	2,630	1,525	1,602	114	-	2
Male	8,021	1,486	3,086	179	-	725	861	114	-	
Female	29,443	3,116	17,161	958	2,630	800	741	-	-	2

Tah	le 6.1	12	Cont'd

Table 0.12	JOHN G									
Region and Sex	Problem Faced to start Own Business									
		I have	Shortage		Problem	Lack of Finance	Lack of working Place	Lack /	Shortage /	
		no	of	Lack of	of work	and	and	Absence	Absence of	D
	Total	problem	Finance	Training	place	Training	Finance	of License	Equipment	Kı
GAMBELLA REGI	ON									
Total	818	136	515	-	33	33	33	-	-	
Male	29	-	29	-	-	-	-	-	-	
Female	790	136	486	-	33	33	33	-	-	
HARARI REGION										
Total	1,974	288	884	-	477	-	98	-	49	
Male	328	-	198	-	41	-	-	-	-	
Female	1,646	288	685	-	436	-	98	-	49	
ADDIS ABABA CI ADMINISTRATION										
Total	81,638	13,337	35,632	346	9,045	2,769	9,817	346	357	1,
Male	20,734	5,672	6,453	-	1,707	1,038	2,737	346	-	7
Female	60,904	7,665	29,179	346	7,338	1,730	7,081	-	357	7
DIRE DAWA										
ADMINISTRATION	N									
Total	9,577	875	5,394	87	701	239	607	-	-	2
Male	2,251	347	786	-	87	152	260	-	-	
Female	7,326	528	4,608	87	615	87	347	-	-	2